## Multistationarity in (bio)chemical reaction networks with mass action kinetics: model discrimination, robustness and beyond

vorgelegt von Diplom-Ingenieur Carsten Conradi aus Ruit auf den Fildern

Von der Fakultät IV - Elektrotechnik und Informatik der Technischen Universität Berlin zur Erlangung des akademischen Grades Doktor der Ingenieurwissenschaften – Dr.-Ing. – genehmigte Dissertation

Promotionsausschuss:

Vorsitzender: Prof. Dr. Wichmann Berichter: Prof. Dr.-Ing. Raisch Berichter: Prof. Dr. Stelling

Tag der wissenschaftlichen Aussprache: 19. 2. 2008

Berlin 2008

D83

Forschungsberichte aus dem Max-Planck-Institut für Dynamik komplexer technischer Systeme

Band 21

## **Carsten Conradi**

Multistationarity in (bio)chemical reaction networks with mass action kinetics: model discrimination, robustness and beyond

D 83 (Diss. TU Berlin)

Shaker Verlag Aachen 2008

#### Bibliographic information published by the Deutsche Nationalbibliothek

The Deutsche Nationalbibliothek lists this publication in the Deutsche Nationalbibliografie; detailed bibliographic data are available in the Internet at http://dnb.d-nb.de.

Zugl.: Berlin, Techn. Univ., Diss., 2008

Copyright Shaker Verlag 2008

All rights reserved. No part of this publication may be reproduced, stored in a retrieval system, or transmitted, in any form or by any means, electronic, mechanical, photocopying, recording or otherwise, without the prior permission of the publishers.

Printed in Germany.

ISBN 978-3-8322-7234-0 ISSN 1439-4804

Shaker Verlag GmbH • P.O. BOX 101818 • D-52018 Aachen Phone: 0049/2407/9596-0 • Telefax: 0049/2407/9596-9 Internet: www.shaker.de • e-mail: info@shaker.de

# Acknowledgments

This work evolved during my time as a research assistant in Magdeburg, at the Max-Planck-Institute Dynamics of Complex Technical Systems and at the Otto-von-Guericke University. I was part of the Systems and Control Theory group headed by Prof. Raisch. Jörg Raisch provided an inspiring and exciting atmosphere for conducting research, giving his students all the freedom they desired, while, at the same time, providing excellent advice, whenever required. Special thanks go to two of my former colleagues from this group, Eckart Mayer and Ulrich Vollmer, whose reliability and helpfulness can not be rated high enough.

My work would not have been successful without close interaction with the Systems Biology group at the Max-Planck-Institute headed by Prof. Gilles. I am especially grateful to Prof. Gilles and all the members of his group, for always making me feel at home in their group. The interest Prof. Gilles has shown in my work was always encouraging to me. I will always remember my discussions about pointed polyhedral cones and Elementary Flux Modes with Steffen Klamt from this group.

It is three people who significantly influenced this work: Jörg Stelling, who pointed me to the work of Martin Feinberg, Karin Gatermann, who explained the exciting connection between Algebraic Geometry and (bio)chemical reaction networks to me in numerous inspiring discussions and Dietrich Flockerzi, whose ever lingering skepticism made me strive for precision in formulating results. Without any of them, this thesis would not be what it is.

During my early time in Magdeburg, I was able to pursue my interest in philosophy together with my friend Roland Waschler. We had a great time visiting philosophical seminars and lectures at the Otto-von-Guericke University.

All in all, I have to thank all my colleagues, former and present, from the University and the Max-Planck-Institute for their cooperation and the wonderful time we had together.

The time in Magdeburg was sweetened by Charlotte Clara, my daughter, whose laughter can help save the day and whose mere presence puts a lot of questions, problems and worries in perspective.

Magdeburg, 2008

Carsten Conradi

To my teacher: Karin Gatermann (1961–2005)

# Contents

1	Introduction							
2	<b>Bio</b> 2.1 2.2 2.3	iochemical reaction networks with mass action kinetics         1 Notation         2 Modeling of biochemical reaction networks with mass action kinetics         2.2.1 Properties of $\phi(x)$ and $v(k, x)$ 3 Chemical Reaction Network Theory (CRNT)						
3	Cor 3.1	Positive solutions to the polynomial equations						
		3.1.2 Parametrizing positive solutions – the equation $Y^{(L)T} \mu = \ln \frac{E\nu}{E\lambda} \dots \dots$	8					
	3.2	Solvability of $Y^{(L)T} \mu = \ln \frac{E\nu}{E\lambda}$ 3.2.1 Solvability I 3.2.1 Solvability I 3.2.2.1 Special form of polynomials 3.2.2 Solvability II 3.2.2.1 L <sup>+</sup> -matrices matrices and sign-central matrices 3.2.2.2 Sign patterns of $Q(\mu)$ determined by linear inequalities 3.2.2.2 Sign patterns of $Q(\mu)$ determined by linear inequalities	11 11 12 13 14 15					
	33	3.2.2.3 An example: application to network $N_1 \dots \dots \dots \dots \dots$	17 21					
	0.0	3.3.1 The linear subspace case	23					
	3.4	Multistationarity in subnetworks	25					
		3.4.1 Multistationarity in subnetworks defined by stoichiometric generators	26					
		<ul> <li>3.4.1.1 Algebraic properties of subnetworks defined by a stoichiometric generator</li> <li>3.4.1.2 Steady states for subnetworks defined by a stoichiometric generator</li> <li>3.4.2 Extension of multistationarity to the overall network</li></ul>	26 28 32					
	3.5	Resume: a program to decide about multistationarity $\hfill \ldots \hfill \ldots \hf$	35					
4	Mu 4.1 4.2 4.3	Itistationarity in the activation of an MAPK(K)       :         Processive vs. distributive phosphorylation       .         Positive solutions for the polynomial equations       .         4.2.1       Positive solutions for $\mathcal{N}_4$ .         4.2.2       Positive solutions for $\mathcal{N}_6$ .         4.2.3       Positive solutions for $\mathcal{N}_6$ .         Positive solutions satisfying the conservation relations       .         4.3.1       Parameterizing Multistationarity for $\mathcal{N}_4$ .	37 39 39 42 43 44 45					
	4.4	An extension: open systems	46					
	4.5	Model discrimination using steady state information	48					
5	Mu 5.1 5.2 5.3 5.4	Itistationarity in cell cycle regulation       Subnetwork analysis         Subnetwork analysis       Analysis of the complete network $\mathcal{N}_{10}$ Analysis of the complete network $\mathcal{N}_{11}$ Conclusions	51 53 59 70 74					

6	Rob	obustness of Multistationarity 75							
	6.1	3.1 Robustness against variations in the rate constants							
	6.2	Robust	mess against concentration fluctuations						
		6.2.1	Analysis of the steady state equations						
			6.2.1.1 Network $N_{12}$						
			6.2.1.2 Network $\mathcal{N}_{13}$						
			6.2.1.3 Network $N_{14}$						
		6.2.2	Comparing robustness against concentration fluctuations						
<b>7</b>	Mu	ltistatio	onarity and Beyond 91						
	7.1	The Ja	cobian matrix						
		7.1.1	Zero eigenvalues of the Jacobian						
		7.1.2	Saddle-node bifurcations						
	7.2	Applica	ation to a double-phosphorylation mechanism						
		7.2.1	Zero eigenvalues						
		7.2.2	Saddle-node bifurcations						
		7.2.3	Bifurcations of higher codimension: Bogdanov-Takens						
~	a								
8	Cor	lclusion	103						
Δ	Mo	dels for	the activation of an $MAPK(K)$ 105						
	A 1	Distrib	utive mechanism for phosphorylation and dephosphorylation 105						
		A 1 1	Species and complexes of network $N_4$ 105						
		A 1 2	Ordinary differential equations 105						
		A 1 3	Structural data 106						
	A.2	Proces	sive phosphorylation and distributive dephosphorylation						
		A.2.1	Species and complexes of network $N_{5}$						
		A.2.2	Ordinary differential equations						
		A.2.3	Structural data						
	A.3	Proces	sive mechanism for phosphorylation and dephosphorylation						
		A.3.1	Species and complexes of network $\mathcal{N}_6$						
		A.3.2	Ordinary differential equations						
		A.3.3	Structural data						
	A.4	Open s	wstems						
		A.4.1	Network $\mathcal{N}_4 + \mathcal{N}_7$						
		A.4.2	Network $\mathcal{N}_4 + \mathcal{N}_8$						
		A.4.3	Network $\mathcal{N}_4 + \mathcal{N}_9$						
		A.4.4	Network $\mathcal{N}_5 + \mathcal{N}_7$						
		A.4.5	Network $\mathcal{N}_5 + \mathcal{N}_8$						
		A.4.6	Network $\mathcal{N}_5 + \mathcal{N}_9$						
		A.4.7	Network $\mathcal{N}_6 + \mathcal{N}_7$						
		A.4.8	Network $\mathcal{N}_6 + \mathcal{N}_8 \dots \dots$						
		A.4.9	Network $\mathcal{N}_6 + \mathcal{N}_9$						
в	Mo	dels for	cell cycle regulation 123						
	B.1	Binary	complex model						
		B.1.1	Species and complexes of network $\mathcal{N}_{10}$						
		B.1.2	Ordinary differential equations						
		B.1.3	Structural data						
	В.2	Ternar	y complex model						
		B.2.1	Species and complexes of network $\mathcal{N}_{11}$						
		B.2.2	Ordinary differential equations						
		B.2.3	Structural data						

$\mathbf{C}$	Models of the signal transduction motifs											
	C.1	Network $\mathcal{N}_{12}^{-}$										
		C.1.1	Species and complexes of network $\mathcal{N}_{12}$	127								
		C.1.2	Ordinary differential equations	127								
		C.1.3	Structural data	127								
	C.2	Netwo	rk $\mathcal{N}_{13}$	129								
		C.2.1	Species for and complexes of network $\mathcal{N}_{13}$	129								
		C.2.2	Ordinary differential equations	129								
		C.2.3	Structural data	129								
	C.3	Netwo	rk $\mathcal{N}_{14}$	130								
		C.3.1	Species and complexes of network $\mathcal{N}_{14}$	130								
		C.3.2	Ordinary differential equations	131								
		C.3.3	Structural data	131								
D	An	An Algorithm to check Lemma 3 133										
	D.1 Preliminary ideas											
	D.2	D.2 Deciding ker $(A) \cap \mathbb{R}^n_{\mathbb{A}} \neq \emptyset$										
	D.3 Determining generators											

# Chapter 1

# Introduction

Multistationarity means the existence of at least two positive steady state solutions to a system of Ordinary Differential Equation (ODEs) derived from a biochemical reaction network. It is an interesting question in its own right: one is looking for the number of positive solutions to a system of polynomial equations in the unknown concentrations with, in most application cases, unknown rate constants. From a mathematical point of view this question is still unanswered. There exists a variety of results that gives an *upper bound* for the number of (positive) solutions to a system of polynomials. However, whether or not a given system of polynomials actually has the predicted number of zeros can usually not be decided (among these results are *Descartes rule of signs* for positive solutions and Bernstein's and Bezout's Theorem for all solutions, including negative and complex; see, for example, [13, 14] and [62]). Moreover, all of these rosults hold for systems of polynomials with *known* coefficients. The question whether or not a system of polynomial equations with unknown coefficients can admit at least two positive solutions has received little attention in mathematics.

In Chemical Engineering, in the so-called Chemical Reaction Network Theory (CRNT) by Martin Feinberg and his co-workers this question has received some attention [30, 24, 25, 26, 55, 28, 22, 21]. Here, the special structure of equations defined by a (bio)chemical reaction network with mass action kinetics is used to derive necessary and sufficient conditions for the existence of at least two positive solutions. These results, however, require that that the network under consideration has certain structural properties. Only recently these results have been recognized in the mathematical literature, for example in [33, 35].

The approach taken in this thesis is inspired by CRNT, even though it has been derived independent of the aforementioned references. In particular the results used to decide about multistationarity presented in Chapter 3 are not based on CRNT (albeit resulting from an attempt to give a self-contained proof of the results given in [25, 27] and [23]). The results described in Chapter 3 are in fact complementary to those presented in the aforementioned references: they can be successfully applied, where CRNT fails, while CRNT might be successfully applied, where the results of Chapter 3 fail (see Chapter 5, where multistationarity is confirmed for two networks, for which CRNT is inconclusive).

Only recently CRNT has been applied to biochemical reaction networks, in a Systems Biology inspired context, that is, in a context, where the analysis of large and complex reaction networks is required and where parameter uncertainty is predominant. Often, in this context, qualitative knowledge about the dynamics of the system exists; for example, it might be known from experimental observations that the system exhibits bistability or some other form of multistationarity. Suppose there are different hypotheses, each corresponding to a different reaction network with uncertain parameters, related to a specific biological process. Suppose, furthermore, that a certain qualitative behaviour has been observed in experiments. It is then natural to ask which of the postulated networks can, for some conceivable parameter vector, exhibit the observed behaviour. To this end, CRNT has been applied in [15, 16, 17, 18]: suppose experimental evidence suggests that the system under consideration admits different steady states, as is, for example, the case in signal transduction networks and cell cycle regulation. Further suppose, all hypotheses lead to reaction networks that can be analysed using CRNT. Then all networks corresponding to hypotheses where multistationarity is excluded for any conceivable parameter vector can be discarded, as these networks can never reproduce the experimental behaviour. With a similar goal CRNT has been applied in [10, 11, 12] (see also Chapter 4, where a similar analysis is performed using the results presented in Chapter 3).

In many cases, a certain qualitative behaviour, like multistationarity has been observed for rate constants that vary over a wide range of values. Traditionally, one therefore concludes that the system is robust with respect to changes in these rate constants. Using the results of Chapter 3, it is, for certain network structures, possible to obtain a parameterization of rate constants that can serve as an explanation of this robustness. Moreover, in these cases, it is possible to obtain analytical expressions of 'critical points' and 'critical parameters', that is points where certain bifurcations occur.

The outline of this thesis is a follows: in Chapter 2, the notation used to describe biochemical reaction networks with mass action kinetics is introduced. In Chapter 3 conditions for multistationarity are derived. Furthermore, an algorithm is presented, whose steps can be applied to any biochemical reaction network to test for multistationarity (but may be inconclusive). If multistationarity is possible, the results of Chapter 3 allow the computation of a pair of steady states and the corresponding vector of rate constants. In some cases, depending on the network structure, the conditions for multistationarity derived in Chapter 3 are necessary and sufficient, while in other cases only sufficient conditions can be obtained. However, a precise description of those network properties that facilitate necessary and sufficient conditions remains open.

In Chapters 4 - 7 the results obtained in Chapter 3 are applied to a variety of reaction networks in order to tackle the questions of multistationarity, robustness of multistationarity and bifurcation points: Chapter 4 deals with model discrimination for a double-phosphorylation mechanism, while in Chapter 5 two reaction networks proposed in cell cycle regulation are analysed with respect to the ability to admit multistationarity. Chapter 6 deals with robustness and Chapter 7 with bifurcation points of the double-phosphorylation mechanism discussed in Chapter 4. The appendices contain the structural data of all networks analysed in this thesis, as well the matlab code of some of the algorithms developed in this thesis to verify/falsify multistationarity.

## Chapter 2

# Biochemical reaction networks with mass action kinetics

#### 2.1 Notation

The following symbols will be used throughout this work:

 $\mathbb{R}^n$  ... the *n*-dimensional Euclidian space,

 $\mathbb{R}^n_{>0}$  ... the positive orthant of  $\mathbb{R}^n$ ,

 $\mathbb{R}^n_{>0}$  ... the nonnegative orthant of  $\mathbb{R}^n$ .

Let  $\overline{p}$  and q be positive integers. Then  $\mathbb{R}^{p \times q}$  is used to denote the set of all  $p \times q$  real valued matrices. If  $A \in \mathbb{R}^{p \times q}$  then  $A^T$  is used to denote its transpose. Vectors  $v \in \mathbb{R}^n$  are considered as column vectors and, in most cases, denoted as transposed row vectors, e.g.  $v = \begin{pmatrix} 1 \\ 0 \\ 1 \end{pmatrix}$  will be displayed as  $v = (1, 0, 1)^T$ . For vectors  $u, v \in \mathbb{R}^n$ , the standard scalar product in  $\mathbb{R}^n$  is denoted by  $\langle u, v \rangle = \sum_{i=1}^n u_i v_i$ . Furthermore, for vectors  $u \in \mathbb{R}^n$ , the following abbreviations are used:

$$u > 0 \quad \Leftrightarrow u_1 > 0, \dots, u_n > 0,$$
 (2.1a)

$$\frac{1}{u} = \left(\frac{1}{u_1}, \dots, \frac{1}{u_n}\right)^T u_i \neq 0$$
(2.1b)

$$\ln u = (\ln u_1, \dots, \ln u_n)^T u > 0$$
(2.1c)

$$e^{u} = (e^{u_{1}}, \dots, e^{u_{n}})^{T}$$
 (2.1d)

The symbol diag (u) is used to denote the  $n \times n$ -diagonal matrix:

diag 
$$(u) := \begin{bmatrix} u_1 & 0 & 0 \\ 0 & \ddots & 0 \\ 0 & 0 & u_n \end{bmatrix}$$
. (2.1e)

Let  $A \in \mathbb{R}^{p \times q}$  be a matrix. Then [A] is used to denote the range (image) of A:

$$[A] := \operatorname{im}(A).$$
 (2.1f)

#### 2.2 Modeling of biochemical reaction networks with mass action kinetics

In this section the notation used to describe biochemical reaction networks with mass action kinetics is introduced. Consider reaction network  $N_1$ :

$$A \underbrace{\frac{\mathbf{k}_{1}}{\mathbf{k}_{2}}}_{2A+B} \frac{\mathbf{k}_{3}}{\mathbf{k}_{4}} B \qquad (\mathcal{N}_{1})$$

$$2A+B \underbrace{\frac{\mathbf{k}_{5}}{\mathbf{k}_{6}}}_{3A} 3A$$

This network consists of two species A and B. The symbol n is used to denote the number of species. Thus n = 2 for network  $\mathcal{N}_1$ . Network  $\mathcal{N}_1$  contains five *complexes*: A, 0, B, 2A + B and 3A, that is, in graph-theoretical terms, the nodes of the graph representing the reaction network are called complexes. The symbol m is used to denote the number of complexes, thus m = 5 for network  $\mathcal{N}_1$ . The zero complex 0 is used to denote that a system is open with respect to a certain species: A and B can enter and leave the system. Throughout this work, it is assumed that the reaction network is in the standard form of CRNT, as defined in, for example, [27]: node labels are unique (i.e. complexes appear exactly once). The complexes are linked by reactions: some of them are  $A \to 0$ , or  $2A + B \to 3A$ . The symbol r is used to denote the number of reactions, thus r = 6 for  $\mathcal{N}_1$ . Associated to every reaction, there is a rate constant:  $k_1, \ldots, k_6$  for  $\mathcal{N}_1$ .

To every species belongs a concentration variable,  $x_1$  for A and  $x_2$  for B. Thus, in general,  $x \in \mathbb{R}^n$ , the n-dimensional Euclidian space. Associate to each species the corresponding unit vector, that is  $e_1$  to A and  $e_2$  to B. Then every complex can be represented by a vector  $y_i$ , which is the sum of its constituent species. In the above example,  $y_1 = e_1$  for A,  $y_2 = 0$ , the two dimensional zero vector for 0,  $y_3 = e_2$  for B,  $y_4 = 2e_1 + e_2$  for 2A + B and  $y_5 = 3e_1$  for 3A. Collect all  $y_i$  in a matrix Y, thus

$$Y = \begin{bmatrix} y_1 & \dots & y_5 \end{bmatrix}$$
(2.2a)  
= 
$$\begin{bmatrix} 1 & 0 & 0 & 2 & 3 \\ 0 & 0 & 1 & 1 & 0 \end{bmatrix}.$$

Thus  $Y \in \mathbb{R}^{2\times 5}$  for  $\mathcal{N}_1$  and  $Y \in \mathbb{R}^{n\times m}$ , in general. The graph of the reaction network is represented by its *incidence matrix*  $I_a$ . Each colum of  $I_a$  represents a reaction and contains exactly two nonzero entries: +1 for the product complex and -1 for the educt complex. Thus  $I_a \in \mathbb{R}^{5\times 6}$  for  $\mathcal{N}_1$  and  $I_a \in \mathbb{R}^{m\times r}$ , in general. Consider, for example, the reaction  $2A + B \rightarrow 3A$ . Here the product complex is 3A and the educt complex is 2A + B. Thus  $I_a$  contains the column vector  $(0, 0, 0, -1, 1)^T$ . For  $\mathcal{N}_1$ one obtains:

$$I_a = \begin{vmatrix} -1 & 1 & 0 & 0 & 0 & 0 \\ 1 & -1 & -1 & 1 & 0 & 0 \\ 0 & 0 & 1 & -1 & 0 & 0 \\ 0 & 0 & 0 & 0 & -1 & 1 \\ 0 & 0 & 0 & 0 & 0 & 1 & -1 \end{vmatrix}$$
(2.2b)

Up to now, the structure of the reaction network has been described. To describe dynamics, a reaction rate is associated to each reaction. As only mass-action kinetics are considered in this thesis, each reaction rate consists of the rate constant multiplied by the product of the species concentrations corresponding to its educt complex. Hence one obtains  $v_1 := k_1 x_1$ ,  $v_2 := k_2$ ,  $v_3 := k_3$ ,  $v_4 := k_4 x_2$ ,  $v_5 := k_5 x_1^2 x_2$  and  $v_6 = k_6 x_1^3$  for  $\mathcal{N}_1$ . To formalize this, let  $Y^{(L)}$  be the matrix of educt complexes, where each educt complex is listed as often, as it is an educt complex of a reaction. For example, for  $\mathcal{N}_1$ , one obtains (using  $y_i$ ,  $i = 1, \ldots, 5$ , as defined above):

$$Y^{(L)} = \begin{bmatrix} y_1 & y_2 & y_2 & y_3 & y_4 & y_5 \end{bmatrix}$$
(2.2c)  
= 
$$\begin{bmatrix} 1 & 0 & 0 & 0 & 2 & 3 \\ 0 & 0 & 0 & 1 & 1 & 0 \end{bmatrix}.$$

Thus one obtains  $Y^{(L)} \in \mathbb{R}^{2 \times 6}$  for  $\mathcal{N}_1$  and  $Y^{(L)} \in \mathbb{R}^{n \times r}$ , in general. Using  $y_i^{(L)}$  to denote column vectors of  $Y^{(L)}$  and the definition

$$x^{y} := x_{1}^{y_{1}} x_{2}^{y_{2}} \dots, x_{n}^{y_{n}} = \prod_{i=1}^{n} x_{i}^{y_{i}},$$
 (2.3a)

for vectors  $x, y \in \mathbb{R}^n$ , one obtains the monomial vector

$$\phi\left(x\right) := \begin{pmatrix} x^{y_1^{(L)}} \\ \vdots \\ x^{y_r^{(L)}} \end{pmatrix}$$
(2.3b)

and the vector of reaction rates

$$v(k, x) = \operatorname{diag}(k) \phi(x),$$
  
= diag(\phi(x)) k, (2.3c)

where  $k := (k_1, \ldots, k_r)^T$ , the vector of rate constants. For the reaction network  $\mathcal{N}_1 \phi(x) = (x_1, 1, 1, 1, x_2, x_1^2 x_2, x_1^3)^T$  and thus  $v(k, x) = (k_1 x_1, k_2, k_3, k_4 x_2, k_5 x_1^2 x_2, k_6 x_1^3)$ . Then the ODEs describing the dynamics of a reaction networks are given as

$$\dot{x}(t) = Y I_a v(k, x(t)).$$
 (2.4)

Note the  $n \times r$ -matrix  $Y I_a$  is sometimes called *stoichiometric matrix* and denoted by  $N := Y I_a$ . Consequently, in the literature on Chemical Reaction Network Theory, the linear subspace spanned by the columns of N is called *stoichiometric subspace*. Usually n (the number of species) is less than r (the number of reactions) and this will be a standing assumption in this thesis. The symbol s is used to denote its dimension, that is  $s := \operatorname{rank}(N)$ . The symbol S is used to denote the stoichiometric subspace, that is  $S := \operatorname{im}(N)$ . If s < n, then the system (2.4) is subject to *conservation relations*. Let  $W \in \mathbb{R}^{n \times (n-s)}$ , with  $W^T N = 0$ . Then the flow of (2.4) is invariant under [W] [58, 34, 35]. Thus one has to consider

$$W^T x = c,$$
 (2.5)

for some  $c \in \mathbb{R}_{>0}^{n-s}$ . Often c is considered as a parameter that has to be determined. Observe that, by elementary linear algebra

$$[W] = S^{\perp}$$
, (2.6)

that is, the range of W is the orthogonal complement of N (which follows naturally from the property  $W^T N = 0$ , [61]). Observe that the ODEs (2.4) are linear in the rate constants, as

$$\dot{x}(t) = Y I_a v(k, x(t)) = Y I_a \operatorname{diag}\left(\phi\left(x\left(t\right)\right)\right) k.$$

For network  $\mathcal{N}_1$ , the stoichiometric matrix is

$$N = \begin{bmatrix} -1 & 1 & 0 & 0 & 1 & -1 \\ 0 & 0 & 1 & -1 & -1 & 1 \end{bmatrix}$$
(2.7)

and s = 2. The ODEs defined by the network are

$$\dot{x}_1 = k_2 - k_1 x_1 - k_6 x_1^3 + k_5 x_1^2 x_2$$
$$\dot{x}_2 = k_3 + k_6 x_1^3 - k_4 x_2 - k_5 x_1^2 x_2$$

#### **2.2.1** Properties of $\phi(x)$ and v(k, x)

Recall the definition of  $\phi(x)$ :

$$\phi\left(x\right) = \begin{pmatrix} x^{y_{1}^{\left(L\right)}} \\ \vdots \\ x^{y_{r}^{\left(L\right)}} \end{pmatrix}.$$

Let  $\alpha \in \mathbb{R}^n$  and observe that  $(\text{diag}(\alpha) \ x) = (\alpha_1 \ x_1, \ldots, \ \alpha_n \ x_n)^T$  and thus

$$(\text{diag}(\alpha) x)^y = (\alpha_1 x_1)^{y_1} (\alpha_2 x_2)^{y_2} \dots (\alpha_n x_n)^{y_n} = \alpha^y x^y,$$

for  $\alpha, x, y \in \mathbb{R}^n$ . Thus one obtains

$$\phi\left(\operatorname{diag}\left(\alpha\right) x\right) = \begin{pmatrix} (\operatorname{diag}\left(\alpha\right) x)^{y_{1}^{(L)}} \\ \vdots \\ (\operatorname{diag}\left(\alpha\right) x)^{y_{r}^{(L)}} \end{pmatrix} = \begin{pmatrix} \alpha^{y_{1}^{(L)} x y_{1}^{(L)}} \\ \vdots \\ \alpha^{y_{r}^{(L)} x y_{r}^{(L)}} \end{pmatrix}$$

$$= \operatorname{diag}\left(\alpha^{y_{1}^{(L)}}, \dots, \alpha^{y_{r}^{(L)}}\right) \begin{pmatrix} x^{y_{1}^{(L)}} \\ \vdots \\ x^{y_{r}^{(L)}} \end{pmatrix} = \operatorname{diag}\left(\phi\left(\alpha\right)\right) \phi\left(x\right).$$

$$(2.8)$$

Using the abbreviation  $e^x := (e^{x_1}, \ldots, e^{x_n})^T$  for vectors  $x \in \mathbb{R}^n$ , one obtains in particular (for  $\mu \in \mathbb{R}^n$ )

$$\phi(e^{\mu}) = \begin{pmatrix} (e^{\mu})^{y_1^{(L)}} \\ \vdots \\ (e^{\mu})^{y_r^{(L)}} \end{pmatrix} = \begin{pmatrix} e^{\langle y_1^{(L)}, \mu \rangle} \\ \vdots \\ e^{\langle y_r^{(L)}, \mu \rangle} \end{pmatrix} = e^{Y^{(L)^T} \mu}.$$
 (2.9)

Further note

$$\frac{1}{\phi(x)} = \begin{pmatrix} \left(x^{y_1^{(L)}}\right)^{-1} \\ \vdots \\ \left(x^{y_r^{(L)}}\right)^{-1} \end{pmatrix} = \begin{pmatrix} \left(x^{-1}\right)^{y_1^{(L)}} \\ \vdots \\ \left(x^{-1}\right)^{y_r^{(L)}} \end{pmatrix} = \phi(x^{-1}).$$
(2.10)

As  $k \in \mathbb{R}^r_{>0}$ , positive values of x will result in positive values of v(k, x), that is

$$x \in \mathbb{R}^n_{>0} \Rightarrow v(k, x) \in \mathbb{R}^r_{>0}$$

and vice versa.

#### 2.3 Chemical Reaction Network Theory (CRNT)

The distinguishing feature of CRNT is its ability to make a connection between the structure of a reaction network and the existence of (multiple) equilibria for the corresponding system of ODEs. Its general idea can be summarised in the following way: for any network, a non-negative integer  $\delta$  called the deficiency can be derived from the network structure alone. For its formal definition one more concept is needed – the linkage class. Network  $\mathcal{N}_1$  consists of two sets of complexes:  $\{A, 0, B\}$  and  $\{2A + B, 3A\}$ . Both sets are internally connected by reactions, while no reactions exist between elements of distinct sets. Sets of complexes that are internally connected by reactions are called linkage classes. Let  $\ell$  be the number of linkage classes in an arbitrary network. Then the deficiency of this network is defined as the non-negative integer [24, 25]

$$\delta = m - \ell - s \qquad (2.11)$$

Note that the deficiency of the network only depends on the network structure and thus, in particular,  $\delta$  is independent of parameter values. For network  $\mathcal{N}_1$ ,  $\delta = 1$ . If  $\delta$  is zero for a particular network, then no system of ODEs endowed with mass action kinetics that can be derived from the network can admit multiple steady states, regardless of the rate constants. Furthermore, sustained periodic oscillations are impossible [24, 27]. If  $\delta$  is one and the network satisfies some mild additional conditions, the so-called Deficiency One Algorithm can be applied to decide whether or not the network can admit multiple steady states. If the deficiency is greater than one, under certain conditions the so-called Advanced Deficiency Theory can be used to decide about the possibility of multistationarity (for the Deficiency One Algorithm see [25, 28] and for Advanced Deficiency Theory see [23, 21, 22]).

For certain networks with  $\delta = 1$ , the deficiency one algorithm can be used to decide about the existence of a parameter vector k such that the corresponding system of ODEs endowed with mass action kinetics admits at least two positive steady-states. For certain networks of higher deficiency, the so-called Advanced Deficiency Theory can be applied. The principle idea of both the Deficiency One Algorithm and the Advanced Deficiency Theory is as follows: for each network where the algorithm or the theory is applicable, several systems of linear equalities and inequalities can be formulated (inequality systems, for short). (Note, whether or not the algorithm or the theory are applicable depends on the network structure alone; see the aforementioned references.) These inequality systems only depend on the network structure and the complexes, that is, the way the species are combined in the complexes and the way the complexes interact. If, for any of these systems, a solution exists and if this solution is sign compatible with the subspace  $\mathcal{S}$ , then multistationarity is possible (and a set of rate constants together with two distinct steady-states can be calculated from this solution). If no such solution exists, then multistationarity is impossible. (Note that, following the notation used in [24], a vector is sign compatible with a linear subspace if this subspace contains at least one vector with the same sign pattern; that is, the vector (1, -1) is sign compatible with the subspace spanned by (1, 0) and (1, -2)because (1, -2) has the same sign pattern as (1, -1), namely (+, -).)

## Chapter 3

## Conditions for multistationarity

Consider the Ordinary Differential Equations describing the dynamics of a biochemical reaction network as introduced in (2.4), (2.5) in Chapter 2:

$$\dot{x}(t) = Y I_a v(k, x(t))$$
  
 $W^T x = c$ 

In this thesis only positive solutions x are considered, thus the corresponding reaction network is said to exhibit multistationarity, if two distinct positive vectors  $a, b \in \mathbb{R}^n_{>0}$ , a positive vector  $k \in \mathbb{R}^r_{>0}$  and values  $c_i, i = 1, \ldots, n - s$ , can be found such that the following conditions hold:

$$N v(k, a) = 0$$
 (3.1a)

$$W^T a = c \tag{3.1b}$$

and

$$N v(k, b) = 0$$
 (3.2a)

$$W^T b = c.$$
 (3.2b)

In the remainder of this chapter (3.1a), (3.1b) and (3.2a), (3.2b) are analysed and transformed into equivalent conditions that are in most cases easier to handle: in a first step the polynomial equations (3.1a) and (3.2a) are transformed and the rate constants are eliminated. This is described in Section 3.1. Solvability of the resulting system of equations is analysed in Section 3.2. As an arbitrary solution obviously can only be expected to satisfy (3.1a) and (3.2a), conditions are derived that guarantee that (3.1b) and (3.2b) hold as well. This is described in Section 3.3. Section 3.4 contains a discussion of the connection between multistationarity in a subnetwork and multistationarity in the overall network. This chapter closes with an algorithm to decide about multistationarity in a biochemical reaction network with mass action kinetics in Section 3.5. This algorithm incorporates the results obtained in Section 3.1 – 3.4.

#### 3.1 Positive solutions to the polynomial equations

For biochemical reaction networks only positive solutions a, b to (3.1a), (3.1b) and (3.2a), (3.2b) are of interest. As discussed in the previous chapter, if k > 0, this requirement is equivalent to v(k, a) and v(k, b) being positive. The equations  $Y I_a v(k, a) = 0$  and  $Y I_a v(k, b) = 0$  with the positivity constraint can be simplified using the pointed polyhedral cone ker $(Y I_a) \cap \mathbb{R}_{\geq 0}^r$  defined by the intersection of the null space of  $Y I_a$ , ker $(Y I_a)$ , with the nonnegative orthant of  $\mathbb{R}^r$ ,  $\mathbb{R}_{\geq 0}^r$ . For this purpose some properties of ker $(Y I_a) \cap \mathbb{R}_{\geq 0}^r$  are discussed in Section 3.1.1, before the equations are transformed in Section 3.1.2.

#### **3.1.1** The cone $\ker(Y I_a) \cap \mathbb{R}^r_{>0}$

The cone ker( $Y I_a$ )  $\cap \mathbb{R}^r_{\geq 0}$  is a well studied object, due to its importance in the (bio)chemical literature, starting with the classical work of Clarke [8]. As a pointed polyhedral cone it can be represented by

non-negative linear combinations of a finite set of extreme rays [52]. The calculation of these rays is in general computationally hard, however, due to the importance of ker $(Y I_a) \cap \mathbb{R}^r_{\geq 0}$ , there exists a variety of algorithms and software tools, for example, [56, 32].

The aforementioned papers are part of the huge body of literature concerned with metabolic flux analysis, where each element of ker( $Y I_a$ )  $\cap \mathbb{R}^{r}_{\geq 0}$  is interpreted as a particular flux, an allocation of values  $v_i \in \mathbb{R}$  to each reaction of the network, such that the overall network is in steady state. An elementary flux is a an element  $v \in \ker(Y I_a) \cap \mathbb{R}^{r}_{\geq 0}$  with a maximum number of zero entries. If every reaction in the network is 'irreversible' (in the terminology used in metabolic flux analysis), then elementary fluxes are equivalent to extreme rays generating ker( $Y I_a$ )  $\cap \mathbb{R}^{r}_{\geq 0}$  [32].

In the literature on metabolic flux analysis the term *irreversible reaction* is defined differently than in the context of this work: forward and backward reaction are represented as a *single edge in the directed graph* representing the reaction network. Thus, in particular, negative reaction rates  $v_i(k, x)$ are possible. These correspond to reactions 'in opposite direction' (relative to the edge in the directed graph). Therefore, in this setup,  $x \in \mathbb{R}_{>0}^n \Rightarrow v_i(k, x) > 0$  is required only for *irreversible reactions*. In the setup considered in this thesis,  $x \in \mathbb{R}_{>0}^n \Rightarrow v_i(k, x) > 0$  for every reaction, thus all *reactions are irreversible in the context of metabolic flux analysis*. As a consequence, it is possible to use the software tools from metabolic flux analysis to calculate the extreme rays of ker( $Y I_a$ )  $\cap \mathbb{R}_{>0}^r$ .

As defined in [32], extreme rays  $E_i$  of ker $(Y I_a) \cap \mathbb{R}^r_{>0}$  satisfy (3.3a), (3.3b) and (3.3c) given below:

$$Y I_a E_i = 0,$$
 (3.3a)

$$E_i \in \mathbb{R}^r_{>0}$$
. (3.3b)

Given  $E_i$ ,  $E_j$  with  $Y I_a E_i = 0$ ,  $Y I_a E_j = 0$  and  $E_i$ ,  $E_j \in \mathbb{R}^r_{>0}$ . Then

$$\operatorname{supp}(E_i) \subseteq \operatorname{supp}(E_j) \Rightarrow E_i = 0 \text{ or } E_j = \alpha E_i, \alpha \in \mathbb{R}_{>0}, \tag{3.3c}$$

where  $\sup (E_j) = \{i \in \{1, ..., r\} | E_{j_i} > 0, \}$  denotes the support of vector  $E_j$ , i.e. the set of indices where  $E_j$  has nonzero values. Let p be the number of extreme rays of  $\ker(Y I_a) \cap \mathbb{R}_{\geq 0}^r$ . (Note that extreme rays need not be linearly independent.) If a set of extreme rays  $\{E_1, ..., E_p\}$  is complete, in the sense that their nonnegative linear combination is all of  $\ker(Y I_a) \cap \mathbb{R}_{\geq 0}^r$ , its elements are called generators of  $\ker(Y I_a) \cap \mathbb{R}_{\geq 0}^r$ .

In metabolic flux analysis the importance of the generators of  $\ker(Y I_a) \cap \mathbb{R}^r_{\geq 0}$  stems from their oneto-one correspondence to the reactions: nonzero entries can be interpreted as 'active' reactions, zero entries as 'inactive' (in steady state). In this sense every generator defines a subnetwork of the original reaction network consisting of all 'active' reactions.

## **3.1.2** Parametrizing positive solutions – the equation $Y^{(L)T} \mu = \ln \frac{E \nu}{E \lambda}$

Let *E* be a matrix, whose columns are generators for ker( $Y I_a$ )  $\cap \mathbb{R}_{\geq 0}^r$ , as described in Section 3.1.1. Then the equations  $Y I_a v(k, a) = 0$  and  $Y I_a v(k, b) = 0$  with the positivity constraint are equal to:

$$v(k, a) = E \lambda, \lambda \in \mathbb{R}^{p}_{>0}, E \lambda > 0,$$
 (3.4a)

$$v(k, b) = E \nu, \nu \in \mathbb{R}^{p}_{>0}, E \nu > 0,$$
 (3.4b)

where p denotes the number of extreme rays of ker(Y  $I_a) \cap \mathbb{R}_{\geq 0}^r$ . Equations (3.4a) and (3.4b) guarantee that  $v(k, a) \in \mathbb{R}_{>0}^r$  and  $v(k, b) \in \mathbb{R}_{>0}^r$ . The definition of  $v(k, \cdot)$  in (2.3c) in turn ensures that  $v(k, a) \in \mathbb{R}_{>0}^r$  and  $v(k, b) \in \mathbb{R}_{>0}^r$  imply  $a, b \in \mathbb{R}_{>0}^n$ .

To streamline the discussion define the set of all nonnegative vectors  $x \in \mathbb{R}^p_{\geq 0}$  such that E x is positive:

$$\Lambda(E) := \left\{ x \in \mathbb{R}^p_{\geq 0} \mid E x > 0 \right\}.$$
(3.5)

Then (3.4a), (3.4b) can be rewritten as

$$v(k, a) = E \lambda,$$
  

$$v(k, b) = E \nu,$$
  

$$\nu, \lambda \in \Lambda (E).$$
(3.6)

**Remark 1.** Some comments about  $\Lambda(E)$ :  $\Lambda(E)$  is the union of cones. It consists of the positive orthant  $\mathbb{R}^{p}_{\geq 0}$  and some of its boundaries. To see this, observe the following points:

- (1) As  $E \in \mathbb{R}_{\geq 0}^{r \times p}$ ,  $x \in \mathbb{R}_{>0}^{p}$  implies E x > 0, that is  $\mathbb{R}_{>0}^{p} \subseteq \Lambda(E)$ .
- (2) Let  $x \in \mathbb{R}^p_{\geq 0}$  with one zero component and the remaining components positive. Denote the zero component with  $x_i$  and let  $n_i^T \in \mathbb{R}^p_{\geq 0}$ ,  $i = 1, \ldots, r$  be the row vectors of E. Then

$$E x = \begin{pmatrix} \langle n_1, x \rangle \\ \vdots \\ \langle n_r, x \rangle \end{pmatrix} > 0,$$

if and only if no scalar product  $\langle n_j, x \rangle$  is a multiple of  $x_i$  (i.e. if and only if  $\langle n_j, x \rangle = \sum_{l=1}^p n_{jl} x_l \neq n_{ji} x_i$ ). That is, if and only if  $i \in \text{supp}(n_j)$  implies  $|\text{supp}(n_j| > 1), j = 1, ..., r$ . A similar argument can be made for  $x \in \mathbb{R}^p_{>0}$  with several components equal to zero.

- (3) In general the constraint ν, λ ∈ Λ (E) is not hard to test: one can, for example, start with looking for positive ν, λ. Only if those cannot be established, one might have to consider ν, λ with some components equal to zero. As such Λ (E) is easily obtained once E is obtained. It is a standing assumption of this thesis that Λ (E) is known.
- To proceed with the transformation, let  $n_i^T \in \mathbb{R}_{\geq 0}^p$ ,  $i = 1, \ldots, r$  be the row vectors of E, that is  $E = \begin{bmatrix} n_i^T \\ \vdots \\ n_r^T \end{bmatrix}$ . Then (3.4a), (3.4b) are  $k_1 a^{y_1^{(L)}} = \langle n_1, \lambda \rangle \qquad \qquad k_1 b^{y_1^{(L)}} = \langle n_1, \nu \rangle$   $\vdots \qquad \qquad \vdots$   $k_r a^{y_r^{(L)}} = \langle n_r, \lambda \rangle \qquad \qquad k_r b^{y_r^{(L)}} = \langle n_r, \nu \rangle$

As all terms are positive, application of ln to both sides of each equation is well justified (using for vectors  $v \in \mathbb{R}^n$  the abbreviation  $\ln v := (\ln v_1, \ldots, \ln v_n)^T$ ):

In short:

$$\ln k_i + \langle y_i^{(L)}, \ln a \rangle = \ln \langle n_i, \lambda \rangle \qquad \qquad \ln k_i + \langle y_i^{(L)}, \ln b \rangle = \ln \langle n_i, \nu \rangle, \ i = 1, \dots, r.$$

Subtracting equations in *a* from equations in *b* yields (using the definition  $\mu := \left(\ln \frac{b_1}{a_1}, \dots, \ln \frac{b_n}{a_n}\right)^T$ ):

$$\langle y_i^{(L)}, \mu \rangle = \ln \frac{\langle n_i, \nu \rangle}{\langle n_i, \lambda \rangle}, i = 1, \dots, r.$$
 (3.7)

Further note that the equations (3.7) are independent of the parameter vector k. These equations can be used to state a lemma connecting the existence of  $a, b \in \mathbb{R}^n_{>0}$  and  $k \in \mathbb{R}^r_{>0}$  to the existence of vectors  $\lambda, \nu, \in \mathbb{R}^p_{>0}$  and  $\mu \in \mathbb{R}^n$ . With the definition

$$\ln \frac{E\,\nu}{E\,\lambda} := \left(\ln \frac{\langle n_1, \nu \rangle}{\langle n_1, \lambda \rangle}, \dots, \ln \frac{\langle n_r, \nu \rangle}{\langle n_r, \lambda \rangle}\right)^T \tag{3.8a}$$

equation (3.7) can be written as

$$Y^{(L)T} \mu = \ln \frac{E\nu}{E\lambda}.$$
(3.8b)

**Lemma 1.** Let  $\dot{x}(t) = Y I_a v(k, x(t))$  be a set of ordinary differential equations derived from a biochemical reaction network endowed with mass action kinetics. Further let  $E = [E_1, \ldots, E_p]$ ,  $E \in \mathbb{R}_{\geq 0}^{r \times p}$  be a matrix, whose columns are generators of ker $(Y I_a) \cap \mathbb{R}_{\geq 0}^r$  and let  $\Lambda(E)$  be the set of all nonnegative vectors  $x \in \mathbb{R}_{\geq 0}^p$  such that E x > 0, as defined in (3.5). Then the following statements are equivalent:

(I) There exist two vectors  $a, b \in \mathbb{R}^n_{>0}$ ,  $a \neq b$  and a vector  $k \in \mathbb{R}^r_{>0}$  with

$$Y I_a v(k, a) = 0$$
  
$$Y I_a v(k, b) = 0$$

(II) There exist two vectors  $\lambda, \nu \in \Lambda(E)$  and a vector  $\mu \in \mathbb{R}^n$  with

$$Y^{(L)^T} \mu = \ln \frac{E \nu}{E \lambda}.$$

*Proof.* (I)  $\Rightarrow$  (II): Let  $a, b \in \mathbb{R}_{>0}^n$  with  $k \in \mathbb{R}_{>0}^r$  with

$$Y I_a v(k, a) = 0 \tag{3.9a}$$

$$Y I_a v(k, b) = 0.$$
 (3.9b)

To obtain  $\mu \in \mathbb{R}^n$  and  $\nu, \lambda \in \Lambda(E)$  proceed as above: (3.9a), (3.9b) and positivity of k imply v(k, a) > 0, v(k, b) > 0 and  $v(k, a) \in \ker(Y I_a) \cap \mathbb{R}^r_{\geq 0}$ ,  $v(k, b) \in \ker(Y I_a) \cap \mathbb{R}^r_{\geq 0}$ . Thus there exists  $\lambda, \nu \in \Lambda(E)$  such that

$$v(k, a) = E \lambda$$
  
 $v(k, b) = E \nu$ 

These equations can be transformed in the same process as described above:

$$Y^{(L)^T} \mu = \ln \frac{E \nu}{E \lambda},$$

where  $\mu := \ln \frac{b}{a}$ .

 $\begin{array}{l} (\underline{\mathrm{II}}) \Rightarrow (\underline{\mathrm{I}}): \text{Assume } \lambda, \nu \in \Lambda(E) \text{ and } \mu \in \mathbb{R}^n \text{ are given such that } {Y^{(L)}}^T \mu = \ln \frac{E\nu}{E\lambda} \text{ holds. Then one} \\ \text{has to show the existence of } a, b \in \mathbb{R}^n_{>0} \text{ and } k \in \mathbb{R}^r_{>0} \text{ such that } Y I_a v(k, a) = 0 \text{ and } Y I_a v(k, b) = 0. \\ \text{To this end choose any vector } a \in \mathbb{R}^n_{>0} \text{ and observe that } a \text{ is a steady state if a positive vector } k \\ \text{exists, such that } v(k, a) = E \lambda, \text{ where } \lambda \in \Lambda(E) \text{ is hnown by assumption. Recall that } v(k, a) = \\ \text{diag}(k) \phi(a) = \text{diag}(\phi(a)) k. \text{ Thus it is straightforward to obtain } k: \end{array}$ 

$$k = \operatorname{diag}\left(\phi\left(a^{-1}\right)\right) E \lambda$$

Then  $b = \text{diag}(e^{\mu}) a$  is a positive steady state as well, where  $\mu \in \mathbb{R}^n$  is fixed by assumption. To see this, one has to show that  $v(k, b) = E \nu$ , with k as above. Note that  $\nu \in \Lambda(E)$  is known by assumption. Recall, that, by assumption as well,  $\nu$ ,  $\lambda$  and  $\mu$  satisfy  $Y^{(L)T} \mu = \ln \frac{E \nu}{E \lambda}$ . Observe the following:

$$\begin{aligned} v(k, b) &= \operatorname{diag}\left(k\right) \phi\left(b\right) = \operatorname{diag}\left(\phi\left(a^{-1}\right)\right) \operatorname{diag}\left(E \lambda\right) \operatorname{diag}\left(\phi\left(e^{\mu}\right)\right) \phi\left(a\right) \\ &= \operatorname{diag}\left(\phi\left(e^{\mu}\right)\right) \operatorname{diag}\left(E \lambda\right) \underbrace{\operatorname{diag}\left(\phi\left(a^{-1}\right)\right) \phi\left(a\right)}_{=1} \\ &= \operatorname{diag}\left(\phi\left(e^{\mu}\right)\right) E \lambda \\ &= \operatorname{diag}\left(e^{Y^{(L)^{T}}\mu}\right) E \lambda \\ &= \operatorname{diag}\left(\frac{E \nu}{E \lambda}\right) E \lambda \\ &= E \nu. \end{aligned}$$

Solutions  $\mu \in \mathbb{R}^n$  and  $\nu, \lambda \in \Lambda(E)$  can be associated with the difference of two positive steady state solutions: if a satisfies  $Y I_a v(k, a) = 0$  (i.e.  $v(k, a) = E \lambda$ ) then  $b = \operatorname{diag}(e^{\mu}) a$  satisfies  $Y I_a v(k, b) = 0$  (i.e.  $v(k, b) = E \nu$ ). Note that, as is shown in the proof of Lemma 1, any positive vector  $a \in \mathbb{R}^n_{>0}$  can be a steady state solution. Simply solve  $v(k, a) = \operatorname{diag}(k) \phi(a)$  for the parameter vector  $k \in \mathbb{R}^r_{>0}$  (note that this equation is *linear in k*):

$$k = \operatorname{diag} \left( \phi \left( a^{-1} \right) \right) E \lambda.$$
 (3.10a)

Thus, a solution  $(\mu, \nu, \lambda)$  defines in fact a continuum of solutions parameterized by  $a \in \mathbb{R}^n_{>0}$ : for a given solution  $(\mu, \nu, \lambda)$  choose  $a \in \mathbb{R}^n_{>0}$  and fix k as in (3.10a). Then, by Lemma 1,

$$b = \operatorname{diag}\left(\operatorname{Exp}\left(\mu\right)\right) a \tag{3.10b}$$

is a steady state solution as well. However, a and b need not satisfy the conservation relations (3.1b) and (3.2b) for the same values  $c_i$ . To find pairs a and b that satisfy these constraints as well, see Section 3.3.

# **3.2** Solvability of $Y^{(L)^T} \mu = \ln \frac{E\nu}{E\lambda}$

In this Section (3.8b) is analysed. Note that

$$Y^{(L)}{}^T \mu = \ln \frac{E \nu}{E \lambda}$$

is solvable for given  $\nu$ ,  $\lambda$ , if and only if

$$U^T Y^{(L)T} = 0 \Rightarrow U^T \ln \frac{E\nu}{E\lambda} = 0.$$
(3.11)

This is a consequence of the fact that (3.8b) is only solvable, if the right hand side is contained in the image of  $Y^{(L)T}$ , that is, if  $\ln \frac{E\nu}{E\lambda} \in [Y^{(L)T}]$ . This is the case, if and only if the orthogonal complement of  $[Y^{(L)T}]$ ,  $[Y^{(L)T}]^{\perp}$ , is orthogonal to  $\ln \frac{E\nu}{E\lambda}$ . Note that by the 'fundamental theorem of linear algebra  $[Y^{(L)T}]^{\perp} = \ker(Y^{(L)})$  [61]. In Section 3.2.1 and Section 3.2.2 two solution strategies based on condition (3.11) are discussed. Recall that  $Y^{(L)} \in \mathbb{R}^{n \times r}$ , where *n* is the number of species and *r* the number of reactions. As, by assumption, r > n, condition (3.11) will exist for almost any biochemical reaction network.

#### 3.2.1 Solvability I

Let U be a basis for ker  $(Y^{(L)})$  with integer coefficients (such a basis can always be obtained by an adequate scaling of basis-vectors, as  $Y^{(L)} \in \mathbb{N}_0^{n \times r}$ ). Consider basis-vector  $u_i$  and let  $u_i = u_i^{\oplus} - u_i^{\ominus}$ , where  $u_i^{\oplus}$ ,  $u_i^{\ominus}$  are vectors with nonnegative integer coefficients. Then

$$u_i^T \ln \frac{E \nu}{E \lambda} = 0$$

evaluates to

$$\begin{aligned} u_i^{\oplus T} \ln \frac{E\nu}{E\lambda} &= u_i^{\oplus T} \ln \frac{E\nu}{E\lambda} \\ \sum_{j \in \text{supp}\left(u_i^{\oplus}\right)} u_{ij}^{\oplus} \ln \frac{\langle n_j, \nu \rangle}{\langle n_j, \lambda \rangle} &= \sum_{l \in \text{supp}\left(u_i^{\oplus}\right)} u_{il}^{\ominus} \ln \frac{\langle n_l, \nu \rangle}{\langle n_l, \lambda \rangle} \\ \sum_{j \in \text{supp}\left(u_i^{\oplus}\right)} \ln \left(\frac{\langle n_j, \nu \rangle}{\langle n_j, \lambda \rangle}\right)^{u_{ij}^{\oplus}} &= \sum_{l \in \text{supp}\left(u_i^{\oplus}\right)} \ln \left(\frac{\langle n_l, \nu \rangle}{\langle n_l, \lambda \rangle}\right)^{u_{il}^{\oplus}}. \end{aligned}$$

Using the summation rule for ln one obtains

$$\ln \prod_{j \in \text{supp}(u_i^{\oplus})} \left( \frac{\langle n_j, \nu \rangle}{\langle n_j, \lambda \rangle} \right)^{u_{ij}^{\oplus}} = \ln \prod_{l \in \text{supp}(u_i^{\ominus})} \left( \frac{\langle n_l, \nu \rangle}{\langle n_l, \lambda \rangle} \right)^{u_{il}^{\ominus}}$$

By removing ln (by taking both sides to  $e^{\cdot}$ ), multiplying with numerators and rearranging terms one finally obtains

$$\prod_{j\in\operatorname{supp}\left(u_{i}^{\oplus}\right)} (\langle n_{j},\nu\rangle)^{u_{ij}^{\oplus}} \prod_{l\in\operatorname{supp}\left(u_{i}^{\ominus}\right)} (\langle n_{l},\lambda\rangle)^{u_{il}^{\ominus}} - \prod_{l\in\operatorname{supp}\left(u_{i}^{\ominus}\right)} (\langle n_{l},\nu\rangle)^{u_{il}^{\oplus}} \prod_{j\in\operatorname{supp}\left(u_{i}^{\oplus}\right)} (\langle n_{j},\lambda\rangle)^{u_{ij}^{\oplus}} = 0,$$

$$i = 1, \dots, r - \operatorname{rank}\left(Y^{(L)}\right).$$
(3.12)

There are  $r - \operatorname{rank}(Y^{(L)})$  basis-vectors for ker  $(Y^{(L)})$  and thus  $r - \operatorname{rank}(Y^{(L)})$  polynomials of the form given in (3.12). If a solution to these polynomials exists, then it is straightforward to solve (3.8b) in terms of the  $\mu_i$ . To see this, let  $\tilde{\nu}, \tilde{\lambda} \in \mathbb{R}_{>0}^p$  be such that (3.12) holds. Then  $\ln \frac{E_{\tilde{\nu}}}{E_{\lambda}} \in \left[Y^{(L)T}\right]$ . Thus

$$Y^{(L)^T} \mu = \ln \frac{E \,\tilde{\nu}}{E \,\tilde{\lambda}}$$

is solvable and, as these equations are linear in  $\mu$ , it is straightforward to obtain  $\mu$ . In this case  $\mu = M \kappa$ , where M is a matrix of appropriate dimension and  $[M] \subseteq \mathbb{R}^n$  is a linear subspace of  $\mathbb{R}^n$ .

**Remark 2.** In this Chapter the question of the existence of more than one positive solution to the system of polynomials  $Y I_a v(k, x) = 0$  has been analysed. In general  $Y I_a v(k, x) = 0$  is a system of n polynomials with r > n unknown coefficients  $k_i$  in the n unknowns  $x_i$ . Lemma 1 and the preceding discussion show that this question is equivalent to the question of the existence of at least one solution  $\nu$ ,  $\lambda \in \Lambda(E)$  to a system of  $r - \operatorname{rank}(Y^{(L)})$  polynomials are easier to handle (in fact, it is possible to examine these polynomials mumerically, even though the rate constants  $k_i$  are unknown). However, even though the polynomials are now accessible numerically it will in general not be easy to find positive solutions, let alone all of them.

#### 3.2.1.1 Special form of polynomials

The matrix  $Y^{(L)}$  as described in Section 2.2 contains all reactant complexes of the network. If a complex is reactant in several reactions (say in  $\gamma$  reactions) then  $Y^{(L)}$  contains exactly  $\gamma$  copies of this complex vector. Then a basis for ker  $(Y^{(L)})$ , the kernel of  $Y^{(L)}$ , exists, that contains  $\gamma - 1$  vectors with exactly one entry +1, one entry -1 and the remaining entries 0. Let  $Y^{(R)} \in \mathbb{R}^{n \times nr}$  of reactant complexes be the matrix of reactant complexes (i.e. the matrix that contains the vectors of each reactant complexes exactly once). If this matrix is of full rank and the number of reactant complexes is less than n, the number of species, then ker  $(Y^{(L)})$  is spanned only by vectors of this form; this is, for example, the case for the networks related to the activation of an MAPK, see Chapter 4. Vectors of this form lead to a special system of polynomials that will be discussed here.

Let  $u_i = (0, ..., 1, 0, ..., -1, 0, ...)^T$  (with supp  $(u_i) = \{j, l\}$ ). Then (3.12) simplifies to

$$\langle n_i, \nu \rangle \langle n_l, \lambda \rangle - \langle n_l, \nu \rangle \langle n_i, \lambda \rangle = 0$$

as  $u_{ij}^{\oplus} = 1$  and  $u_{il}^{\ominus} = 1$ . This is equivalent to

$$\left(\sum_{s \in \operatorname{supp}(n_j)} n_{js} \,\nu_s\right) \,\left(\sum_{t \in \operatorname{supp}(n_t)} n_{lt} \,\lambda_t\right) - \left(\sum_{s \in \operatorname{supp}(n_j)} n_{js} \,\lambda_s\right) \,\left(\sum_{t \in \operatorname{supp}(n_l)} n_{lt} \,\nu_t\right) = 0.$$

Let  $\Gamma_i := \operatorname{supp}(n_j) \times \operatorname{supp}(n_l)$ , be the set of all ordered pairs of elements of  $\operatorname{supp}(n_j)$  and  $\operatorname{supp}(n_l)$ . Then

$$\left(\sum_{s \in \operatorname{supp}(n_j)} n_{js} \, \nu_s\right) \, \left(\sum_{t \in \operatorname{supp}(n_l)} n_{lt} \, \lambda_t\right) = \sum_{(s,t) \in \Gamma_i} n_{js} \, n_{lt} \, \nu_s \, \lambda_t$$

and

$$\left(\sum_{s \in \operatorname{supp}(n_j)} n_{js} \, \lambda_s\right) \, \left(\sum_{t \in \operatorname{supp}(n_l)} n_{lt} \, \nu_t\right) = \sum_{(s,t) \in \Gamma_i} n_{js} \, n_{lt} \, \nu_t \, \lambda_s.$$

This leads to the following special form for (3.12):

$$\langle n_j, \nu \rangle \langle n_l, \lambda \rangle - \langle n_l, \nu \rangle \langle n_j, \lambda \rangle = \sum_{(s,t) \in \Gamma_i} n_{js} n_{lt} \left( \nu_s \lambda_t - \nu_t \lambda_s \right).$$
(3.13)

Let  $\alpha_i = (n_{js} n_{lt})_{(s,t)\in\Gamma_i}$  and  $\beta_i(\nu, \lambda) = (\nu_s \lambda_t - \nu_t \lambda_s)_{(s,t)\in\Gamma_i}$ . Then (3.13) is equivalent to  $\langle \alpha_i, \beta_i(\nu, \lambda) \rangle = 0$ . Clearly each  $u_i$  of the form discussed here yields vectors  $\alpha_i$  and  $\beta_i$ . By collecting all binomials of the form  $\nu_s \lambda_t - \nu_t \lambda_s$  in a vector  $\beta(\nu, \lambda)$  and, after an appropriate filling with zeros, of all  $\alpha_i$  as row vectors of a matrix A one obtains the following compact representation:

$$A \beta (\nu, \lambda) = 0.$$
 (3.14)

#### 3.2.2 Solvability II

Instead of solving the polynomials associated with  $U^T \ln \frac{E\nu}{E\lambda} = 0$  as described in the previous section the equation  $Y^{(L)T} \mu = \ln \frac{E\nu}{E\lambda}$  is transformed in the following way: note that (3.8b) is equivalent to

$$\operatorname{diag}\left(e^{Y^{(L)^{T}}\,\mu}\right)\,E\,\lambda=E\,\nu$$

and, after rearranging terms,

$$\begin{bmatrix} E & -\operatorname{diag}\left(e^{Y^{(L)^{T}}\mu}\right) E\end{bmatrix} \begin{pmatrix} \nu\\ \lambda \end{pmatrix} = 0, \nu, \lambda \in \Lambda(E).$$
(3.15)

Recall that  $E \in \mathbb{R}_{\geq 0}^{r \times p}$ , r the number of reactions and note that in general r > p. Thus, for the remainder of Section 3.2.2 the following assumption will be made:

**Assumption 1.** The number of reactions r is greater than p the number of generators of ker $(Y I_a) \cap \mathbb{R}^r_{>0}$ .

Recall that  $\nu$ ,  $\lambda \in \Lambda(E)$ , as they are coordinates of points in the interior of a pointed polyhedral cone. Let let  $U \in \mathbb{R}^{r \times r}$  be a matrix such that UE is in row reduced echelon form:

$$UE = \begin{bmatrix} I_o & E_o \\ 0 & 0 \end{bmatrix},$$

where  $o = \operatorname{rank}(E)$  and  $\begin{bmatrix} E_0 \\ -I_{r-o} \end{bmatrix}$  is a basis for ker (E). The row reduced echelon form is a result of *Gauss-Jordan elimination* (after a possible reordering of variables). Thus, such a matrix U can always be obtained [61]. Let  $U = \begin{bmatrix} U_1 \\ U_2 \end{bmatrix}$  and multiply (3.15) by U to obtain

$$\underbrace{\begin{bmatrix} I_o & E_0 & -U_1 \operatorname{diag} \left( e^{Y^{(L)^T} \mu} \right) E \\ 0 & 0 & -U_2 \operatorname{diag} \left( e^{Y^{(L)^T} \mu} \right) E \\ =: Q(\mu) \end{bmatrix}}_{=:Q(\mu)} \begin{pmatrix} \nu \\ \lambda \end{pmatrix} = 0 , \nu, \lambda \in \Lambda(E) .$$
(3.16)

Observe the following facts:

(i) Equation (3.16) is feasible, only if  $Q_2(\mu) := U_2 \operatorname{diag}\left(e^{Y^{(L)^T}\mu}\right) E$  has a kernel vector  $\lambda \in \Lambda(E)$ , that is only if

$$Q_2(\mu) \lambda = 0, \lambda \in \Lambda(E)$$

is feasible for some  $\mu \in \mathbb{R}^n$ .

(ii) However  $\nu \in \Lambda(E)$  has to hold as well. Thus, in general, one has to consider the complete matrix  $Q(\mu)$ . There might be cases, where one must not consider all of  $Q(\mu)$ : if  $Q_1(\mu) := \begin{bmatrix} e_0 - U_1 \operatorname{diag}\left(e^{Y^{(L)T}\mu}\right)E\end{bmatrix}$  contains nonpositive rows. Let  $Q_1(\mu) = \begin{bmatrix} q_1(\mu) \\ \vdots \\ q_i(\mu) \end{bmatrix}$  and let  $\mathcal{I} \subseteq \{1, \ldots, o\}$  be the set of indices such that  $q_i(\mu) \leq 0$ ,  $q_i(\mu) \neq 0$ , for all  $\mu \in \mathbb{R}^n$ . That is,  $q_i(\mu)$  is a nonpositive row vector of  $Q_1(\mu)$ , not identically zero. Split  $\binom{\nu}{\lambda} = \binom{\overline{\nu}}{\nu_0}$  accordingly. Then

$$\tilde{\nu} = -\begin{bmatrix} E_0 & -U_1 \operatorname{diag} \left( e^{Y^{(L)T} \mu} \right) E \end{bmatrix} \begin{pmatrix} \nu_0 \\ \lambda \end{pmatrix} = -Q_1 \left( \mu \right) \begin{pmatrix} \nu_0 \\ \lambda \end{pmatrix}$$

Obviously, if  $q_i(\mu) \leq 0$ ,  $q_i(\mu) \neq 0$  then  $\tilde{\nu}_i = -q_i(\mu) \binom{\nu_0}{\lambda} > 0$ , if  $\binom{\nu_0}{\lambda} > 0$ . Thus the row  $q_i(\mu)$  can be excluded, as  $\tilde{\nu}_i$  will be positive, if  $\binom{\nu_0}{\lambda} > 0$ .

(iii) Suppose Q<sub>1</sub>(μ) as defined above contains a nonnegative row (for all μ ∈ ℝ<sup>n</sup>), that is, q<sub>j</sub>(μ) ≥ 0, for some j ∈ {1,..., o}. Then ν̃<sub>i</sub> = -q<sub>i</sub>(μ) (<sup>ν<sub>0</sub></sup><sub>λ</sub>) ≤ 0, if (<sup>ν<sub>0</sub></sup><sub>λ</sub>) > 0 and multistationarity is excluded, as by Lemma 1, the system cannot admit two positive steady state solutions.

As a consequence of the previous discussion, for the remainder of Section 3.2.2, the following assumption will be made:

Assumption 2. No row of  $Q_1(\mu)$  as defined above is nonnegative (for all  $\mu \in \mathbb{R}^n$ ).

As a consequence of and the above assumption, it suffices to consider those rows of  $Q_1(\mu)$  that contain positive and negative elements. Let  $\mathcal{I}^{\pm} \subseteq \{1, \ldots, o\}$  be the set of indices that correspond to rows of  $Q_1(\mu)$  containing positive and negative entries and let  $pm := |\mathcal{I}^{\pm}|$ . Then one has to examine the following submatrix of  $Q(\mu)$ :

$$\tilde{Q}\left(\mu\right) := \begin{bmatrix} I_{pm} & (q_{i}\left(\mu\right))_{i \in \mathcal{I}^{\pm}} \\ 0 & Q_{2}\left(\mu\right) \end{bmatrix}.$$

Note that this submatrix is defined mainly for convenience. The algorithms presented in the remainder of this Chapter depend on the size of this matrix. Thus a removal of those rows of  $Q(\mu)$  that always have a positive null vector can simplify the analysis. Thus, from now on,  $Q(\mu)$  will be used to denote the matrix, where all nonpositive rows of  $Q_1(\mu)$  gave been removed.

Further note that (3.16) is linear in  $\nu$  and  $\lambda$ . For fixed values of  $\mu$  the computation of  $\nu$ ,  $\lambda$  is straightforward, by determining the extreme rays of  $\ker(Q(\mu)) \cap \mathbb{R}_{\geq 0}^{2p}$ . However,  $\ker(Q(\mu)) \cap \mathbb{R}_{\geq 0}^{2p}$  might be the empty set for a particular  $\mu$ . In some cases it is known, that for any choice of  $\mu$  the cone  $\ker(Q(\mu)) \cap \mathbb{R}_{\geq 0}^{2p}$  has nonempty interior, that is there exists a positive kernel vector for  $Q(\mu)$ . This is the case, if  $Q(\mu)$  is a so-called  $L^+$ -matrix [44].

#### 3.2.2.1 L<sup>+</sup>-matrices matrices and sign-central matrices

In some cases it is possible to decide about the existence of positive kernel vectors by the sign pattern of a matrix. In many applications a system of linear equations of the form Q y = 0 has to be solved, where the entries  $Q_{ij}$  of Q are known only approximately, if at all. It is however frequently the case that the sign of the  $Q_{ij}$  is known. There exist certain matrices Q where the sign pattern of the solutions y to Q y = 0 can be determined by the sign pattern of Q (see e.g. [6] for a general introduction, [40] and [41] or [3] for nonnegative solutions and [44] for positive solutions). In the context of this section, the results of [44] and of [3] are of the most importance: in [44]  $L^+$ -matrices are introduced and in [3] sign-central matrices are introduced. If Q is an  $L^+$ -matrix, then every matrix with the same sign pattern has a positive kernel vector. If Q is a sign-central matrix, then every matrix with the same sign pattern has a nonnegative kernel vector.

To state the main results, some additional notation is necessary: let  $A \in \{-1, 0, 1\}^{m \times n}$  be a sign pattern. A nonzero vector  $\sigma \in \{-1, 0, 1\}^m$  is called a signing and a nonzero vector  $\sigma \in \{-1, 1\}^m$  is called a strict signing. Furthermore, in [44, 3], the 'requires' and 'allows' terminology is used: let P be a property that a matrix can or cannot have. Then the sign pattern A requires P if each matrix with the same sign pattern has property P, and A allows P if there exists some matrix with the same sign pattern with property P. Consider [44, Thereoem 2.4, p. 6]:

**Theorem 1.** Let A be an m by n sign pattern. Then the following are equivalent:

#### 3.2. SOLVABILITY OF $Y^{(L)T} \mu = \operatorname{LN} \frac{E \nu}{E \lambda}$

- (a) A is an L<sup>+</sup>-matrix.
- (b) A requires a positive null vector and A has no zero row.
- (c) For each signing  $\sigma$ , some column of diag ( $\sigma$ ) A is nonzero and nonnegative.
- (d) For each signing  $\sigma$ , some column of diag ( $\sigma$ ) A is nonzero and nonpositive.
- and [3, Theorem 2.1, p. 286]:

**Theorem 2.** Let A be an m by n sign pattern. Then the following are equivalent:

- (a) A is a sign-central matrix.
- (b) For every strict signing,  $\sigma$ , some column of diag ( $\sigma$ ) A is nonzero and nonnegative.
- (c) For each strict signing,  $\sigma$ , some column of diag ( $\sigma$ ) A is nonzero and nonpositive.

Recall that by definition a sign-central matrix has a nonnegative kernel vector [3, p. 284]. Note that Theorem 1 and 2 already contain primitive algorithms to determine whether or not a given sign pattern is an  $L^+$ - or a sign-central matrix, respectively.

In general  $Y^{(L)T}$   $\mu$  and thus the sign of the entries  $Q_{ij}$  of  $Q(\mu)$  is unknown, however, in certain cases, the signs of the  $Q_{ij}$  can be determined by systems of linear inequalities. In the remainder of this section it is in a first step shown under what conditions the sign-pattern of  $Q(\mu)$  can be determined by linear

inequality systems, before in a second step  $L^+$ -matrices are used to solve  $Q(\mu) \begin{pmatrix} \nu \\ \lambda \end{pmatrix} = 0, \nu, \lambda \in \Lambda(E).$ 

#### **3.2.2.2** Sign patterns of $Q(\mu)$ determined by linear inequalities

Consider

$$Q\left(\mu\right) = \left[ \begin{array}{cc} I_o & E_0 \\ 0 & 0 \end{array} \right. - U \operatorname{diag}\left(e^{Y^{(L)^T}\mu}\right) E \right]$$

as defined above. Split  $Q(\mu) = \begin{bmatrix} Q_f & -Q_v(\mu) \end{bmatrix}, Q_f \in \mathbb{R}_{>0}^{r \times p}, Q_v(\mu) \in \mathbb{R}^{r \times p}$ , where

$$Q_{f} := \begin{bmatrix} I_{o} & E_{0} \\ 0 & 0 \end{bmatrix}, Q_{v}(\mu) := U \operatorname{diag}\left(e^{Y^{(L)^{T}}\mu}\right) E.$$
(3.17)

Obviously sign  $(Q_f)$ , the sign pattern of  $Q_f$ , is fixed, while sign  $(Q_v(\mu))$ , the sign pattern of  $Q_v(\mu)$ , depends on the vector  $\mu$ . Under some conditions it is possible to determine all sign patterns sign  $(Q_v(\mu))$ , that  $Q_v(\mu)$  can admit, for any conceivable vector  $\mu \in \mathbb{R}^n$ . This is examined in this section. Consider  $Q_v(\mu) = U$  diag  $(e^{Y^{(L)T}\mu}) E$  and observe that

$$Q_{ij} = U_i^T \operatorname{diag}\left(e^{Y^{(L)^T}\mu}\right) E_j, \ i, j = 1, \dots, r,$$

where  $U_i^T$  is the i-th row vector of U and  $E_j$  the j-th column vector of E. Then

$$Q_{ij} = \sum_{l=1}^{r} e^{\langle y_l^{(L)}, \mu \rangle} U_{il} E_{jl}, \ i, j = 1, \dots, r.$$

Obviously  $e^{\langle y_l^{(L)}, \mu \rangle} U_{il} E_{jl} = 0$ , if  $U_{il} = 0$  or  $E_{jl} = 0$ . Thus

$$Q_{ij} = \sum_{l \in \operatorname{supp}(U_i^T) \cap \operatorname{supp}(E_j)} e^{\langle y_l^{(L)}, \mu \rangle} U_{il} E_{jl}.$$

Now assume  $\sup (U_i^T) \cap \sup (E_j) = \{s_{ij}, t_{ij}\}$ , that is  $\sup (U_i^T)$  and  $\sup (E_j)$  have only two elements in common. Then

$$Q_{ij} = e^{\langle y_{s_{ij}}^{(L)}, \mu \rangle} U_{is_{ij}} E_{js_{ij}} + e^{\langle y_{t_{ij}}^{(L)}, \mu \rangle} U_{it_{ij}} E_{jt_{ij}}.$$

If either

$$e^{\langle y_{s_{ij}}^{(L)},\,\mu\rangle}\,U_{is_{ij}}\,E_{js_{ij}}>0 \quad \text{and} \ e^{\langle y_{t_{ij}}^{(L)},\,\mu\rangle}\,U_{it_{ij}}\,E_{jt_{ij}}>0$$

or

$$e^{\langle y_{s_{ij}}^{(L)}, \mu \rangle} U_{is_{ij}} E_{js_{ij}} < 0 \text{ and } e^{\langle y_{t_{ij}}^{(L)}, \mu \rangle} U_{it_{ij}} E_{jt_{ij}} < 0$$

then trivially  $Q_{ij} > 0$  or  $Q_{ij} < 0$ , respectively. Thus assume

$$e^{\langle y_{s_{ij}}^{(L)}, \mu \rangle} U_{is_{ij}} E_{js_{ij}} > 0 \quad \text{and} \ e^{\langle y_{t_{ij}}^{(L)}, \mu \rangle} U_{it_{ij}} E_{jt_{ij}} < 0.$$

Then

$$\operatorname{sign}\left(Q_{ij}\right) = \begin{cases} > 0, \text{ if } e^{\langle y_{ij}^{(L)}, \mu \rangle} U_{is_{ij}} E_{js_{ij}} > e^{\langle y_{ij}^{(L)}, \mu \rangle} U_{it_{ij}} E_{jt_{ij}} \\ < 0, \text{ if } e^{\langle y_{ij}^{(L)}, \mu \rangle} U_{is_{ij}} E_{js_{ij}} < e^{\langle y_{ij}^{(L)}, \mu \rangle} U_{it_{ij}} E_{jt_{ij}} \\ = 0, \text{ if } e^{\langle y_{sij}^{(L)}, \mu \rangle} U_{is_{ij}} E_{js_{ij}} = e^{\langle y_{ij}^{(L)}, \mu \rangle} U_{it_{ij}} E_{jt_{ij}} \end{cases}$$
(3.18)

It is easy to see that the conditions in (3.18) are linear: simply apply  $\ln(\cdot)$  to obtain

$$\begin{aligned} Q_{ij} &< 0, \text{ if } \langle y_{s_{ij}}^{(L)}, \mu \rangle + \ln \left( U_{is_{ij}} E_{js_{ij}} \right) < \langle y_{t_{ij}}^{(L)}, \mu \rangle + \ln \left( U_{it_{ij}} E_{jt_{ij}} \right), \\ Q_{ij} &= 0, \text{ if } \langle y_{s_{ij}}^{(L)}, \mu \rangle + \ln \left( U_{is_{ij}} E_{js_{ij}} \right) = \langle y_{t_{ij}}^{(L)}, \mu \rangle + \ln \left( U_{it_{ij}} E_{jt_{ij}} \right), \\ Q_{ij} &> 0, \text{ if } \langle y_{s_{ij}}^{(L)}, \mu \rangle + \ln \left( U_{is_{ij}} E_{js_{ij}} \right) > \langle y_{t_{ij}}^{(L)}, \mu \rangle + \ln \left( U_{it_{ij}} E_{jt_{ij}} \right). \end{aligned}$$

Thus one obtains

$$\operatorname{sign}\left(Q_{ij}\right) = \operatorname{sign}\left(\langle v_{ij}, \, \mu \rangle + \beta_{ij}\right),\tag{3.19a}$$

where

$$v_{ij} := y_{s_{ij}}^{(L)} - y_{t_{ij}}^{(L)},$$
 (3.19b)

$$\beta_{ij} := \ln \frac{U_{is_{ij}} E_{is_{ij}}}{U_{is_{ij}} E_{is_{ij}}}.$$
(3.19c)

This motivates the following Lemma:

**Lemma 2.** Let  $U = \begin{bmatrix} U_1^T \\ \vdots \\ U_r^T \end{bmatrix}$  and  $E = \begin{bmatrix} E_1 \dots E_p \end{bmatrix}$ . Then the sign pattern of  $Q(\mu)$  as defined in (3.16)

can be determined by linear inequality systems if supp  $(U_i^T) \cap$  supp  $(E_j) \leq 2, \forall i, j \in 1, \ldots, r$ .

**Remark 3.** Note that if for  $Q_{ij}$  either all terms  $e^{\langle \cdot, \cdot \rangle}$  have the same sign or supp  $(U_i^T) \cap \text{supp}(E_j) \leq 1$ , it is straightforward to determine the sign of  $Q_{ij}$ .

Next determine all entries  $Q_{ij}$ , where the sign has to be determined by a linear inequality, that is define the index set

$$\mathcal{I} := \left\{ (i,j) \in \{1, \dots, r\} \times \{1, \dots, r\} | \operatorname{supp} (U_i) \cap \operatorname{supp} (E_j) = \{s_{ij}, t_{i,j}\} \\ \operatorname{and} (U_{is_{ij}} E_{js_{ij}}) (U_i t_{ij} E_{jt_{ij}}) < 0 \right\}.$$
(3.20)

Let  $s_0$  be the number of entries  $Q_{ij}$ , where the sign has to be determined by a linear inequality (i.e.  $s_0 = |\mathcal{I}|$ ) and collect all  $v_{ij}$  as defined in (3.19b) as row vectors in a matrix  $V \in \mathbb{R}^{s_0 \times n}$ :

$$V := \begin{bmatrix} v_{ij}^T \end{bmatrix}_{(i,j)\in\mathcal{I}}$$
(3.21a)

and all  $\beta_{ij}$  as defined in (3.19c) in a vector  $\beta \in \mathbb{R}^{s_0}$ :

$$\beta := (\beta_{ij})_{(i,j) \in I}$$
, (3.21b)

using  $\mathcal{I}$  as in (3.20). Note that one can determine all possible values of sign  $(V \mu + \beta)$ : let  $\sigma \in \{-1, 0, 1\}^{s_0}$  be a signing and observe that sign  $(V \mu + \beta) = \sigma$ , if and only if

$$V \mu + \beta - \operatorname{diag}\left(\sigma\right) \, s = 0, s > 0 \tag{3.22}$$

has a solution  $(\mu, s)$ , s > 0. Thus, by checking feasibility of (3.22), for every  $\sigma \in \{-1, 0, 1\}^{s_0}$ , one can determine all signings that sign  $(Q(\mu))$  can assume. A signing  $\sigma$ , such that (3.22) is feasible is called a *feasible signing*. Observe the following fact:

Fact 1. Suppose  $\beta_{ij} = 0, \forall (i, j) \in \mathcal{I}$ . In this case, if  $(\bar{\mu}, \bar{s})$  is solution to (3.22) for  $\sigma$ , then  $(-\bar{\mu}, \bar{s})$  is solution for  $-\sigma$ . Thus, if  $\sigma$  is a feasible signing for (3.22), then  $-\sigma$  a feasible signing as well.

To see this, suppose  $(\bar{\mu}, \bar{s})$  has been established as a solution to (3.22), that is

 $V \bar{\mu} - \text{diag}(\sigma) \bar{s} = 0$ 

holds. Observe that

 $V(-\bar{\mu}) - \operatorname{diag}(-\sigma) \,\bar{s} = -V\,\bar{\mu} + \operatorname{diag}(\sigma) \,\bar{s} = 0.$ 

The following algorithm can be used to determine all feasible signings:

#### Algorithm 1.

- (1) Let  $s_0$  be the number of inequalities. Create all signings  $\Sigma := \{-1, 0, 1\}^{s_0}$ .
- (2)  $\forall \sigma \in \Sigma$ :
  - (a) Solve

$$\begin{bmatrix} V & \beta & -\operatorname{diag} \sigma \end{bmatrix} \begin{pmatrix} \mu \\ \bar{s} \\ s \end{pmatrix} = 0, \quad s > 0, \bar{s} = 1.$$

(b) If  $(\mu, \bar{s}, s)$ , with  $\bar{s} = 1$ , s > 0 exists, then  $\sigma$  is a feasible signing.

**Remark 4.** For Algorithm 1 the system (3.22) has to be solved  $(3^{s_0} - 1)$  times (if the signing  $\sigma = 0$  is excluded). If  $\beta = 0$ , the system has only to be solved  $\frac{1}{2}(3^{s_0} - 1)$  times (as in this case  $(\mu, s)$  is a solution to (3.22) for  $\sigma$  implies by Fact 1 that  $(-\mu, s)$  is a solution to (3.22) for  $-\sigma$ ).

**Remark 5.** The existence of a signing such that the sign pattern of  $Q(\mu)$  is an  $L^+$ -matrix is necessary, but not sufficient for the existence of a positive solution to (3.16).

**Remark 6.** If one cannot establish signings, such that  $Q(\mu)$  is an  $L^+$ -matrix, it might still be possible to establish signings, such that  $Q(\mu)$  is a sign-central matrix. In this case some elements of  $\nu$  and  $\lambda$ and thus of the vector of rate constants k can equal zero. In this case one can try to find values for the zero elements of k using the results discussed in Section 3.4 (This approach has been successfully applied to one of the models for cell cycle regulation discussed in Chapter 5).

#### 3.2.2.3 An example: application to network $N_1$

As an illustration the matrix  $Q(\mu)$  for network  $\mathcal{N}_1$  is determined. Recall that for the example network

$$Y I_a = \begin{bmatrix} -1 & 1 & 0 & 0 & 1 & -1 \\ 0 & 0 & 1 & -1 & -1 & 1 \end{bmatrix}$$

The generators of ker $(Y I_a) \cap \mathbb{R}^6_{\geq 0}$  are given as the column vectors of the matrix

For  $\Lambda(E)$  as defined in (3.5) one obtains

$$\begin{split} \Lambda\left(E\right) &= \Big\{ x \in R_{\geq 0}^{\leq} \mid x_{1} = 0\&(x_{2} = 0\&(x_{3} = 0\&x_{4} > 0\&x_{5} > 0) \mid x_{3} > 0\&x_{4} > 0\&x_{5} > 0) \mid \\ & x_{2} > 0\&(x_{3} = 0\&x_{4} > 0\&x_{5} > 0) \mid x_{3} > 0\&x_{4} > 0\&x_{5} > 0) \mid \\ & x_{1} > 0\&(x_{2} = 0\&(x_{3} = 0\&x_{4} > 0\&x_{5} > 0) \mid x_{3} > 0\&x_{4} > 0\&x_{5} > 0) \mid \\ & x_{2} > 0\&(x_{3} = 0\&x_{4} > 0\&x_{5} > 0) \mid x_{3} > 0\&x_{4} \ge 0\&x_{5} \ge 0) \Big\}. \end{split}$$

And the transformation matrix U, with  $U\,E=\left[\begin{smallmatrix} I_o & E_o \\ 0 & 0 \end{smallmatrix}\right]$  is given by

$$U = \begin{bmatrix} 0 & 1 & 0 & 0 & 0 & 0 \\ 0 & 0 & 0 & 1 & 0 & 0 \\ 0 & 0 & 0 & 0 & 0 & 1 \\ 0 & 0 & 0 & 0 & 1 & -1 \\ 1 & -1 & 0 & 0 & -1 & 1 \\ 0 & 0 & 1 & -1 & -1 & 1 \end{bmatrix}$$

Therefore one obtains

$$\begin{aligned} Q_{\mathcal{N}_1} &= \begin{bmatrix} I_o & E_0 & -U \operatorname{diag} \left( e^{Y^{(L)^T} \mu} \right) E \end{bmatrix} \\ &= \begin{bmatrix} 1 & 0 & 0 & 0 & 1 & 1 \\ 0 & 1 & 0 & 0 & 1 & 0 & -e^{\mu_2} & 0 & 0 & -e^{\mu_2} \\ 0 & 0 & 1 & 0 & 1 & 0 & 0 & -e^{3\mu_1} & 0 & -e^{3\mu_1} \\ 0 & 0 & 0 & 1 & -1 & 0 & 0 & q_1 & -e^{2\mu_1+\mu_2} & e^{3\mu_1} \\ 0 & 0 & 0 & 0 & 0 & q_2 & 0 & -q_1 & q_3 & q_4 \\ 0 & 0 & 0 & 0 & 0 & 0 & 0 & q_5 & -q_1 & q_6 & q_7 \end{bmatrix}, \end{aligned}$$

with

$$q_{1} = e^{3\mu_{1}} - e^{2\mu_{1}+\mu_{2}}$$

$$q_{2} = 1 - e^{\mu_{1}}$$

$$q_{3} = -e^{\mu_{1}} + e^{2\mu_{1}+\mu_{2}}$$

$$q_{4} = 1 - e^{3\mu_{1}}$$

$$q_{5} = -1 + e^{\mu_{2}}$$

$$q_{6} = -1 + e^{2\mu_{1}+\mu_{2}}$$

$$q_{7} = -e^{3\mu_{1}} + e^{\mu_{2}}.$$

As every row of  $Q(\mu)$  contains positive and negative entries, no row can be discarded. To determine the sign pattern of  $Q(\mu)$ , the signs of the  $q_i$  have to be determined. This can be done by solving systems of linear inequalities:  $sign(q_1) = sign(\mu_1 - \mu_2)$ ,  $sign(q_2) = sign(-\mu_1)$ ,  $sign(q_3) = sign(\mu_1 + \mu_2)$ , .... Let  $q = (q_1, \ldots, q_7)^T$ . Then

$$\operatorname{sign}\left(q\right) = \operatorname{sign}\left(V\,\mu\right),$$

where

$$V = \begin{bmatrix} 1 & -1 \\ -1 & 0 \\ 1 & 1 \\ -3 & 0 \\ 0 & 1 \\ 2 & 1 \\ -3 & 1 \end{bmatrix}.$$

Let  $\sigma \in \{-1, 0, 1\}^7$ . The signing  $\sigma = (1, 1, 1, 1, 1, 1)^T$ , for example, corresponds to  $q_1 > 0, \ldots, q_7 > 0$ and thus  $V \mu > 0$ . Each signing defines an inequality system that is feasible if an only if the system

$$\begin{bmatrix} V & -\operatorname{diag}\left(\sigma\right) \end{bmatrix} \begin{pmatrix} \mu \\ s \end{pmatrix} = 0, s > 0 \tag{3.23}$$

is feasible. (Note that  $\beta = 0$  in this case.) Solutions to (3.23) form a pointed polyhedral cone, whose extreme rays can be computed by efficient algorithms. Let the columns of  $E = \begin{bmatrix} A_1 \\ A_2 \end{bmatrix}$  be generators of this cone (where  $A_1$  and  $A_2$  are matrices of appropriate dimension). Then  $\mu = A_1 \kappa$ ,  $\kappa > 0$  is a parametrization of all  $\mu$  that satisfy (3.23) for a particular signing.

The inequality system defined by  $\sigma = (1, 1, 1, 1, 1)^T$  has no solution, as the inequality  $-\mu_2 > 0$  (a consequence of the first two inequalities  $\mu_1 - \mu_2 > 0$  and  $-\mu_1 > 0$ ) is not compatible with the fifth inequality  $\mu_2 > 0$ . Thus the sign pattern defined by  $\sigma = (1, 1, 1, 1, 1)^T$ 

is not a feasible sign pattern for  $Q(\mu)$ . The sign pattern

is feasible. It corresponds to the signing  $\sigma^* = (-1, 1, -1, 1, 1, -1, 1)$  that defines the feasible inequality system

$$\mu_1 - \mu_2 < 0 \\ -\mu_1 > 0 \\ \mu_1 + \mu_2 < 0 \\ -3 \mu_1 > 0 \\ \mu_2 > 0 \\ 2 \mu_1 + \mu_2 < 0 \\ -3 \mu_1 + \mu_2 > 0$$

The pointed polyhedral cone defined by the corresponding equalities

$$\begin{bmatrix} V & -\operatorname{diag}\left(\sigma^*\right) \end{bmatrix} \binom{\mu}{s} = 0, s > 0 \tag{3.24}$$

is generated by

$$E_{\sigma^*} = \begin{bmatrix} -1 & -1 \\ 0 & 1 \\ \hline 1 & 2 \\ 1 & 1 \\ 1 & 0 \\ 3 & 3 \\ 0 & 1 \\ 2 & 1 \\ 3 & 4 \end{bmatrix}.$$

Thus all  $\mu = A_1 \kappa, \kappa > 0$ , with

$$A_1 = \begin{bmatrix} -1 & -1 \\ 0 & 1 \end{bmatrix}.$$
(3.25)

are solutions to the inequality system. Equation (3.24) is feasible for the following signings  $\sigma \in$ 

 $\{-1, 0, 1\}^7$ 

$\sigma_1 = (-1, -1)$	-1,	1,	-1,	1,	1,	$(-1)^{T}$	
$\sigma_2 = (-1, -1)$	-1,	1,	-1,	1,	1,	$(0)^{T}$	
$\sigma_3 = (-1, -1)$	-1,	1,	-1,	1,	1,	$(1)^{T}$	
$\sigma_4 = (-1, -1)$	1,	-1,	1,	-1,	-1,	$(1)^{T}$	
$\sigma_5 = (-1, -1)$	1,	-1,	1,	0,	-1,	$(1)^{T}$	
$\sigma_6 = (-1, -1)$	1,	-1,	1,	1,	-1,	$(1)^{T}$	
$\sigma_7 = (-1, -1)$	1,	0,	1,	1,	-1,	$(1)^{T}$	
$\sigma_8 = (-1, -1)$	1,	1,	1,	1,	-1,	$(1)^{T}$	
$\sigma_9 = (-1, -1)$	1,	1,	1,	1,	1,	$(1)^{T}$	
$\sigma_{10} = (0,$	-1,	1,	-1,	1,	1,	$(-1)^{T}$	(3.26)
$\sigma_{11} = (0,$	1,	-1,	1,	-1,	-1,	$(1)^{T}$	(3.20)
$\sigma_{12} = (1,$	-1,	-1,	-1,	-1,	-1,	$(-1)^{T}$	
$\sigma_{13} = (1,$	-1,	-1,	-1,	-1,	1,	$(-1)^{T}$	
$\sigma_{14} = (1,$	-1,	0,	-1,	-1,	1,	$(-1)^{T}$	
$\sigma_{15} = (1,$	-1,	1,	-1,	-1,	1,	$(-1)^{T}$	
$\sigma_{16} = (1,$	-1,	1,	-1,	0,	1,	$(-1)^{T}$	
$\sigma_{17} = (1,$	-1,	1,	-1,	1,	1,	$(-1)^{T}$	
$\sigma_{18} = (1,$	1,	-1,	1,	-1,	-1,	$(-1)^{T}$	
$\sigma_{19} = (1,$	1,	-1,	1,	-1,	-1,	$(0)^{T}_{-}$	
$\sigma_{20} = (1,$	1,	-1,	1,	-1,	-1,	$(1)^{T}$	

Of these twenty signings only the signings

$$\sigma_6 = (-1, 1, -1, 1, 1, -1, 1)^T \sigma_{15} = (1, -1, 1, -1, -1, 1, -1)^T$$
(3.27)

define sign patterns of  $Q(\mu)$  that are  $L^+$ -matrices. This is easy to verify using Theorem 1. The signings given in (3.27) correspond to the following sign patterns of  $Q(\mu)$ :

for  $\sigma_6$  and

for  $\sigma_{15}$ . Consider  $\sigma_6$ , the solutions to the corresponding inequality systems are given above as  $\mu_1 = -\kappa_1 - \kappa_2$ ,  $\mu_2 = \kappa_2$ ,  $\kappa_1, \kappa_2 > 0$ . Choose, for example,  $\kappa_1 = \kappa_2 = 1$ . Then  $\mu = (-2, 1)^T$  and

$$Q\left(-2,1\right) = \begin{bmatrix} 1 & 0 & 0 & 0 & 1 & -1 & 0 & 0 & 0 & 0 & -1 \\ 0 & 1 & 0 & 0 & 1 & 0 & -e & 0 & 0 & -e \\ 0 & 0 & 1 & 0 & 1 & 0 & 0 & -\frac{1}{e^6} & 0 & -\frac{1}{e^6} \\ 0 & 0 & 0 & 1 & -1 & 0 & 0 & \frac{1}{e^6} - \frac{1}{e^3} & -\frac{1}{e^3} & -\frac{1}{e^3} & -\frac{1}{e^6} \\ 0 & 0 & 0 & 0 & 0 & 1 - \frac{1}{e^2} & 0 & -\frac{1}{e^6} + \frac{1}{e^3} & \frac{1}{e^3} - \frac{1}{e^2} & 1 - \frac{1}{e^6} \\ 0 & 0 & 0 & 0 & 0 & 0 & -1 + e & -\frac{1}{e^6} + \frac{1}{e^3} & -1 + \frac{1}{e^3} - \frac{1}{e^6} + e \end{bmatrix}$$

The cone ker  $(Q(-2,1)) \cap \mathbb{R}^{10}_{>0}$  is generated by

```
4.0342879e + 02 \quad 0.0000000e + 00 \quad 1.0000000e + 00
                                                   4.0242879e + 02 \quad 0.0000000e + 00
                                                                                       0.0000000e + 00
6.4346646e + 03 3.1936309e + 02 1.5193499e + 01 6.4336646e + 03 3.1936309e + 02
                                                                                       3.0516959e + 02
1.0000000e + 00 \quad 0.0000000e + 00
                                  0.0000000e + 00 0.0000000e + 00 0.0000000e + 00
                                                                                       1.0000000e + 00
2.3320418e + 02 3.1696090e + 01
                                  5.0321472e - 01 2.3420418e + 02 3.1696090e + 01
                                                                                       3.0192875e + 01
0.0000000e + 00 \quad 1.0000000e + 00
                                  0.0000000e + 00
                                                    1.0000000e + 00
                                                                     1.0000000e + 00
                                                                                       0.0000000e + 00
0.0000000e + 00 \quad 0.0000000e + 00
                                  1.0000000e + 00
                                                    0.0000000e + 00
                                                                     1.0000000e + 00
                                                                                       0.0000000e \pm 00
1.9637520e + 03
                 1.1685499e + 02
                                  5.5893760e + 00
                                                    1.9637520e + 03
                                                                      1.1785499e + 02
                                                                                       1.1226562e + 02
0.0000000e + 00
                 4.0242879e + 02
                                  0.0000000e + 00
                                                    0.0000000e + 00
                                                                      4.0342879e + 02
                                                                                       4.0342879e + 02
4.7041168e + 03 2.3420418e + 02
                                   1.0107338e + 01
                                                    4.7041168e + 03
                                                                      2.3320418e + 02
                                                                                       2.2309685e + 02
4.0342879e + 02
                1.0000000e + 00
                                  0.0000000e + 00
                                                   4.0342879e + 02 \quad 0.0000000e + 00
                                                                                       0.0000000e + 00
```

A solution for  $\nu$ ,  $\lambda$  is

$$\nu = (8.0685759e + 01, 1.3827419e + 03, 2.000000e - 01, 5.6149664e + 01, 3.000000e - 01)^{T} (3.28a)$$
  

$$\lambda = (2.0000000e - 01, 4.2800691e + 02, 1.2092864e + 02, 1.0108846e + 03, 8.0785759e + 01)^{T} (3.28b)$$
  
(3.28b)

A pair of steady states is given by a = (1, 1) and  $b = (e^{-2}, e^{1})$ . The corresponding parameter vector is given by

$$k = (1.0110846e + 03, 8.0985759e + 01, 1.4388915e + 03, 5.0879266e + 02, 1.1318133e + 03, 2.0171440e + 02)^{T}.$$
(3.28c)

#### 3.3 Positive solutions satisfying the conservation relations

Recall the present situation: the goal is to find two positive steady state solutions  $a, b \in \mathbb{R}^{n}_{>0}$  and a parameter vector  $k \in \mathbb{R}^{r}_{>0}$ , such that  $Y I_{a} v(k, a) = Y I_{a} v(k, b) = 0$  and  $W^{T} a = W^{T} b$ . As a result of Section 3.1, it is known how  $a, b \in \mathbb{R}^{n}_{>0}$  and  $k \in \mathbb{R}^{r}_{>0}$  can be established: by obtaining solutions to

$$Y^{(L)^{T}} \mu = \ln \frac{E \lambda}{E \nu},$$

where  $\nu, \lambda \in \Lambda(E)$  and  $\mu \in \mathbb{R}^n$ . If a solution  $\nu, \lambda, \mu$  exists, then

$$a \in \mathbb{R}^{n}_{>0}, \text{ arbitrary}$$
$$b = \operatorname{diag}(e^{\mu}) a$$
$$k = \operatorname{diag}(\phi(a^{-1})) E \lambda$$

satisfy  $Y I_a v(k, a) = Y I_a v(k, b) = 0$ . However, the condition  $W^T a = W^T b$  will in general not hold for arbitrary vectors a. But, as a can be chosen freely,  $W^T a = W^T b$  can be checked independent of the rate constants k. Note that  $W^T a = W^T b$  is equivalent to  $W^T (b - a)$  and thus to  $b - a \in S$ , where  $S = [Y I_a]$ . Let  $M_1$  and  $M_2$  be two subsets of  $\mathbb{R}^n$ , not necessarily linear subspaces. The following lemma gives necessary and sufficient conditions for the existence of two positive vectors  $p \in \mathbb{R}^n_{>0}$  and  $q \in \mathbb{R}^n_{>0}$  with the following properties:

$$\ln \frac{q}{p} \in M_1, \tag{3.29a}$$

where, as usual,  $\ln \frac{q}{p} := \left(\ln \frac{q_1}{p_1}, \dots, \ln \frac{q_n}{p_n}\right)^T$  and

$$q - p \in M_2$$
. (3.29b)

The following Lemma formalizes the discussion in [25], or [23, p. 187-188]:

**Lemma 3.** Let  $M_1 \subseteq \mathbb{R}^n$  and  $M_2 \subseteq \mathbb{R}^n$  be two nontrivial subsets of  $\mathbb{R}^n$  and define  $M_3 := \{ (m_1, m_2) \in M_1 \times M_2 | \operatorname{sign}(m_1) = \operatorname{sign}(m_2) \}$  as the set of all ordered pairs  $(m_1, m_2)$  of elements  $m_1 \in M_1$  and

 $m_2 \in M_2$  with the same sign pattern. Two positive vectors p and q with the properties given in (3.29a) and (3.29b) exist, if and only if  $M_3 \neq \emptyset$ . Moreover, any pair  $(m_1, m_2) \in M_3$  defines a pair p, q:

$$(p_i)_{i=1,\dots,n} = \begin{cases} \frac{m_{2i}}{e^{m_{1i}}-1}, \text{ if } m_{1i} \neq 0\\ \bar{p}_i > 0, \text{ arbitrary, if } m_{1i} = 0 \end{cases}$$
(3.30a)

and

$$(q_i)_{i=1,\dots,n} = e^{m_{1i}} p_i. \tag{3.30b}$$

Proof. Suppose p and q with the properties given in (3.29a) and (3.29b) exist. One has to show that  $M_3 \neq \emptyset$ . As  $\ln \frac{q_i}{p_i} < 0$  iff  $q_i - p_i < 0$ ,  $\ln \frac{q_i}{p_i} > 0$  iff  $q_i - p_i > 0$  and  $\ln \frac{q_i}{p_i} = 0$  iff  $q_i - p_i = 0$  one has sign  $\left(\ln \frac{q}{p}\right) = \text{sign}(q-p)$ . As, by assumption,  $\ln \frac{q}{p} \in M_1$  and  $q-p \in M_2$  the desired result  $M_3 \neq \emptyset$  follows.

Suppose, on the other hand,  $M_3 \neq \emptyset$ . Then it remains to show that p and q as defined in (3.30a) and (3.30b) have the properties defined in (3.29a) and (3.29b). Pick any pair  $(m_1, m_2) \in M_3$  and suppose  $m_{1i} \neq 0, i = 1, ..., n$ . Then

$$q_i - p_i = e^{m_{1i}} p_i - p_i = p_i \ (e^{m_{1i}} - 1) = m_{2i}$$
 and  $\ln \frac{q_i}{p_i} = m_{1i}$ .

Otherwise, if  $m_{1i} = 0$  for some  $i \in \{1, \ldots, n\}$ , then  $q_i = p_i = \bar{p}_i$ . Moreover,  $\operatorname{sign}(m_1) = \operatorname{sign}(m_2)$  ensures  $m_{2i} = q_i - p_i = 0$  and  $\ln \frac{q_i}{p_i} = m_{1i} = 0$ .

Thus  $q - p = m_2 \in M_2$  and  $\frac{q}{p} = m_1 \in M_1$ , by definition.  $p \in \mathbb{R}^n_{>0}$  follows from sign  $(m_2) = \text{sign}(m_1)$  and the fact that sign  $(e^{m_{1i}} - 1) = \text{sign}(m_{1i}), i = 1, ..., n$ , positivity of q follows from positivity of p.

As an easy consequence of Lemma 3 necessary and sufficient conditions for the existence of multiple positive steady state solutions can be derived:

**Theorem 3.** Consider a biochemical reaction network with mass action kinetics and the matrices  $Y \in \mathbb{R}^{n \times m}$ ,  $I_a \in \mathbb{R}^{m \times r}$  and  $Y^{(L)} \in \mathbb{R}^{n \times r}$  as defined in Chapter 2. Let the columns of  $E \in \mathbb{R}^{r \times p}$  be generators of ker $(Y I_a) \cap \mathbb{R}^r_{>0}$ . Obtain

$$\mathcal{M} = \left\{ \mu \in \mathbb{R}^n | \exists \nu, \lambda \in \Lambda(E) \text{ such that } Y^{(L)^T} \mu = \ln \frac{E \nu}{E \lambda} \right\}.$$
(3.31)

Let  $S = [Y I_a]$  denote the stoichiometric subspace and define the set of all ordered pairs  $(\mu, v)$  of elements  $\mu \in \mathcal{M}$  and  $v \in S$  with the same sign pattern:

$$\mathcal{T} := \{(\mu, v) \in \mathcal{M} \times \mathcal{S} | \operatorname{sign}(\mu) = \operatorname{sign}(v) \}.$$
(3.32)

The ODEs derived from the biochemical reaction network admit multistationarity, if and only if  $\mathcal{T} \neq \emptyset$ . Moreover, any element of T defines a pair of steady state solutions  $a, b \in \mathbb{R}^n_{\geq 0}$  and a vector of rate constants  $k \in \mathbb{R}^r_{\geq 0}$  with

$$Y I_a v(k, a) = Y I_a v(k, b) = 0$$
$$W^T b = W^T a$$

in the following way:

$$(a_i)_{i=1,\dots,n} = \begin{cases} \frac{v_i}{v_{i-1}}, & \text{if } \mu_i \neq 0\\ \bar{a}_i > 0, & \text{arbitrary, if } \mu_i = 0 \end{cases}$$
(3.33a)

$$b = \operatorname{diag}\left(e^{\mu}\right) a \tag{3.33b}$$

$$k = \operatorname{diag}\left(\phi\left(a^{-1}\right)\right) E \lambda, \tag{3.33c}$$

for some  $\lambda \in \Lambda(E)$ , such that  $Y^{(L)^T} \mu = \ln \frac{E\nu}{E\lambda}$ .

#### Proof.

- ⇒ Suppose  $\mathcal{T} \neq \emptyset$ . Therefore  $\mathcal{M} \neq \emptyset$ . Then there exist  $\mu \in \mathbb{R}^n$  and  $\nu, \lambda \in \Lambda(E)$  such that  $Y^{(L)T} \mu = \ln \frac{E\nu}{E\lambda}$ . Thus, by Lemma 1, for arbitrary positive a, vectors  $b \in \mathbb{R}^n_{>0}$  as in (3.33b) and  $k \in \mathbb{R}^r_{>0}$  as in (3.33c) satisfy  $Y I_a v(k, a) = Y I_a v(k, b) = 0$ . As  $\mathcal{T} \neq 0$ , Lemma 3 implies that  $b a \in S$  and thus  $W^T a = W^T b$ , if a as in (3.33a).
- $\leftarrow \text{ Suppose } a, b \in \mathbb{R}_{>0}^n \text{ and } k \in \mathbb{R}_{>0}^r \text{ exist with } Y I_a \operatorname{diag}(k) \phi(a) = Y I_a \operatorname{diag}(k) \phi(b) = 0 \text{ and } W^T a = W^T b. \text{ As } a, b > 0, \text{ the assumption } Y I_a \operatorname{diag}(k) \phi(a) = Y I_a \operatorname{diag}(k) \phi(b) = 0 \text{ implies } \operatorname{diag}(k) \phi(a) = E \lambda \text{ and } \operatorname{diag}(k) \phi(b) = E \nu, \text{ for some } \nu, \lambda \in \Lambda(E). \text{ From Lemma 1 follows that } \mu := \ln \frac{b}{a} \text{ satisfies } Y^{(L)^T} \mu = \ln \frac{E \nu}{E \lambda}, \text{ for } \nu, \lambda \text{ as above. Thus } \mathcal{M} \neq \emptyset. \text{ Observe that } W^T a = W^T b \text{ implies } b a \in \mathcal{S} \text{ and thus } \mathcal{T} \neq \emptyset \text{ by Lemma 3. }$

**Corollary 1.** Suppose dim (S) = n, that is, the system has no conservation relations. Then the ODEs derived from the biochemical reaction network admit multistationarity, if and only if  $\mathcal{M} \neq \emptyset$ , with  $\mathcal{M}$  as defined in (3.31). Any vector  $\mu \in \mathcal{M}$  defines  $a, b \in \mathbb{R}^n_{>0}$  and  $k \in \mathbb{R}^r_{>0}$  with

$$Y I_a v(k, a) = Y I_a v(k, b) = 0$$

in the following way:

$$a \in \mathbb{R}_{>0}, arbitrary$$
$$b = \operatorname{diag}(e^{\mu}) a$$
$$k = \operatorname{diag}(\phi (a^{-1})) E \lambda,$$

for some  $\lambda \in \Lambda(E)$ , such that  $Y^{(L)^{T}} \mu = \ln \frac{E\nu}{E\lambda}$ .

Moreover, based on Lemma 3, it is, at least in principle, possible to prove that a certain network structure cannot admit multiple (positive) steady state solutions. This motivates the following corollary:

**Corollary 2.** Suppose  $\mathcal{M} \neq \emptyset$ ,  $\mathcal{M}$  as in (3.31) and  $\mathcal{T} = \emptyset$ ,  $\mathcal{T}$  as in (3.32). Then multistationarity is **impossible** (in the sense, that no pair  $a, b \in \mathbb{R}_{>0}^n$  of positive steady state solutions satisfies  $W^T b = W^T a$ ).

Results similar to Lemma 3 have been known in the literature on Feinberg's Chemical Reaction Network Theory for some time (see, for example, the informal discussion in Elison's Ph.D. thesis [23, p. 187-188]). In fact, both the Deficiency One Algorithm and the Advanced Deficiency Algorithm utilize the way p and q are constructed in (3.30a) and (3.30b). Note that multistationarity does not depend on specific parameter values. This motivates the following fact.

Fact 2. For a biochemical reaction network with mass action kinetics multistationarity is a **network** property, that is, it can be tested without any knowledge of parameter values whatsoever. Conclusiveness of a test depends on the existence of vectors  $\nu$ ,  $\lambda \in \Lambda(E)$  and a vector  $\mu$  that satisfy the nonlinear equation (3.8b) and a vector  $v \in [Y I_a]$  with sign  $(v) = \text{sign}(\mu)$ .

#### 3.3.1 The linear subspace case

Suppose that  $\mathcal{M}$  as given in (3.31) defines a linear subspace of  $\mathbb{R}^n$ . In this case it is straightforward (but may be computationally demanding) to check whether or not  $\mathcal{T} \neq \emptyset$  (i.e. to check whether or not there exists  $\mu \in \mathcal{M}$  and  $v \in \mathcal{S}$  with  $\operatorname{sign}(\mu) = \operatorname{sign}(v)$ ): to see this, suppose, for a particular  $\mu \in \mathcal{M}$ , a vector  $v \in \mathcal{S}$  with  $\operatorname{sign}(v) = \operatorname{sign}(\mu)$  can be determined. Then all vectors  $\alpha v$  with  $\alpha > 0$  also satisfy  $\operatorname{sign}(v) = \operatorname{sign}(\mu)$ . In fact, every vector  $v \in \mathcal{S}$  that is contained in the same orthant as  $\mu$  also satisfies  $\operatorname{sign}(v) = \operatorname{sign}(\mu)$ . And vice versa: suppose for a given vector  $v \in \mathcal{S}$  there exists a  $\mu \in \mathcal{M}$  with  $\operatorname{sign}(\mu) = \operatorname{sign}(v)$ . Then every vector  $\mu \in \mathcal{M}$  that is contained in the same orthant as v also satisfies  $\operatorname{sign}(\mu) = \operatorname{sign}(v)$ . Thus, to show that  $\mathcal{T} \neq \emptyset$ , it suffices to find an orthant that intersects  $\mathcal{M}$  as well as  $\mathcal{S}$  (provided that  $\mathcal{M}$  is a linear subspace of  $\mathbb{R}^n$ ).

To formalize this discussion, let  $\delta \in \{-1, 0, 1\}^n$  be a vector composed of entries +1, 0 and -1. It is used to denote a particular orthant  $\mathbb{R}^n_{\delta} := \{x \in \mathbb{R}^n | \delta = \operatorname{sign}(x), j = 1, \ldots, n\}$ , with  $\delta$  the

signature of the orthant. In this notation the positive orthant, for example, is represented by the vector  $\delta = (1, \ldots, 1)^T$ , entirely composed of entries +1, and  $\mathbb{R}^n_{\delta}$ ,  $\delta = (0, 1, \ldots, 1)^T$  is the part of the hyperplane  $x_1 = 0$  separating  $\mathbb{R}^n_{\delta}$  with  $\bar{\delta} = (1, \ldots, 1)^T$  from  $\mathbb{R}^n_{\delta}$  with  $\tilde{\delta} = (-1, 1, \ldots, 1)^T$ . Using this notation the previous discussion can be summarized as follows:

Fact 3. Recall that S is a linear subspace of  $\mathbb{R}^n$ . Suppose  $\mathcal{M} \subseteq \mathbb{R}^n$  as given in (3.31) is a linear subspace of  $\mathbb{R}^n$ . Then  $\mathcal{T}$  as defined in (3.32), has  $\mathcal{T} \neq \emptyset$ , if and only if there exists an orthant  $\mathbb{R}^n_{\delta}$  as defined above with  $\mathcal{M} \cap \mathbb{R}^n_{\delta} \neq \emptyset$  and  $S \cap \mathbb{R}^n_{\delta} \neq \emptyset$ . Note that in this case  $\mathcal{M} \cap \mathbb{R}^n_{\delta_o} \neq \emptyset$  and  $S \cap \mathbb{R}^n_{\delta_o} \neq \emptyset$ , with  $\delta_o = -\delta$  as well.

Proof. Recall that  $\mathcal{T} \neq \emptyset$ , if there exists  $v \in \mathcal{S}$  and  $\mu \in \mathcal{M}$  with sign  $(v) = \text{sign}(\mu) = \delta$ , which implies  $\mathcal{M} \cap \mathbb{R}^n_{\delta} \neq \emptyset$  and  $\mathcal{S} \cap \mathbb{R}^n_{\delta} \neq \emptyset$ . As  $\mathcal{S}$  and  $\mathcal{M}$  are linear vector spaces,  $-v \in \mathcal{S}$  and  $-\mu \in \mathcal{M}$  holds. Obviously sign  $(-v) = \text{sign}(-\mu) = -\delta =: \delta_o$ . Thus  $\mathcal{M} \cap \mathbb{R}^n_{\delta_o} \neq \emptyset$  and  $\mathcal{S} \cap \mathbb{R}^n_{\delta_o} \neq \emptyset$ .

Note that a, b and k can be recast using the cones  $\mathcal{M} \cap \mathbb{R}^n_{\delta}$  and  $\mathcal{S} \cap \mathbb{R}^n_{\delta}$ . Let  $\Delta := \{\delta \in \{-1, 0, 1\}^n | \mathcal{M} \cap \mathbb{R}^n_{\delta} \neq \emptyset$  and  $\mathcal{S} \cap \mathbb{R}^n_{\delta} \neq \emptyset$ . Using elements  $\delta \in \Delta$ , a representation of  $a^{\delta}, b^{\delta}, k^{\delta}$  can be derived in terms of the generators of  $\mathcal{M} \cap \mathbb{R}^n_{\delta}$  and  $\mathcal{S} \cap \mathbb{R}^n_{\delta}$ : let  $E_1^{S^{\delta}}, \dots, E_{p_{S^{\delta}}}^{S^{\delta}}$  be a set of generators for the cone  $\mathcal{S} \cap \mathbb{R}^n_{\delta}$  and define the matrix  $E^{S^{\delta}} = \left[E_1^{S^{\delta}}, \dots, E_{p_{S^{\delta}}}^{S^{\delta}}\right]$ . Further let  $E_1^{\mathcal{M}^{\delta}}, \dots, E_{p_{\mathcal{M}^{\delta}}}^{\mathcal{M}^{\delta}}$  be a set of generators for the cone  $\mathcal{M} \cap \mathbb{R}^n_{\delta}$  and define the matrix  $E^{\mathcal{M}^{\delta}} = \left[E_1^{\mathcal{M}^{\delta}}, \dots, E_{p_{\mathcal{M}^{\delta}}}^{\mathcal{M}^{\delta}}\right]$ . Then, for a particular orthant  $R^n_{\delta}$ , all  $\mu \in \mathcal{M} \cap \mathbb{R}^n_{\delta}$ , can be represented by

$$\mu^{\delta} = E^{\mathcal{M}^{\delta}} \alpha^{\delta}, \ \alpha^{\delta} \in \mathbb{R}^{p_{\mathcal{M}^{\delta}}}_{>0}, \tag{3.34a}$$

and all  $v \in S \cap \mathbb{R}^n_{\delta}$  can be represented by

$$v^{\delta} = E^{S^{\delta}} \beta^{\delta}, \ \beta^{\delta} \in \mathbb{R}^{p_{S^{\delta}}}_{>0}.$$
 (3.34b)

Using (3.34a) and (3.34b) in (3.33a) a can be parametrized in terms of  $\alpha^{\delta}$  and  $\beta^{\delta}$ . The symbol  $a^{\delta}$  is used to denote this parametrization of a for each  $\delta \in \Delta$ . Using  $a^{\delta}$  in (3.33b) and (3.33c) yields parametrizations of  $b^{\delta}$  and  $k^{\delta}$  in terms of  $\alpha^{\delta}$  and  $\beta^{\delta}$ . Thus, for each orthant  $\delta \in \Delta$  a different representation of a, b and k can be derived.

It seems worthwhile to discuss the following property of  $\mathcal{M}$  and S: whenever  $v \in \mathcal{S}$   $(\mu \in \mathcal{M})$  holds,  $-v \in \mathcal{S}$   $(-\mu \in \mathcal{M})$  holds as well. In terms of orthants this is equivalent to the fact, that whenever  $\delta \in \Delta$ , then  $-\delta \in \Delta$  as well. It is sufficient to consider either  $\delta$  or  $-\delta$ , as all representations  $a^{-\delta}$ ,  $b^{-\delta}$ ,  $k^{-\delta}$  can be obtained from representations  $a^{\delta}$ ,  $b^{\delta}$ ,  $k^{\delta}$ . To be more precise, let  $\mu_1^{\delta} \in \mathcal{M} \cap \mathbb{R}^n_{\delta}$ ,  $v_1^{\delta} \in \mathcal{S} \cap \mathbb{R}^n_{\delta}$ and  $\mu_2^{-\delta} \in \mathcal{M} \cap \mathbb{R}^n_{-\delta}$ ,  $v_2^{-\delta} \in \mathcal{S} \cap \mathbb{R}^n_{-\delta}$  with  $\mu_2^{-\delta} = -\mu_1^{\delta}$  and  $v_2^{-\delta} = -v_1^{\delta}$ . Without loss of generality assume  $\mu_{1i}^{\delta} \neq 0$ ,  $\mu_{2i}^{-\delta} \neq 0$ ,  $v_{1i}^{-\delta} \neq 0$ ,  $v_{2i}^{-\delta} \neq 0$ ,  $i = 1, \ldots, n$ . Then  $a_{1i}^{\delta} = \frac{v_{1i}^{\delta}}{e^{\mu_{1i}^{\delta}-1}}$  and  $b_{1i} = e^{\mu_{1i}^{\delta}} a_{1i}$ . Using  $v_{2i}^{-\delta} = -v_{1i}^{\delta}$  and  $\mu_{2i}^{-\delta} = -\mu_{1i}^{\delta}$  in  $a_{2i}^{-\delta}$  yields

$$a_{2i}^{-\delta} = \frac{v_{2i}^{-\delta}}{e^{\mu_{2i}^{-\delta}} - 1} = \frac{-v_{1i}^{\delta}}{e^{-\mu_{1i}^{\delta}} - 1} = \frac{-v_{1i}^{\delta}}{e^{-\mu_{1i}^{\delta}} \left(1 - e^{\mu_{1i}^{\delta}}\right)} = \frac{1}{e^{-\mu_{1i}^{\delta}}} \frac{v_{1i}^{\delta}}{e^{\mu_{1i}^{\delta}} - 1} = e^{\mu_{1i}^{\delta}} \frac{v_{1i}^{\delta}}{e^{\mu_{1i}^{\delta}} - 1} = b_{1i}^{\delta}.$$

In a similar way  $b_{2i}^{-\delta} = a_{1i}^{\delta}$  can be derived. As k is determined by a, all representations  $a^{-\delta}$ ,  $b^{-\delta}$ ,  $k^{-\delta}$  can be derived from  $a^{\delta}$ ,  $b^{\delta}$ ,  $k^{\delta}$  as  $a^{-\delta} = b^{\delta}$ ,  $b^{-\delta} = a^{\delta}$  and  $k^{-\delta} = k^{\delta}$ .

Let *B* and *W* be two matrices of appropriate dimension with  $[B] = \mathcal{M}^{\perp}$  and  $[W] = \mathcal{S}^{\perp}$  and note that  $x \in \mathbb{R}^n_{\delta}$  is equivalent to  $x = \text{diag}(\delta) \xi, \xi \in \mathbb{R}^n_{>0}$ . Observe that  $\mathcal{M} \cap \mathbb{R}^n_{\delta} \neq \emptyset$  and  $\mathcal{S} \cap \mathbb{R}^n_{\delta} \neq \emptyset$  hold for a particular  $\delta$ , if and only if the system

$$B^T \operatorname{diag}\left(\delta\right) \xi = 0 \tag{3.35a}$$

$$W^T \operatorname{diag}(\delta) \xi = 0 \tag{3.35b}$$

has a positive solution  $\xi \in \mathbb{R}^n_{>0}$ . The following algorithm can be used to obtain orthants with  $\mathcal{M} \cap \mathbb{R}^n_{\delta} \neq \emptyset$  and  $\mathcal{S} \cap \mathbb{R}^n_{\delta} \neq \emptyset$  (see Appendix D, where (3.35b) and (3.35a) are used):

Algorithm 2.

#### 3.4. MULTISTATIONARITY IN SUBNETWORKS

#### (1) Assumptions:

- (i) In solving (3.8b),  $\nu$ ,  $\lambda \in \Lambda(E)$  can be determined independently of  $\mu \in \mathbb{R}^n$ .
- (ii)  $\mathcal{M}$  as defined in (3.31) is a linear subspace of  $\mathbb{R}^n$ .
- (2) Determine all  $\delta \in \{-1, 0, 1\}^n$  with  $S \cap \mathbb{R}^n_{\delta} \neq \emptyset$  (e.g. using the algorithms discussed in Appendix D):

$$\Delta^{\mathcal{S}} := \{ \delta \in \{-1, 0, 1\}^n \, | \mathcal{S} \cap \mathbb{R}^n_\delta \neq \emptyset \}$$
(3.36a)

(3) Determine all  $\delta \in \{-1, 0, 1\}^n$  with  $\mathcal{M} \cap \mathbb{R}^n_{\delta} \neq \emptyset$  (e.g. using the algorithms discussed in Appendix D).

$$\Delta^{\mathcal{M}} := \{\delta \in \{-1, 0, 1\}^n | \mathcal{M} \cap \mathbb{R}^n_{\delta} \neq \emptyset\}$$
(3.36b)

(4) Determine  $\Delta := \Delta^{S} \cap \Delta^{\mathcal{M}}$ . If and only if

$$\Delta \neq \emptyset$$
 (3.36c)

multistationarity is possible. This is a consequence of Fact 3 and Theorem 3.

(5) If Δ ≠ Ø, use (3.34a) and (3.34b) in (3.33a), (3.33b) and (3.33c) to obtain parametrizations for a pair of steady state a<sup>δ</sup>, b<sup>δ</sup> and the corresponding vector of rate constants k<sup>δ</sup>. Note that one obtains a different parameterization for each δ ∈ Δ.

There exists a variety of algorithms and software tools to check feasibility of (3.35b) and (3.35a). However, as up to  $\frac{1}{2}3^n$  inequality systems have to be checked, it is computationally hard, to use Fact 3 to check sufficient conditions for multistationarity. For systems of moderate size, as those presented in Section 4, the algorithms presented in Appendix D can be used.

#### 3.4 Multistationarity in subnetworks

Consider again the pointed polyhedral cone ker( $Y I_a$ )  $\cap \mathbb{R}_{\geq 0}^r$  and its generators. Let  $E_i$  be a generator. Then two different types of generators can be distinguished: trivial generators with  $I_a E_i = 0$  and stoichiometric generators with  $I_a E_i \neq 0.^{1,2}$  As explained in Section 3.1.1 every generator corresponds to a subnetwork of the overall network. The ODEs corresponding to any subnetwork can be obtained from the ODEs of the overall network by setting those rate constants to zero, that correspond to reactions not contained in the subnetwork. Thus, if  $J \subseteq \{1, \ldots, r\}$  is the index set of the reactions contained in a subnetwork, then  $k_i = 0$ ,  $i \notin J$  (if the subnetwork is defined by a stoichiometric generator  $E_i$ , this is equivalent to  $k_i = 0$ , if  $i \notin \sup (E_i)$ ).

The motivation of this section is based on the idea that steady states in a subnetwork are connected to steady states in the overall network. Intuitively, if  $x^{(1)}$ ,  $x^{(2)}$  are steady states of a subnetwork and if in the overall network those  $k_i$  previously set to zero are assigned arbitrarily small numbers, then the overall network should have steady states 'close' to  $x^{(1)}$ ,  $x^{(2)}$ . In Section 3.4.1 multistationarity in subnetworks defined by a stoichiometric generator are examined. For these subnetworks it is particularly easy to establish multistationarity, as this can often be done using the Deficiency One Algorithm of CRNT. This means, in particular, that multistationarity can be established by analysis of systems of linear inequalities. Furthermore, as a consequence of the results presented in Section 3.1 and 3.3 it is straightforward to obtain a parametrization of pairs of steady states and the corresponding parameter vectors.

[35] describes how steady states of the overall network deform to steady states of subnetworks defined by stoichiometric generators, if the  $k_i$  not contained in the subnetwork approach zero. In Section 3.4.2 the other direction is pursued: starting with (multiple) equilibria in a subnetwork, conditions are derived that guarantee that these steady states are steady states of the overall network as well. Moreover, by testing these conditions for a specific (pair of) steady state(s) it is possible to obtain values for those parameters of the overall network that are not contained in the subnetwork.

<sup>&</sup>lt;sup>1</sup>Note that trivial generators correspond to cycles in the directed graph defined by I<sub>a</sub>.

<sup>&</sup>lt;sup>2</sup>The term stoichiometric generators is used in [35]

#### 3.4.1 Multistationarity in subnetworks defined by stoichiometric generators

In this Section a subnetwork of a (bio)chemical reaction network that is defined by a stoichiometric generator of the larger network is considered. To distinguish subnetwork and overall network the symbol  $\hat{i}$  is used to mark all entities that belong to the overall network:  $\hat{I}_a$  for the incidence matrix,  $\hat{E}$  for the stoichiometric generator that defines the subnetwork and  $\hat{Y}^{(L)}$  for the matrix of exponent vectors. Both subnetwork and overall network involve the same complexes, thus Y is used for both networks. Note that the incidence matrix  $I_a$  and the matrix of exponent vectors of the subnetwork consist of column vectors  $\hat{I}_{a,i}$  and  $\hat{y}_i^{(L)}$  of the overall network, namely those with  $i \in \text{supp}\left(\hat{E}\right)$ . Let  $\hat{r}$  be the number of reactions contained in the overall network and r be the number of reactions in the subnetwork. Let E be the vector obtained from  $\hat{E}$  by deleting all zero entries; obviously E > 0. It follows that  $Y I_a E = 0$  and  $I_a E \neq 0$ .

This Section contains two parts. In the first part algebraic properties of subnetworks defined by a stoichiometric generator are derived, before in the second part parametrizations of pairs of steady states and the corresponding parameter vectors are derived.

#### 3.4.1.1 Algebraic properties of subnetworks defined by a stoichiometric generator

Recall the properties of generators  $E_i$ ,  $E_j$  of ker $(Y I_a) \cap \mathbb{R}^r_{>0}$ :

$$Y I_a E_i = 0, \quad Y I_a E_j = 0$$
$$E_i \in \mathbb{R}_{>0}^r.$$

Given  $E_i$ ,  $E_j$  with  $Y I_a E_i = 0$  and  $Y I_a E_j = 0$ . Then

$$\operatorname{supp}(E_i) \subseteq \operatorname{supp}(E_i) \Rightarrow E_i = 0 \text{ or } E_i = \alpha E_i.$$

Further note the well-known result from algebraic graph theory concerning incidence matrices of arbitrary (oriented) graphs (for details see, for example, [36, Theorem 8.3.1, p. 168]). It is given here using the notation introduced in [24, 25].

**Lemma 4.** Let  $I_a$  be the incidence matrix of a graph with m vertices and l connected components (i.e. linkage classes). Then rank  $I_a = m - l$ .

The following facts can be established for the matrix product  $Y I_a$  (the matrices associated to the subnetwork defined by  $\hat{E}$ ):

**Lemma 5.** Let  $\hat{E}$  be a stoichiometric generator and E a vector obtained by deleting all zero entries of  $\hat{E}$ . Further let Y and  $I_a$  be the matrices associated with the subnetwork, as described above. Then the following holds:

(i) E is a generator of ker $(Y I_a) \cap \mathbb{R}^r_{>0}$  and it is the only generator, that is

$$\ker(Y I_a) \cap \mathbb{R}^r_{>0} = \alpha E, \alpha \ge 0$$

- (*ii*) dim (ker  $(Y I_a)$ ) = 1
- (iii) I<sub>a</sub> has full column rank
- (iv)  $I_a$  contains m l columns
- (v) rank  $(Y I_a) = m l 1$
- Proof. (i.) Follows from the fact that  $\hat{E}$  is a generator of ker  $(\hat{Y} \hat{I}_a) \cap \mathbb{R}_{\geq 0}^{\hat{c}}$  and positivity of E. Let  $E_0$  be a nonzero nonnegative vector that satisfies (3.3a) and (3.3b). Positivity of E implies  $\supp(E_0) \subseteq supp(E)$ . Let  $\hat{E}_0 \in \mathbb{R}_{\geq 0}^{\hat{c}}$  be obtained from  $E_0$  by an appropriate padding with zeros. Then  $\hat{Y} \hat{I}_a \hat{E}_0 = 0$  and  $supp(\hat{E}_0) \subseteq supp(\hat{E})$  follow from the construction of  $\hat{E}_0$ . As  $\hat{E}$  is by definition a generator of ker  $(\hat{Y} \hat{I}_a) \cap \mathbb{R}_{\geq 0}^{\hat{c}}$  we conclude that  $\hat{E}_0 = \alpha \hat{E}, \alpha > 0$  and thus  $E_0 = \alpha E, \alpha > 0$ . Thus E is a generator of ker $(Y I_a) \cap \mathbb{R}_{\geq 0}^{\hat{c}}$ .
That *E* is the only generator follows again from positivity of *E*: any nonzero nonnegative vector  $E_0$  that satisfies (3.3a) and (3.3b) will inevitably satisfy  $\sup (E_0) \subseteq \sup (E)$ . As *E* is a generator of ker(*Y I<sub>a</sub>*)  $\cap \mathbb{R}^r_{\geq 0}$  all  $E_0$  satisfying (3.3a) and (3.3b) must satisfy  $E_0 = \alpha E$ ,  $\alpha > 0$ . Thus *E* is the only generator of ker(*Y I<sub>a</sub>*)  $\cap \mathbb{R}^r_{\geq 0}$ .

- (ii.) The proof is similar to the one for (i): suppose there exists a nonzero vector E<sub>0</sub> ∈ ℝ<sup>r</sup>, E<sub>0</sub> ∉ [E] with Y I<sub>a</sub> E<sub>0</sub> = 0. That E<sub>0</sub> contains negative entries follows from (i): otherwise E<sub>0</sub> = α E, α > 0 and dim (ker (Y I<sub>a</sub>)) = 1 from (i). Let Ẽ = α E + E<sub>0</sub>. Then Ẽ > 0, for sufficiently large α and Ẽ = β E by (i), a contradiction. Thus dim (ker (Y I<sub>a</sub>)) = 1.
- (iii.) As  $I_a E \neq 0$  by definition of E and dim (ker  $(Y I_a)$ ) = 1 by (ii),  $I_a$  must have full column rank.
- (iv.) By Lemma 4 rank  $(I_a) = m l$ . By (iii)  $I_a$  has full column rank. Thus  $I_a$  must have m l columns.
- (v.) By a standard result of linear algebra for any matrix A (see e.g. [61]) dim (ker (A)) = nr. of columnsrank (A). As dim (ker  $(Y I_a)$ ) = 1 by (ii) and nr. of columns = m - l by (iv) the desired result rank  $(Y I_a) = m - l - 1$  follows.

Suppose the subnetwork defined by  $\hat{E}$  contains several linkage classes and let  $\ell$  be the number of linkage classes of the subnetwork with  $\ell \geq 2$ . Let  $J_i \subseteq 1, \ldots, m$  be the set of indices whose complexes are part of linkage class *i*. Each linkage class can be considered as a graph of its own with an incidence matrix  $I_a^{I_i}$ , that contains those columns of  $I_a$  that are indexed by  $J_i$ . The following result can be established for the rank of the matrix product  $Y I_a^{I_i}$ :

**Corollary 3.** Assume the subnetwork defined by  $\tilde{E}$  has  $\ell \geq 2$  linkage classes. Consider one of these linkage classes. Let  $J_i \subseteq 1, ..., m$  be the set of indices whose complexes are part of that linkage class and let  $I_a^{J_i}$  the incidence matrix associated to the linkage class (as described above,  $I_a^{J_i}$ , that contains those columns of  $I_a$  that are indexed by  $J_i$ ). Let  $m_i$  be the number of complexes in the linkage class. Then

$$\operatorname{rank}\left(Y I_{a}^{J_{i}}\right) = m_{i} - 1.$$

Proof. By assumption the linkage class contains  $m_i$  nodes. By definition a linkage class is a graph with one connected component. Thus rank  $(I_a^{J_i}) = m_i - 1$  by Lemma 4. By Lemma 5, (ii) one has for the subnetwork defined by  $\hat{E}$ , that dim (ker  $(Y I_a)) = 1$ . As a consequence  $Y I_a^{J_i}$  must have full column rank. To see this, assume that  $Y I_a^{J_i}$  does not have full column rank, that is, there exists a vector  $E_0$  of appropriate dimension with  $Y I_a^{J_i} E_0 = 0$ . Let  $\tilde{E}_0$  be a suitable padding of  $E_0$  with zeros. One has  $Y I_a \tilde{E}_0 = 0$  for the complete subnetwork and  $\tilde{E}_0 \notin [E]$  (as E is a positive vector and  $\tilde{E}_0$  contains zero entries). Thus dim (ker  $(Y I_a)) = 2$ , a contradiction. Thus  $Y I_a^{J_i}$  has full column rank. As  $Y I_a^{J_i}$  has full column rank and as rank  $(I_a^{J_i}) = m_i - 1$  one obtains rank  $(Y I_a^{J_i}) = m_i - 1$ .

Subnetworks defined by stoichiometric generators are intimately connected to the Deficiency One Theorem of CRNT. To see this, recall the definition of network deficiency  $\mathcal{D}$ , as given in, for example, [24, 25, 27, 28]:

$$\mathcal{D} = m - \ell - s.$$

Recall that s is the dimension of the stoichiometric subspace, that is  $s = \operatorname{rank}(Y I_a)$  for the subnetwork defined by the stoichiometric generator  $\hat{E}$ . As a consequence of Lemma 5 and Corollary 3 the following result can be established:

**Corollary 4.** Consider a subnetwork with  $\ell \geq 2$  linkage classes defined by a stoichiometric generator  $\hat{E}$ . For the deficiency  $\mathcal{D}$  of the subnetwork as well as for the deficiencies  $\mathcal{D}_i$  of the linkage classes of the subnetwork the following holds:

(*i*.) D = 1

(*ii.*)  $\mathcal{D}_i = 0, i = 1, \ldots, \ell$ 

Proof. Lemma 5 (v) and (2.11) imply (i), Corollary 3 and (2.11) imply (ii).

In CRNT the Deficiency One Algorithm can be used to decide about multistationarity. It is applicable to biochemical reaction networks with deficiency  $\mathcal{D} = 1$  that satisfy certain additional constraints. To state these a concept of graph theory is needed: the strongly connected component. A directed graph is called strongly connected if for every pair of nodes u and v there is a path from u to v and a path from v to u. The strongly connected components of a directed graph are its maximal strongly connected subgraphs. In the notation introduced in [27, 28], strongly connected components are called strong linkage classes. If no edge from a node inside a strong linkage classes to a node outside exists, this strong linkage class is called terminal. The Deficiency One Algorithm is applicable to biochemical reaction networks satisfying the the following requirements [25, 28]:

- (I) The network deficiency is  $\mathcal{D} = 1$
- (II) The deficiency of the linkage classes is  $\mathcal{D}_i = 0, i = 1, \ldots, \ell$
- (III) There exists a positive vector E with  $Y I_a E = 0$ .
- (IV) The terminal strong linkage classes do not contain any cycles
- (V) Each linkage class contains only one terminal strong linkage class

Note that (I) and (II) hold by Corollary 4 and that (III) holds by Lemma 5. Further note that the graph cannot contain any cycles, as by Lemma 5  $I_a$  has full column rank. Thus (IV) holds as well. However, no information about (V) is obtainable using Lemma 5. This motivates the following fact:

Fact 4. Consider a biochemical reaction network that is a subnetwork of a larger network defined by a stoichiometric generator of the larger network. Assume that the subnetwork is displayed in the standard form of CRNT. Then the following holds: if every linkage class of the subnetwork contains only one terminal strong linkage class, the Deficiency One Algorithm is applicable. Thus, in particlar, only systems of **linear inequalities** have to be considered to decide about multistationarity.

**Remark 7.** Suppose the subnetwork defined by  $\hat{E}$  consists of a single linkage class (i.e.  $\ell = 1$ ). Clearly Corollary 3 does not hold in this case. However the deficiency of the subnetwork is still one. To see this, note that by Lemma 5 rank (Y  $I_a$ ) =  $m - \ell - 1 = m - 2$  in this case. And for the deficiency  $\mathcal{D}$  one obtains

$$\mathcal{D} = m - \ell - s = m - 1 - (m - 2) = 1.$$

#### 3.4.1.2 Steady states for subnetworks defined by a stoichiometric generator

From Lemma 5 it follows that  $\ker(Y I_a) \cap \mathbb{R}^r_{\geq 0}$  is spanned by a single positive vector E, that is  $\ker(Y I_a) \cap \mathbb{R}^r_{>0} = \alpha \ E, E \in \mathbb{R}^r_{>0}, \alpha > 0$ . Then the conditions  $v(k, a) = E \lambda$  and  $v(k, b) = E \nu$  become

$$k_i a^{y_i^{(L)}} = n_i \lambda$$
  $k_i b^{y_i^{(L)}} = n_i \nu, \ i = 1, \dots, r,$ 

where  $n_i$  is the *i*-th component of *E*. Apply ln ()

$$\ln k_i + \langle y_i^{(L)}, \ln a \rangle = \ln n_i + \ln \lambda \qquad \qquad \ln k_i + \langle y_i^{(L)}, \ln b \rangle = \ln n_i + \ln \nu, \ i = 1, \dots, r.$$

Subtracting equations in a from those in b one obtains (using, as before,  $\mu := \ln \frac{b}{a}$ ):

$$\langle y_i^{(L)}, \mu \rangle = \ln \frac{\nu}{\lambda}, \ i = 1, \dots, r.$$

Using  $\underline{1} = (1, ..., 1)^T$ , (3.8b) becomes

$$Y^{(L)T} \mu = \ln \frac{\nu}{\lambda} \underline{1}, \qquad (3.37)$$

 $\nu$ ,  $\lambda > 0$ . Solvability of (3.37) can be established by means of linear algebra. Let  $\tilde{\mu} = \left( \ln \frac{\nu}{\lambda}, \mu \right)$  and transform (3.37) to

$$\begin{bmatrix} -\underline{1} & Y^{(L)T} \end{bmatrix} \tilde{\mu} = 0. \tag{3.38}$$

Any solution  $\tilde{\mu} \neq 0$  to (3.38) can be used to determine two vectors  $a, b \in \mathbb{R}_{>0}^n$  and  $k \in \mathbb{R}_{>0}^r$  with  $Y I_a v(k, a) = Y I_a v(k, b) = 0$ . Note that the only requirement is that  $\tilde{\mu} \neq 0$  and not that  $\ln \frac{\nu}{\lambda} \neq 0$ . Thus, in particular,  $\ln \frac{\nu}{\lambda} = 0$  and thus  $\nu = \lambda$  is allowed. One obtains the following Lemma: Lemma 6. Consider equation (3.8b)

$$Y^{(L)^T} \mu = \ln \frac{\nu}{\lambda} \underline{1},$$

with  $\nu$ ,  $\lambda > 0$ . Let  $\tilde{\mu} = \left( \ln \frac{\nu}{\lambda}, \mu_0 \right)$  be a nonzero solution to

$$\begin{bmatrix} -\underline{1} & Y^{(L)^T} \end{bmatrix} \tilde{\mu} = 0.$$

Let  $\alpha > 0$  and  $a \in \mathbb{R}^n_{>0}$ . Choose

$$k = \operatorname{diag} \left( \phi \left( a^{-1} \right) \right) \left( \alpha E \right)$$

$$b = \operatorname{diag} \left( e^{\mu_0} \right) a.$$
(3.39a)
(3.39b)

Then  $Y I_a v(k, a) = Y I_a v(k, b) = 0.$ 

Proof. Suppose the first component of  $\tilde{\mu}$  is nonzero. Then  $\ln \frac{\nu}{\lambda} = \beta$ ,  $\beta \in \mathbb{R} \setminus \{0\}$ . Let  $\nu = e^{\beta} \lambda$ ,  $\lambda > 0$  and choose  $\alpha = \lambda$  in (3.39a). In this case the desired result follows from Lemma 1.

Suppose the first component of  $\tilde{\mu}$  is zero. Then  $\ln \frac{\nu}{\lambda} = 0$  and  $Y^{(L)^T} \mu_0 = 0$  (as, by assumption  $\tilde{\mu}$  is a solution to (3.38)). As a consequence,  $\phi(e^{\mu_0}) = e^{Y^{(L)^T} \mu_0} = \underline{1}$ . Consider the steady state equations

$$\begin{array}{l} Y \ I_a \ v(k, \ a) = Y \ I_a \ \operatorname{diag}\left(\alpha \ E\right) \underbrace{\operatorname{diag}\left(\phi \ \left(a^{-1}\right)\right) \phi \left(a\right)}_{=\underline{1}} \\ = Y \ I_a \ \left(\alpha \ E\right) = 0 \\ Y \ I_a \ v(k, \ b) = Y \ I_a \ \operatorname{diag}\left(\alpha \ E\right) \ \operatorname{diag}\left(\phi \ \left(a^{-1}\right)\right) \phi \left(\operatorname{diag}\left(e^{\mu_0}\right) a\right) \\ = Y \ I_a \ \operatorname{diag}\left(\alpha \ E\right) \ \operatorname{diag}\left(\phi \ \left(a^{-1}\right)\right) \ \operatorname{diag}\left(\phi \ \left(a^{-1}\right)\right) \\ = I_r \\ = Y \ I_a \ \operatorname{diag}\left(\alpha \ E\right) \ \operatorname{diag}\left(\phi \ \left(a^{-1}\right)\right) \\ = Y \ I_a \ \operatorname{diag}\left(\alpha \ E\right) \ \underline{1} \\ = Y \ I_a \ \left(\alpha \ E\right) = 0. \end{array}$$

Let ker  $\left(\left[-\frac{1}{4}Y^{(L)^{T}}\right]\right) = \left[\frac{M_{1}}{M_{2}}\right]$ . Then  $\mu = M_{2} \kappa$ ,  $\kappa$  a vector of appropriate dimension, is a representation of all vectors that can satisfy (3.38) (i.e. all vectors  $\mu \in \mathbb{R}^{n}$ , with  $\exists \alpha \in \mathbb{R}$  such that  $(\alpha, \mu)^{T}$  is solution to (3.38)). To determine  $a, b \in \mathbb{R}^{n}_{>0}$  and  $k \in \mathbb{R}^{n}_{>0}$  that satisfy the conservation relations as well (if rank (Y Ia) < n, otherwise  $a \in \mathbb{R}^{n}_{>0}$  is free) Lemma 3 has to be considered. Thus one has to find vectors  $v \in S$  and  $\mu \in [M_{2}]$  with sign  $(\mu) = \text{sign}(v)$ . If such a pair  $\mu, v$  can exists, equation (3.33a) from Theorem 3 and (3.39a), (3.39b) yield the desired  $a, b \in \mathbb{R}^{n}_{>0}$  and  $k \in \mathbb{R}^{r}_{>0}$ . Note that by Lemma 6 the parameter  $\alpha > 0$  can be chosen independent of  $\mu$ .

**Remark 8.** Note that  $\ln \frac{\nu}{\lambda} = 0$  will always occur, if the subnetwork defined by a stoichiometric generator is open with respect to certain species. If a species enters the system, then  $Y^{(L)}$  contains the zero column. Assume that only one species can enter the system. Then  $Y^{(L)} = \begin{bmatrix} 0 & \tilde{Y} \end{bmatrix}$ , where  $\tilde{Y}$  contains only nonzero column vectors. Equation (3.37) becomes

$$\begin{bmatrix} 0\\ \tilde{Y}^T \end{bmatrix} \mu = \ln \frac{\nu}{\lambda} E.$$

and  $\ln \frac{\nu}{\lambda} = 0$  follows immediately.

As an example consider the following network involving a protein A, its phosphorylated form  $A_p$ , a catalyst C and various combinations of these species (network  $N_2$  is structurally equivalent to a subnetwork of network  $\mathcal{N}_{11}$  discussed in Section 5):

$$\begin{split} \mathbf{A}_{\mathbf{p}} & \xrightarrow{\mathbf{k}_{3}} \mathbf{0} \underbrace{\stackrel{\mathbf{k}_{1}}{\overleftarrow{\mathbf{k}_{2}}} \mathbf{A}} \\ \mathbf{A} + \mathbf{C} & \underbrace{\stackrel{\mathbf{k}_{4}}{\overleftarrow{\mathbf{k}_{5}}} \mathbf{A} \mathbf{C}} \\ \mathbf{A} \mathbf{C} + \mathbf{C} & \underbrace{\stackrel{\mathbf{k}_{6}}{\overleftarrow{\mathbf{k}_{7}}} \mathbf{C} \mathbf{A} \mathbf{C} \xrightarrow{\mathbf{k}_{8}} \mathbf{A}_{\mathbf{p}} \mathbf{C} + \mathbf{C}} \\ \mathbf{A}_{\mathbf{p}} \mathbf{C} & \underbrace{\stackrel{\mathbf{k}_{9}}{\overleftarrow{\mathbf{k}_{10}}} \mathbf{A}_{\mathbf{p}} + \mathbf{C}} \end{split}$$
( $\mathcal{N}_{2}$ )

Network  $\mathcal{N}_2$  involves the species A,  $A_p$ , C, AC, CAC and  $A_pC$ . Use  $x_1$  for the concentration of A,  $x_2$  for the concentration of  $A_p$ ,  $x_3$  for the concentration of C,  $x_4$  for the concentration of AC,  $x_5$  for the concentration of CAC and  $x_6$  for the concentration of  $A_pC$ . The complexes are the zero complex  $y_1 = 0$ , A with  $y_2 = e_1$ ,  $A_p$  with  $y_3 = e_2$ , A + C with  $y_4 = e_1 + e_3$ , AC with  $y_5 = e_4$ , AC + C with  $y_6 = e_4 + e_3$ , CAC with  $y_7 = e_5$ ,  $A_pC + C$  with  $y_8 = e_3 + e_6$ ,  $A_pC$  with  $y_9 = e_6$  and  $A_p + C$  with  $y_{10} = e_2 + e_3$ . The columns of the  $6 \times 10$ -matrix Y are:

$$Y = \begin{bmatrix} 0 & e_1 & e_2 & e_1 + e_3 & e_4 & e_4 + e_3 & e_5 & e_3 + e_6 & e_6 & e_2 + e_3 \end{bmatrix},$$

the columns of the  $6 \times 10$ -matrix  $Y^{(L)}$  are

$$\tilde{Y}^{(L)} = \begin{bmatrix} 0 & e_1 & e_2 & e_1 + e_3 & e_4 & e_4 + e_3 & e_5 & e_5 & e_6 & e_2 + e_3 \end{bmatrix}$$

and the 10  $\times$  10-matrix  $\hat{I}_a$  is given by

$$\hat{I}_a = \text{diag}\left(I_a^{(1)}, I_a^{(2)}, I_a^{(3)}, I_a^{(2)}\right)$$

with

$$\hat{I}_{a}^{(1)} = \begin{bmatrix} -1 & 1 & 1 \\ 1 & -1 & 0 \\ 0 & 0 & -1 \end{bmatrix}$$

$$\hat{I}_{a}^{(2)} = \begin{bmatrix} -1 & 1 \\ 1 & -1 \end{bmatrix}$$

$$\hat{I}_{a}^{(3)} = \begin{bmatrix} -1 & 1 & 0 \\ 1 & -1 & -1 \\ 0 & 0 & 1 \end{bmatrix}.$$

The stoichiometric matrix is

and the conservation relation is given by

$$W = \begin{pmatrix} 0, & 0, & 1, & 1, & 2, & 1 \end{pmatrix}^T$$

The cone  $\ker(Y\,I_a)\cap I\!\!R^r_{>0}$  is generated by the column vectors of

$$E_{\mathcal{N}_2} = \left| \begin{array}{ccccccc} 1 & 0 & 0 & 0 & 1 \\ 1 & 0 & 0 & 0 & 0 \\ 0 & 0 & 0 & 0 & 1 \\ 0 & 1 & 0 & 0 & 1 \\ 0 & 1 & 0 & 0 & 0 \\ 0 & 0 & 1 & 0 & 1 \\ 0 & 0 & 0 & 1 & 0 \\ 0 & 0 & 0 & 1 & 1 \\ 0 & 0 & 0 & 1 & 0 \end{array} \right|$$

The first for columns are trivial generators, while the fifth column vector is the only stoichiometric generator of network  $\mathcal{N}_2$ , thus  $\hat{E} = (1, 0, 1, 1, 0, 1, 0, 1, 1, 0)^T$ . The subnetwork defined by  $\hat{E}$  is displayed below:

$$A_{p} \xrightarrow{k_{3}} 0 \xrightarrow{k_{1}} A$$

$$A + C \xrightarrow{k_{4}} A C$$

$$A C + C \xrightarrow{k_{6}} C A C \xrightarrow{k_{8}} A_{p} C + C$$

$$A_{p} C \xrightarrow{k_{9}} A_{p} + C$$

$$(\mathcal{N}_{3})$$

The matrix  $Y^{(L)}$  of the subnetwork  $\mathcal{N}_3$  is given by

$$Y^{(L)} = \begin{bmatrix} 0 & e_2 & e_1 + e_3 & e_3 + e_4 & e_5 & e_6 \end{bmatrix}$$

Thus (3.37) is given by

$$\begin{bmatrix} 0 \\ \mu_2 \\ \mu_1 + \mu_3 \\ \mu_3 + \mu_4 \\ \mu_5 \\ \mu_6 \end{bmatrix} = \ln \frac{\nu}{\lambda} \begin{bmatrix} 1 \\ 1 \\ 1 \\ 1 \\ 1 \\ 1 \end{bmatrix}$$

Obviously any solution must have  $\ln \frac{\nu}{\lambda} = 0$ . All solutions are given by

$$\mu = \kappa \left( \begin{array}{c} 1\\ 0\\ -1\\ 1\\ 0\\ 0 \end{array} \right)$$

There are two orthants, where sign  $(\mu) = \text{sign}(v), v \in [S]$  and  $\mu$  as defined above:

$$\delta = (1, 0, -1, 1, 0, 0)^T$$

and  $-\delta$ . Using the generators of  $S \cap \mathbb{R}^{\delta}_{\delta}$  all pairs of positive steady states and the corresponding vectors of rate constants are can be parameterized by  $\alpha_1, \ldots, \alpha_5 > 0$  and  $\kappa \in \mathbb{R}, \kappa \neq 0$  (where, as usual,  $S = [Y I_a]$ , the stoichiometric subspace of the subnetwork defined by  $\hat{E}$ ):

$$\begin{split} & a = \left(\frac{\alpha_1}{e^{\kappa} - 1}, \, \alpha_3, \, -\frac{\alpha_2}{e^{-\kappa} - 1}, \, \frac{\alpha_2}{e^{\kappa} - 1}, \, \alpha_4, \, \alpha_5\right)^T \\ & b = \left(\frac{e^{\kappa}\alpha_1}{e^{\kappa} - 1}, \, \alpha_3, \, -\frac{e^{-\kappa}\alpha_2}{e^{-\kappa} - 1}, \, \frac{e^{\kappa}\alpha_2}{e^{\kappa} - 1}, \, \alpha_4, \, \alpha_5\right)^T \\ & k = \left(1, \, \alpha_3^{-1}, \, -\frac{(e^{\kappa} - 1)(e^{-\kappa} - 1)}{\alpha_1 \alpha_2}, \, -\frac{(e^{\kappa} - 1)(e^{-\kappa} - 1)}{\alpha_2^2}, \, \alpha_4^{-1}, \, \alpha_5^{-1}\right)^T \end{split}$$

#### 3.4.2 Extension of multistationarity to the overall network

If multistationarity can be established for a subnetwork, one is of course interested in extending those solutions to the overall network. In this section conditions are presented that guarantee that (multiple) steady states in a subnetwork can be extended to (multiple) steady states in the overall network. Recall that the ODEs defined by a subnetwork can be obtained from those of the overall network by setting certain rate constants to zero. To carry over a pair of steady states and a vector of rate constants from a subnetwork to the overall network one therefore has to determine values for those rate constants that are zero in the subnetwork.

**Remark 9.** All results given in this subsection and especially Theorem 4 stem from a cooperation with D. Flockerzi and have been published in [9]. The proof of Theorem 4 is entirely D. Flockerzi's work and is given here only for the sake of completeness.

To begin with, consider a system of ODEs in the form  $\dot{x} = N v(k, x)$ , with  $N \in \mathbb{R}^{n \times \hat{r}}$ ,  $x \in \mathbb{R}^n$ and  $v, k \in \mathbb{R}^{\hat{r}}$  (using, as before, the symbol `to denote quantities belonging to the overall network). Recall that  $v(k, x) = \operatorname{diag}(k) \phi(x) = \operatorname{diag}(\phi(x)) k$ , where  $\phi(x) = \left(x^{y_1^{(L)}}, \ldots, x^{y_r^{(L)}}\right)^T$ . Let  $J \subseteq \{1, \ldots, \hat{r}\}$  be the set of reactions defining the subnetwork (if, for example, the subnetwork is defined by a stoichiometric generator  $\hat{E}$  of the overall network, then  $J = \operatorname{supp}(\hat{E})$ ). J can be used to split the parameter vector: let  $\hat{k}_E$ ,  $\hat{k}_c \in \mathbb{R}_{\geq 0}^{\hat{r}}$  and collect all parameters belonging to reactions contained in the subnetwork in  $\hat{k}_E$  (i.e.  $\hat{k}_{E,i} = k_i$  for  $i \in \operatorname{supp}(E)$  and  $\hat{k}_{E,i} = 0$  otherwise) and the remaining ones in  $\hat{k}_c$ (i.e.  $\hat{k}_c = k - \hat{k}_E$ ). For example, for the subnetwork  $\mathcal{N}_3$  of the overall network  $\mathcal{N}_2$  one has  $J = \{1, 3, 4, 6, 8, 9\}$  and thus  $\hat{k}_E = (k_1, 0, k_3, k_4, 0, k_6, 0, k_8, k_9, 0)^T$  and  $\hat{k}_c = (0, k_2, 0, 0, k_5, 0, k_7, 0, 0, k_{10})^T$ . Since v is linear in k one obtains

$$\dot{x} = N v(k, x) = N v(\dot{k}_E, x) + N v(\dot{k}_c, x).$$
 (3.40)

Obtain vectors  $k_E \in \mathbb{R}^r_{>0}$  and  $k_c \in \mathbb{R}^{\hat{r}-r}_{>0}$  by removing zero elements from  $\hat{k}_E$ ,  $\hat{k}_c$  as

$$k_E = (k_i)_{i \in J}$$
 and  $k_c = (k_i)_{i \notin J}$  resp.

For the subnetwork  $\mathcal{N}_3$  of the overall network  $\mathcal{N}_2$  one has  $k_E = (k_1, k_3, k_4, k_6, k_8, k_9)^T$  and  $k_c = (k_2, k_5, k_7, k_{10})^T$ . Let  $v_E(k_E, x) \in \mathbb{R}^r$  and  $v_c(k_c, x) \in \mathbb{R}^{\hat{r}-r}$  be given by

$$\begin{split} v_E\left(k_E,\,x\right) &:= \left(v(\hat{k}_E,\,x)\right)_{i\in J} = \operatorname{diag}\phi_E\left(x\right)k_E\\ v_c\left(k_c,\,x\right) &:= \left(v(\hat{k}_c,\,x)\right)_{i\notin J} = \operatorname{diag}\phi_c\left(x\right)k_c \;. \end{split}$$

Let  $N_i \in \mathbb{R}^n$  denote the columns of N and define

$$N_E = (N_i)_{i \in J}$$
 and  $N_c = (N_i)_{i \notin J}$ 

Then the ODE (3.40) can be rewritten as

$$\dot{x} = N_E v_E (k_E, x) + N_c v_c (k_c, x)$$
  
(3.41)

and the ODEs for the subnetwork defined by J are given by

$$\dot{x} = N_E v_E (k_E, x).$$
 (3.42)

For the subnetwork  $\mathcal{N}_3$  of the overall network  $\mathcal{N}_2$  one has  $v_E = (k_1, k_3 x_2, k_4 x_1 x_3, k_6 x_3 x_4, k_8 x_5, k_9 x_3 x_6)^T$ and  $v_c = (k_2 x_1, k_5 x_4, k_7 x_5, k_1 0 x_6)^T$  and the matrices

$$N_E = \begin{bmatrix} 1 & 0 & -1 & 0 & 0 & 0 \\ 0 & -1 & 0 & 0 & 0 & 1 \\ 0 & 0 & -1 & -1 & 1 & 1 \\ 0 & 0 & 1 & -1 & 0 & 0 \\ 0 & 0 & 0 & 0 & 1 & -1 \end{bmatrix} \quad \text{and} \quad N_c = \begin{bmatrix} -1 & 1 & 0 & 0 \\ 0 & 0 & 0 & -1 \\ 0 & 1 & 1 & -1 \\ 0 & -1 & 1 & 0 \\ 0 & 0 & -1 & 0 \\ 0 & 0 & 0 & 1 \end{bmatrix}.$$

The ODEs defined by the overall network  $N_2$  and by the subnetwork  $N_3$  are (the terms on the right hand side of the ODEs defined by the subnetwork are marked with a gray background):

$$\dot{x} = N_E v_E \left(k_E, x\right) + N_c v_c \left(k_c, x\right)$$

$$= \begin{vmatrix} k_1 - k_4 x_1 x_3 & -k_2 x_1 + k_5 x_4 \\ -k_3 x_2 + k_9 x_3 x_6 & -k_1 x_6 \\ -k_4 x_1 x_3 - k_6 x_3 x_4 + k_8 x_5 + k_9 x_3 x_6 \\ k_4 x_1 x_3 - k_6 x_3 x_4 \\ k_6 x_3 x_4 - k_8 x_5 & -k_5 x_4 + k_7 x_5 \\ k_8 x_5 - k_9 x_3 x_6 & -k_7 x_5 \\ k_{10} x_6 \end{vmatrix}$$

Let  $\hat{s}$  denote rank (N) and s denote rank  $(N_E)$ . Observe that rank (N) need not equal rank  $(N_E)$ , that is, in general rank  $(N_E) \leq \operatorname{rank}(N)$  and thus  $s \leq \hat{s}$  holds. Let  $W \in \mathbb{R}^{(n-\hat{s}) \times n}$  be an orthonormal basis for the left kernel of N and let  $W_E \in \mathbb{R}^{(n-s) \times n}$  with  $W_E^T = \begin{pmatrix} W_{add}^T \\ W_{add}^T \end{pmatrix}$  be an orthonormal basis for the left kernel of  $N_E$  (using  $W_{add}$  to accommodate additional basis vectors, if  $s < \hat{s}$ ;  $W_{add}$  empty if  $s = \hat{s}$ ). Let  $S_E$  be an orthonormal basis for im  $(N_E)$  and let  $T = (S_E, W_{add}, W)$  be an orthonormal transformation with  $x = x(\xi, \eta, \rho) = T \begin{pmatrix} \xi \\ \rho \\ p \end{pmatrix}$  and  $\xi = S_E^T x, \eta = W_{add}^T x, \rho = W^T x$ . Then (3.40) reads

$$\begin{aligned} \dot{\xi} &= S_E^T N_E \, v_E \, (k_E, \, x) + S_E^T \, N_c \, v_c \, (k_c, \, x) \\ \dot{\eta} &= W_{add}^T \, N_c \, v_c \, (k_c, \, x) \\ \dot{\rho} &= 0. \end{aligned} \tag{3.43}$$

Suppose a  $k_E^* > 0$  and steady states  $x_{1,2}^* > 0$  for the subnetwork (3.42) defined by J have been established, that is

$$N_E v_E \left( k_E^*, \, x_{1,2}^* \right) = 0$$
$$W_E^T x_1^* = W_E^T x_2^*$$

Let  $\left(\xi_{1,2}^*,\,\eta^*,\,\rho^*\right)^T=T^T\,x_{1,2}^*$  and observe that

$$\begin{split} \dot{\xi} &= \underbrace{S_E^T N_E v_E \left(k_E^*, x\left(\xi_{1,2}^*, \eta^*, \rho^*\right)\right)}_{=0} + S_E^T N_c v_c \left(k_c, x\left(\xi_{1,2}^*, \eta^*, \rho^*\right)\right) \\ \dot{\eta} &= W_{add}^T N_c v_c \left(k_c, x\left(\xi_{1,2}^*, \eta^*, \rho^*\right)\right) \\ \dot{\rho} &= 0. \end{split}$$

Fix  $k_E^* > 0$  and  $\rho^*$ , define  $\hat{k}_E^*$  as

$$(\hat{k}_{E}^{*})_{i\notin J} = 0 \text{ and } (\hat{k}_{E}^{*})_{i\in J} = k_{E}^{*},$$
 (3.44)

so that  $\hat{k}_E^*$  is obtained from  $k_E^*$  by a suitable padding with zeros and let  $k := \hat{k}_E^* + \epsilon \hat{k}_c$ . For steady states of (3.43) one needs to determine  $\xi$ ,  $\eta$  and  $\hat{k}_c$  such that

$$\begin{split} S_E^T N_E \, v_E \left( k_E^*, \, x \left( \xi, \, \eta, \, \rho^* \right) \right) &+ \epsilon \, S_E^T N_c \, v_c \left( k_c, \, x \left( \xi, \, \eta, \, \rho^* \right) \right) = 0 \\ & \epsilon \, W_{add}^T N_c \, v_c \left( k_c, \, x \left( \xi, \, \eta, \, \rho^* \right) \right) = 0. \end{split}$$

Observe that the last equation is equivalent to

$$W_{add}^T N_c \operatorname{diag}\left(\phi_c\left(x\left(\xi, \eta, \rho^*\right)\right)\right) k_c = 0.$$

With F(k, x) := Nv(k, x) and its Jacobian  $F_x := D_x F(k, x)$  one can outline the following program:

**Algorithm 3.** (1) Compute  $S_E^T F_x\left(\hat{k}_E^*, x_{1,2}^*\right)$   $(S_E, W_{add}) =: (A_{1,2}^*, B_{1,2}^*),$ ask for

regular  $A_{1,2}^{*}$  (3.45a)

and solve

$$A_{1,2}^* X_{1,2}^* + B_{1,2}^* = 0$$
 (3.45b)

for  $X_{1,2}^* = -[A_{1,2}^*]^{-1}B_{1,2}^*$ .

#### (2) Compute a positive $\kappa_0$ with

for

$$\mathcal{G}_c^* \kappa_0 = 0 \tag{3.45c}$$

$$\mathcal{G}_{c}^{*} := \begin{bmatrix} W_{add}^{T} N_{c} \operatorname{diag}\left(\phi_{c}\left(x_{1}^{*}\right)\right) \\ W_{add}^{T} N_{c} \operatorname{diag}\left(\phi_{c}\left(x_{2}^{*}\right)\right) \end{bmatrix}.$$
(3.45d)

and define  $\hat{k}_c^* \in I\!\!R^{\hat{r}}$  by

$$\left(\hat{k}_{c}^{*}\right)_{i \in \operatorname{supp}(E)} = 0 \quad and \quad \left(\hat{k}_{c}^{*}\right)_{i \notin \operatorname{supp}(E)} = \kappa_{0}.$$

$$(3.45e)$$

(3) Compute  $W_{add}^T F_x\left(\hat{k}_c^*, x_{1,2}^*\right)$   $(S_E, W_{add}) =: (C_{1,2}^*, D_{1,2}^*)$  and ask for

regular  $\mathcal{D}_{1,2}^* := D_{1,2}^* + C_{1,2}^* X_{1,2}^*.$  (3.45f)

The following Theorem 4 shows that this program leads to a pair of positive steady states  $\tilde{x}_{1,2}$  of (3.40) near  $x_{1,2}^*$  for  $\hat{k}_E^* + \epsilon \hat{k}_c^*$  if  $\epsilon > 0$  is sufficiently small. To obtain values for  $\tilde{x}_{1,2}$ , fix a sufficiently small  $\epsilon$  and solve  $N v(\hat{k}_E^* + \epsilon \hat{k}_c^*, \tilde{x}_{1,2}) = 0$  for  $\tilde{x}_{1,2}$  near  $x_{1,2}^*$ .

Theorem 4. Suppose the following conditions are fulfilled:

- (i) There exist  $x_{1,2}^* > 0$ ,  $k_E^* > 0$  with  $N_E v_E \left( k_E^*, x_{1,2}^* \right) = 0$ .
- (ii) There exists a  $\kappa_0 > 0$  with  $\mathcal{G}_c^* \kappa_0 = 0$ , cf. (3.45d).
- (iii) Both  $A_1^*$  and  $A_2^*$  are regular, cf. (3.45a).
- (iv) Both  $\mathcal{D}_1^*$  and  $\mathcal{D}_2^*$  are regular, cf. (3.45f).

Then there exist  $\epsilon_0 > 0$  and  $\delta_0 > 0$  such that

$$0 < \epsilon < \epsilon_0$$
 and  $|\rho - \rho^*| < \delta_0$ 

imply the existence of different positive hyperbolic steady states

$$x_{1,2}\left(\rho,\,\epsilon\right) = \left(S_E,\,W_{add},\,W\right)\,\begin{pmatrix} \Xi_{1,2}\left(\rho,\,\epsilon\right)\\ H_{1,2}\left(\rho,\,\epsilon\right)\\ \rho\end{pmatrix}$$

of (3.40) with

$$W^T x_1(\rho, \epsilon) = W^T x_2(\rho, \epsilon)$$

for the positive  $k(\epsilon) = \hat{k}_E^* + \epsilon \hat{k}_c^* - cf.$  (3.44) and (3.45e).

*Proof.* Apply the orthonormal transformation  $\xi = S_E^T x$ ,  $\eta = W_{add}^T x$ ,  $\rho = W^T x$  to obtain

$$\begin{split} & \dot{\xi} = S_E^T N \, v(\dot{k}_E^* + \epsilon \, \dot{k}_c^*, \, x) \\ & = S_E^T N \, v(\hat{k}_E^*, \, x) + \epsilon \, S_E^T N \, v(\hat{k}_c^*, \, x) \\ & \dot{\eta} = W_{add}^T N \, v(\hat{k}_E^* + \epsilon \, \hat{k}_c^*, \, x) = \epsilon \, W_{add}^T N \, v(\hat{k}_c^*, \, x) \\ & \dot{\dot{\eta}} = 0. \end{split}$$

For fixed  $x_j^*$ ,  $k_E^*$  and regular  $A_j^*$  (see assumption (i) and (iii)) the equation  $\dot{\xi} = 0$  has a locally unique solution  $\xi_j = \Xi_j \left(\eta, \, \rho, \, \epsilon \hat{k}_c^*\right)$  near  $(\eta^*, \rho^*, 0)$  with  $A_j^* X_j + B_j^* = 0$ ,  $X_j := \frac{\partial}{\partial \eta} \Xi_j (\eta^*, \, \rho^*, 0)$ , by the implicit function theorem. To find steady states with common  $\rho$ -components we need to solve

$$\begin{split} W^T_{add} N \, v(\hat{k}^*_c, \, S_E \, \Xi_1 \left( \eta, \, \rho, \, \epsilon \, \hat{k}^*_c \right) + W_{add} \, \eta + W \, \rho) &= 0 \\ W^T_{add} \, N \, v(\hat{k}_c, \, S_E \, \Xi_2 \left( \tilde{\eta}, \, \rho, \, \epsilon \, \hat{k}^*_c \right) + W_{add} \, \tilde{\eta} + W \, \rho) &= 0 \end{split}$$

for  $\begin{pmatrix} \eta \\ \eta \end{pmatrix} = \begin{pmatrix} H_1(\rho, \epsilon) \\ H_2(\rho, \epsilon) \end{pmatrix}$  near the solution  $(\eta^*, \eta^*, \rho^*, 0)$  (cf. assumption (ii) and (3.45e)). By assumption (iv) and the implicit function theorem such functions  $H_1$  and  $H_2$  exist. Thus locally, there exist steady states

$$\begin{aligned} \xi_1 &= \Xi_1 \left( H_1 \left( \rho, \, \epsilon \right), \, \rho, \, \epsilon \right) =: \Xi_1 \left( \rho, \, \epsilon \right), \, \eta = H_1 \left( \rho, \, \epsilon \right), \, \rho \\ \xi_2 &= \Xi_2 \left( H_2 \left( \rho, \, \epsilon \right), \, \rho, \, \epsilon \right) =: \Xi_2 \left( \rho, \, \epsilon \right), \, \eta = H_2 \left( \rho, \, \epsilon \right), \, \rho \end{aligned}$$

of (3.43) for sufficiently small  $\epsilon > 0$ . The corresponding positive steady states of (3.40) near  $x_{1,2}^*$  are given by the  $x_{1,2}(\rho, \epsilon)$  of the theorem.

**Remark 10.** Note that Theorem 4 makes no use of the fact, that the subnetwork is defined by a stoichiometric generator. Thus it can be applied to any subnetwork. The same holds of course for Algorithm 3.

#### 3.5 Resume: a program to decide about multistationarity

In this section the results of this chapter are combined to a program that can be applied to decide about multistationarity. One has to distinguish two cases: (i)  $\nu$ ,  $\lambda \Lambda(E)$  can be determined independent of  $\mu \in \mathbb{R}^n$  (see Section 3.2.1) and (ii)  $\nu$ ,  $\lambda \in \Lambda(E)$  and  $\mu \in \mathbb{R}^n$  have to be determined together (see Section 3.2.2). In the first case Algorithm 2 given in Section 3.3.1, in the second case Algorithm 4 given below can be applied.

#### Algorithm 4.

#### (1) Assumptions:

- (i) The sign of the elements of Q<sub>ij</sub> the matrix Q<sub>v</sub>(μ), sign (Q<sub>ij</sub>) can be determined by linear inequalities (i.e. the conditions given in Lemma 2 hold).
- (ii) Let  $s_0$  be the number of linear inequalities that determine the sign pattern of  $Q(\mu)$ .
- (iii) Let V ∈ ℝ<sup>s<sub>0</sub>×n</sup> be the coefficient matrix of the inequalities that determine the sign pattern of Q (μ), as defined in (3.21a).
- (iv) Let  $\beta \in \mathbb{R}^{s_0}$  be as defined in (3.21b).
- (2) Determine all  $\delta \in \{-1, 0, 1\}^n$  with  $S \cap \mathbb{R}^n_{\delta} \neq \emptyset$ :

$$\Delta^{\mathcal{S}} := \{\delta \in \{-1, 0, 1\}^n | \mathcal{S} \cap \mathbb{R}^n_{\delta} \neq \emptyset\}$$

$$(3.46)$$

(3) Determine all feasible signings  $\sigma \in \{-1, 0, 1\}^{s_0}$  for  $Q(\mu)$ :

$$\Sigma := \left\{ \sigma \in \{-1, 0, 1\}^n | \begin{bmatrix} V & \beta & -\operatorname{diag}(\sigma) \end{bmatrix} \begin{pmatrix} \mu \\ \bar{s} \\ s \end{pmatrix} = 0, \ \bar{s} = 1, \ s > 0, \ has \ a \ solution \right\}$$
(3.47)

(4) (a) Determine all  $\sigma \in \Sigma$  such that  $Q(\mu)$  is an  $L^+$  -matrix (e.g. using Theorem 1):

$$\Sigma^{+} := \{ \sigma \in \Sigma | Q(\mu) \text{ is an } L^{+}\text{-matrix} \}$$

$$(3.48)$$

(b) Determine all  $\sigma \in \Sigma$  such that  $Q(\mu)$  is a sign-central matrix (e.g. using Theorem 2):

$$\Sigma^{0} := \{\sigma \in \Sigma | Q(\mu) \text{ is an sign-central matrix}\}$$
  
(3.49)

(5) Determine all orthants δ ∈ Δ<sup>S</sup> that contain a feasible signing σ ∈ Σ<sup>+</sup> (or, if Σ<sup>+</sup> = Ø, σ ∈ Σ<sup>0</sup>). That is, identify all orthants, with S ∩ ℝ<sup>n</sup><sub>δ</sub> ≠ Ø that contain a solution to V μ + β − diag (σ) s = 0, s > 0:

$$\mathcal{P}^{+,0} := \left\{ \begin{array}{ll} (\sigma,\,\delta) \in \Sigma^{+,0} \times \Delta | \left[ V \operatorname{diag}\left(\delta\right) \quad \beta \quad -\operatorname{diag}\left(\sigma\right) \right] \begin{pmatrix} \xi \\ \bar{s} \\ s \end{pmatrix} = 0, \, \xi, \, s > 0, \, \bar{s} = 1 \right\}$$
(3.50)

If  $\mathcal{P}^+ \neq \emptyset$  then multistationarity is guaranteed. If  $\mathcal{P}^+ = \emptyset$  and  $\mathcal{P}^0 \neq \emptyset$ , then multistationarity is guaranteed for a subnetwork. Observe that  $\mathcal{P}^0 \neq \emptyset$  means that  $Q(\mu)$  is a sign-central matrix, thus it is only guaranteed that  $Q(\mu)$  has a **nonnegative** kernel vector. That is, some components of  $\nu$ ,  $\lambda \in \mathbb{R}^p_{>0}$  might be zero. In this case one can still try to establish multistationarity for the overall network using the results given in Section 3.4; see (vi) below.

- (6) If  $\mathcal{P}^+ \neq \emptyset$  obtain a pair of steady states in the following way:
  - (i) Solve

$$\underbrace{\begin{bmatrix} V \operatorname{diag}(\delta) & \beta & -\operatorname{diag}(\sigma) \end{bmatrix}}_{=:H} \begin{pmatrix} \xi \\ \bar{s} \\ s \end{pmatrix} = 0, \ \xi, s > 0, \ \bar{s} = 1$$

to obtain generators of ker  $(H) \cap \mathbb{R}^{n+1+s_0}_{>0}$ . Let the columns of  $M = \begin{bmatrix} M_1 \\ M_2 \end{bmatrix}$  be generators. Then  $\mu = \text{diag}(\delta) \ M_1 \kappa, \kappa > 0$  is a representation of all  $\mu \in \mathbb{R}^n_{\delta}$  with sign  $(V \mu + \beta) = \sigma$ .

- (ii) Fix  $\bar{\kappa} > 0$  to obtain  $\bar{\mu} = M_1 \bar{\kappa}$ .
- (iii) Solve  $Q(\bar{\mu}) \begin{pmatrix} \nu \\ \lambda \end{pmatrix} = 0, \ \nu, \lambda \in \Lambda(E) \ for \ \nu, \lambda.$
- (iv) Choose any  $v \in S \cap \mathbb{R}^n_{\delta}$ .
- (v) Use (3.33a), (3.33b) and (3.33c) to obtain values for a, b, k (using  $\bar{\mu}$  and  $\nu$ ,  $\lambda$  obtained above).
- (vi) If  $\mathcal{P}^+ = \emptyset$  and  $\mathcal{P}^0 \neq \emptyset$  use the Algorithm 3 to try to extend multistationarity to the overall network.

**Remark 11.** The condition  $\mathcal{P}^+ \neq \emptyset$  is necessary for multistationarity, the conditions  $\mathcal{P}^+ = \emptyset$  and  $\mathcal{P}^0 \neq \emptyset$  are necessary for multistationarity in a subnetwork (and can be extended to necessary conditions for the overall network if combined with the conditions given in Algorithm 3). As a consequence  $\mathcal{P}^+ = \emptyset$  and  $\mathcal{P}^0 = \emptyset$  does not imply that multistationarity cannot occur in a biochemical reaction network.

## Chapter 4

# Multistationarity in the activation of an MAPK(K)

Mitogen-activated protein kinase (MAPK) cascades are well studied systems in cell biology. The most common form of an MAPK cascade is that of a three tiered cascade, as depicted in Fig. 4.1. Its building blocks are an MAPK, an MAPK kinase (MAPKK) and an MAPKK kinase (MAPKKK) together with the respective mono- and double-phosphorylated forms (-P denoting single phosphorylation and -PP double phosphorylation). Mono-phosphorylated MAPKKK catalyzes phosphorylation of MAPKK, whose double-phosphorylated form in turn catalyzes phosphorylation of MAPK.  $E_1$  stands for a stimulus that triggers the cascade, MAPKK'ase, MAPKK'ase and  $E_2$  are phosphatases. (If the mammalian ERK-cascade was considered,  $E_1$ , for example, would be RAS, the MAPKKK Raf-1 or Mos and the MAPK ERK-1 [57]).

Mathematical models describing the dynamics of an MAPK cascade have been known for some time (see e.g. [31, 37] or [54] for a more general overview of quantitative models for signal transduction networks in general, as well as [39] and the references therein). These models have been extensively studied in the literature using numerical tools like bifurcation analysis. It is therefore known that models of an MAPK cascade can exhibit all sorts of complex dynamical behaviour like bistability and oscillations (see e.g. [4, 5] or [38, 39]). Only recently [45] showed that, surprisingly, bistability can even occur on layer two or three of Fig. 4.1 alone, provided a distributive, multi-collision mechanism is used for both, phosphorylation and dephosphorylation of the MAPK or the MAPKK [39, 45].

In principle, both phosphorylation and dephosphorylation could follow either a distributive or processive mechanism (see Section 4.1 for details). One way to decide which mechanism is employed is to experimentally verify multiple steady states on a single layer. This idea was already suggested in [45, 50]. However, to **safely discard** different candidate mechanisms, a **mathematical proof** is required that these systems cannot exhibit multistationarity for any conceivable parameter vector. This can be accomplished using CRNT, as shown in [10, 11, 12]. However, using the approach presented in Chapter 3 will not only provide additional information (that is discussed in Chapter 6) but will also lead to a better understanding of the phenomenon multistationarity itself.

This chapter is organised as follows: in Section 4.1 distributive and processive mechanisms are introduced. In Section 4.2 and 4.3 the methods developed in Chapter 3 are applied to the reaction networks presented in Section 4.1. In Section 4.4 the influence of the assumption that some species are **not** subject to a conservation relation on the existence of multiple steady states is examined, while Section 4.5 contains a brief discussion of the implications of the results obtained in Section 4.2 and 4.3 for model discrimination. The chapter closes with a discussion of the results obtained in Chapter 3 in the light of the results obtained for the activation of an MAPK(K).

## 4.1 Processive vs. distributive phosphorylation

In this section three candidate network structures for a single layer of the reaction scheme displayed in Fig. 4.1 are discussed. Every scheme is a realization of the double-phosphorylation process sketched in Fig. 4.2. Both phosphorylation and dephosphorylation can follow either a distributive or processive mechanism. By a processive mechanism the kinase (phosphatase) carries out two phosphorylation or



Figure 4.1: Scheme of an MAPK cascade (c.f. [37]). MAPK denotes the mitogen-activated protein kinase, MAPKK the MAPK kinase and MPAKKK the MAPKK kinase, -P mono- and -PP double-phosphorylation.  $E_1$  is a stimulus, MAPK'ase, MAPKK'ase and  $E_2$  are phosphatases.

dephosphorylation steps, before the final product is released. In a distributive mechanism, monophosphorylated intermediates are released before conversion to the final product occurs, after a second binding of kinase (phosphatase) and intermediate. In the following scenarios, A is the MAPK, subscript  $_p$  and  $_{pp}$  denote single and double phosphorylation,  $E_1$  corresponds to MEK and  $E_2$  to the phosphatase. The following networks involving different combinations of processive and distributive mechanisms are analyzed:

• A distributive mechanism both for phosphorylation and dephosphorylation:

$$\begin{split} \mathbf{A} + \mathbf{E}_{1} \underbrace{\stackrel{\mathbf{k}_{1}}{\overset{}{\overset{}{\mathbf{k}_{2}}}} \mathbf{A} \mathbf{E}_{1} \xrightarrow{\overset{\mathbf{k}_{3}}{\longrightarrow}} \mathbf{A}_{\mathbf{p}} + \mathbf{E}_{1} \underbrace{\stackrel{\mathbf{k}_{4}}{\overset{}{\overset{}{\mathbf{k}_{5}}}} \mathbf{A}_{\mathbf{p}} \mathbf{E}_{1} \xrightarrow{\overset{\mathbf{k}_{6}}{\longrightarrow}} \mathbf{A}_{\mathbf{pp}} + \mathbf{E}_{1} \\ \mathbf{A}_{\mathbf{pp}} + \mathbf{E}_{2} \underbrace{\stackrel{\mathbf{k}_{7}}{\overset{}{\overset{}{\mathbf{k}_{8}}}} \mathbf{A}_{\mathbf{pp}} \mathbf{E}_{2} \xrightarrow{\overset{\mathbf{k}_{9}}{\longrightarrow}} \mathbf{A}_{\mathbf{p}} + \mathbf{E}_{2} \underbrace{\stackrel{\mathbf{k}_{10}}{\overset{}{\overset{}{\overset{}{\mathbf{k}_{11}}}} \mathbf{A}_{\mathbf{p}} \mathbf{E}_{2} \xrightarrow{\overset{\mathbf{k}_{12}}{\longrightarrow}} \mathbf{A} + \mathbf{E}_{2} \end{split} (\mathcal{N}_{4})$$

• A processive mechanism for phosphorylation and a distributive mechanism for dephosphorylation:

$$\mathbf{A} + \mathbf{E}_{1} \underbrace{\overset{\mathbf{k}_{1}}{\overleftarrow{\mathbf{k}_{2}}}}_{\mathbf{k}_{2}} \mathbf{A} \mathbf{E}_{1} \underbrace{\overset{\mathbf{k}_{3}}{\overleftarrow{\mathbf{k}_{4}}}}_{\mathbf{k}_{4}} \mathbf{A}_{p} \mathbf{E}_{1} \underbrace{\overset{\mathbf{k}_{5}}{\overleftarrow{\mathbf{k}_{5}}}}_{\mathbf{k}_{p}\mathbf{p}} + \mathbf{E}_{1}$$

$$\mathbf{A}_{pp} + \mathbf{E}_{2} \underbrace{\overset{\mathbf{k}_{6}}{\overleftarrow{\mathbf{k}_{7}}}}_{\mathbf{k}_{7}} \mathbf{A}_{pp} \mathbf{E}_{2} \underbrace{\overset{\mathbf{k}_{8}}{\overleftarrow{\mathbf{k}_{1}}}}_{\mathbf{k}_{1}\mathbf{p}} \mathbf{A}_{p} + \mathbf{E}_{2} \underbrace{\overset{\mathbf{k}_{9}}{\overleftarrow{\mathbf{k}_{10}}}}_{\mathbf{k}_{10}\mathbf{p}} \mathbf{A}_{p} \mathbf{E}_{2} \underbrace{\overset{\mathbf{k}_{11}}{\overleftarrow{\mathbf{k}_{1}}}}_{\mathbf{k}_{1}\mathbf{p}} \mathbf{A}_{p} \mathbf{E}_{2}$$

$$(\mathcal{N}_{5})$$



Figure 4.2: Scheme of activation of an MAPK (A). Enzyme  $E_1$  (e.g. the kinase MEK) double phosphorylates the MAPK A, enzyme  $E_2$  (a phosphatase) dephosphorylates both mono- and double- phosphorylated MAPK (i.e.  $A_p$  and  $A_{pp}$ ).

From a mathematical perspective this network is structurally equal to a network incorporating a distributive mechanism for phosphorylation and a processive mechanism for dephosphorylation (as the latter one would be the result of a mere change of names:  $A \rightarrow A_{pp}$ ,  $E_1 \rightarrow E_2$  and  $E_2 \rightarrow E_1$ ).

• A processive mechanism both for phosphorylation and dephosphorylation:

$$\mathbf{A} + \mathbf{E}_{1} \underbrace{\overset{\mathbf{k}_{1}}{\underset{\mathbf{k}_{2}}{\overset{\mathbf{k}_{3}}{\overset{\mathbf{k}_{4}}{\overset{\mathbf{k}_{3}}{\overset{\mathbf{k}_{4}}{\overset{\mathbf{k}_{5}}{\overset{\mathbf{k}_{5}}{\overset{\mathbf{k}_{5}}{\overset{\mathbf{k}_{5}}{\overset{\mathbf{k}_{5}}{\overset{\mathbf{k}_{5}}{\overset{\mathbf{k}_{5}}{\overset{\mathbf{k}_{5}}{\overset{\mathbf{k}_{1}}{\overset{\mathbf{k}_{5}}{\overset{\mathbf{k}_{1}}{\overset{\mathbf{k}_{5}}{\overset{\mathbf{k}}{\overset{\mathbf{k}_{5}}{\overset{$$

The structural data and the ODEs for each network are given in Appendix A.

## 4.2 Positive solutions for the polynomial equations

In this Section the equation  $Y^{(L)T} \mu = \ln \frac{E\nu}{E\lambda}$  (see (3.8b) in Chapter 3) is derived and solved for  $\mathcal{N}_4$ ,  $\mathcal{N}_5$  and  $\mathcal{N}_6$ .

Remark 12. From hereon, when analyzing the equation

$$Y^{(L)^{T}} \mu = \ln \frac{E \nu}{E \lambda},$$

the constraint  $\nu$ ,  $\lambda > 0$  is considered instead of the constraint  $\nu$ ,  $\lambda \in \Lambda(E)$ . This is done because the constraint enhances  $\nu$ ,  $\lambda > 0$  readability and because for **all but one** of the networks considered in the remaining chapters  $\Lambda(E)$  is equal to the positive orthant. The only exception is network  $\mathcal{N}_{10}$ . For this network however, it was possible to establish multistationarity by considering  $\nu$ ,  $\lambda > 0$ . Thus there was no need to consider all of  $\nu$ ,  $\lambda \in \Lambda(E)$  (recall that  $\Lambda(E)$  is a union of cones, multistationarity was established for one cone, thus it was not necessary to consider the remaining ones).

#### 4.2.1 Positive solutions for $N_4$

For this network the equation

$$Y^{(L)^{T}} \mu = \ln \frac{E \nu}{E \lambda}$$

reads

$$\mu_1 + \mu_2 = \ln \frac{\nu_1 + \nu_5}{\lambda_1 + \lambda_5} \tag{4.1a}$$

$$\mu_3 = \ln \frac{\nu_1}{\lambda_1} \tag{4.1b}$$

$$\mu_3 = \ln \frac{\gamma_3}{\lambda_5} \tag{4.1c}$$

$$\mu_2 + \mu_4 = \ln \frac{\nu_2 + \nu_6}{\lambda_2 + \lambda_6}$$
(4.1d)  
$$\mu_5 - \ln \frac{\nu_2}{\lambda_2 + \lambda_6}$$
(4.1d)

$$\mu_5 = \ln \frac{\nu_6}{\lambda_c} \tag{4.1f}$$

$$\mu_6 + \mu_7 = \ln \frac{\nu_3 + \nu_6}{\lambda_3 + \lambda_6} \qquad (4.1g)$$

$$\mu_8 = \ln \frac{\nu_3}{\lambda_3} \tag{4.1h}$$

$$\mu_8 = \ln \frac{\nu_6}{\lambda_6} \tag{4.1i}$$

$$\mu_4 + \mu_7 = \ln \frac{\nu_4 + \nu_5}{\lambda_4 + \lambda_5} \tag{4.1j}$$

$$\mu_9 = \ln \frac{\nu_4}{\lambda_4} \tag{4.1k}$$

$$\mu_9 = \ln \frac{\nu_5}{\lambda_5}.\tag{4.11}$$

These equations are solvable, if and only if

$$\ln \frac{\nu_1}{\lambda_1} = \ln \frac{\nu_5}{\lambda_5}, \qquad \qquad \ln \frac{\nu_2}{\lambda_2} = \ln \frac{\nu_6}{\lambda_6}, \qquad \qquad \ln \frac{\nu_3}{\lambda_3} = \ln \frac{\nu_6}{\lambda_6}, \qquad \qquad \ln \frac{\nu_4}{\lambda_4} = \ln \frac{\nu_5}{\lambda_5}$$

holds (note that this is equivalent to condition (3.11)). Thus one obtains the following polynomials (as  $\nu_i > 0$  and  $\lambda_i > 0$ , i = 1, ..., 6):

$$\begin{aligned} \lambda_5 \,\nu_1 &- \lambda_1 \,\nu_5 = 0, & \lambda_6 \,\nu_2 &- \lambda_2 \,\nu_6 = 0 \\ \lambda_6 \,\nu_3 &- \lambda_3 \,\nu_6 = 0, & \lambda_5 \,\nu_4 &- \lambda_4 \,\nu_5 = 0. \end{aligned}$$

Solve for  $\nu_1$ ,  $\nu_2$ ,  $\nu_5$  and  $\nu_6$  to obtain:

$$\nu_1 = \lambda_1 \frac{\nu_4}{\lambda_4}, \qquad \qquad \nu_2 = \lambda_2 \frac{\nu_3}{\lambda_3}, \qquad \qquad \nu_5 = \lambda_5 \frac{\nu_4}{\lambda_4}, \qquad \qquad \nu_6 = \lambda_6 \frac{\nu_3}{\lambda_3}.$$

This yields a representation of the vector  $\nu$  in terms of  $\lambda_1, \ldots, \lambda_6$  and  $\nu_3$  and  $\nu_4$ :

$$\nu^{T} = \left(\lambda_{1} \frac{\nu_{4}}{\lambda_{4}}, \lambda_{2} \frac{\nu_{3}}{\lambda_{3}}, \nu_{3}, \nu_{4}, \lambda_{5} \frac{\nu_{4}}{\lambda_{4}}, \lambda_{6} \frac{\nu_{3}}{\lambda_{3}}\right).$$
(4.4)

Thus it is possible to determine  $\nu, \lambda \in \mathbb{R}^6_{>0}$  independent of  $\mu \in \mathbb{R}^9$ . Note that for  $\nu$  as in (4.4), one obtains

$$\ln \frac{\nu_1 + \nu_5}{\lambda_1 + \lambda_5} = \ln \frac{\nu_1}{\lambda_1} = \ln \frac{\nu_5}{\lambda_5} = \ln \frac{\nu_4 + \nu_5}{\lambda_4 + \lambda_5} = \ln \frac{\nu_4}{\lambda_4}$$

and

$$\ln \frac{\nu_2 + \nu_6}{\lambda_2 + \lambda_6} = \ln \frac{\nu_2}{\lambda_2} = \ln \frac{\nu_6}{\lambda_6} = \ln \frac{\nu_3 + \nu_6}{\lambda_3 + \lambda_6} = \ln \frac{\nu_3}{\lambda_3}.$$

And thus the following system of linear equations:

$$\begin{split} \mu_1 + \mu_2 &= \ln \frac{\lambda_4}{\lambda_4} \\ \mu_3 &= \ln \frac{\nu_4}{\lambda_4} \\ \mu_2 + \mu_4 &= \ln \frac{\nu_3}{\lambda_3} \\ \mu_5 &= \ln \frac{\nu_3}{\lambda_3} \\ \mu_6 + \mu_7 &= \ln \frac{\nu_3}{\lambda_3} \\ \mu_8 &= \ln \frac{\nu_3}{\lambda_3} \\ \mu_4 + \mu_7 &= \ln \frac{\nu_4}{\lambda_4} \\ \mu_9 &= \ln \frac{\nu_4}{\lambda_4} \end{split}$$

...

Solving for  $\mu$  finally yields:

$$\mu = \mu_7 \ (-1, 1, 0, -1, 0, -1, 1, 0, 0)^{'} + \ln \frac{\nu_3}{\lambda_3} \ (-1, 1, 0, 0, 1, 1, 0, 1, 0)^{'} \\ + \ln \frac{\nu_4}{\lambda_4} \ (2, -1, 1, 1, 0, 0, 0, 0, 1)^{'} .$$

$$(4.5)$$

If in (4.5)  $\kappa_1 := \ln \frac{\nu_3}{\lambda_3}$  and  $\kappa_2 := \ln \frac{\nu_4}{\lambda_4}$  are interpreted as free parameters  $\mu$  is given by

$$\mu = M_{\mathcal{N}_4} \begin{pmatrix} \mu_7\\ \kappa_1\\ \kappa_2 \end{pmatrix}$$

where

$$M_{\mathcal{N}_4} := \begin{bmatrix} -1 & -1 & 2 \\ 1 & 1 & -1 \\ 0 & 0 & 1 \\ -1 & 0 & 1 \\ 0 & 1 & 0 \\ -1 & 1 & 0 \\ 1 & 0 & 0 \\ 0 & 1 & 0 \\ 0 & 0 & 1 \end{bmatrix}.$$
 (4.6a)

Consider the column space of  $M_{\mathcal{N}_4}$ :  $\mathcal{M}_{\mathcal{N}_4} = [M_{\mathcal{N}_4}]$ . From Chapter 3 it is known that any element  $\mu \in \mathcal{M}_{\mathcal{N}_4}$  can be associated with the difference of two positive steady state solutions: if a satisfies  $Y I_a v(a, k) = 0$  and if  $\mu \in \mathcal{M}_{\mathcal{N}_4}$ , then  $b = \text{diag}(\text{Exp}(\mu)) a$  satisfies  $Y I_a v(b, k) = 0$ . As discussed in Section 3.1, any vector  $a \in \mathbb{R}_{>0}^9$  is a steady state solution for the vector of rate constants k defined in (3.10a). For network  $\mathcal{N}_4$  one obtains the following rate constants:

$$k_1 = \frac{\lambda_1 + \lambda_5}{a_1 a_2} \qquad \qquad k_2 = \frac{\lambda_1}{a_3} \tag{4.7a}$$

$$k_3 = \frac{\lambda_5}{a_3} \qquad \qquad k_4 = \frac{\lambda_2 + \lambda_6}{a_2 a_4} \tag{4.7b}$$

$$k_5 = \frac{\lambda_2}{a_5} \qquad \qquad k_6 = \frac{\lambda_6}{a_5} \qquad (4.7c)$$

$$\lambda_2 + \lambda_c \qquad \qquad \lambda_3$$

$$k_7 = \frac{\lambda_3 + \lambda_6}{a_6 a_7} \qquad \qquad k_8 = \frac{\lambda_3}{a_8} \qquad (4.7d)$$

$$k_9 = \frac{\lambda_6}{a_8} \qquad \qquad k_{10} = \frac{\lambda_4 + \lambda_5}{a_4 \, a_7} \tag{4.7e}$$

$$k_{11} = \frac{\lambda_4}{a_9}$$
  $k_{12} = \frac{\lambda_5}{a_9}.$  (4.7f)

## 4.2.2 Positive solutions for $N_5$

For this network the equation

$$Y^{(L)T} \mu = \ln \frac{E \nu}{E \lambda}$$

reads

$$\mu_1 + \mu_2 = \ln \frac{\nu_1 + \nu_5}{\lambda_1 + \lambda_5} \tag{4.8a}$$

$$\mu_3 = \ln \frac{\nu_1}{\lambda_1} \tag{4.8b}$$

$$\mu_3 = \ln \frac{\nu_2 + \nu_3}{\lambda_2 + \lambda_5} \tag{4.8c}$$

$$\mu_4 = \ln \frac{z}{\lambda_2} \tag{4.8d}$$

$$\mu_4 = \ln \frac{1}{\lambda_5} \tag{4.8e}$$

$$\mu_5 + \mu_6 = \ln \frac{\lambda_3 + \lambda_5}{\lambda_3 + \lambda_5} \tag{4.8f}$$

$$\mu_7 = \ln \frac{\nu_5}{\lambda_3}$$

$$\mu_7 = \ln \frac{\nu_5}{\lambda_3}$$
(4.8g)
(4.8h)

$$\mu_6 + \mu_8 = \ln \frac{\nu_4 + \nu_5}{\nu_4 + \lambda_5} \tag{4.8i}$$

$$\mu_9 = \ln \frac{\nu_4}{\lambda_4} \tag{4.8j}$$

$$\mu_9 = \ln \frac{\nu_5}{\lambda_5} \tag{4.8k}$$

These equations are solvable, if and only if

$$\ln \frac{\nu_1}{\lambda_1} = \ln \frac{\nu_2 + \nu_5}{\lambda_2 + \lambda_5}, \qquad \qquad \ln \frac{\nu_2}{\lambda_2} = \ln \frac{\nu_5}{\lambda_5}, \qquad \qquad \ln \frac{\nu_3}{\lambda_3} = \ln \frac{\nu_5}{\lambda_5}, \qquad \qquad \ln \frac{\nu_4}{\lambda_4} = \ln \frac{\nu_5}{\lambda_5}.$$

One obtains

$$\nu_2 = \lambda_2 \frac{\nu_5}{\lambda_5}, \qquad \qquad \nu_3 = \lambda_3 \frac{\nu_5}{\lambda_5}, \qquad \qquad \nu_4 = \lambda_4 \frac{\nu_5}{\lambda_5}$$

and thus  $\ln \frac{\nu_2 + \nu_5}{\lambda_2 + \lambda_5} = \frac{\nu_5}{\lambda_5}$ . Therefore  $\nu_1 = \lambda_1 \frac{\nu_5}{\lambda_5}$  and

$$\nu = \frac{\nu_5}{\lambda_5} \,\lambda,$$

 $\lambda \in \mathbb{R}^{5}_{>0}$  and  $\nu_{5} > 0$  free; thus, as for network  $\mathcal{N}_{4}$ , vectors  $\nu$  and  $\lambda \in \mathbb{R}^{5}_{>0}$  can be determined independent of  $\mu \in \mathbb{R}^{8}$ . Further note that

$$\ln \frac{\nu_1 + \nu_5}{\lambda_1 + \lambda_5} = \ln \frac{\nu_3 + \nu_5}{\lambda_3 + \lambda_5} = \ln \frac{\nu_4 + \nu_5}{\lambda_4 + \lambda_5} = \ln \frac{\nu_5}{\lambda_5}.$$

That is, all terms on the right hand side of (4.8a) - (4.8k) equal

$$\kappa_1 := \ln \frac{\nu_5}{\lambda_5}.\tag{4.9}$$

Solving (4.8a) - (4.8k) for  $\mu$  yields

$$\mu = M_{\mathcal{N}_5} \begin{pmatrix} \mu_2 \\ \mu_8 \\ \kappa_1 \end{pmatrix} \tag{4.10a}$$

where

$$M_{\mathcal{N}_{5}} := \begin{bmatrix} 0 & -1 & 1 \\ 0 & 1 & 0 \\ 0 & 0 & 1 \\ 1 & 0 & 0 \\ -1 & 0 & 1 \\ 1 & 0 & 0 \\ -1 & 0 & 1 \\ 1 & 0 & 0 \\ 0 & 0 & 1 \end{bmatrix}.$$
 (4.10b)

Again, any element of the linear subspace  $\mathcal{M}_{\mathcal{N}_5} = [M_{\mathcal{N}_5}]$  can be interpreted as the difference of two steady states: for any  $\mu \in \mathcal{M}_{\mathcal{N}_5}$ , follows from Chapter 3 that  $b = \operatorname{diag}(e^{\mu}) a, a \in \mathbb{R}^9_{>0}$ , free, is a steady state solution, if a is a steady state solution. Note that any vector  $a \in \mathbb{R}^9_{>0}$  is a steady state solution, if the following  $k \in \mathbb{R}^{11}_{>0}$ :

$$k_1 = \frac{\lambda_1 + \lambda_5}{a_1 a_2}$$
  $k_2 = \frac{\lambda_1}{a_3}$  (4.11a)

$$k_3 = \frac{\lambda_2 + \lambda_5}{a_3} \qquad \qquad k_4 = \frac{\lambda_2}{a_4} \tag{4.11b}$$

$$k_5 = \frac{\lambda_5}{a_4} \qquad \qquad k_6 = \frac{\lambda_3 + \lambda_5}{a_5 a_6} \qquad (4.11c)$$

$$\lambda_2 \qquad \qquad \lambda_5$$

$$k_7 = \frac{\lambda_3}{a_7} \qquad \qquad k_8 = \frac{\lambda_5}{a_7} \tag{4.11d}$$

$$k_9 = \frac{\lambda_4 + \lambda_5}{a_6 a_8} \qquad \qquad k_{10} = \frac{\lambda_4}{a_9} \qquad (4.11e)$$

$$k_{11} = \frac{\lambda_5}{a_9}.\tag{4.11f}$$

## 4.2.3 Positive solutions for $\mathcal{N}_6$

For this network the equation

$$Y^{(L)T} \mu = \ln \frac{E\nu}{E\lambda}$$

reads

$$\mu_1 + \mu_2 = \ln \frac{\nu_1 + \nu_5}{\lambda_1 + \lambda_5} \tag{4.12a}$$

$$\mu_3 = \ln \frac{\nu_1}{\lambda_1} \tag{4.12b}$$

$$\mu_3 = \ln \frac{\nu_2 + \nu_5}{\lambda_2 + \lambda_5} \tag{4.12c}$$

$$\mu_4 = \ln \frac{1}{\lambda_2} \tag{4.12d}$$

$$\mu_5 = \ln^{\nu_5} \tag{4.12e}$$

$$\mu_4 - m \frac{1}{\lambda_5}$$

$$\mu_5 + \mu_6 = \ln \frac{\nu_3 + \nu_5}{\lambda_5}$$
(4.12c)
(4.12f)

$$\mu_7 = \ln \frac{\nu_3}{\lambda_2} \tag{4.12g}$$

$$\mu_7 = \ln \frac{\nu_4 + \nu_5}{\lambda_4 + \lambda_5} \tag{4.12h}$$

$$\mu_8 = \ln \frac{\nu_4}{\lambda_4} \tag{4.12i}$$

$$\mu_8 = \ln \frac{\nu_5}{\lambda_5} \tag{4.12j}$$

A solution exists, if and only if

$$\ln\frac{\nu_1}{\lambda_1} = \ln\frac{\nu_2 + \nu_5}{\lambda_2 + \lambda_5}, \qquad \ln\frac{\nu_2}{\lambda_2} = \ln\frac{\nu_5}{\lambda_5}, \qquad \ln\frac{\nu_3}{\lambda_3} = \frac{\nu_4 + \nu_5}{\lambda_4 + \lambda_5}, \qquad \ln\frac{\nu_4}{\lambda_4} = \ln\frac{\nu_5}{\lambda_5}.$$

One immediately obtains  $\nu_2 = \lambda_2 \frac{\nu_5}{\lambda_5}$  and  $\nu_4 = \lambda_4 \frac{\nu_5}{\lambda_5}$ . Note that this implies

$$\ln\frac{\nu_2+\nu_5}{\lambda_2+\lambda_5} = \ln\frac{\nu_4+\nu_5}{\lambda_4+\lambda_5} = \ln\frac{\nu_5}{\lambda_5},$$

which in turn implies  $\ln \frac{\nu_1}{\lambda_1} = \ln \frac{\nu_3}{\lambda_3} = \ln \frac{\nu_5}{\lambda_5}$ . Thus one obtains  $\nu_1 = \lambda_1 \frac{\nu_5}{\lambda_5}$ ,  $\nu_3 = \lambda_3 \frac{\nu_5}{\lambda_5}$  and thus

$$\nu = \frac{\nu_5}{\lambda_5} \,\lambda,$$

 $\lambda \in \mathbb{R}_{>0}^5$  and  $\nu_5 > 0$ , free. That is, one can obtain  $\nu$ ,  $\lambda \in \mathbb{R}_{>0}^5$  independent of  $\mu \in \mathbb{R}^9$ . Note that all terms on the right hand side of (4.12a) – (4.12j) are equal to

$$\kappa_1 := \ln \frac{\nu_5}{\lambda_5}.\tag{4.13}$$

Solving (4.12a) - (4.12j) for  $\mu$  yields:

 $k_{3} =$ 

$$\mu = M_{\mathcal{N}_6} \begin{pmatrix} \mu_2 \\ \mu_6 \\ \kappa_1 \end{pmatrix} \tag{4.14a}$$

where

$$M_{\mathcal{N}_{6}} := \begin{bmatrix} -1 & 0 & 1 \\ 1 & 0 & 0 \\ 0 & 0 & 1 \\ 0 & 0 & 1 \\ 0 & -1 & 1 \\ 0 & 1 & 0 \\ 0 & 0 & 1 \\ 0 & 0 & 1 \end{bmatrix}$$
(4.14b)

As in the previous sections, any element of the linear subspace  $\mathcal{M}_{\mathcal{N}_6} = [\mathcal{M}_{\mathcal{N}_6}]$  can be interpreted as the difference of two steady states: for any  $\mu \in \mathcal{M}_{\mathcal{N}_6}$ , follows from Chapter 3 that  $b = \text{diag}(e^{\mu}) a$ ,  $a \in \mathbb{R}^8_{>0}$ , free, is a steady state solution, if a is a steady state solution. Note that any vector  $a \in \mathbb{R}^8_{>0}$ is a steady state solution, if the following  $k \in \mathbb{R}^{10}_{>0}$ :

$$k_1 = \frac{\lambda_1 + \lambda_5}{a_1 a_2}$$
  $k_2 = \frac{\lambda_1}{a_3}$  (4.15a)

$$\frac{\lambda_2 + \lambda_5}{a_3} \qquad \qquad k_4 = \frac{\lambda_2}{a_4} \tag{4.15b}$$

$$k_5 = \frac{\lambda_5}{a_4} \qquad \qquad k_6 = \frac{\lambda_3 + \lambda_5}{a_5 a_6} \tag{4.15c}$$

$$k_7 = \frac{\lambda_3}{a_7}$$
  $k_8 = \frac{\lambda_4 + \lambda_5}{a_7}$  (4.15d)

$$k_9 = \frac{\lambda_4}{a_8} \qquad \qquad k_{10} = \frac{\lambda_5}{a_8} \tag{4.15e}$$

## 4.3 Positive solutions satisfying the conservation relations

Recall that for networks  $\mathcal{N}_4$ ,  $\mathcal{N}_5$  and  $\mathcal{N}_6$  it is possible to determine  $\nu$  and  $\lambda$  independent of  $\mu$ . Thus Algorithm 2 can be applied to decide about multistationarity. As a consequence of Theorem 3, the existence of an orthant  $\delta$  with  $\mathcal{S}_i \cap \mathbb{R}^n_\delta \neq \emptyset$  and  $\mathcal{M}_i \cap \mathbb{R}^n_\delta \neq \emptyset$  (i.e. with  $\mathcal{T} \neq \emptyset$ ,  $\mathcal{T}$  as defined in (3.32)) is necessary and sufficient for multistationarity,  $i = \mathcal{N}_4$ ,  $\mathcal{N}_5$ ,  $\mathcal{N}_6$ . For network  $\mathcal{N}_4$  14 orthants were found with  $\mathcal{S}_{\mathcal{N}_4} \cap \mathbb{R}^n_\delta \neq \emptyset$  and  $\mathcal{M}_{\mathcal{N}_4} \cap \mathbb{R}^n_\delta \neq \emptyset$ , while for network  $\mathcal{N}_5$  and  $\mathcal{N}_6$  no orthants were found. This leads to the following conclusion: **Fact 5.** Only network  $N_4$  can admit multistationarity. That is, in particular, for  $N_5$  and  $N_6$  no parameter vector exists, such that the corresponding ODEs exhibit multistationarity.

Using the algorithms presented in Appendix D all orthants with  $S_{N_4} \cap \mathbb{R}^9_{\delta} \neq \emptyset$  and  $\mathcal{M}_{N_4} \cap \mathbb{R}^9_{\delta} \neq \emptyset$  have been determined:

$$\begin{split} \delta_1 &= (-1, -1, -1, -1, 1, 1, -1, -1, 1, -1)^T \\ \delta_2 &= (-1, 0, -1, 1, 1, 1, -1, 1, -1)^T \\ \delta_3 &= (-1, 1, -1, -1, 1, 1, -1, 1, -1)^T \\ \delta_4 &= (-1, 1, -1, -1, 1, 1, 0, 1, -1)^T \\ \delta_5 &= (-1, 1, -1, -1, 1, 1, 1, 1, 1, -1)^T \\ \delta_6 &= (-1, 1, -1, -1, 1, 1, 1, -1, 1)^T \\ \delta_7 &= (-1, 1, -1, -1, 1, 1, 1, -1, 1)^T \end{split}$$
(4.16)

Recall that if  $S_{\mathcal{N}_4} \cap \mathbb{R}^8_{\delta} \neq \emptyset$  and  $\mathcal{M}_{\mathcal{N}_4} \cap \mathbb{R}^8_{\delta} \neq \emptyset$ , then this holds for  $-\delta$  as well; thus there are in fact 14 orthants where  $S_{\mathcal{N}_4}$  and  $\mathcal{M}_{\mathcal{N}_4}$  are sign compatible.

#### 4.3.1 Parameterizing Multistationarity for $N_4$

As  $\mathcal{M}_{\mathcal{N}_4}$  is a linear subspace of  $\mathbb{R}^9$ , (3.34a) and (3.34b) can be used to represent  $\mu \in \mathcal{M}_{\mathcal{N}_4}$  and  $v \in S_{\mathcal{N}_4}$ in the orthants given in (4.16). Thus vectors  $\mu^{\delta_i}$  and  $v^{\delta_i}$  have been determined using generators of  $\mu \cap \mathbb{R}^9_{\delta_i}$  and  $S \cap \mathbb{R}^9_{\delta_i}$  for each orthant  $\mathbb{R}^9_{\delta_i}$  given by (4.16). Table 4.1 contains the  $\mu^{\delta_i}$ ,  $i = 1, \ldots, 7$ , Table 4.2 the  $v^{\delta_i}$ ,  $i = 1, \ldots, 7$ . Using this data, a set  $(a^{\delta_i}, b^{\delta_i}, k^{\delta_i})$  of steady states and rate constants can be assembled for each orthant  $\mathbb{R}^9_{\delta_i}$ .

Using Table 4.1 and 4.2 it is therefore possible to give analytical expressions for any pair of positive steady states  $a^{\delta_i}$ ,  $b^{\delta_i}$  and, for each pair, all parameter vectors  $k^{\delta_i}$  that ensure that the ODEs (A.1a) – (A.1i) and the conservation relations (A.2a) – (A.2c) given in Appendix A.1.2 admit  $a^{\delta_i}$  and  $b^{\delta_i}$  as steady state solutions.

As an example the parametrizations  $a^{\delta_5}$ ,  $b^{\delta_5}$  and  $k_1^{\delta_5}$ , ...,  $k_{12}^{\delta_5}$  are given. From Table 4.1 the cone  $\mathcal{M}_{\mathcal{N}_4} \cap \mathbb{R}^9_{\delta_5}$  can be parametrized as  $\mu^{\delta_5} = (-\alpha_1 - 2\alpha_2 - 2\alpha_3, \alpha_1 + 2\alpha_2 + \alpha_3, -\alpha_3, -\alpha_2 - \alpha_3, \alpha_1 + \alpha_2, \alpha_1, \alpha_2, \alpha_1 + \alpha_2, -\alpha_3)^T$ . From Table 4.2 follows that  $v \in \mathcal{S}_{\mathcal{N}_4} \cap \mathbb{R}^9_{\delta_5}$  can be parametrized as  $v^{\delta_5} = (-\beta_1, \beta_2, -\beta_2 - \beta_4, -\beta_3, \beta_4, \beta_1 + \beta_2 + \beta_3 + \beta_5, \beta_5, \beta_6, -\beta_5 - \beta_6)^T$ . Inserting  $\mu^{\delta_5}$  in (3.33a) yields a parametrization of  $a^{\delta_5}$  in terms of  $\alpha_1, \alpha_2, \alpha_3$  and  $\beta_1, \ldots, \beta_6, \alpha_i > 0, \beta_i > 0$ :

$$a^{\delta_{5}} = \left(-\frac{\beta_{1}}{-1+e^{-\alpha_{1}-2\alpha_{2}-2\alpha_{3}}}, \frac{\beta_{2}}{-1+e^{\alpha_{1}+2\alpha_{2}+\alpha_{3}}}, \frac{-\beta_{2}-\beta_{4}}{-1+e^{-\alpha_{3}}}\right) \\ -\frac{\beta_{3}}{-\frac{\beta_{4}}{-1+e^{-\alpha_{2}-\alpha_{3}}}, \frac{\beta_{4}}{-1+e^{\alpha_{1}+\alpha_{2}}}, \frac{\beta_{1}+\beta_{2}+\beta_{3}+\beta_{5}}{-1+e^{\alpha_{1}}} \\ -\frac{\beta_{5}}{-1+e^{\alpha_{2}}}, \frac{\beta_{6}}{-1+e^{\alpha_{2}+\alpha_{2}}}, \frac{-\beta_{5}-\beta_{6}}{-1+e^{-\alpha_{3}}}\right)^{T}$$
(4.17a)

Using (4.17a) and  $\mu^{\delta_5}$  in (3.33b) yields the following parametrization of  $b^{\delta_5}$ :

$$b^{\delta_{5}} = \left(-\frac{\beta_{1}e^{-\alpha_{1}-2\alpha_{2}-2\alpha_{3}}}{-1+e^{-\alpha_{1}-2\alpha_{2}-2\alpha_{3}}}, \frac{\beta_{2}e^{\alpha_{1}+2\alpha_{2}+\alpha_{3}}}{-1+e^{\alpha_{1}+2\alpha_{2}+\alpha_{3}}}, \frac{(-\beta_{2}-\beta_{4})e^{-\alpha_{3}}}{-1+e^{-\alpha_{3}}}\right) \\ -\frac{\beta_{3}e^{-\alpha_{2}-\alpha_{3}}}{-1+e^{-\alpha_{2}-\alpha_{3}}}, \frac{\beta_{4}e^{\alpha_{1}+\alpha_{2}}}{-1+e^{\alpha_{1}+\alpha_{2}}}, \frac{(\beta_{1}+\beta_{2}+\beta_{3}+\beta_{5})e^{\alpha_{1}}}{-1+e^{\alpha_{1}}} \\ \frac{\beta_{5}e^{\alpha_{2}}}{-1+e^{\alpha_{2}}}, \frac{\beta_{6}e^{\alpha_{1}+\alpha_{2}}}{-1+e^{\alpha_{1}+\alpha_{2}}}, \frac{(-\beta_{5}-\beta_{6})e^{-\alpha_{3}}}{-1+e^{-\alpha_{3}}}\right)^{T}$$
(4.17b)

Using (4.17a) in (3.33c) yields a parametrization of  $k_1^{\delta_5}, \ldots, k_{12}^{\delta_5}$  in terms of  $\alpha_1, \alpha_2, \alpha_3, \beta_1, \ldots, \beta_6$ ,

 $\alpha_i \geq 0, \ \beta_i \geq 0 \ \text{and} \ \lambda_1, \ \ldots, \ \lambda_6, \ \lambda_i > 0$ :

$$k_{1}^{\delta_{5}} = -\frac{\left(-1 + e^{-\alpha_{1} - 2\alpha_{2} - 2\alpha_{3}}\right)\left(-1 + e^{\alpha_{1} + 2\alpha_{2} + \alpha_{3}}\right)\left(\lambda_{1} + \lambda_{5}\right)}{\beta_{1}\beta_{2}}$$
(4.18a)

$$k_{2}^{\delta_{5}} = \frac{(-1+e^{-\delta})\lambda_{1}}{-\beta_{2}-\beta_{4}}$$

$$(4.18b)$$

$$k_{2}^{\delta_{5}} = (-1+e^{-\alpha_{3}})\lambda_{5}$$

$$(4.18c)$$

$$k_{3}^{5} = \frac{-\beta_{2} - \beta_{4}}{(4.18c)}$$

$$k_{4}^{5} = -\frac{(-1 + e^{-\alpha_{2} - \alpha_{3}})(-1 + e^{\alpha_{1} + 2\alpha_{2} + \alpha_{3}})(\lambda_{2} + \lambda_{6})}{\alpha_{4} - \alpha_{4}}$$
(4.18d)

$$k_{5}^{\delta_{5}} = \frac{(-1 + e^{\alpha_{1} + \alpha_{2}})\lambda_{2}}{\beta_{4}}$$
(4.18e)

$$k_6^{\delta_5} = \frac{(-1 + e^{\alpha_1 + \alpha_2})\lambda_6}{\beta_4}$$
(4.18f)

$$k_7^{\delta_5} = \frac{(-1+e^{\alpha_1})(-1+e^{\alpha_2})(\lambda_3+\lambda_6)}{\beta_5(\beta_1+\beta_2+\beta_3+\beta_5)}$$
(4.18g)

$$k_8^{\delta_5} = \frac{(-1 + e^{\alpha_1 + \alpha_2})\lambda_3}{\beta_6} \tag{4.18h}$$

$$k_9^{\delta_5} = \frac{(-1 + e^{-1/\delta_2})\lambda_6}{\beta_6} \tag{4.18i}$$

$$k_{10}^{\delta_5} = -\frac{(-1+e^{\alpha_2})(-1+e^{-\alpha_2-\alpha_3})(\lambda_4+\lambda_5)}{\beta_3\,\beta_5} \tag{4.18j}$$

$$k_{11}^{\delta_5} = \frac{(-1+e^{-\alpha_3})\lambda_4}{-\beta_5 - \beta_6} \tag{4.18k}$$

$$k_{12}^{\delta_5} = \frac{(-1+e^{-\alpha_3})\,\lambda_5}{-\beta_5 - \beta_6} \tag{4.181}$$

$I\!\!R^9_{\delta_i}$	$p_{\mu^{\delta_i}}$	Representation of $\mu^{\delta_i} \in \mathcal{M}_{\mathcal{N}_4} \cap \mathbb{R}^9_{\delta_i}$
$I\!\!R^9_{\delta_1}$	3	$\mu^{\delta_1} = \left(-\alpha_1, -\alpha_2, -\alpha_1 - \alpha_2, \alpha_2 + \alpha_3, \alpha_3, \alpha_1 + 2\alpha_2 + 2\alpha_3, -\alpha_1 - 2\alpha_2 - \alpha_3, \alpha_3, -\alpha_1 - \alpha_2\right)^T$
$\mathbb{R}^9_{\delta_2}$	3	$\mu^{\delta_{2}} = \left(-\alpha_{3}, 0, -\alpha_{3}, \alpha_{2}, \alpha_{2}, 2\alpha_{2} + \alpha_{3}, -\alpha_{2} - \alpha_{3}, \alpha_{2}, -\alpha_{3}\right)^{T}$
$I\!\!R^9_{\delta_3}$	3	$\mu^{\delta_3} = \left(-2\alpha_1 - \alpha_2 - \alpha_3, \alpha_1 + \alpha_2, -\alpha_1 - \alpha_3, -\alpha_1, \alpha_2, \alpha_2 + \alpha_3, -\alpha_3, \alpha_2, -\alpha_1 - \alpha_3\right)^T$
$I\!\!R^9_{\delta_4}$	3	$\mu^{\delta_4} = \left( -\alpha_1 - 2\alpha_3, \alpha_1 + \alpha_3, -\alpha_3, -\alpha_3, \alpha_1, \alpha_1, 0, \alpha_1, -\alpha_3 \right)^T$
$I\!\!R^9_{\delta_5}$	3	$\mu^{\delta_{5}} = \left(-\alpha_{1} - 2\alpha_{2} - 2\alpha_{3}, \alpha_{1} + 2\alpha_{2} + \alpha_{3}, -\alpha_{3}, -\alpha_{2} - \alpha_{3}, \alpha_{1} + \alpha_{2}, \alpha_{1}, \alpha_{2}, \alpha_{1} + \alpha_{2}, -\alpha_{3}\right)^{T}$
$I\!\!R^9_{\delta_6}$	3	$\mu^{\delta_{6}} = \left(-\alpha_{1} - \alpha_{3}, \alpha_{1}, -\alpha_{3}, 0, \alpha_{1}, \alpha_{1} + \alpha_{3}, -\alpha_{3}, \alpha_{1}, -\alpha_{3}\right)^{T}$
$I\!\!R^9_{\delta_7}$	3	$\mu^{\delta_{7}} = \left(-\alpha_{1} - \alpha_{3}, \alpha_{1}, -\alpha_{3}, \alpha_{2}, \alpha_{1} + \alpha_{2}, \alpha_{1} + 2\alpha_{2} + \alpha_{3}, -\alpha_{2} - \alpha_{3}, \alpha_{1} + \alpha_{2}, -\alpha_{3}\right)^{T}$

Table 4.1: Representations of  $\mu^{\delta_i} \in \mathcal{M}_{\mathcal{N}_4}$  in the orthants  $\mathbb{R}^9_{\delta_i}$  that contain both,  $\mathcal{M}_{\mathcal{N}_4}$  and  $\mathcal{S}_{\mathcal{N}_4}$   $(i = 1, \ldots, 7, \text{ as defined in (4.16)}).$ 

## 4.4 An extension: open systems

Common to network  $\mathcal{N}_4 - \mathcal{N}_6$  is that all concentrations are subject to at least one conservation relation. But in reality, none of these reaction networks can be observed in isolation, as  $\mathcal{N}_4 - \mathcal{N}_6$  each describe a single layer of a signal transduction network that consists of several similar layers and  $E_1$  and  $E_2$ are subject to regulation from different layers. Thus, the assumption that the total concentration of  $E_1$  and  $E_2$  is constant, is debatable and the question, whether or not conservation of moiety with respect to enzymes  $E_1$  and  $E_2$  is necessary for multistationarity is natural – especially, if the focus is on model discrimination. (If conservation of moiety with respect to enzymes  $E_1$  and  $E_2$  is necessary for multistationarity, this phenomenon might be hard to observe experimentally).

$I\!\!R^9_{\delta_i}$	$p_{S^{\delta_i}}$	Representation of $v^{\delta_i} \in S_{\mathcal{N}_4} \cap \mathbb{R}^9_{\delta_i}$
$I\!\!R^9_{\delta_1}$	6	$v^{\delta_{1}} = \left(-\beta_{1} - \beta_{3} - \beta_{4} - \beta_{5}, -\beta_{1}, -\beta_{2}, \beta_{3}, \beta_{1} + \beta_{2}, \beta_{4}, -\beta_{5}, \beta_{5} + \beta_{6}, -\beta_{6}\right)^{T}$
$I\!\!R^9_{\delta_2}$	6	$v^{\delta_2} = ig( -eta_3 - eta_4 - eta_5, 0, -eta_2, eta_3, eta_2, eta_4, -eta_5, eta_5 + eta_6, -eta_6 ig)^T$
$I\!\!R^9_s$	8	$v^{\delta_3} = \left(-eta_3 - eta_6, eta_4 + eta_7, -eta_1 - eta_4 - eta_7, -eta_5 - eta_8, eta_1, eta_6 + eta_7 + eta_8, eta_7, eta_8, eta_8$
10 <sub>03</sub>		$-eta_3-eta_4-eta_5,eta_2+eta_3+eta_4+eta_5,-eta_2\Big)^T$
$I\!\!R^9_{\delta_4}$	6	$v^{\delta_4} = ig( -eta_1, eta_2, -eta_2 - eta_4, -eta_3, eta_4, eta_1 + eta_2 + eta_3, 0, eta_6, -eta_6 ig)^T$
$I\!\!R^9_{\delta_5}$	6	$v^{\delta_5} = ig( -eta_1, eta_2, -eta_2 - eta_4, -eta_3, eta_4, eta_1 + eta_2 + eta_3 + eta_5, eta_5, eta_6, -eta_5 - eta_6 ig)^T$
$I\!\!R^9_{\delta_6}$	7	$v^{\delta_{6}} = \left(-\beta_{4} - \beta_{6}, \beta_{5} + \beta_{7}, -\beta_{2} - \beta_{5} - \beta_{7}, 0, \beta_{2}, \beta_{6} + \beta_{7}, -\beta_{4} - \beta_{5}, \beta_{3} + \beta_{4} + \beta_{5}, -\beta_{3}\right)^{T}$
$\mathbb{R}^9_{c}$	8	$v^{\delta_7} = \left(-eta_3 - eta_5 - eta_6, eta_4 + eta_7 + eta_8, -eta_1 - eta_4 - eta_7 - eta_8, eta_5 + eta_7, eta_1, -eta_8, eta_8, eta_$
± 087	5	$eta_6+eta_8,-eta_3-eta_4,eta_2+eta_3+eta_4,-eta_2ig)^T$

Table 4.2: Representations of  $v^{\delta_i} \in S_{\mathcal{N}_4} \cap \mathbb{R}^9_{\delta_i}$  in orthants  $\mathbb{R}^9_{\delta_i}$  that contain both,  $\mathcal{M}_{\mathcal{N}_4}$  and  $\mathcal{S}_{\mathcal{N}_4}$   $(i = 1, \dots, 7, \text{ as defined in (4.16)}).$ 

To tackle this question the assumption that the system is open with respect to total concentrations of  $E_1$  and/or  $E_2$  is considered, by introducing 0, the 'zero'–complex In this Section the same strategy as in Section 4.2 and 4.3 is used to establish multistationarity. The following networks are analysed, where

$$0 = \frac{\kappa_1}{\kappa_2} E_1,$$
 (N7)

for example, incorporates the fact that  $E_1$  is fed into the system with a constant rate  $\kappa_1$  and is removed from the system with a rate that is proportional to its concentration ( $\kappa_2$  being the rate constant). Thus, in this case, the 'zero'-complex represents the effect of the other layers of the signal transduction network on  $E_1$ . In a similar way

$$0 = \frac{\kappa_3}{\kappa_4} E_2, \qquad (\mathcal{N}_8)$$

and

$$E_1 \xrightarrow{\kappa_2} 0 \xrightarrow{\kappa_3} E_2 \qquad (\mathcal{N}_9)$$

cover the fact that either  $E_2$  alone or  $E_1$  and  $E_2$  together are introduced to the system. In the remainder of this Section the nine networks that can be composed by combining one network of  $\mathcal{N}_4 - \mathcal{N}_6$  with one network of  $\mathcal{N}_7 - \mathcal{N}_9$  will be analysed with respect to multistationarity.

Adding one of the networks  $\mathcal{N}_7 - \mathcal{N}_9$  to one of the networks  $\mathcal{N}_4 - \mathcal{N}_6$  will affect the number of conserved moieties in the combined network: in the case of network  $\mathcal{N}_7$  or  $\mathcal{N}_8$  only two conserved moieties exist: total concentration  $A_{tot}$  of A and  $E_{1tot}$  of  $E_1$  for combinations with  $\mathcal{N}_8$  and  $A_{tot}$  and  $E_{2tot}$ , the total concentration of  $E_2$ , for combinations with  $\mathcal{N}_7$ . For all combinations involving  $\mathcal{N}_9$ , only one conserved moiety remains,  $A_{tot}$ . Using the same steps as in Section 4.2, the equation  $Y^{(L)}{}^T \mu = \ln \frac{E\nu}{E\lambda}$  was analysed for each of the nine networks. As in Chapter 3, let p be the number of generators of ker( $Y I_a$ )  $\cap \mathbb{R}_{\geq 0}^r$ . As in Section 4.2,  $\nu$  and  $\lambda \in \mathbb{R}_{\geq 0}^p$  can be determined independent of  $\mu \in \mathbb{R}^n$  and  $\mathcal{M}$ , as defined in (3.31) is a linear subspace of  $\mathbb{R}^n$ , for each network. Table 4.3 contains for each network a matrix M whose image is  $\mathcal{M}$  (i.e.  $\mathcal{M} = [M]$ ). If multistationarity is possible, Table 4.3 also contains the orthant where  $\mathcal{M} \cap \mathbb{R}_6^n \neq \emptyset$  and  $S_i \cap \mathbb{R}_\delta^n \neq \emptyset$ . In Appendix A.4 the equations  $Y^{(L)T} \mu = \ln \frac{E\nu}{E\lambda}$  are solved for each network in Table 4.3.

Not very surprisingly, for all networks involving  $N_5$  and  $N_6$  multistationarity is excluded and for networks  $N_4+N_7$  and  $N_4+N_8$  multistationarity is possible. Interestingly for network  $N_4+N_9$  multistationarity is **impossible**, even though the difference to  $N_4 + N_7$  and  $N_4 + N_8$  seems very small. This result is obvious, however, if  $\mathcal{M}_{\mathcal{N}_4+\mathcal{N}_9}$  and  $S_{\mathcal{N}_4+\mathcal{N}_9}$  are analysed:  $S_{\mathcal{N}_4+\mathcal{N}_9}$  is defined by

$$W_{N_4+N_9}^T x = 0$$
 (4.19a)

with

$$W_{\mathcal{N}_4+\mathcal{N}_9} = (1, 0, 1, 1, 1, 1, 0, 1, 1)^T$$
 (4.19b)

Network	$M_i$ with $[M_i] = \mathcal{M}_i$	Sign compatible orthants $\delta$	multistationarity
$\mathcal{N}_4 + \mathcal{N}_7$	$M^{T} = \begin{bmatrix} 1, 0, 1, 0, 0, -1, 1, 0, 1\\ 0, 0, 0, 1, 1, 2, -1, 1, 0 \end{bmatrix}$	$\delta^T = \left[ \begin{array}{c} -1, 0, -1, 1, 1, 1, -1, 1, -1 \end{array} \right]$	Yes
$\mathcal{N}_4 + \mathcal{N}_8$	$M^T = \begin{bmatrix} 2, -1, 1, 1, 0, 0, 0, 0, 1\\ -1, 1, 0, 0, 1, 1, 0, 1, 0 \end{bmatrix}$	$\delta^T = \left[ \begin{array}{c} -1, 1, -1, -1, 1, 1, 0, 1, -1 \end{array} \right]$	Yes
$\mathcal{N}_4 + \mathcal{N}_9$	$M^T = \left[ \begin{array}{c} 1, 0, 1, 1, 1, 1, 0, 1, 1 \end{array} \right]$	$\delta = \emptyset$	No
$\mathcal{N}_5 + \mathcal{N}_7$	$M^{T} = \begin{bmatrix} 0, 0, 0, 0, 1, -1, 0, 1, 0\\ 1, 0, 1, 1, 0, 1, 1, 0, 1 \end{bmatrix}$	$\delta = \emptyset$	No
$\mathcal{N}_5 + \mathcal{N}_8$	$M^{T} = \begin{bmatrix} -1, 1, 0, 0, 0, 0, 0, 0, 0 \\ 1, 0, 1, 1, 1, 0, 1, 1, 1 \end{bmatrix}$	$\delta = \emptyset$	No
$N_5+N_9$	$M^T = \begin{bmatrix} 1, 0, 1, 1, 1, 0, 1, 1, 1 \end{bmatrix}$	$\delta = \emptyset$	No
$\mathcal{N}_6 + \mathcal{N}_7$	$M^T = \begin{bmatrix} 0, 0, 0, 0, -1, 1, 0, 0\\ 1, 0, 1, 1, 1, 0, 1, 1 \end{bmatrix}$	$\delta = \emptyset$	No
$\mathcal{N}_6 + \mathcal{N}_8$	$M^{T} = \begin{bmatrix} -1, 1, 0, 0, 0, 0, 0, 0 \\ 1, 0, 1, 1, 1, 0, 1, 1 \end{bmatrix}$	$\delta = \emptyset$	No
$N_6+N_9$	$M^T = \left[ \begin{array}{c} 1, 0, 1, 1, 1, 0, 1, 1 \end{array} \right]$	$\delta = \emptyset$	No

Table 4.3: Solutions for all combinations of  $\mathcal{N}_4$ ,  $\mathcal{N}_5$  and  $\mathcal{N}_6$  with  $\mathcal{N}_7$ ,  $\mathcal{N}_8$  and  $\mathcal{N}_9$ .

From Table 4.3 it follows that  $[W_{\mathcal{N}_4+\mathcal{N}_9}] = \mathcal{M}_{\mathcal{N}_4+\mathcal{N}_9}$  and thus  $\mathcal{M}_{\mathcal{N}_4+\mathcal{N}_9} \perp S_{\mathcal{N}_4+\mathcal{N}_9}$ . Thus  $\mathcal{M}_{\mathcal{N}_4+\mathcal{N}_9}$ and  $S_{\mathcal{N}_4+\mathcal{N}_9}$  **cannot** be sign compatible: recall that  $\mathcal{M}_{\mathcal{N}_4+\mathcal{N}_9}$  and  $S_{\mathcal{N}_4+\mathcal{N}_9}$  are sign compatible, if and only if  $\exists \mu \in \mathcal{M}_{\mathcal{N}_4+\mathcal{N}_9}$  and  $\exists v \in S_{\mathcal{N}_4+\mathcal{N}_9}$  with sign  $(\mu) = \text{sign}(v)$ . In this case  $\langle \mu, v \rangle \geq 0$  must hold (and equality holds iff  $\mu = 0$  and/or v = 0). Thus, as  $\langle \mu, v \rangle = 0$ , for all  $\mu \in \mathcal{M}_{\mathcal{N}_4+\mathcal{N}_9}, v \in S_{\mathcal{N}_4+\mathcal{N}_9}$  and  $\mu$ ,  $v \neq 0$ ,  $\mu$  and v cannot be sign compatible, and multistationarity is excluded by Corollary 2.

#### 4.5 Model discrimination using steady state information

Up to this point it has been established that only network  $\mathcal{N}_4$  can exhibit multistationarity. In this section a method to use this result for model discrimination is discussed, based on Theorem 3. Theorem 3 can on the one hand be used to discard  $\mathcal{N}_5$  and  $\mathcal{N}_6$ , while on the other hand it can be used to falsify  $\mathcal{N}_4$ , given the necessary experimental data. However, the results obtained in Section 4.4 can explain, why this data might be hard to obtain.

Using Theorem 3 for model discrimination Already [45] have argued that the fact that among  $\mathcal{N}_4, \mathcal{N}_5, \mathcal{N}_6$  only  $\mathcal{N}_4$  can exhibit multistationarity can be used for model discrimination: their suggestion and a consequence of Section 4.3 is that if multistationarity was found experimentally for the activation of an MAPK(K), the mechanism involved has to be the one represented by  $\mathcal{N}_4$  (as the other hypotheses cannot give rise to multistationarity). However, [45] use numerical tools to show multistationarity for  $\mathcal{N}_4$  and fail to show multistationarity for  $\mathcal{N}_5$  and  $\mathcal{N}_6$  using numerical tools as well. Thus their conclusion is only valid because of the results obtained in Theorem 3: as for  $\mathcal{N}_5$  and  $\mathcal{N}_6$  no orthant with  $\mathcal{M}_j \cap \mathbb{R}_6^n \neq \emptyset$  and  $S_j \cap \mathbb{R}_6^n \neq \emptyset$ ,  $j = \mathcal{N}_5, \mathcal{N}_6$  exists, it is guaranteed that multistationarity can be excluded for these networks for any conceivable parameter vector. Using numerical tools alone, this result cannot be obtained.

But even if multistationarity is verified experimentally, it is still possible to falsify  $\mathcal{N}_4$  using Theorem 3: to this end let p and q be two positive steady states for  $\mathcal{N}_4$ , with  $W_{\mathcal{N}_4}^T p = W_{\mathcal{N}_4}^T q$ . From Section 4.2.1 follows that the vector  $\mu$  defined by p and q as  $\mu_i = \ln \frac{q_i}{p_i}$ ,  $i = 1, \ldots, 9$  is an element of the linear

subspace  $\mathcal{M}_{\mathcal{N}_4}$ . From Theorem 3 follows that there exists a vector  $v \in \mathcal{S}_{\mathcal{N}_4}$  with  $\operatorname{sign}(v) = \operatorname{sign}(\mu)$ . Now suppose two steady states a and b have been obtained experimentally for  $\mathcal{N}_4$  with  $W_{\mathcal{N}_4}^T a = W_{\mathcal{N}_4}^T b$ . Then, as mentioned in [45] and discussed in [11, 12] neither phosphorylation nor dephosphorylation nor both can be realized using a processive mechanism. However, the conclusion that in this case a distributive mechanism is used for both, phosphorylation and dephosphorylation, as suggested in the aforementioned references, is only valid if there exists a vector  $v \in \mathcal{S}_{\mathcal{N}_4}$  with  $\operatorname{sign}(v) = \operatorname{sign}(\mu)$  for  $\mu$  defined as  $\mu_i = \ln \frac{b_i}{a_i}$ ,  $i = 1, \ldots, 9$ . If this was not the case,  $\mathcal{N}_4$  would not be compatible with experimental data and could thus also be discarded. A similar argument has been used to discard hypotheses regarding catalytic mechanisms in [21, 22, 23]. **Consequences of Section 4.4** Section 4.4 implies that multistationarity might be hard to observe experimentally: for multistationarity to occur, total concentration of either kinase  $E_1$  or phosphatase  $E_2$  has to be constant, that is the system must be closed either with respect to the kinase or the phosphatase.

Classically, signal transduction has been considered as a process driven by kinases, while phosphatases just act as passive, housekeeping enzymes and most mathematical models are set up accordingly. According to this paradigm total concentration of  $E_2$  is constant and multistationarity is thus possible (if the mechanism involved is the one described by  $\mathcal{N}_4$ ). However, recent studies indicate that phosphatases play a major role, and that they are tightly and actively regulated [2, 49]. This implies that the system has to be open with respect to  $E_2$  as well, thus challenging the possibility of the existence of multistationarity.

## Chapter 5

# Multistationarity in cell cycle regulation

In this chapter two network hypotheses regarding control of the transition from G1 to S phase in the cell cycle of *Saccharomyces cerevisiae* are analysed. The key components and their proposed interaction are displayed in Fig. 5.1. In a very simple form, the biochemical network consists of two regulators that



Figure 5.1: Network structure for the G1/S model. Arrows indicate biochemical reactions between cell cycle regulators (boxes). Solid lines represent elementary reactions and dotted lines catalyzed reactions (composite reactions). Kinetic parameters  $k_i$  are displayed next to the arrows. Free Clb2-CDK complexes have been omitted. Components in gray denote degradation products. An unspecified external signal controls the activity of Cdc14p phosphatase. Clb2-CDK phosphorylates the inhibitor Sic1p either in its unbound form (left,  $k_{19-21}$ ), or when it is bound to a second molecule of the kinase complex (right,  $k_{10-12}$ ).

mutually inhibit each other. Cyclin-dependent kinase (CDK), when associated with the mitotic cyclin Clb2p, promotes entry into mitosis through phosphorylation of its target proteins. Simultaneously, this activity prevents the exit from mitosis and subsequent passage to the G1 phase of the cell cycle. The competitive CDK inhibitor Sic1p is one component responsible for inactivating Clb2-CDK at the end of mitosis. Mitotic kinase activity, however, can phosphorylate Sic1p, thus targeting the inhibitor for rapid, proteasome-dependent degradation. The transition to a G1 state with high Sic1p concentration and low Clb2-CDK activity therefore requires activation of the phosphatase Cdc14p and concomitant stabilization of Sic1p [46].

Importantly, the transition between the cell cycle phases requires only transient Cdc14 activity as an input (trigger) signal for the bi-stable switch. Hysteresis, which means that at least two stable steadystate output signals of the system exist for an identical input signal, underlies many cellular switches [63]. It depends on the system's history, whether, for instance, low or high Sic1p concentration will be established. For the G1/S system, a transient activation of the phosphatase Cdc14p should move the system from the mitotic branch of low Sic1p concentration to the upper branch representing a G1 state with high Sic1p abundance and, consequently, low mitotic kinase activity. Any potentially valid network hypothesis has to represent this qualitative behavior, that is the ODEs must admit two stable positive steady states; one representing the G1- and one representing the S-phase. Thus the ODEs derived from any valid network hypothesis have to admit multistationarity for some conceivable parameter vector (recall that multistationarity in the sense of this work requires the existence of at least two positive steady states).

Here, two similar network structures are considered. In one alternative, Clb2-CDK phosphorylates free Sic1p (binary complex model, see network  $\mathcal{N}_{10}$ ), whereas in the other alternative, the already bound inhibitor is a substrate for a second kinase molecule (ternary complex model, see network  $\mathcal{N}_{11}$ ): that is, in network  $\mathcal{N}_{10}$  phosphorylated Sic1 is produced by the reaction  $Clb+Sic1 = Sic1 \cdot Clb \rightarrow Sic1P+Clb$ , while in  $\mathcal{N}_{11}$  phosphorylated Sic1 is produced by the reactions  $Clb \cdot Sic1 + Clb = Sic1 \cdot Clb \rightarrow Sic1 \cdot Clb \rightarrow Clb \cdot Sic1P + Clb$  and  $Clb + Sic1P = Clb \cdot Sic1P$ . (Note that even though  $\mathcal{N}_{10}$  contains the reaction  $Clb + Sic1P = Clb \cdot Sic1P$  as well, the source of Sic1P-production is the reaction  $Sic1 \cdot Clb \rightarrow Clb + Sic1P$ ) Both alternatives are biologically plausible, yet hard to distinguish experimentally.

Each network has n = 9 species and m = 17 complexes in r = 18 reactions. Note that the zero-complex 0 is associated with a nine dimensional zero vector  $y_1 = (0, ..., 0)$ . It incorporates that each system is open with respect to Sic1 and its phosphorylated form Sic1P: Sic1 can enter and leave the system, Sic1P can leave the system (see Fig. 5.1).

(A) Binary complex model:

$$\begin{split} [\operatorname{Sic1P}] \xrightarrow{\mathbf{k}_{3}} [0] \xrightarrow{\mathbf{k}_{1}}_{\mathbf{k}_{2}} [\operatorname{Sic1}] \\ [\operatorname{Sic1} \cdot \operatorname{Clb}] \xrightarrow{\mathbf{k}_{4}}_{\mathbf{k}_{5}} [\operatorname{Clb}] + [\operatorname{Sic1}] \xrightarrow{\mathbf{k}_{6}}_{\mathbf{k}_{7}} [\operatorname{Clb} \cdot \operatorname{Sic1}] \\ & & & & & & & \\ & & & & & \\ & & & & & & \\ & & & & & \\ & & & & & \\ & & & & & & \\ & & & & & \\ & & & & & \\ & & & & & \\ & & & & & \\ & & & & & \\ & & & & & \\ & & & & & \\ & & & & & \\ & & & & & \\ & & & & & \\ & & & & & \\ & & & & & & \\ & & & & & & \\ & & & & & & \\ & & & & & & \\ & & & & & & \\ & & & & & & \\ & & & & & & \\ & & & & & & \\ & & & & & & \\ & & & & & & \\ & & & & & & \\ & & & & & & \\ & & & & & & \\ & & & & & & \\ & & & & & & \\ & & & & & & \\ & & & & & & \\ & & & & & \\ & & & & & & & \\ & & & & & & & \\ & & & & & & & \\ & & & & & & & \\ & & & & & & & \\ & & & & & & & \\ & & & & & & &$$

(B) Ternary complex model:

$$\begin{split} & [\operatorname{Sic1P}] \xrightarrow{k_3} [0] \xrightarrow{k_1}_{k_2} [\operatorname{Sic1}] \\ & [\operatorname{Clb}] + [\operatorname{Sic1}] \xrightarrow{k_4}_{k_5} [\operatorname{Clb} \cdot \operatorname{Sic1}] \\ & [\operatorname{Clb}] + [\operatorname{Sic1P}] \xrightarrow{k_7}_{k_8} [\operatorname{Clb} \cdot \operatorname{Sic1P}] \\ & [\operatorname{Clb}] + [\operatorname{Sic1P}] \xrightarrow{k_7}_{k_8} [\operatorname{Clb} \cdot \operatorname{Sic1P}] \\ & [\operatorname{Clb} \cdot \operatorname{Sic1}] + [\operatorname{Clb}] \xrightarrow{k_{13}}_{k_{14}} [\operatorname{Clb} \cdot \operatorname{Sic1} \cdot \operatorname{Clb}] \xrightarrow{k_{12}}_{\rightarrow} [\operatorname{Clb} \cdot \operatorname{Sic1P}] + [\operatorname{Clb}] \\ & [\operatorname{Sic1P}] + [\operatorname{Cdc14}] \xrightarrow{k_{13}}_{k_{14}} [\operatorname{Sic1P} \cdot \operatorname{Cdc14}] \xrightarrow{k_{15}}_{\rightarrow} [\operatorname{Sic1}] + [\operatorname{Cdc14}] \\ & [\operatorname{Clb} \cdot \operatorname{Sic1P}] + [\operatorname{Cdc14}] \xrightarrow{k_{16}}_{k_{17}} [\operatorname{Clb} \cdot \operatorname{Sic1P} \cdot \operatorname{Cdc14}] \xrightarrow{k_{18}}_{\rightarrow} [\operatorname{Clb} \cdot \operatorname{Sic1}] + [\operatorname{Cdc14}] \end{split}$$

The ternary complex network as defined above has deficiency  $\delta = 5$ , the binary complex model has deficiency  $\delta = 4$ . Hence, the advanced deficiency algorithm has to be applied for its analysis. As it turns out, the implementation in the *Chemical Reaction Network Toolbox* cannot decide about multistationarity. The algorithm returns that the system may or may not have multiple steady states, as nonlinear inequalities have to be considered in both cases. Thus, in a first step, the subnetworks defined by the stoichiometric generators of  $\mathcal{N}_{10}$  and  $\mathcal{N}_{11}$  are analysed to decide about multistationarity. This is described in Section 5.1. In Section 5.2 and 5.3 networks  $\mathcal{N}_{10}$  and  $\mathcal{N}_{11}$  are analysed using the methods described in Chapter 3. This chapter closes with some concluding remarks in Section 5.4.

**Remark 13.** Note that even though the CRNT toolbox cannot decide about multistationarity as it needs to solve nonlinear equations, the methods described in Chapter 3 can, by analysis of linear inequalities.

### 5.1 Subnetwork analysis

The generators of ker(Y  $I_a$ )  $\cap \mathbb{R}_{\geq 0}^{18}$  for network  $\mathcal{N}_{10}$  and  $\mathcal{N}_{11}$  are given in (5.1) and (5.2), respectively. Visual inspection shows that  $E_{\mathcal{N}_{10}}$  contains six stoichiometric generators (columns 7–12), while  $E_{\mathcal{N}_{11}}$  contains five stoichiometric generators (columns 7–11). Figures 5.2 and 5.3 contain the subnetworks defined by the stoichiometric generators of network  $\mathcal{N}_{10}$ , while Figures 5.4 and 5.5 depict those of network  $\mathcal{N}_{11}$ . Visual inspection of the subnetworks confirms that each linkage class contains exactly one terminal strong linkage class. Thus, by Fact 4, the *Deficiency One Algorithm* is applicable. Using the CRNT toolbox [29] one can establish that for network  $\mathcal{N}_{10}$  no subnetwork admits multistationarity, while for network  $\mathcal{N}_{11}$  the subnetworks corresponding to generators  $E_9$ ,  $E_{10}$  and  $E_{11}$  (c.f. (5.2) and Figure 5.5) can have multiple steady states. In contrast, it is guaranteed that those networks defined by  $E_7$  and  $E_8$  cannot admit multistationarity. Thus, by analyzing the subnetworks defined by stoichiometric generators one can establish that multistationarity is possible for network  $\mathcal{N}_{11}$ , while this is still unclear for network  $\mathcal{N}_{10}$ . To decide about multistationarity for network  $\mathcal{N}_{10}$  the methods described in Chapter 3 are applied in Section 5.2 (for completeness, these methods are also applied to network  $\mathcal{N}_{11}$  in Section 5.3).

53

(5.1)

### CHAPTER 5. MULTISTATIONARITY IN CELL CYCLE REGULATION

	[1]	0	0	0	0	0	1	0	0	1	1 ]
	1	0	0	0	0	0	0	0	0	0	0
	0	0	0	0	0	0	0	0	0	0	1
	0	1	0	0	0	0	1	0	1	1	1
	0	1	0	0	0	0	0	0	0	0	0
	0	0	0	0	0	0	1	0	0	0	0
	0	0	1	0	0	0	0	0	0	0	0
	0	0	1	0	0	0	0	0	1	0	1
F	0	0	0	0	0	0	0	0	0	1	0
$L_{N_{11}} -$	0	0	0	1	0	0	0	1	1	1	1
	0	0	0	1	0	0	0	0	0	0	0
	0	0	0	0	0	0	0	1	1	1	1
	0	0	0	0	1	0	0	0	1	0	0
	0	0	0	0	1	0	0	0	0	0	0
	0	0	0	0	0	0	0	0	1	0	0
	0	0	0	0	0	1	0	1	0	0	0
	0	0	0	0	0	1	0	0	0	0	0
	0	0	0	0	0	0	0	1	0	0	0
	LU	0	0	0	0	0	0	T	0	0	

(5.2)



 $E_7 = (1, 0, 0, 0, 0, 1, 0, 1, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0)^T$ 





 $E_9 = (0, 0, 0, 0, 1, 0, 0, 0, 1, 0, 0, 0, 1, 0, 1, 0, 0, 0)^T$ 



Figure 5.2: Subnetworks corresponding to stoichiometric generators  $E_7 - E_9$  for the binary complex model. Subnetworks are denoted in black, while gray color indicates reactions not used.



 $E_{10} = (0, 0, 0, 0, 1, 0, 1, 0, 1, 1, 0, 0, 0, 0, 0, 1, 0, 1)^T$ 

 $E_{11} = (1, 0, 0, 0, 1, 0, 0, 0, 1, 1, 0, 1, 0, 0, 0, 0, 0, 0)^T$ 



 $E_{12} = (1, 0, 0, 0, 1, 0, 0, 1, 1, 1, 0, 0, 0, 0, 0, 1, 0, 1)^T$ 



Figure 5.3: Subnetworks corresponding to stoichiometric generators  $E_{10} - E_{12}$  for the binary complex model. Subnetworks are denoted in black, while gray color indicates reactions not used.



 $E_7 = (1, 0, 0, 1, 0, 1, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0)^T$ 





Figure 5.4: Subnetworks corresponding to stoichiometric generators  $E_7$  and  $E_8$  for the ternary complex model. Subnetworks are denoted in black, while gray color indicates reactions not used.

 $E_9 = (0, 0, 0, 1, 0, 0, 0, 1, 0, 1, 0, 1, 1, 0, 1, 0, 0, 0)^T$ 



 $E_{10} = (1, 0, 0, 1, 0, 0, 0, 0, 1, 1, 0, 1, 0, 0, 0, 0, 0, 0)^T$ 





Figure 5.5: Subnetworks corresponding to the stoichiometric generators  $E_9 - E_{11}$  for the ternary complex model. Subnetworks are denoted in black, while gray color indicates reactions not used.

## 5.2 Analysis of the complete network $N_{10}$

As described in Section 5.1, analysis of the subnetworks of  $\mathcal{N}_{10}$  defined by stoichiometric generators using CRNT established that multistationarity is excluded for all subnetworks. As multistationarity is still possible for the overall network, this is analysed using the methods described in Chapter 3. For network  $\mathcal{N}_{10}$  it was not possible to determine  $\nu, \lambda \in \mathbb{R}^{12}_{>0}$  independent of  $\mu \in \mathbb{R}^9$ . (Note that this is no guarantee, that no such solution exists.) Thus the equation

$$Y^{(L)T} \mu = \ln \frac{E \lambda}{E \nu}$$

is considered in the following form:

$$\begin{bmatrix} E & -\operatorname{diag}\left(e^{Y^{(L)^{T}}\mu}\right) E\end{bmatrix} \begin{pmatrix} \nu \\ \lambda \end{pmatrix}, \ \begin{pmatrix} \nu \\ \lambda \end{pmatrix} > 0$$

Using an unimodular matrix  $U \in \mathbb{R}^{18 \times 18}$  with  $UE = \begin{bmatrix} I_{11} & v \\ 0_{7 \times 11} & 0_{7 \times 1} \end{bmatrix}$ , where ker  $(E) = \operatorname{span}\left(\begin{pmatrix} v \\ -1 \end{pmatrix}\right)$ , one obtains the following system of equations

$$\underbrace{U\left[E - \operatorname{diag}\left(e^{Y^{(L)^{T}}\mu}\right)E\right]}_{=:H(\mu)} \begin{pmatrix}\nu\\\lambda\end{pmatrix} = 0, \ \begin{pmatrix}\nu\\\lambda\end{pmatrix} > 0,$$

where  $H(\mu)$  is given at the end of this Section on page 67. Note that the nonzero elements of rows 14, 17 and 18 of  $H(\mu)$  always have the same sign, for every  $\mu$ . Thus, positive  $\nu$ ,  $\lambda$  only exists for those values of  $\mu$ , where rows 14, 17 and 18 of  $H(\mu)$  are identical to zero, thus  $\mu$  must satisfy

$$\mu_9 = \mu_1 + \mu_3$$
 (5.3a)

$$\mu_7 = \mu_2 + \mu_6 \tag{5.3b}$$

$$\mu_8 = \mu_5 + \mu_6$$
 (5.3c)

Note that (5.3a), (5.3b) and (5.3c) are the only restrictions on the  $\mu_i$ . Thus  $\mu$  can be represented as (using new variables  $\kappa \in \mathbb{R}^6$ ):

$$\mu = M_{N_{10}} \kappa$$
, (5.4a)

where

$$M_{\mathcal{N}10} = \begin{bmatrix} 1 & 0 & 0 & 0 & 0 & 0 \\ 0 & 1 & 0 & 0 & 0 & 0 \\ 0 & 0 & 1 & 0 & 0 & 0 \\ 0 & 0 & 0 & 1 & 1 & 0 & 0 \\ 0 & 0 & 0 & 0 & 1 & 1 \\ 0 & 1 & 0 & 0 & 0 & 1 \\ 1 & 0 & 1 & 0 & 0 & 0 \end{bmatrix} \kappa.$$
(5.4b)

Apart from  $h_{12,23} = -1 + e^{\mu_4} - e^{\mu_8} + e^{\mu_9}$  the signs of all nonzero elements of  $H(\mu)$  can be determined by linear inequalities in the  $\mu_i$ . By subtracting column 23 from column 20 one obtains a matrix  $\tilde{H}$ , where the signs of all elements can be determined by linear inequalities. Note that this corresponds to a change of variables  $\begin{pmatrix} \nu \\ \lambda \end{pmatrix} = T \begin{pmatrix} \tilde{\nu} \\ \tilde{\lambda} \end{pmatrix}$  where

 $T = \begin{bmatrix} e_1 & \dots & e_{22} & -e_{23} + e_{20} & e_{24} \end{bmatrix}$ 

Thus

$$H\left(\mu\right) \; T \; \begin{pmatrix} \tilde{\nu} \\ \tilde{\lambda} \end{pmatrix} = 0, \; T \; \begin{pmatrix} \tilde{\nu} \\ \tilde{\lambda} \end{pmatrix} > 0,$$

where  $T\begin{pmatrix} \tilde{\nu}\\ \tilde{\lambda} \end{pmatrix} > 0$  is equivalent to  $\nu, \lambda > 0$  and

$$\lambda_8 - \lambda_{11} > 0.$$
 (5.5)

Note that this condition can easily be included by adding a new variable  $s_0 > 0$ . Then one has to determine positive solutions to

$$\underbrace{\begin{bmatrix} H\left(\mu\right) T & 0_{24\times 1} \\ e_{20}^T - e_{23}^T & -1 \end{bmatrix}}_{=:\tilde{H}(\mu)} \begin{pmatrix} \nu \\ \lambda \\ s_0 \end{pmatrix} = 0$$

The matrix  $\tilde{H}(\mu)$  is given at the end of this section on page 68. Note that equations (5.3a) – (5.3c) have been incorporated  $\tilde{H}(\mu)$ . Further note that each of row 1, 2, 4, 5, 6, 9, 10 contains exactly two nonzero entries, thus they are equal to

$$\nu_i - e^{\langle y_j^{(L)}, \mu \rangle} \lambda_i = 0,$$

for some linear form  $\langle y_j^{(L)}, \mu \rangle$ . Further note that rows 13–18 are independent of the  $\nu_i, i = 1, ..., 12$ . Thus it is straightforward to obtain

$$\nu_1 = e^{\mu_1} \lambda_1$$
 (5.6a)

$$\nu_2 = e^{\mu_1 + \mu_3} \,\lambda_2 \tag{5.6b}$$

$$\nu_4 = e^{\mu_5} \lambda_4 \tag{5.6c}$$

$$\nu_5 = e^{\mu_2 + \mu_6} \,\lambda_5 \tag{5.6d}$$

$$\nu_6 = e^{\mu_5 + \mu_6} \,\lambda_6 \tag{5.6e}$$

$$\nu_9 = e^{\mu_2 + \mu_6} \,\lambda_9 \tag{5.61}$$

$$\nu_{10} = e^{\mu_5} \lambda_{10} \tag{5.6g}$$

Further note that, if  $\mu$  satisfies (5.3a) – (5.3c), then  $\lambda_2$ ,  $\lambda_5$  and  $\lambda_6$  only occur in row 2, 5 and 6, respectively. That is,  $\lambda_2$ ,  $\lambda_5$  and  $\lambda_6$  are only involved in equations (5.6b), (5.6d) and (5.6e). Thus  $\lambda_2$ ,  $\lambda_5$  and  $\lambda_6$  can be chosen freely in  $\mathbb{R}_{>0}$ . Therefore it remains to determine  $x = (\nu_3, \nu_7, \nu_8, \nu_{11}, \nu_{12}, \lambda_1, \lambda_3, \lambda_4, \lambda_7, \lambda_8, \lambda_9, \lambda_{10}, \lambda_{11}, \lambda_{12}, s_0)^T$ . This amounts to solving a system of equations defined by row 3, 7, 8, 11–19 and columns 3, 7, 8, 11, 12, 13, 15, 16, 19–26 of matrix  $\tilde{H}(\mu)$ . The resulting submatrix  $Q(\mu)$  is given on page 69 at the end of this Section. Using the unimodular transformation matrix

$$TT = \begin{bmatrix} 1 & 0 & 0 & 0 & 0 & 0 & 0 & 0 & 0 \\ 0 & 1 & 0 & 1 & 0 & 0 & 0 & 0 & 0 \\ 0 & 0 & 1 & 0 & 0 & 0 & 0 & 0 & 0 \\ 0 & 0 & 1 & 1 & 0 & 0 & 0 & 0 & 0 \\ 0 & 1 & 1 & 1 & 1 & 0 & 0 & 0 & 0 \\ 0 & 0 & 1 & 0 & 0 & 1 & 0 & 0 & 0 \\ 1 & 1 & 0 & 1 & 0 & 0 & 1 & 0 & 0 \\ 0 & 0 & 0 & 1 & 0 & 0 & 0 & 1 & 0 \\ 0 & 0 & 0 & 0 & 0 & 0 & 0 & 0 & 1 \end{bmatrix}$$
(5.7)

one obtains the much simpler matrix  $Q_s(\mu) := TT Q(\mu)$ , displayed on page 69. Using

$$q_1 = -e^{\mu_1 + \mu_3} + e^{\mu_2 + \mu_6} \qquad (5.8a)$$

$$q_2 = -e^{\mu_1 + \mu_3} + e^{\mu_5} \qquad (5.8b)$$

$$q_3 = -e^{\mu_1 + \mu_3} + e^{\mu_5 + \mu_6} \tag{5.8c}$$

$$q_4 = -1 + e^{\mu_1} \tag{5.8d}$$

$$q_5 = -1 + e^{\mu_5} \tag{5.8e}$$

$$q_6 = -e^{\mu_2 + \mu_3} + e^{\mu_5} \tag{5.8f}$$

 $Q_s(\mu)$  can be represented as:

Thus any feasible inequality system

$$\begin{bmatrix} V & -\operatorname{diag}\left(\sigma\right) \end{bmatrix} \binom{\kappa}{s} = 0, \ s > 0 \tag{5.10}$$

with  $\sigma \in \{-1, 0, 1\}^6$  and

$$V := \begin{bmatrix} -1 & 1 & -1 & 0 & 0 & 1 \\ -1 & 0 & -1 & 0 & 1 & 0 \\ -1 & 0 & -1 & 0 & 1 & 1 \\ 1 & 0 & 0 & 0 & 0 & 0 \\ 0 & 0 & 0 & 0 & 1 & 0 \\ 0 & -1 & -1 & 0 & 1 & 0 \end{bmatrix},$$
(5.11)

defines a feasible signing for  $Q_s(\mu)$ . No signing was found such that  $Q_s(\mu)$  is an  $L^+$ -matrix. However,  $Q_s(\mu)$  is a sign-central matrix for some signings. Thus the results of Section 3.2.2 and 3.4 can be applied.

Note that all  $\mu$  as defined in (5.4a) are contained in the linear subspace  $\mathcal{M}_{\mathcal{N}_{10}} = [\mathcal{M}_{\mathcal{N}_{10}}]$ . Thus, with Fact 3 in mind, all orthants with  $\mathcal{S}_{\mathcal{N}_{10}} \cap \mathbb{R}^9_{\delta} \neq \emptyset$  were determined. Of those all orthants with  $\mathcal{M}_{\mathcal{N}_{10}} \cap \mathbb{R}^9_{\delta} \neq \emptyset$  and (5.10) is feasible for some  $\sigma$  have been determined. That is, all orthants, where

$$\begin{bmatrix} \operatorname{diag}\left(\delta\right) & -M_{\mathcal{N}_{10}} & 0\\ 0 & V & -\operatorname{diag}\left(\sigma\right) \end{bmatrix} \begin{pmatrix} \xi\\ \kappa\\ s \end{pmatrix} = 0, \ \xi, \ s > 0 \tag{5.12}$$

is feasible. Finally all signings, where (5.10) is feasible for some orthant and  $Q(\mu)$  is a sign-central matrix were determined. The orthants are given in (5.13a):

$$\begin{split} \delta_1 &= (-1, -1, 1, -1, 1, -1, -1, 1, -1)^T \\ \delta_2 &= (-1, -1, 1, -1, 1, -1, -1, 1, 1)^T \\ \delta_3 &= (-1, -1, 1, -1, 1, 0, -1, 1, -1)^T \\ \delta_4 &= (-1, -1, 1, -1, 1, 0, -1, 1, -1)^T \\ \delta_5 &= (-1, -1, 1, -1, 1, 1, -1, 1, -1, 1)^T \\ \delta_6 &= (-1, -1, 1, -1, 1, 1, -1, 1, -1)^T \\ \delta_7 &= (-1, -1, 1, 1, 1, 1, -1, -1, 1, -1)^T \\ \delta_9 &= (-1, -1, 1, 1, 1, 1, -1, -1, 1, -1)^T \\ \delta_{10} &= (-1, 0, 1, -1, 1, -1, -1, 1, -1)^T \\ \delta_{11} &= (-1, 0, 1, -1, 1, -1, -1, 1, -1)^T \\ \delta_{12} &= (-1, -1, 1, 1, 1, -1, -1, 1, -1)^T \\ \delta_{11} &= (-1, 0, 1, -1, 1, -1, -1, 1, -1)^T \\ \delta_{11} &= (-1, 0, 1, -1, 1, -1, -1, 1, -1)^T \\ \delta_{14} &= (-1, 1, 1, -1, 1, -1, -1, 1, -1)^T \\ \delta_{15} &= (-1, 1, 1, -1, 1, -1, 1, 1, 1)^T \\ \delta_{16} &= (-1, 1, 1, -1, 1, -1, 1, 1, 1)^T \\ \delta_{16} &= (-1, 1, 1, 1, -1, 1, -1, 1, -1)^T \end{split}$$

the signings in (5.13b):

$$\begin{aligned} & \tau_1 = (-1, 1, 1, -1, 1, 0)^T \\ & \tau_2 = (-1, 0, -1, -1, 1, 0)^T \\ & \tau_3 = (-1, 1, -1, -1, 1, 0)^T \\ & \tau_4 = (-1, 1, 0, -1, 1, 0)^T \\ & \tau_5 = (-1, 0, 1, -1, 1, 0)^T \end{aligned}$$
(5.13b)

Note that if (5.10) is feasible for  $(\delta, \sigma)$ , then it is also feasible for  $(-\delta, -\sigma)$ . Thus there are in fact 32 orthants and five signings. Further note, that not all signings need to hold in every orthant. It is only guaranteed that in every orthant some signing holds. Recall that a sign-central matrix has the property that every matrix with the same sign pattern has a nonnegative nullvector. A nonnegative nullvector yields a nonnegative parameter vector, where some rate constants are zero. In some cases it is possible to derive values for these rate constants using the ideas discussed in Section 3.4. This is the strategy pursued in the remainder of this section. Further analysis of these signings reveals that  $Q_s(\mu)$  contains sub-matrices that are L<sup>+</sup>-matrices. Obviously, by an appropriate filling with zeros, a positive nullvector of a submatrix yields a nonnegative nullvector of the overall matrix. Thus, for each submatrix that is an  $L^+$ -matrix, one obtains a nullvector of  $Q_s(\mu)$ , with known zero pattern. To see this, let  $\mathcal{I}^+ \subseteq 1, \ldots, 15$  contain the indices of those columns of  $Q(\mu)$  that form an  $L^+$ -matrix and let  $\bar{Q} := Q_s(\mu) [\mathcal{I}^+]$  be the corresponding submatrix (after a removal of zero rows, if necessary). Recall that an  $L^+$ -matrix has a positive nullvector, thus let x be a vector of appropriate dimension with  $\bar{Q}x = 0$ . As  $\bar{Q}$  is an  $L^+$  matrix supp  $(x) = \mathcal{I}^+$  follows. Let  $\hat{x} \in \mathbb{R}^{15}_{\geq 0}$  with  $\hat{x}_i = x_i$ , if  $i \in \mathcal{I}^+$ and  $\hat{x}_i = 0$ , otherwise. Clearly  $Q_s(\mu) \hat{x} = 0$  and  $\operatorname{supp}(\hat{x}) = \mathcal{I}^+$ , thus the zero pattern of  $\hat{x}$  is known. Table 5.1 contains, for each signing, the column indices that form  $L^+$ -matrices, after a removal of zero

Column indices	$\sigma_1$	$\sigma_2$	$\sigma_3$	$\sigma_4$	$\sigma_5$	$\sigma_6$	$\sigma_7$	$\sigma_8$	$\sigma_9$	$\sigma_{10}$
$\{3, 6, 8, 10, 11, 12, 15\}$	+		+	+		+	+	+		
$\{3, 6, 10, 11, 12, 15\}$	+		+	+		+	+	+		
$\{6, 8, 11, 12\}$	+		+	+		+	+	+		
$\{6, 11, 12\}$	+		+	+		+	+	+		
$\{6, 8, 12\}$		+			+				+	+
$\{6, 12\}$		+			+				+	+

Table 5.1: Signings and column indices that are  $L^+$ -matrices

rows, were necessary. In total, there are six different sub-matrices that are  $L^+$ -matrices. Consider, for example,  $\sigma_1 = (-1, 1, 1, -1, 1, 0)^T$ . Here the following sub-matrices are  $L^+$ -matrices (note that for each matrix all zero rows have been removed):

$$\begin{aligned} Q\left(\mu\right)[3,\,6,\,8,\,10,\,11,\,12,\,15] = \begin{bmatrix} 1 & 0 & 0 & -1 & -1 & 1 & 0 \\ 1 & -1 & 0 & -1 & 0 & 1 & 0 \\ 1 & 0 & 0 & -1 & 0 & 0 & 0 \\ 0 & 0 & 0 & 1 & 0 & 0 & -1 \end{bmatrix} \\ Q\left(\mu\right)[3,\,6,\,10,\,11,\,12,\,15] = \begin{bmatrix} 1 & 0 & -1 & -1 & 1 & 0 \\ 1 & -1 & -1 & 0 & 1 & 0 \\ 1 & 0 & -1 & 0 & 0 & 0 \\ 0 & 0 & 1 & 0 & 0 & -1 \end{bmatrix} \\ Q\left(\mu\right)[6,\,8,\,11,\,12] = \begin{bmatrix} 0 & 0 & -1 & 1 \\ -1 & 0 & 0 & 1 \end{bmatrix} \\ Q\left(\mu\right)[6,\,11,\,12] = \begin{bmatrix} 0 & -1 & 1 \\ -1 & 0 & 1 \end{bmatrix} \end{aligned}$$

For the signing  $\sigma_2 = (-1, 0, -1, -1, 1, 0)$  the following sub-matrices are  $L^+$ -matrices:

$$Q(\mu)[6, 8, 12] = \begin{bmatrix} -1 & 0 & 1 \end{bmatrix} \qquad \qquad Q(\mu)[6, 12] = \begin{bmatrix} -1 & 1 \end{bmatrix}$$
$\nu_7$ ,  $\nu_8$ ,  $\nu_{11}$ ,  $\nu_{12}$ ,  $\lambda_1$ ,  $\lambda_3$ ,  $\lambda_4$ ,  $\lambda_7$ ,  $\lambda_8$ ,  $\lambda_9$ ,  $\lambda_{10}$ ,  $\lambda_{11}$ ,  $\lambda_{12}$ ,  $s_0$ )<sup>T</sup> and that  $\lambda_2$ ,  $\lambda_5$  and  $\lambda_6$  are free. Thus one obtains

$$\lambda_i \neq 0, \quad \nu_i \neq 0, \quad i = 1, 2, 4, 5, 6, 8, 9, 10.$$

Recall that  $k = \text{diag}(\phi(a^{-1})) E \lambda$ . Using this definition one obtains for k:

$$k = \left(\lambda_1 + \lambda_8 + \lambda_{10}, \frac{\lambda_1}{a_1}, \frac{\lambda_8}{a_2}, \frac{\lambda_2}{a_9}, \frac{\lambda_2 + \lambda_8 + \lambda_9 + \lambda_{10}}{a_1 a_3}, 0, 0, 0, 0, \frac{\lambda_8 + \lambda_9 + \lambda_{10}}{a_9}, \frac{\lambda_4 + \lambda_{10}}{a_2 a_3}, \frac{\lambda_4}{a_5}, \frac{\lambda_{10}}{a_5}, \frac{\lambda_5 + \lambda_9}{a_2 a_6}, \frac{\lambda_5}{a_7}, \frac{\lambda_9}{a_7}, \frac{\lambda_6}{a_5 a_6}, \frac{\lambda_6}{a_8}, 0\right)^T.$$

Thus  $k_i = 0, i = 6, 7, 8, 18$ . To find values for these  $k_i$ , the procedure described in Section 3.4 is applied. To this end it is necessary to find a positive vector v with

$$G_{c} v = 0$$

$$G_{c} = \begin{bmatrix} W_{add}^{T} N_{c} \phi(a) \\ W_{add}^{T} N_{c} \phi(b) \end{bmatrix}$$

where  $b = \text{diag}(e^{\mu}) a$ , with  $\mu = M_{N_{10}} \kappa$  and  $W_{add}^T = (0, 0, 0, 1, 0, 0, 0, 0, 0)$ . Then

$$G_c = \begin{bmatrix} a_1 a_3 & -a_4 & -a_4 & a_8 \\ a_1 a_3 e^{\mu_1 + \mu_3} & -e^{\mu_4} a_4 & -e^{\mu_4} a_4 & e^{\mu_5 + \mu_6} a_8 \end{bmatrix}.$$

Solutions to  $G_c v = 0$  are defined by

$$a_1 a_3 v_1 - a_4 v_2 - a_4 v_3 + a_8 v_4 = 0$$
  
$$a_1 a_3 e^{\mu_1 + \mu_3} v_1 - a_4 e^{\mu_4} v_2 - a_4 e^{\mu_4} v_3 + a_8 e^{\mu_5 + \mu_6} v_4 = 0$$

The first equation is equivalent to

$$a_4 (v_2 + v_3) = a_1 a_3 v_1 + a_8 v_4.$$

Inserting in the second equation yields

$$a_1 a_3 (e^{\mu_1 + \mu_3} - e^{\mu_4}) v_1 + a_8 (-e^{\mu_4} + e^{\mu_5 + \mu_6}) v_4 = 0.$$

The last equation has a positive solution, if and only if either

$$e^{\mu_1 + \mu_3} - e^{\mu_4} > 0$$
 and  $e^{\mu_5 + \mu_6} - e^{\mu_4} < 0$  (5.14a)

or

$$e^{\mu_1 + \mu_3} - e^{\mu_4} < 0 \text{ and } e^{\mu_5 + \mu_6} - e^{\mu_4} > 0.$$
 (5.14b)

Note that (5.14a) and (5.14b) are equivalent to the linear inequalities

$$\mu_1 + \mu_3 - \mu_4 > 0$$
 and  $-\mu_4 + \mu_5 + \mu_6 < 0$  (5.15a)

or

$$\mu_1 + \mu_3 - \mu_4 < 0 \text{ and } -\mu_4 + \mu_5 + \mu_6 > 0.$$
 (5.15b)

Thus multistationarity can be extended to the overall network, if  $\mu$  satisfies the inequalities defined by the signing  $\sigma_1$  together with these additional constraints. For the other sub-matrices given in Table 5.1 it is possible to show that multistationarity can be extended to the overall network, if (5.15a) and (5.15b) hold. There exists exactly one signing such that (5.15a) and (5.15b) hold in some of the orthants given in (5.13a):

$$\sigma_1 = (-1, 1, 1, -1, 1, 0.)'$$
  
(5.16)

 $\sigma_1$  holds in the following orthants:

$$\begin{split} \delta_{1} &= (-1, -1, 1, -1, 1, -1, -1, 1, -1)^{T} \\ \delta_{2} &= (-1, -1, 1, -1, 1, 1, -1, 1, -1)^{T} \\ \delta_{3} &= (-1, -1, 1, 1, 1, 1, -1, -1, 1, -1)^{T} \\ \delta_{4} &= (-1, -1, 1, 1, 1, 1, -1, -1, 1, -1)^{T} \\ \delta_{5} &= (-1, 1, 1, 1, 1, 1, -1, -1, 1, -1)^{T} \\ \delta_{5} &= (-1, 1, 1, 1, 1, 1, -1, -1, 1, -1)^{T} \\ \delta_{7} &= (1, -1, -1, -1, -1, 1, 1, -1, 1)^{T} \\ \delta_{8} &= (1, -1, -1, 1, -1, 1, 1, -1, 1)^{T} \\ \delta_{9} &= (1, 1, -1, -1, -1, -1, 1, 1, -1, 1)^{T} \\ \delta_{10} &= (1, 1, -1, -1, -1, -1, 1, 1, -1, 1)^{T} \\ \delta_{11} &= (1, 1, -1, 1, -1, -1, 1, 1, -1, 1)^{T} \\ \delta_{12} &= (1, 1, -1, 1, -1, -1, 1, 1, -1, 1)^{T} \\ \delta_{12} &= (1, 1, -1, 1, -1, 1, 1, -1, 1)^{T} \end{split}$$

Consider  $\delta_5 = (-1, 1, 1, -1, 1, -1, -1, 1, -1)^T$ . To determine  $\mu$ , the following equations are solved for nonnegative  $\xi$ , s and unconstraint  $\kappa$ :

with V as in (5.11) and

$$V_{add} = \begin{bmatrix} 1 & 0 & 1 & -1 & 0 & 0 \\ 0 & 0 & 0 & -1 & 1 & 1 \end{bmatrix}.$$
 (5.18)

A solution for  $\kappa$  is given by

$$\kappa = \begin{bmatrix} 0 & -2 & -1 & -1 & -2 \\ 1 & 0 & 0 & 0 & 0 \\ 0 & 1 & 1 & 1 & 1 \\ 0 & -1 & 0 & 0 & 0 \\ 1 & 1 & 1 & 1 & 1 \\ -1 & -1 & 0 & -1 & -1 \end{bmatrix} \alpha, \quad \alpha \in \mathbb{R}^5_{>0}.$$

$$(5.19)$$

Choose  $\alpha_i = 1$  to obtain  $\bar{\mu} = (-6, 1, 4, -1, 5, -4, -3, 1, -2)$  and thus

The cone  $\ker\left(Q\left(\bar{\mu}\right)\right)\cap I\!\!R^{15}_{\geq 0}$  is generated by

Choose, for example,

$$\begin{split} x &= E \begin{pmatrix} 1 \\ 1 \\ 1 \end{pmatrix} \\ &= \begin{pmatrix} 0, 0, e, 0, 0, -\frac{-1+e}{-1+\frac{1}{e^6}} - \frac{-1+e^5}{-1+\frac{1}{e^6}}, 0, 1, 0, 1, \\ &e \left(1+e+e^2\right) + e \left(1+e+e^2+e^3+e^4+e^5+e^6\right), 1, 0, 0, 1 \end{pmatrix}^T. \end{split}$$

Thus one obtains

$$\begin{split} \lambda_1 &= -\frac{-1+e}{-1+\frac{1}{e^6}} - \frac{-1+e^5}{-1+\frac{1}{e^6}} \\ \lambda_4 &= 1, \\ \lambda_8 &= 1, \\ \lambda_9 &= e\left(1+e+e^2\right) + e\left(1+e+e^2+e^3+e^4+e^5+e^6\right), \\ \lambda_{10} &= 1. \end{split}$$

Recall that  $\lambda_2, \lambda_5, \lambda_6$  are free, thus choose  $\lambda_2 = \lambda_5 = \lambda_6 = 1$  to obtain

$$\begin{split} \hat{k}_{E}^{*} &= \left(2 - \frac{-1 + e}{-1 + \frac{1}{e^{0}}} - \frac{-1 + e^{5}}{-1 + \frac{1}{e^{0}}}, \frac{e^{6}\left(2 + e + e^{2} + e^{3} + e^{4}\right)}{a_{1}\left(1 + e + e^{2} + e^{3} + e^{4} + e^{5}\right)}, \frac{1}{a_{2}}, \frac{1}{a_{9}}, \\ &\frac{3 + 2e + 2e^{2} + 2e^{3} + e^{4} + e^{5} + e^{6} + e^{7}}{a_{1}a_{3}}, 0, 0, 0, \frac{2 + 2e + 2e^{2} + 2e^{3} + e^{4} + e^{5} + e^{6} + e^{7}}{a_{9}} \\ &\frac{2}{a_{2}a_{3}}, \frac{1}{a_{5}}, \frac{1}{a_{5}}, \frac{1 + 2e + 2e^{2} + 2e^{3} + e^{4} + e^{5} + e^{6} + e^{7}}{a_{2}a_{6}}, \frac{1}{a_{7}}, \frac{2(2 + 2e + 2e^{2} + e^{3} + e^{4} + e^{5} + e^{6})}{a_{7}}, \frac{1}{a_{5}a_{6}}, \frac{1}{a_{8}}, 0\right)^{T}. \end{split}$$

Using  $v \in S \cap \mathbb{R}^9_{\delta_5}$ ,  $v = (-1, 1, 2, -4, 2, -2, -2, 4, -4)^T$  one obtains

and

$$G_c = \left[ \begin{array}{ccc} \frac{2e^6}{1-e^4-e^6+e^{10}} & -\frac{4e}{-1+e} & -\frac{4e}{-1+e} & \frac{4}{-1+e} \\ \frac{2e^4}{1-e^4-e^6+e^{10}} & -\frac{4}{-1+e} & -\frac{4}{-1+e} & \frac{4e}{-1+e} \end{array} \right]$$

For the nullspace of  $G_c$  one obtains

$$\ker \left(G_c\right) = \left[ \begin{array}{ccc} \frac{2\left(-1+e\right)\left(1+e\right)^3\left(1+2\,e^2+2\,e^4+e^6\right)}{e^5} & 1+\frac{1}{e}+e & 0 & 1\\ 0 & -1 & 1 & 0 \end{array} \right]$$

Choosing

$$\kappa_{0} = \left(\frac{2\left(-1+e\right)\left(1+e\right)^{3}\left(1+2e^{2}+2e^{4}+e^{6}\right)}{e^{5}}, \frac{1}{e}+e, 1, 1\right)^{T}$$

one obtains for  $\hat{k}_c$ :

$$\hat{k}_{c} = (0, 0, 0, 0, 0, \frac{2(-1+e)(1+e)^{3}(1+2e^{2}+2e^{4}+e^{6})}{e^{5}}, \frac{1}{e} + e, 1, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 1)^{T}$$

Using  $k(\epsilon) = \hat{k}_E^* + \epsilon \hat{k}_c$  as parameter vector and  $x_1$  as initial value for a numerical continuation with, for example,  $c_1$  (total concentration of Cdc14) as a bifurcation parameter one obtains results displayed in Fig. 5.6.



Figure 5.6: Numerical Continuation of  $x_1$  using  $c_1$  (total concentration of Cdc14) as bifurcation parameter. Parameter vector  $k(\epsilon) = k^* + \epsilon k_c$ , as given in the text. For  $\epsilon \in [0, 19]$  the system shows multistationarity.





			· · · · · · · · · · · · · · · · · · ·
			0 0 0 0 0 0 0 1
			$\begin{array}{c} -e^{\mu}4\\ -e^{\mu}1+\mu 3+e^{\mu}5+\mu 6\\ -e^{\mu}6+\mu 6\\ e^{\mu}5+\mu 6\\ e^{\mu}5+\mu 0\\ 0\\ 0\\ -e^{\mu}2+\mu 3\end{array}$
		$\mu_6$ 0 0 1 6 0 0 1 6 0 0 1 6 0 0 1 1 6 0 0 1 1 1 1	$\begin{array}{c} & 0 \\ & -e^{\mu}4 \\ & e^{\mu}5 + \mu6 \\ & -e^{\mu}5 + \mu6 \\ & e^{\mu}5 + \mu6 \\ & e^{\mu}2 \\ & e^{\mu}2 \\ & -e^{\mu}2 + \mu3 \\ & -e^{\mu}2 + \mu3 \end{array}$
	$-e^{\mu}4_{4}$ $-e^{\mu}4_{4}$ $-1+e^{\mu}4_{4}$ $-e^{\mu}1+\mu_{3}+e^{\mu}4_{4}$ 0	$\begin{array}{c} -e^{\mu}4\\ -e^{\mu}3+\mu 6\\ -e^{\mu}1+\mu 3+e^{\mu}5+\mu 6\\ -e^{\mu}2+\mu 6\\ e^{\mu}1+\mu 3-e^{\mu}5+\mu 6\\ e^{\mu}1+\mu 3-e^{\mu}5+\mu 6\\ -e^{\mu}5+\mu 3+e^{\mu}5+\mu 3\\ -e^{\mu}2+\mu 3+e^{\mu}5+\mu 6\end{array}$	$\begin{array}{c} 0 \\ -e^{\mu_1} + \mu_3 + e^{\mu_5} \\ -1 + e^{\mu_5} \\ 0 \\ -e^{\mu_2} + \mu_3 + e^{\mu_5} \end{array}$
	$\begin{smallmatrix} & 0 \\ & 0 \\ & 0 \\ & 0 \\ & 0 \\ & -e^{\mu_2} + \mu_3 + e^{\mu_5} \\ & -e^{\mu_5} \end{smallmatrix}$	$\begin{array}{c} -e^{\mu A}+e^{\mu 5}+\mu 6\\ e^{\mu 5}+\mu 6\\ -e^{\mu 5}+\mu 6\\ -e^{\mu 5}+\mu 6\\ e^{\mu 4}-e^{\mu 5}+\mu 6\\ e^{\mu 2}-e^{\mu 5}+\mu 6\\ e^{\mu 4}-e^{\mu 5}+\mu 6\\ -e^{\mu 2}+\mu 3+e^{\mu 5}+\mu 6\\ -e^{\mu 2}+\mu 3+e^{\mu 5}+\mu 6\end{array}$	$\begin{array}{c} 0 \\ -e^{\mu}1 + \mu 3 \\ + e^{\mu}2 + \mu 6 \\ 0 \\ 0 \\ 0 \\ 0 \\ 0 \\ 0 \\ 0 \\ 0 \\ 0 \\$
$l = \begin{bmatrix} a_1 & a_2 \end{bmatrix}$	$-e^{\mu \cdot 4}$ 0 0 0 -e^{\mu \cdot 1} $+\mu_3 + e^{\mu_1}$ 0 0 0 0	$\begin{array}{c} 0\\ e^{\mu}1+\mu 3 + e^{\mu}5\\ 0\\ -1+e^{\mu}1+\mu 3\\ \mu 1+\mu 3 - e^{\mu}5\\ e^{\mu}2+\mu 3 + e^{\mu}5\\ 0\end{array}$	$\begin{array}{c} & 0 \\ & -e^{\mu}1 + \mu_3 \\ & 0 \\ & -1 \\ & -e^{\mu}2 \\ & -e^{\mu}2 \\ & 3 \\ & 0 \\ & 1 \end{array}$
0	$\begin{array}{c} 0 \\ 0 \\ 0 \\ 0 \\ 0 \\ 0 \\ 0 \\ 0 \\ 0 \\ 0 $	2+μ6 - 2+μ6 - 2+μ6 -	$-e^{\mu_4}$ 0 0 -1 $-e^{\mu_1}+\mu$ 0 0
	$\begin{array}{cccccccccccccccccccccccccccccccccccc$	$\begin{array}{c} 0\\ -e^{\mu}1+\mu 3 + e\\ e^{\mu}1+\mu 3 - e^{\mu}\\ e^{\mu}1+\mu 3 - e^{\mu}\\ 0 \end{array}$	$e^{0} + e^{\mu_{2}} + e^{\mu_{5}} + e^{\mu_{5}}$
	$\mathcal{O}_1 = \begin{bmatrix} 1 & 0 & 0 \\ 0 & 0 & 0 & 0 \\ 0 & 0 & 0 & 0$	$\begin{array}{c} 0 \\ -e^{\mu}1 + \mu 3 \\ -1 + e^{\mu}1 + \mu 3 \\ -e^{\mu}2 + e^{\mu}1 + \mu 3 \\ -e^{\mu}2 + e^{\mu}1 + \mu 3 \\ 0 \\ 1 \end{array}$	$-e^{\mu}4$ 0 0 0 $-e^{\mu}1+\mu 3$ 0 0 0
	G	22 =	$\begin{array}{c} 0 \\ 0 \\ -1 + e^{\mu} 1 \\ 0 \\ 0 \end{array}$
		0	$\begin{smallmatrix} -1 & 0 & -1 \\ 0 & -1 & 0 \\ 0 & -1 \\$
			0 - 0 - 0 0 - 0
			0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0
			Ő.

### 5.3 Analysis of the complete network $\mathcal{N}_{11}$

In Section 5.1 it has been established that multistationarity is possible for three subnetworks of  $\mathcal{N}_{11}$ . For the sake of completeness the overall network is analysed using the methods established in Chapter 3. As for network  $\mathcal{N}_{10}$ , it was not possible for network  $\mathcal{N}_{11}$  to determine  $\nu$ ,  $\lambda \in \mathbb{R}^{11}_{>0}$  independent of  $\mu \in \mathbb{R}^9$ . Thus the equation

$$Y^{(L)}{}^T \mu = \ln \frac{E \lambda}{E \nu}$$

is considered in the following form:

$$\begin{bmatrix} E & -\operatorname{diag}\left(e^{Y^{(L)^{T}}\mu}\right) E\end{bmatrix} \begin{pmatrix}\nu\\\lambda\end{pmatrix}, \ \begin{pmatrix}\nu\\\lambda\end{pmatrix} > 0.$$

Using an unimodular matrix  $U \in \mathbb{R}^{18 \times 18}$  with  $UE = \begin{bmatrix} I_{11} \\ 0_{7 \times 11} \end{bmatrix}$  one obtains the following system of equations

$$\underbrace{U\left[E - \operatorname{diag}\left(e^{Y^{(L)^{T}}\mu}\right)E\right]}_{=:H(\mu)} \begin{pmatrix}\nu\\\lambda\end{pmatrix} = 0, \ \begin{pmatrix}\nu\\\lambda\end{pmatrix} > 0,$$

where  $H(\mu)$  is given at the end of this Section on page 73. Note that the nonzero elements of rows 16 - 18 of  $H(\mu)$  always have the same sign, for every  $\mu$ . Thus, positive  $\nu$ ,  $\lambda$  only exists for those values of  $\mu$ , where rows 16 - 18 of  $H(\mu)$  are identical to zero, that is, for  $\mu$  that satisfy

$$\mu_9 = \mu_3 + \mu_4 \tag{5.20a}$$

$$\mu_7 = \mu_2 + \mu_6 \tag{5.20b}$$

$$\mu_8 = \mu_5 + \mu_6. \tag{5.20c}$$

Note that (5.20a), (5.20b) and (5.20c) are the only restriction on  $\mu \in \mathbb{R}^9$ . Thus  $\mu$  can be represented as (using new variables  $\kappa \in \mathbb{R}^6$ ):

$$\mu = M_{N_{11}}$$
 (5.21a)

where

$$M_{\mathcal{N}_{11}} := \begin{bmatrix} 1 & 0 & 0 & 0 & 0 & 0 \\ 0 & 1 & 0 & 0 & 0 & 0 \\ 0 & 0 & 1 & 0 & 0 & 0 \\ 0 & 0 & 0 & 1 & 0 & 0 \\ 0 & 0 & 0 & 0 & 1 & 0 \\ 0 & 1 & 0 & 0 & 0 & 1 \\ 0 & 0 & 0 & 0 & 1 & 1 \\ 0 & 0 & 1 & 1 & 0 & 0 \end{bmatrix}.$$
(5.21b)

Note that each of rows 1, 2, 4, 5, 6, 8, 9, 10, 11 contains exactly two entries, thus they are equal to

ν

$$\nu_i - e^{\langle y_j^{(L)}, \, \mu \rangle} \, \lambda_i = 0,$$

for some linear form  $\langle y_j^{(L)}, \mu \rangle$ . Further note that rows 12–18 are independent of the  $\nu_i, i = 1, ..., 11$ . Thus it is straightforward to obtain

$$\nu_1 = e^{\mu_1} \lambda_1$$
 (5.22a)

$$\nu_2 = e^{\mu_4} \lambda_2$$
 (5.22b)

$$a_4 = e^{\mu_9} \lambda_4 \tag{5.22c}$$

$$\nu_5 = e^{\mu_7} \lambda_5 \tag{5.22d}$$

- $\nu_6 = e^{\mu_8} \lambda_6 \tag{5.22e}$
- $\nu_8 = e^{\mu_4} \lambda_8 \tag{5.22f}$
- $\nu_9 = e^{\mu_5} \lambda_9 \tag{5.22g}$
- $\nu_{10} = e^{\mu_7} \lambda_{10} \tag{5.22h}$

$$\nu_{11} = e^{\mu_8} \lambda_{11}. \tag{5.22i}$$

Thus it remains to determine  $\nu_3$ ,  $\nu_7$ ,  $\lambda_1$ , ...,  $\lambda_{11}$ . This corresponds to a system of equations that involves rows 3, 7 and 12 – 15 and columns 3, 7 and 12 – 22 of  $H(\mu)$ . Note that  $\lambda_4$ ,  $\lambda_5$  and  $\lambda_6$  do not occur in these equations anymore, thus the corresponding columns of  $H(\mu)$  can be left out. Then the following system of equations remains:

$$Q(\mu)\,\tilde{\lambda} = 0, \,\,\tilde{\lambda} > 0,$$

where  $Q(\mu) = H(\mu)[\{3, 7, 12, 13, 14, 15\}, \{12, 13, 14, 18, 19, 20, 21, 22\}]$  and  $\tilde{\lambda} = (\nu_3, \nu_7, \lambda_1, \lambda_2, \lambda_3, \lambda_7, \lambda_8, \lambda_9, \lambda_{10}, \lambda_{11})^T$ .  $Q(\mu)$  is of the form

$$Q(\mu) = \begin{bmatrix} 1 & 0 & 0 & 0 & -1 & q_1 & 0 & q_1 & q_1 & q_2 \\ 0 & 1 & 0 & 0 & 0 & -1 & 0 & -q_1 & q_3 & -q_2 \\ 0 & 0 & q_4 & 0 & 0 & q_5 & q_6 & q_5 & -q_3 & q_2 \\ 0 & 0 & 0 & 0 & 0 & q_7 & 0 & q_1 & -q_3 & q_2 \\ 0 & 0 & 0 & q_8 & 0 & q_9 & q_8 & q_9 & q_2 \\ 0 & 0 & 0 & 0 & q_{10} & -q_1 & 0 & -q_1 & -q_2 \end{bmatrix}$$
(5.23)

with

$$\begin{array}{ll} q_1 = e^{\mu_3 + \mu_4} - e^{\mu_5 + \mu_6}, & q_2 = e^{\mu_3 + \mu_4} - e^{\mu_5 + \mu_6}, & q_3 = -e^{\mu_3 + \mu_4} + e^{\mu_2 + \mu_6}, \\ q_4 = -1 + e^{\mu_1}, & q_5 = -1 + e^{\mu_3 + \mu_4}, & q_6 = -1 + e^{\mu_4}, \\ q_7 = -e^{\mu_2} + e^{\mu_3 + \mu_4}, & q_8 = -e^{\mu_1 + \mu_3} + e^{\mu_4}, & q_9 = -e^{\mu_1 + \mu_3} + e^{\mu_3 + \mu_4}, \\ q_{10} = -e^{\mu_2 + \mu_3} + e^{\mu_5}. \end{array}$$

Thus any feasible inequality system

$$\begin{bmatrix} V & -\operatorname{diag}\left(\sigma\right) \end{bmatrix} \binom{\kappa}{s} = 0, s > 0$$

with  $\sigma \in \{-1, 0, 1\}^{10}$  and

$$V = \begin{bmatrix} 0 & 1 & 0 & 0 & -1 & 1 \\ 0 & 0 & -1 & -1 & 1 & 0 \\ 0 & 1 & -1 & -1 & 0 & 1 \\ 0 & 0 & -1 & -1 & 1 & 1 \\ 1 & 0 & 0 & 0 & 0 & 0 \\ 0 & 0 & 0 & 1 & 0 & 0 \\ 0 & 0 & 0 & 0 & 1 & 0 \\ -1 & 0 & -1 & 1 & 0 & 0 \\ 0 & -1 & -1 & 0 & 1 & 0 \end{bmatrix}$$
(5.24)

defines a feasible signing for  $Q(\mu)$ . Following Algorithm 4, first  $\Delta^{S_{N_{11}}}$ , the set of all orthants  $\delta \in \{-1,0,1\}^9$  with  $\mathcal{S}_{N_{11}} \cap \mathbb{R}^8_{\theta} \neq \emptyset$  has been obtained. Then the set  $\Delta^+ = \{\delta \in \{-1,0,1\}^9 | \mathcal{M}_{N_{11}} \cap \mathbb{R}^8_{\theta} \neq \emptyset$  and  $\exists \sigma \in \{-1,0,1\}^{10}$  such that  $Q(\mu)$  is an  $L^+$ -matrix  $\}$  has been obtained. As  $\Delta^{S_{N_{11}}} \cap \Delta^+ \neq \emptyset$ , multistationarity has been established by Algorithm 4. The elements of  $\Delta^{S_{N_{11}}} \cap \Delta^+$  are

$$\begin{split} \delta_1 &= (-1, 0, 1, -1, 1, -1, -1, 1, 0)^T \\ \delta_2 &= (-1, 1, 1, -1, 1, -1, -1, 1, 0)^T \\ \delta_3 &= (-1, 1, 1, -1, 1, -1, -1, 1, 0)^T \\ \delta_4 &= (-1, 1, 1, -1, 1, -1, 1, -1, 1)^T \\ \delta_5 &= (-1, 1, 1, -1, 1, -1, 1, 1, 1)^T \\ \delta_5 &= (-1, 1, 1, -1, 1, -1, 1, 1, 1)^T \\ \delta_5 &= (-1, 1, 1, 1, 1, 1, -1, 1, 1, -1, 1)^T \end{split}$$
(5.25)

All signings such that  $Q(\mu)$  is an  $L^+$  in one of the orthants given in (5.25), are

Table 5.2 shows, for which orthant  $\delta$  as in (5.25) and which signing  $\sigma$  as in (5.26), the system

$$\begin{bmatrix} \operatorname{diag}\left(\delta\right) & -M_{\mathcal{N}_{11}} & 0_{10\times10} \\ 0_{9\times9} & V & -\operatorname{diag}\left(\sigma\right) \end{bmatrix} \begin{pmatrix} \xi \\ \kappa \\ s \end{pmatrix}, \ \xi, \ s > 0$$

is feasible.

	$\sigma_1$	$\sigma_2$	$\sigma_3$	$\sigma_4$	$\sigma_5$	$\sigma_6$	$\sigma_7$	$\sigma_8$	$\sigma_9$	$\sigma_{10}$	$\sigma_{11}$	$\sigma_{12}$	$\sigma_{13}$	$\sigma_{14}$
$\delta_1$	+													
$\delta_2$		+												
$\delta_3$			+	+	+	+	+	+	+	+				
$\delta_4$							+	+			+	+		
$\delta_5$			+	+	+	+	+	+	+	+	+	+		
$\delta_6$													+	+

Table 5.2: Orthants and sign conditions for the ternary complex model



### 5.4 Conclusions

In this chapter two equally plausible hypotheses describing the G1/S transition in Saccharomyces cerevisiae have been analysed. For both network structures multistationarity has been established, however, there is a difference: in contrast to the ternary complex model, where individual subunits can establish the desired switch–like behavior, the integrity of the binary complex model needs to be preserved for this function. This has important consequences for the robustness of the two models. Robustness, often, is defined as the resistance of qualitative network behavior to perturbations, for instance, in network structure or parameter values [60]. Clearly, the ternary model is more robust in this sense than the binary model, although capturing this effect quantitatively is difficult. Previously, it has been proposed that robustness can also serve as a measure of plausibility for biological network models because, in fluctuating environments, robustness of functions should have been key for evolutionary selection [47]. In this regard, subnetwork analysis could be used for model discrimination—the ternary complex mechanism would be considered as the more plausible one.

### Chapter 6

## **Robustness of Multistationarity**

A variety of biochemical reaction networks give rise to qualitative properties that are robust with respect to certain parameter variations (see [42, 48] and [60] and the references therein). Nowadays robustness is often considered as an inherent principle of cellular functions (and, of course, of the biochemical reaction networks that implement this functionality). However, robustness never comes alone. It is almost always accompanied by 'fragility' with respect to variations of different parameter sets: in [59], for example, it has been shown that the existence of cyclic trajectories is robust to large variations in one group of parameters, whereas small variations in a second, different group of parameters lead to a destruction of this qualitative property. This 'robust yet fragile' nature has been observed frequently both in biochemical and in man-made networks like the Internet [19, 20].

In this chapter robustness of multistationarity in the Activation of an MAPK(K) and in a variety of other signal transduction motifs is analysed. For the Activation of an MAPK(K) robustness (and fragility) of multistationarity with respect to variations in the rate constants is examined in Section 6.1. For a variety of signal transduction motifs, which are the building blocks of many signal transduction networks robustness (and fragility) of multistationarity with respect to fluctuations in the total concentrations is examined in Section 6.2.

### 6.1 Robustness against variations in the rate constants

Multistationarity in the activation of an MAPK is a qualitative property with a *robust yet fragile* nature. This is a direct consequence of the results presented in Chapter 4, in particular of the definition of k, the parameter vector, as given in (4.7a) – (4.7f). Even for a specific steady state a the  $k_i$  are not unique, as the vector  $\lambda \in \mathbb{R}^{6}_{>0}$  can still be chosen freely. Thus, a specific pair of steady states is compatible with an infinite set of parameter values.

Multistationarity is robust with respect to variations in the  $k_i$ , as long as the  $k_i$  satisfy the equations (4.7a) – (4.7f) for positive  $\lambda_i$ . Note that for a fixed vector a (4.7a) – (4.7f) are linear. In this case the right hand side of the equations defines a pointed polyhedral cone in  $\mathbb{R}_{\geq 0}^{12}$ . The generators of this 'parameter cone' can be obtained from the generators of ker $(Y I_a) \cap \mathbb{R}_{\geq 0}^{12}$  via the linear transformation

$$P = \operatorname{diag} \left( \phi \left( a^{-1} \right) \right) E, \tag{6.1}$$

where E is a matrix composed of the generators of  $\ker(Y I_a) \cap \mathbb{R}^{12}_{\geq 0}$ . For a given vector a the vector of rate constants is given by

$$k = P \lambda, \lambda \in \mathbb{R}^6_{>0}$$
. (6.2)

Thus one can conclude that the property multistationarity is robust with respect to variations of k within the 'parameter cone' P. Moreover, as long as k is in the interior of P, the values of the steady state pair a, b are not affected. To make explicit the *robust* nature a random set of 197 parameter vectors was created and simulations for each of these vectors were performed using the initial conditions given in Table 6.1<sup>1</sup>. The results are displayed in Fig. 6.1 (for the initial condition  $x_0^{(1)}$  in the upper half and for the initial condition  $x_0^{(2)}$  in the lower half). The figure shows the temporal evolution of  $x_1(t)$ 

<sup>&</sup>lt;sup>1</sup>Tables and Figures in this chapter are reprinted from [65] with permission from Elsevier

	$x_0^{(1)}$	$x_0^{(2)}$	$x_0^{up}$
$x_{10}$	1.4047	4.1339	1.0199
$x_{20}$	0.12618	2.1585	0.11185
$x_{30}$	0.41462	0.54691	0.38576
$x_{40}$	0.84691	0.91462	0.89271
$x_{50}$	2.3169	0.15236	2.3601
$x_{60}$	0.73726	0.70497	1.058
$x_{70}$	0.013662	0.74596	0.0096985
$x_{80}$	3.9701	1.4605	4.0441
$x_{90}$	0.27818	2.0555	0.20815

Table 6.1: Initial conditions for the dynamic simulations in Fig. 6.1.

and  $x_6(t)$ . In in the upper half all trajectories converge to  $x^{(1)}$  and in the lower half all trajectories converge to  $x^{(2)}$ , the stable steady states obtained for  $k^*$ . Table 6.2 contains the stable steady states  $x^{(1)}$  and  $x^{(2)}$  and the unstable steady state  $x^*$  obtained for  $k^*$ . Changing  $\lambda$  (and thus changing k within P) does not change the steady states of the ODEs displayed in (A.1a) – (A.1i).

	$x^{(1)}$	$x^*$	$x^{(2)}$
$x_1$	1.3199	1.7047	4.6339
$x_2$	0.41185	0.42618	1.1585
$x_3$	0.085761	0.11462	0.84691
$x_4$	0.89271	0.84691	0.11462
$x_5$	2.3601	2.3169	0.85236
$x_6$	1.058	0.73726	0.0049676
$x_7$	0.0096985	0.013662	0.74596
$x_8$	4.0441	3.9701	1.4605
$x_9$	0.20815	0.27818	2.0555

Table 6.2: Lower stable steady state  $x^{(1)}$ , unstable steady state  $x^*$  and upper stable steady state  $x^{(2)}$  ( $x^*$  and  $x^{(2)}$  have been published in [11]).

The fragility of multistationarity Multistationarity is fragile with respect to variations of k perpendicular to P (or, more precise, to variations of k in the orthogonal complement of the subspace of  $\mathbb{R}^{12}$  that contains P). To demonstrate this, the parameter vector was perturbed in the following way:

$$k = k^* + \epsilon \, u^\perp,$$

where

$$u^{\perp} = (-1, 1, 1, -1, 1, 1, -1, 1, 1, -1, 1, 1)^{T}$$

Note that  $\langle u^{\perp}, E_i \rangle = 0$ ,  $i = 1, \ldots, 6$ , thus changing  $k^*$  along  $u^{\perp}$  corresponds to changes transversal to P. To illustrate this, a bifurcation analysis was performed using  $\epsilon$  as bifurcation parameter (the bifurcation was performed using *Matcont*, see e.g. [1]). The result is displayed in Fig. 6.2. This analysis established upper and lower bifurcation parameters (at  $\epsilon \approx 0.0017$  and  $\epsilon \approx -0.1993$ , respectively). Visual inspection alone shows that  $\epsilon$  can vary only in a very small interval, without losing multistationarity. The parameter vectors  $k_{\epsilon}^{up}$  (corresponding to  $\epsilon \approx 0.0017$ ) and  $k_{\epsilon}^{low}$  (corresponding to  $\epsilon \approx -0.1993$ ) are displayed in Table 6.3, as well. Dynamic simulations displayed in Fig. 6.3 confirm that multistationarity vanishes, once  $\epsilon$  is outside the interval (-0.1993, 0.0017). As long as the parameter vector is 'sufficiently close' to P, the system retains multistationarity (albeit for different steady states). If the parameter vector is 'far' from P multistationarity is no longer exhibited. Fig. 6.3 shows the effect of changing  $\epsilon$  on  $x_1(t)$ . In the upper half  $\epsilon$  varies from 0 to -0.25 and the initial condition  $x_0^{(2)}$  is used. As long as  $\epsilon > -0.1993$  the trajectories reach a steady state that is close to  $x^{(2)}$  (i.e. the trajectories varies from 0 to 0.0017 and the initial condition  $x_0^{up}$  is used. As long as  $\epsilon < 0.0017$  the trajectories reach a steady state solutions displayed in Fig. 6.2). In the lower half  $\epsilon$  varies from 0 to 0.0017 and the initial condition  $x_0^{up}$  is used. As long as  $\epsilon < 0.0017$  the trajectories reach a steady state solutions displayed in Fig. 6.2).



Figure 6.1: Temporal evolution of  $x_1(t)$  and  $x_6(t)$  for a set of randomly generated parameter vectors, using the initial condition  $x_0^{(1)}$ . The dashed red line indicates the temporal evolution obtained for the reference parameter vector  $k^*$ , the delimiting trajectories correspond to parameter sets labeled 'upper' and 'lower'.

reach a steady state that is close to  $x^{(1)}$  (i.e. the trajectories converge to the lower branch of the steady state solutions displayed in Fig. 6.2). (Note that when integrating the ODEs using matlab<sup>®</sup>'s ode15s the bifurcation occurs somewhere between  $\epsilon = 0.0018$  and  $\epsilon = 0.0019$ , whereas the bifurcation software *Matcont* locates the bifurcation at  $\epsilon = 0.0017$ .)

**Quantification of robustness** To quantify 'robustness' of multistationarity (or lack thereof) the parameter sets corresponding to those trajectories marked green in Fig. 6.1 were used to approximate the parameter range where multistationarity can occur. (This is, of course a poor approximation. But it is sufficient to back the argument). For each parameter an *elasticity coefficient* was defined that measures the difference between the upper and the lower value of the parameter in relation to the



Figure 6.2: Bifurcation along  $u^{\perp}$ .



Figure 6.3: Dynamic simulation 'close' to bifurcation points for  $\epsilon \approx -0.1993$  and  $\epsilon \approx 0.0017$ . The bifurcation occurs between the temporal evolution displayed in green.

'nominal' value  $k_i^*$ :

$$\eta_i := \frac{k_i^{up} - k_i^{low}}{k_i^*}$$
(6.3)

	$k^*$	$k^{up}$	$k^{low}$	$k_{\epsilon}^{up}$	$k_{\epsilon}^{low}$
$k_1$	10.9293	8.3595	12.6521	10.9275	11.1285
$k_2$	34.6381	30.44	46.1592	34.6399	34.4389
$k_3$	34.6381	22.5474	34.0373	34.6399	34.4389
$k_4$	21.9991	31.4783	14.6527	21.9973	22.1983
$k_5$	1.7135	1.3566	1.604	1.7153	1.5143
$k_6$	1.7135	3.5471	0.67864	1.7153	1.5143
$k_7$	788.272	1211.7563	727.2734	788.2703	788.4713
$k_8$	1	1.0042	1.4491	1.0017	0.80077
$k_9$	1	2.0701	0.39605	1.0017	0.80077
$k_{10}$	686.2147	430.6848	647.1921	686.2129	686.4139
$k_{11}$	14.2718	8.6237	12.8949	14.2736	14.0726
$k_{12}$	14.2718	9.2901	14.0243	14.2736	14.0726

Table 6.3: Reference parameter vector  $k^*$  (as given in [11]), parameter vectors  $k^{up}$  and  $k^{low}$  obtained by random variations  $\lambda$ , with  $\sum_i \lambda_i = \lambda_0^*$  and parameter vectors  $k_{\epsilon}^{up}$  and  $k_{\epsilon}^{low}$  obtained by varying kalong  $\epsilon u^{\perp}$ .

Here  $k^{up}$  corresponds to the upper delimiting trajectory in the upper left part of Fig. 6.1 and  $k^{low}$  to the lower delimiting trajectory in the upper left part of Fig. 6.1. Both parameter vectors are given in Table 6.3. Using  $k_i^{up}$  and  $k_i^{low}$  in formula (6.3), the elasticities for perturbations along  $u^{\perp}$  have been determined as well (see Table 6.4, second colum). The results are displayed in Fig. 6.4: the elasticities corresponding to variations within P as empty bars and those corresponding to variations along  $u^{\perp}$ as filled bars. This figure nicely reflects the robust yet fragile nature of multistationarity: on the one hand, each  $k_i$  can vary by large amounts, if variation occurs within the parameter cone P. On the other hand, if the parameter cone P is left (e.g. by adding  $\epsilon u^{\perp}$ ), then, for relatively small variations of the  $k_i$ , multistationarity is lost. This is best illustrated using  $k_7$ . From Table 6.3 it follows that  $k_7$  can vary between 727.2734 and 1211.7563 if variation occurs within P (i.e.  $k_7$  can vary by at least 60% of its nominal value  $k_7^* = 788.272$  and multistationarity is retained). If  $k_7$  is varied along  $u^{\perp}$ , then again by Table 6.3 it can vary between 788.2703 and 788.4713 (i.e. by at most 0.025% relative to  $k_7^* = 788.272$ ) without losing multistationarity.

	$\tfrac{k_i^{up}-k_i^{low}}{k_i^*}$	$\frac{k_{\epsilon_i}^{up} - k_{\epsilon_i}^{low}}{k_i^*}$
$\eta_1$	-0.39276	-0.018391
$\eta_2$	-0.45381	0.0058029
$\eta_3$	-0.33171	0.0058029
$\eta_4$	0.76483	-0.0091367
$\eta_5$	-0.14438	0.1173
$\eta_6$	1.674	0.1173
$\eta_7$	0.61461	-0.00025499
$\eta_8$	-0.4449	0.20093
$\eta_9$	1.6741	0.20093
$\eta_{10}$	-0.31551	-0.00029291
$\eta_{11}$	-0.29927	0.014084
$\eta_{12}$	-0.33172	0.014084

Table 6.4: Elasticities obtained for a random sample of parameter vectors (column one) and for variation along  $u^{\perp}$  (column two).

A note on quantifying robustness When quantifying 'robustness', some care has to be exercised: as P is a pointed polyhedral cone, it is in principle possible to obtain deviations from  $k^*$  that are arbitrarily large without losing multistationarity. To see this, consider two vectors  $\tilde{k}^{up}$ ,  $\tilde{k}^{low}$  and the vector  $k^*$ . Let  $\tilde{k}^{up} = \lambda_0 k^*$  and  $\tilde{k}^{low} = \frac{1}{\lambda_0} k^*$ , where  $\lambda_0 \gg 1$ . As P is a cone and  $k^* \in P$ , any positive scalar multiple of  $k^*$  is in the cone as well. Thus the system will exhibit multistationarity for



Figure 6.4: Robustness vs. fragility of multistationarity for the activation of an MAPK. Empty bars: Elasticities obtained for a random sample of parameters contained in the *parameter cone* P. Filled bars Elasticities obtained for a set of parameters created along  $\epsilon u^{\perp}$ .

 $\tilde{k}^{up} = \lambda_0 k^*$  and  $\tilde{k}^{low} = \frac{1}{\lambda_0} k^*$  and the elasticity coefficients will be  $\tilde{\eta}_i = \frac{\lambda_o k_i^* - \frac{1}{\lambda_0} k_i^*}{k_i^*} = \lambda_0 - \frac{1}{\lambda_0} \approx \lambda_0$  (using  $\tilde{k}^{low}$ ,  $\tilde{k}^{up}$  and  $k^*$  in formula (6.3)).

However such an  $\tilde{\eta}_i$  is not meaningful, as  $\tilde{k}^{up}$  and  $\tilde{k}^{low}$  correspond to a mere scaling of time that does not change the qualitative dynamic behaviour of the system, hence it is 'obvious' that the system retains multistationarity: to see this, let  $k = \lambda_0 \tilde{k}$  and recall that the ODEs derived from any biochemical reaction network endowed with mass action kinetics are linear in the vector of rate constants k:

$$\dot{x} = Y I_a \operatorname{diag}(k) \phi(x) = Y I_a \operatorname{diag}(\phi(x)) k$$

Thus using  $k = \lambda_0 \tilde{k}$  yields

$$\dot{x} = \lambda_0 Y I_a \operatorname{diag}(\phi(x)) \dot{k}.$$

Clearly this corresponds to a scaling of time t with  $\lambda_0$ . To circumvent this scaling of time and thus to exclude this trivial robustness, these parameter deviations have to be ruled out. To do this, note that if  $k^* \in P$ , then there exists a  $\lambda^* \in \mathbb{R}^6_{>0}$ , such that  $k^* = P\lambda^*$ . Then  $\lambda_0 k^* = \lambda_0 P \lambda^* = P(\lambda_0 \lambda^*)$  (i.e. multiplying  $k^*$  by  $\lambda_0$  is equivalent to multiplying  $\lambda^*$  by  $\lambda_0$ ). Thus, to deal with the effect of time scaling we consider only  $\lambda_i$  contained in the affine linear subspace  $\langle \underline{1}, \lambda \rangle = \lambda_0$  for a fixed value  $\lambda_0$ . Elements of this subspace are given by  $\lambda = \lambda_0 (\lambda_1, \ldots, \lambda_5, 1 - \sum_{i=1}^5 \lambda_i), \lambda_i \in [0, 1], i = 1, \ldots, 5$ . Then  $\lambda_0$  is fixed as  $\lambda_0^* = 23.8207$ , the value calculated for  $k^*$ , as given in Table 6.3 (using e.g. the steady state  $x^*$  from Table 6.2, the vector  $\lambda$  can be calculated as  $\lambda = \lambda_0^* (0.1667, 0.1667, 0.1667, 0.1667, 0.1667)$ . To demonstrate the effect of  $\lambda_0$ , Fig. 6.1 contains the trajectories of  $x_1(t)$  obtained using  $100 \lambda_0^*$ .

# 6.2 Robustness against concentration fluctuations for different signal transduction motifs

In [53] a variety of small reaction networks called motifs that can be found as subnetworks in many signal transduction networks is presented and analysed. Among them are network  $\mathcal{N}_4$  described in Section 4 and  $\mathcal{N}_{12}$ ,  $\mathcal{N}_{13}$  and  $\mathcal{N}_{14}$  given below. Here \* is used to denote the activated form of protein A and  $E_i$  denotes activating or deactivating enzymes.

• Autocatalytic activation:

$$A + E_1 \xrightarrow[\mathbf{k_1}]{\mathbf{k_2}} AE_1 \xrightarrow{\mathbf{k_3}} A^* + E_1$$

$$A + A^* \xrightarrow[\mathbf{k_6}]{\mathbf{k_5}} AA^* \xrightarrow{\mathbf{k_6}} A^* + A^* \qquad (\mathcal{N}_{12})$$

$$A^* + E_2 \xrightarrow[\mathbf{k_7}]{\mathbf{k_8}} A^* E_2 \xrightarrow{\mathbf{k_9}} A + E_2$$

• Distributive activation by enzyme  $E_1$  and distributive deactivation by two different enzymes  $E_2$  and  $E_3$ :

$$A + E_1 \xrightarrow{\mathbf{k_1}} A E_1 \xrightarrow{\mathbf{k_3}} A^* + E_1 \xrightarrow{\mathbf{k_4}} A^* E_1 \xrightarrow{\mathbf{k_6}} A^* E_1$$

$$A^{**} + E_3 \xrightarrow{\mathbf{k_7}} A^{**} E_3 \xrightarrow{\mathbf{k_9}} A^* + E_3$$

$$A^* + E_2 \xrightarrow{\mathbf{k_{10}}} A^* E_2 \xrightarrow{\mathbf{k_{12}}} A + E_2$$

$$(\mathcal{N}_{13})$$

• Autocatalytic activation and distributive deactivation by two different enzymes E2 and E3:

$$\begin{aligned} A + E_1 & \xrightarrow{\mathbf{k_1}} A E_1 \xrightarrow{\mathbf{k_3}} A^* + E_1 \\ A^* + E_3 & \xrightarrow{\mathbf{k_4}} A^* E_3 \xrightarrow{\mathbf{k_6}} A + E_3 \\ 2 A^* & \xrightarrow{\mathbf{k_7}} A^* A^* \xrightarrow{\mathbf{k_9}} A^{**} + A^* \xrightarrow{\mathbf{k_{10}}} A^* A^{**} \xrightarrow{\mathbf{k_{12}}} 2 A^{**} \\ A^{**} + E_2 & \xrightarrow{\mathbf{k_{13}}} A^{**} E_2 \xrightarrow{\mathbf{k_{15}}} A^* + E_2 \end{aligned}$$

$$(\mathcal{N}_{14})$$

Network  $\mathcal{N}_4$ ,  $\mathcal{N}_{12}$ ,  $\mathcal{N}_{13}$  and  $\mathcal{N}_{14}$  were the only networks considered in [53] that showed multistationarity. While in Section 6.1 robustness against variations in the  $k_i$  has been analysed, in this section robustness against variations in total concentrations is compared among networks  $\mathcal{N}_4$ ,  $\mathcal{N}_{12}$ ,  $\mathcal{N}_{13}$  and  $\mathcal{N}_{14}$ . This is based on the results of Section 3. In Section 6.2.1 the steady state equations are analysed in the way described in Section 3 and in Section 6.2.2 robustness against variations in total concentrations is compared.

### 6.2.1 Analysis of the steady state equations

### 6.2.1.1 Network $N_{12}$

The equations

$$Y^{(L)^{T}} \mu = \ln \frac{E \nu}{E \lambda}$$

for this network are

$$\mu_1 + \mu_2 = \ln \frac{\nu_1 + \nu_4}{\lambda_1 + \lambda_4} \tag{6.4a}$$

$$\mu_3 = \ln \frac{\nu_1}{\lambda_1} \tag{6.4b}$$

$$\mu_3 = \ln \frac{\nu_4}{\lambda_4} \tag{6.4c}$$

$$\mu_1 + \mu_4 = \ln \frac{\nu_2 + \nu_5}{\lambda_2 + \lambda_5} \tag{6.4d}$$

$$\mu_5 = \ln \frac{\nu_2}{\lambda_2} \tag{6.4e}$$

$$\mu_5 = \ln \frac{\nu_5}{\lambda_5} \tag{6.4f}$$

$$\mu_4 + \mu_6 = \ln \frac{\nu_3 + \nu_4 + \nu_5}{\lambda_3 + \lambda_4 + \lambda_5} \tag{6.4g}$$

$$\mu_7 = \ln \frac{\nu_3}{\lambda_3} \tag{6.4h}$$

$$\mu_7 = \ln \frac{\nu_4 + \nu_5}{\lambda_4 + \lambda_5} \tag{6.4i}$$

These equations are solvable, if and only if

$$\ln \frac{\nu_1}{\lambda_1} = \ln \frac{\nu_4}{\lambda_4}, \qquad \qquad \ln \frac{\nu_2}{\lambda_2} = \ln \frac{\nu_5}{\lambda_5}, \qquad \qquad \ln \frac{\nu_3}{\lambda_3} = \ln \frac{\nu_4 + \nu_5}{\lambda_4 + \lambda_5}$$

Thus one obtains  $\nu_1 = \lambda_1 \frac{\nu_4}{\lambda_4}$ ,  $\nu_2 = \lambda_2 \frac{\nu_5}{\lambda_5}$  and  $\nu_3 = \lambda_3 \frac{\nu_4 + \nu_5}{\lambda_4 + \lambda_5}$  and thus

$$\nu = \left(\lambda_1 \frac{\nu_4}{\lambda_4}, \lambda_2 \frac{\nu_5}{\lambda_5}, \lambda_3 \frac{\nu_4 + \nu_5}{\lambda_4 + \lambda_5}, \nu_4, \nu_5\right), \tag{6.5}$$

with  $\lambda \in \mathbb{R}^5_{>0}$  and  $\nu_4, \nu_5 > 0$ . Using  $\nu$  as above, one obtains

$$\ln\frac{\nu_1+\nu_4}{\lambda_1+\lambda_4}=\ln\frac{\nu_4}{\lambda_4}, \qquad \qquad \ln\frac{\nu_2+\nu_5}{\lambda_2+\lambda_5}=\ln\frac{\nu_5}{\lambda_5}, \qquad \qquad \ln\frac{\nu_3+\nu_4+\nu_5}{\lambda_3+\lambda_4+\lambda_5}=\ln\frac{\nu_4+\nu_5}{\lambda_4+\lambda_5}.$$

Using the definitions

$$\kappa_1 := \ln \frac{\nu_4}{\lambda_4} \tag{6.6a}$$

$$\kappa_2 := \ln \frac{\nu_5}{\lambda_5} \tag{6.6b}$$

$$\kappa_3 := \frac{\nu_4 + \nu_5}{\lambda_4 + \lambda_5} \tag{6.6c}$$

and solving for  $\mu$ , one obtains

$$\mu = \underbrace{\begin{bmatrix} 1 & 0 & 0 & 0 \\ -1 & 1 & 0 & 0 \\ 0 & 1 & 0 & 0 \\ -1 & 0 & 1 & 0 \\ 0 & 0 & 1 & 0 \\ 1 & 0 & -1 & 1 \\ 0 & 0 & 0 & 1 \end{bmatrix}}_{=:M_{N_{12}}} \begin{pmatrix} \mu_1 \\ \kappa_1 \\ \kappa_2 \\ \kappa_3 \end{pmatrix}.$$
(6.7)

If the  $\kappa_i$  are considered as free variables, (6.7) defines a linear subspace of  $\mathbb{R}^7$ . Note that in this case  $\nu_4, \nu_5, \lambda_4$  and  $\lambda_5$  are determined by (6.6a) – (6.6c): from (6.6a) follows  $\nu_4 = \lambda_4 e^{\kappa_1}$  and from (6.6b)  $\nu_5 = \lambda_5 e^{\kappa_2}$ . Inserting in (6.6c) and solving for  $\lambda_5$  yields

$$\lambda_5 = \lambda_4 \frac{e^{\kappa_1} - e^{\kappa_3}}{e^{\kappa_3} - e^{\kappa_2}}.$$
(6.8a)

Note that  $\lambda_5 > 0$ , if one of the following conditions holds

$$\kappa_1 > \kappa_3 > \kappa_2$$
 (6.8b)

$$\kappa_1 < \kappa_3 < \kappa_2$$
. (6.8c)

Note that (6.7) together with either (6.8b) or (6.8c) is a parametrization of positive steady states in the following sense: suppose (i) the  $\kappa_i$  are chosen such that either (6.8b) or (6.8c) holds, (ii)  $\lambda_5$  is chosen such that (6.8a) holds and (iii) the remaining  $\lambda_i$  are assigned some positive values. Further suppose the vector a is a steady state. Then, by Lemma 1,  $b = \text{diag}(\text{Exp}(\mu)) a$  is a steady state as well. To ensure that a is indeed a steady state apply again (3.10a) to obtain:

$$k_1 = \frac{\lambda_1 + \lambda_4}{a_1 a_2}$$
  $k_2 = \frac{\lambda_1}{a_3}$   $k_3 = \frac{\lambda_4}{a_3}$  (6.9a)

$$k_4 = \frac{\lambda_2 + \lambda_5}{a_1 a_4} \qquad \qquad k_5 = \frac{\lambda_2}{a_5} \qquad \qquad k_6 = \frac{\lambda_5}{a_5} \tag{6.9b}$$

$$k_7 = \frac{\lambda_3 + \lambda_4 + \lambda_5}{a_4 \, a_6} \qquad \qquad k_8 = \frac{\lambda_3}{a_7} \qquad \qquad k_9 = \frac{\lambda_4 + \lambda_5}{a_7} \tag{6.9c}$$

To find pairs of steady states that satisfy the conservation relations for the same  $c_i$ , Lemma 3 is used. To this end, let  $\kappa := (\mu_2, \kappa_1, \kappa_2, \kappa_3)^T$  and  $\mathcal{M}_{\mathcal{N}_{12}} := \{ \mu \in \mathbb{R}^7 \mid \mu = M_{\mathcal{N}_{12}} \kappa$  and either  $\kappa_1 > \kappa_3 > \kappa_2$  or  $\kappa_1 < \kappa_3 < \kappa_2 \}$ , with  $M_{\mathcal{N}_{12}}$  as in (6.7). Further let  $\mathcal{S}_{\mathcal{N}_{12}}$  be the stoichiometric subspace for  $\mathcal{N}_{12}$ . Using Algorithm 2, it is straightforward to establish that there are two orthants with  $\mathcal{M}_{\mathcal{N}_{12}} \cap \mathbb{R}_{\delta}^7 \neq \emptyset$  and  $\mathcal{S}_{\mathcal{N}_{12}} \cap \mathbb{R}_{\delta}^7 \neq \emptyset$ :

$$\delta_1 = (1, -1, 1, -1, -1, 1, -1)^T 
\delta_2 = (-1, 1, -1, 1, 1, -1, 1)^T$$
(6.10)

However,  $\delta_1 = -\delta_2$ , thus these orthants give rise to the same pair of steady states (see Section 3.3) and only one of them need to be considered. Thus multistationarity has been established by Theorem 3. To obtain a pair of steady states, consider  $\delta_1$ . All  $\mu \in \mathcal{M}_{\mathcal{N}_{12}} \cap \mathbb{R}^7_{\delta_1}$  are given by

$$\mu_1 = \alpha_1 + \alpha_2$$
 (6.11a)

$$\mu_2 = -\alpha_1$$
 (6.11b)

$$\mu_3 = \alpha_2$$
 (6.11c)

$$\mu_4 = -\alpha_1 - \alpha_2 - \alpha_3 - \alpha_4 \tag{6.11d}$$

$$\mu_5 = -\alpha_3 - \alpha_4 \tag{6.11e}$$

$$\mu_6 = \alpha_1 + \alpha_2 + \alpha_4 \tag{6.11f}$$

$$\mu_7 = -\alpha_3, \tag{6.11g}$$

for  $\alpha_i > 0$ , i = 1, 2, 3. The  $\kappa_i$ , i = 1, 2, 3 can, of course, be expressed in terms of the  $\alpha_i$ :

$$\kappa = \begin{pmatrix} -\alpha_1 \\ \alpha_2 \\ -\alpha_3 - \alpha_4 \\ -\alpha_3 \end{pmatrix}. \tag{6.12a}$$

Thus one obtains the following expression for  $\lambda_5$ :

$$\lambda_5 = \lambda_4 \frac{(e^{\alpha_2} - e^{-\alpha_3})}{e^{-\alpha_3} - e^{-\alpha_3 - \alpha_4}}.$$
(6.12b)

All  $v \in \mathcal{S}_{\mathcal{N}_{12}} \cap \mathbb{R}^7_{\delta_1}$  are given by

$$v_1 = \beta_1 + \beta_5 + 2\,\beta_6 \tag{6.13a}$$

$$v_2 = -\beta_2 - 2\,\beta_3 - \beta_4 \tag{6.13b}$$

$$v_3 = \beta_2 + 2\,\beta_3 + \beta_4 \tag{6.13c}$$

$$v_4 = -\beta_4 - \beta_5 \tag{6.13d}$$

$$v_5 = -\beta_3 - \beta_6 \tag{6.13e}$$

$$v_6 = \beta_1 + \beta_2 \tag{6.131}$$

$$v_7 = -\beta_1 - \beta_2,$$
 (6.13g)

for  $\beta_i > 0$ , i = 1, ..., 6. As described in Section 3.3, it is possible to state a and b using the generators of  $\mathcal{M}_{\mathcal{N}_{12}} \cap \mathbb{R}^7_{\delta}$  and  $\mathcal{SN}_{12} \cap \mathbb{R}^7_{\delta}$  in (3.30a). This leads to the following formulas for the steady state a

$$a_1 = \frac{\beta_1 + \beta_5 + 2\beta_6}{-1 + e^{\alpha_1 + \alpha_2}} \qquad \qquad a_2 = \frac{-\beta_2 - 2\beta_3 - \beta_4}{-1 + e^{-\alpha_1}} \tag{6.14a}$$

$$a_{3} = \frac{\beta_{2} + 2\beta_{3} + \beta_{4}}{-1 + e^{\alpha_{2}}} \qquad a_{4} = \frac{-\beta_{4} - \beta_{5}}{-1 + e^{-\alpha_{1} - \alpha_{2} - \alpha_{3} - \alpha_{4}}}$$
(6.14b)  
$$a_{5} = \frac{-\beta_{3} - \beta_{6}}{-1 + e^{-\alpha_{3} - \alpha_{4}}} \qquad a_{6} = \frac{\beta_{1} + \beta_{2}}{-1 + e^{\alpha_{1} + \alpha_{2} + \alpha_{4}}}$$
(6.14c)

$$a_7 = \frac{-\beta_1 - \beta_2}{-1 + e^{-\alpha_3}} \tag{6.14d}$$

and the steady state b

$$b_1 = \frac{(\beta_1 + \beta_5 + 2\beta_6)e^{\alpha_1 + \alpha_2}}{-1 + e^{\alpha_1 + \alpha_2}} \qquad b_2 = \frac{(-\beta_2 - 2\beta_3 - \beta_4)e^{-\alpha_1}}{-1 + e^{-\alpha_1}}$$
(6.15a)

$$b_{3} = \frac{(\beta_{2} + 2\beta_{3} + \beta_{4})e^{\alpha_{2}}}{-1 + e^{\alpha_{2}}} \qquad b_{4} = \frac{(-\beta_{4} - \beta_{5})e^{-\alpha_{1} - \alpha_{2} - \alpha_{3} - \alpha_{4}}}{-1 + e^{-\alpha_{1} - \alpha_{2} - \alpha_{3} - \alpha_{4}}} \qquad (6.15b)$$

$$b_{5} = \frac{(-\beta_{3} - \beta_{6})e^{-\alpha_{3} - \alpha_{4}}}{-1 + e^{-\alpha_{3} - \alpha_{4}}} \qquad b_{6} = \frac{(\beta_{1} + \beta_{2})e^{\alpha_{1} + \alpha_{2} + \alpha_{4}}}{-1 + e^{\alpha_{1} + \alpha_{2} + \alpha_{4}}} \qquad (6.15c)$$

$$b_{7} = \frac{(-\beta_{1} - \beta_{2})e^{-\alpha_{3}}}{-1 + e^{-\alpha_{3}}}. \qquad (6.15d)$$

$$\frac{\alpha_3 - \alpha_4}{-\alpha_4} \qquad b_6 = \frac{(\beta_1 + \beta_2)e^{\alpha_1 + \alpha_2 + \alpha_4}}{-1 + e^{\alpha_1 + \alpha_2 + \alpha_4}} \tag{6.15c}$$

### 6.2.1.2 Network $\mathcal{N}_{13}$

The equations

$$Y^{(L)T} \mu = \ln \frac{E \nu}{E \lambda}$$

for this network are

$$\begin{split} \mu_1 + \mu_2 &= \ln \frac{\nu_1 + \nu_5}{\lambda_1 + \lambda_5} \\ \mu_3 &= \ln \frac{\nu_1}{\lambda_1} \\ \mu_3 &= \ln \frac{\nu_2}{\lambda_5} \\ \mu_2 + \mu_4 &= \ln \frac{\nu_2 + \nu_6}{\lambda_2 + \lambda_6} \\ \mu_5 &= \ln \frac{\nu_2}{\lambda_2} \\ \mu_5 &= \ln \frac{\nu_6}{\lambda_6} \\ \mu_6 + \mu_7 &= \ln \frac{\nu_3 + \nu_6}{\lambda_3 + \lambda_6} \\ \mu_8 &= \ln \frac{\nu_3}{\lambda_3} \\ \mu_8 &= \ln \frac{\nu_6}{\lambda_6} \\ \mu_4 + \mu_9 &= \ln \frac{\nu_4 + \nu_5}{\lambda_4 + \lambda_5} \\ \mu_{10} &= \ln \frac{\nu_5}{\lambda_5} \end{split}$$

These equations are solvable, if and only if

$$\ln \frac{\nu_1}{\lambda_1} = \ln \frac{\nu_5}{\lambda_5}, \qquad \qquad \ln \frac{\nu_2}{\lambda_2} = \ln \frac{\nu_6}{\lambda_6}, \qquad \qquad \ln \frac{\nu_3}{\lambda_3} = \ln \frac{\nu_6}{\lambda_6}, \qquad \qquad \ln \frac{\nu_4}{\lambda_4} = \ln \frac{\nu_5}{\lambda_5}.$$

Thus one obtains

$$\nu = \left(\lambda_1 \frac{\nu_5}{\lambda_5}, \lambda_2 \frac{\nu_6}{\lambda_6}, \lambda_3 \frac{\nu_6}{\lambda_6}, \lambda_4 \frac{\nu_5}{\lambda_5}, \nu_5, \nu_6\right)^T$$
(6.16)

with  $\lambda \in I\!\!R^6_{>0}$  and  $\nu_5$ ,  $\nu_6 > 0$ , free. Note that using  $\nu$  as above, one obtains

$$\ln \frac{\nu_1 + \nu_5}{\lambda_1 + \lambda_5} = \ln \frac{\nu_4 + \nu_5}{\lambda_4 + \lambda_5} = \ln \frac{\nu_5}{\lambda_5}$$

and

$$\ln\frac{\nu_2+\nu_6}{\lambda_2+\lambda_6} = \ln\frac{\nu_3+\nu_6}{\lambda_3+\lambda_6} = \ln\frac{\nu_6}{\lambda_6}$$

Using  $\kappa_1 := \ln \frac{\nu_5}{\lambda_5}$  and  $\kappa_2 := \ln \frac{\nu_6}{\lambda_6}$  and solving for  $\mu$ , one finally obtains

$$\mu = \underbrace{\begin{bmatrix} -1 & 0 & 1 & 0 \\ 1 & 0 & 0 & 0 \\ 0 & 0 & 1 & 0 \\ -1 & 0 & 0 & 1 \\ 0 & 1 & 0 & 0 \\ 0 & -1 & 0 & 1 \\ 0 & 0 & 0 & 1 \\ 1 & 0 & 1 & -1 \\ 0 & 0 & 1 & 0 \end{bmatrix}}_{=:M_{N_{13}}} \begin{pmatrix} \mu_2 \\ \mu_6 \\ \kappa_1 \\ \kappa_2 \end{pmatrix}$$
(6.17)

Let  $\mathcal{M}_{\mathcal{N}_{13}} = [\mathcal{M}_{\mathcal{N}_{13}}]$  and observe that  $\mathcal{M}_{\mathcal{N}_{13}} \cap \mathbb{R}^{10}_{\delta} \neq \emptyset$  and  $\mathcal{S}_{\mathcal{N}_{13}} \cap \mathbb{R}^{10}_{\delta} \neq \emptyset$  holds in the orthant  $\delta_1$ :

$$\delta_1 = (-1, 1, -1, -1, 1, 1, -1, 1, 1, -1)^T$$
(6.18)

Thus multistationarity has been established by Theorem 3. Using generators of  $\mathcal{M}_{\mathcal{N}_{13}} \cap \mathbb{R}^{10}_{\delta}$  and  $\mathcal{S}_{\mathcal{N}_{13}} \cap \mathbb{R}^{10}_{\delta}$  a parametrization of all pairs of steady states a, b that satisfy  $W^T b = W^T a$  is obtained:

$$\begin{array}{rcl} a_{1} & = & \frac{-\beta_{2} - \beta_{3}}{-1 + e^{-\alpha_{1}} - \alpha_{3} - 2\alpha_{4}}, & a_{2} & = & \frac{\beta_{4} + \beta_{3}}{-1 + e^{-\alpha_{1}} + \alpha_{3} + \alpha_{4}}, \\ a_{3} & = & \frac{-\beta_{3} - \beta_{4} - \beta_{9}}{-1 + e^{-\alpha_{4}}}, & a_{4} & = & \frac{-\beta_{1} - \beta_{2}}{-1 + e^{-\alpha_{3}} - \alpha_{4}}, \\ a_{5} & = & \frac{-\beta_{3}}{-1 + e^{-\alpha_{1}}}, & a_{6} & = & \frac{\beta_{2} + \beta_{5} + \beta_{3} + \beta_{6}}{-1 + e^{-1} + \alpha_{2}}, \\ a_{7} & = & \frac{-\beta_{1} - \beta_{4} - \beta_{6} - \beta_{8}}{-1 + e^{-\alpha_{2}}}, & a_{8} & = & \frac{\beta_{1} + \beta_{4} + \beta_{6} + \beta_{8}}{-1 + e^{-1} + \alpha_{3}}, \\ a_{9} & = & \frac{\beta_{5} + \beta_{6}}{-1 + e^{\alpha_{3}}} & a_{10} & = & -\frac{\beta_{5} - \beta_{6}}{-1 + e^{-\alpha_{4}}}. \end{array}$$
(6.19a)

Using the  $a_i$  given in (6.19a) one obtains a parametrization of the rate constants that ensure that  $a = (a_i)$  and  $b = (b_i)$ , i = 1, ..., 10 are steady states. Note that here  $\lambda_1 > 0, ..., \lambda_6 > 0$  can be can be chosen freely.

$$\begin{aligned} k_1 &= \frac{\lambda_1 + \lambda_2}{a_1 a_2} \quad k_2 &= \frac{\lambda_1}{a_3}, \quad k_3 &= \frac{\lambda_5}{a_3}, \\ k_4 &= \frac{\lambda_2 + \lambda_6}{a_2 - \lambda_4}, \quad k_5 &= \frac{\lambda_2}{a_5}, \quad k_6 &= \frac{\lambda_6}{a_5}, \\ k_7 &= \frac{\lambda_3 + \lambda_6}{a_6 a_7}, \quad k_8 &= \frac{\lambda_3}{a_8}, \quad k_9 &= \frac{\lambda_6}{a_8}, \\ k_{10} &= \frac{\lambda_4 + \lambda_5}{a_4 a_9}, \quad k_{11} &= \frac{\lambda_4}{a_4 a_9}, \quad k_{12} &= \frac{\lambda_5}{a_{10}} \end{aligned}$$

$$(6.20a)$$

### 6.2.1.3 Network $N_{14}$

The equations

$$Y^{(L)^T} \mu = \ln \frac{E \nu}{E \lambda}$$

for this network are

$$\begin{split} \mu_1 + \mu_2 &= \ln \frac{\nu_1 + \nu_6}{\lambda_1 + \lambda_6} \\ \mu_3 &= \ln \frac{\nu_1}{\lambda_1} \\ \mu_3 &= \ln \frac{\nu_6}{\lambda_6} \\ \mu_4 + \mu_5 &= \ln \frac{\nu_2 + \nu_6}{\lambda_2 + \lambda_6} \\ \mu_6 &= \ln \frac{\nu_2}{\lambda_2} \\ \mu_6 &= \ln \frac{\nu_6}{\lambda_6} \\ 2 \mu_4 &= \ln \frac{\nu_3 + \nu_7}{\lambda_3 + \lambda_7} \\ \mu_7 &= \ln \frac{\nu_3}{\lambda_3} \\ \mu_7 &= \ln \frac{\nu_7}{\lambda_7} \\ \mu_4 + \mu_8 &= \ln \frac{\nu_4 + \nu_8}{\lambda_4 + \lambda_8} \\ \mu_9 &= \ln \frac{\nu_8}{\lambda_6} \\ \mu_{10} + \mu_8 &= \ln \frac{\nu_5 + \nu_7 + \nu_8}{\lambda_5 + \lambda_7 + \lambda_8} \\ \mu_{11} &= \ln \frac{\nu_7 + \nu_8}{\lambda_7 + \lambda_8} \\ \mu_{11} &= \ln \frac{\nu_7 + \nu_8}{\lambda_7 + \lambda_8} \end{split}$$

These equations are solvable, if and only if

$$\ln \frac{\nu_1}{\lambda_1} = \ln \frac{\nu_6}{\lambda_6}, \qquad \ln \frac{\nu_2}{\lambda_2} = \ln \frac{\nu_6}{\lambda_6}, \qquad \ln \frac{\nu_3}{\lambda_3} = \ln \frac{\nu_7}{\lambda_7}, \qquad \ln \frac{\nu_4}{\lambda_4} = \ln \frac{\nu_8}{\lambda_8}, \qquad \ln \frac{\nu_5}{\lambda_5} = \ln \frac{\nu_7 + \nu_8}{\lambda_7 + \lambda_8}.$$

Thus one obtains  $\nu_1 = \lambda_1 \frac{\nu_6}{\lambda_6}$ ,  $\nu_2 = \lambda_2 \frac{\nu_6}{\lambda_6}$ ,  $\nu_3 = \lambda_3 \frac{\nu_7}{\lambda_7}$ ,  $\nu_4 = \lambda_4 \frac{\nu_8}{\lambda_8}$  and  $\nu_5 = \lambda_5 \frac{\nu_7 + \nu_8}{\lambda_7 + \lambda_8}$ . Thus one finally obtains

$$\nu = \left(\lambda_1 \frac{\nu_6}{\lambda_6}, \lambda_2 \frac{\nu_6}{\lambda_6}, \lambda_3 \frac{\nu_7}{\lambda_7}, \lambda_4 \frac{\nu_8}{\lambda_8}, \lambda_5 \frac{\nu_7 + \nu_8}{\lambda_7 + \lambda_8}, \nu_6, \nu_7, \nu_8\right)^T$$

Note that using  $\nu$  as above, one obtains

$$\begin{split} \ln \frac{\nu_1 + \nu_6}{\lambda_1 + \lambda_6} &= \ln \frac{\nu_2 + \nu_6}{\lambda_2 + \lambda_6} = \ln \frac{\nu_6}{\lambda_6},\\ \ln \frac{\nu_3 + \nu_7}{\lambda_3 + \lambda_7} &= \ln \frac{\nu_7}{\lambda_7},\\ \ln \frac{\nu_4 + \nu_8}{\lambda_4 + \lambda_8} &= \ln \frac{\nu_8}{\lambda_8}, \end{split}$$

and

$$\ln \frac{\nu_{5} + \nu_{7} + \nu_{8}}{\lambda_{5} + \lambda_{7} + \lambda_{8}} = \ln \frac{\nu_{7} + \nu_{8}}{\lambda_{7} + \lambda_{8}}.$$

Using

$$\kappa_1 := \ln \frac{\nu_6}{\lambda_6} \tag{6.21a}$$

$$\kappa_2 := \ln \frac{\nu_7}{\lambda_7} \tag{6.21b}$$

$$\kappa_3 := \ln \frac{1}{\lambda_8} \tag{6.21c}$$

$$\kappa_4 := \ln \frac{\lambda_1 + \lambda_8}{\lambda_7 + \lambda_8} \tag{6.21d}$$

and solving for  $\mu$ , one obtains

$$\mu = \underbrace{\left(\begin{array}{cccccc} -1 & 1 & 0 & 0 & 0 \\ 1 & 0 & 0 & 0 & 0 \\ 0 & 1 & 0 & 0 & 0 \\ 0 & 0 & \frac{1}{2} & 0 & 0 \\ 0 & 1 & -\frac{1}{2} & 0 & 0 \\ 0 & 0 & 1 & 0 & 0 \\ 0 & 0 & -\frac{1}{2} & 1 & 0 \\ 0 & 0 & 0 & 1 & 0 \\ 0 & 0 & 0 & \frac{1}{2} & -1 & 1 \\ 0 & 0 & 0 & 0 & 1 \end{array}\right)}_{=:M_{NA}} \begin{pmatrix} \mu_2 \\ \kappa_1 \\ \kappa_2 \\ \kappa_3 \\ \kappa_4 \end{pmatrix}}$$
(6.22)

If the  $\kappa_i$  are considered as free variables, (6.22) defines a linear subspace of  $\mathbb{R}^{11}$ . Note that in this case  $\nu_7$ ,  $\nu_8$ ,  $\lambda_7$  and  $\lambda_8$  are determined by (6.21b) – (6.21d): from (6.21b) follows  $\nu_7 = \lambda_7 e^{\kappa_2}$  and from (6.21c)  $\nu_8 = \lambda_8 e^{\kappa_3}$ . Inserting in (6.21d) and solving for  $\lambda_8$  yields

$$\lambda_8 = \frac{e^{\kappa_4} - e^{\kappa_2}}{e^{\kappa_3} - e^{\kappa_4}} \lambda_7 \tag{6.23}$$

Thus  $\lambda_8 > 0$ , if the  $\kappa_i$  have to satisfy either one of the relations

$$\kappa_3 > \kappa_4 > \kappa_2$$
 (6.24a)

$$\kappa_3 < \kappa_4 < \kappa_2 \tag{6.24b}$$

Let  $\kappa := (\mu_2, \kappa_1, \kappa_2, \kappa_3, \kappa_4)^T$  and  $\mathcal{M}_{\mathcal{N}_{14}} := \{\mu \in \mathbb{R}^{11} | \mu = M_{\mathcal{N}_{14}} \kappa$  and either  $\kappa_3 > \kappa_4 > \kappa_2$  or  $\kappa_3 < \kappa_4 < \kappa_2\}$ . Observe that  $\mathcal{S}_{\mathcal{N}_{14}} \cap \mathbb{R}_{\delta}^{11} \neq \emptyset$  and  $\mathcal{M}_{\mathcal{N}_{14}} \cap \mathbb{R}_{\delta}^{11} \neq \emptyset$  hold for  $\delta_1$  and  $-\delta_1$  as given below:

$$\delta_1 = (-1, 1, -1, -1, 1, -1, -1, 1, 1, -1, 1)^T$$
(6.25)

The pairs of steady states that satisfy the conservation relations are given by:

$$\begin{array}{rcl} a_{1} & = \frac{-\beta_{1} - \beta_{8} - 2\beta_{0}}{-1 + e^{-\alpha_{1}} - \alpha_{3}}, & a_{2} & = \frac{\beta_{10} + 2\beta_{11} + \beta_{2}}{-1 - 1 + e^{-\alpha_{1}}}, \\ a_{3} & = \frac{-\beta_{10} - 2\beta_{11} - \beta_{2}}{-1 - 1 - e^{-\alpha_{3}}}, & a_{4} & = \frac{-\beta_{12} - 2\beta_{13} - \beta_{3}}{-1 + e^{-\alpha_{2} - 2\alpha_{2}}}, \\ a_{5} & = \frac{\beta_{14} + 2\beta_{15} + \beta_{4}}{-1 + e^{-\alpha_{3}} + \frac{1}{2}(\alpha_{2} + 2\alpha_{3})}, & a_{6} & = \frac{-\beta_{14} - 2\beta_{15} - \beta_{4}}{-1 + e^{-\alpha_{3}}}, \\ a_{7} & = \frac{-\beta_{5} - \beta_{6} - \beta_{7}}{-1 + e^{-\alpha_{2}} - 2\alpha_{3}}, & a_{8} & = \frac{\beta_{10} + \beta_{12} + \beta_{14} + 2\beta_{7} + \beta_{8}}{-1 + e^{\frac{1}{2}(\alpha_{2} + 2\alpha_{3}) + \alpha_{4} + \alpha_{5}}}, \\ a_{9} & = \frac{\beta_{11} + \beta_{13} + \beta_{3} + \beta_{6}}{-1 + e^{\alpha_{4} + \alpha_{5}}}, & a_{10} & = \frac{-\beta_{1} - \beta_{2} - \beta_{3} - \beta_{4} - 2\beta_{5}}{-1 + e^{\frac{1}{2}(-2\alpha_{2} - 2\alpha_{3}) - \alpha_{5}}}, \\ a_{11} & = \frac{\beta_{1} + \beta_{2} + \beta_{3} + \beta_{4} + 2\beta_{5}}{-1 + e^{\alpha_{4}}}. \end{array}$$

$$(6.26a)$$

$$b_{1} = \frac{(-\beta_{1} - \beta_{8} - 2\beta_{0})e^{-\alpha_{1}} + \beta_{3}}{-1 + e^{-\alpha_{3}}}, \qquad b_{2} = \frac{(\beta_{10} + 2\beta_{11} + \beta_{2})e^{\alpha_{1}}}{-1 + e^{\alpha_{1}}}, \\ b_{3} = \frac{(-\beta_{10} - 2\beta_{11} - \beta_{2})e^{-\alpha_{3}}}{-1 + e^{-\alpha_{3}}}, \qquad b_{4} = \frac{(-\beta_{12} - 2\beta_{13} - \beta_{3})e^{\frac{1}{2}(-2\alpha_{2} - 2\alpha_{3})}}{-1 + e^{\frac{1}{2}(-2\alpha_{2} - 2\alpha_{3})}}, \\ b_{5} = \frac{(\beta_{11} + 2\beta_{15} + \beta_{4})e^{-\alpha_{3} + \frac{1}{2}(2\alpha_{2} + 2\alpha_{3})}}{-1 + e^{-\alpha_{3}} + \frac{1}{2}(2\alpha_{2} + 2\alpha_{3})}, \qquad b_{6} = \frac{(-\beta_{14} - 2\beta_{15} - \beta_{4})e^{-\alpha_{3}}}{-1 + e^{-\alpha_{3}}}, \\ b_{7} = \frac{(-\beta_{5} - \beta_{6} - \beta_{7})e^{-2\alpha_{2} - 2\alpha_{3}}}{-\alpha_{3}}, \qquad b_{8} = \frac{(\beta_{10} + \beta_{12} + \beta_{14} + 2\beta_{7} + \beta_{8})e^{\frac{1}{2}(2\alpha_{2} + 2\alpha_{3}) + \alpha_{4} + \alpha_{5}}}{(6.26b)}.$$

$$b_{9} = \frac{(\beta_{11} + \beta_{13} + \beta_{65} + \beta_{6}) e^{\alpha_{4} + \alpha_{5}}}{(1 + \alpha_{4} + \alpha_{5})}, \qquad b_{10} = \frac{(-\beta_{1} - \beta_{2} - \beta_{3} - \beta_{4} - 2\beta_{5}) e^{\frac{1}{2}(-2\alpha_{2} - 2\alpha_{3}) - \alpha_{5}}}{-1 + e^{\frac{1}{2}(-2\alpha_{2} - 2\alpha_{3}) - \alpha_{5}}},$$

$$b_{11} = \frac{(\beta_1 + \beta_2 + \beta_3 + \beta_4 + 2\beta_5) e^{\alpha_4}}{-1 + e^{\alpha_4}}.$$

Using the  $a_i$  given in (6.19a) one obtains a parametrization of rate constants that ensure that  $a = (a_i)$ and  $b = (b_i)$ , i = 1, ..., 11 from (6.26a) and (6.26b) are steady states. Note that  $\lambda_i > 0$ , i = 1, ..., 7can be chosen freely, but  $\lambda_8$  must satisfy (6.23). To this end a parametrization of the  $\kappa_i$  associated with  $\mu \in S_{\mathcal{N}_14} \cap \mathbb{R}^{11}_{61}$  is obtained:

$$\kappa = \begin{pmatrix} \alpha_1 \\ -\alpha_3 \\ -2\alpha_2 - 2\alpha_3 \\ \alpha_4 + \alpha_5 \\ \alpha_4 \end{pmatrix}.$$
(6.27a)

Then one obtains the following expression for  $\lambda_8$ :

$$\lambda_8 = \lambda_7 \frac{-e^{-2\,\alpha_2 - 2\,\alpha_3} + e^{\alpha_4}}{-e^{\alpha_4} + e^{\alpha_4 + \alpha_5}}.$$
(6.27b)

The rate constants are then given by

$$\begin{array}{rcl} k_1 & = \frac{\lambda_1 \pm \lambda_6}{a_1, a_2}, & k_2 & = \frac{\lambda_1}{a_3}, & k_3 & = \frac{\lambda_6}{a_3}, \\ k_4 & = \frac{\lambda_2 \pm \lambda_6}{a_4 a_5}, & k_5 & = \frac{\lambda_2}{a_6}, & k_6 & = \frac{\lambda_6}{a_6}, \\ k_7 & = \frac{\lambda_3 \pm \lambda_7}{a_4^2}, & k_8 & = \frac{\lambda_3}{a_7}, & k_9 & = \frac{\lambda_7}{a_7}, \\ k_{10} & = \frac{\lambda_4 \pm \lambda_8}{a_4 a_8}, & k_{11} & = \frac{\lambda_4}{a_9}, & k_{12} & = \frac{\lambda_8}{a_9}, \\ k_{13} & = \frac{\lambda_5 \pm \lambda_7 \pm \lambda_8}{a_8 a_{10}}, & k_{14} & = \frac{\lambda_5}{a_{11}}, & k_{15} & = \frac{\lambda_7 \pm \lambda_8}{a_{11}}. \end{array}$$

$$(6.28a)$$

### 6.2.2 Comparing robustness against concentration fluctuations

In this section the robustness of multistationarity with respect to variations in  $A_T$ , the total concentration of protein A is analysed. This is done to asses the usability of  $\mathcal{N}_4$ ,  $\mathcal{N}_{12}$ ,  $\mathcal{N}_{13}$  and  $\mathcal{N}_{14}$  as a switching device. Since concentrations typically fluctuate, a useful switch should maintain bistability over a wide range of concentrations. To this end, the following steps have been performed:

- (1) For each of the parametrizations of steady state pairs obtained for N<sub>4</sub>, N<sub>12</sub>, N<sub>13</sub> and N<sub>14</sub> in Section 6.2.1, the vectors α and β were chosen randomly (for each network approximately 1000 vectors α and β were created). Thus one obtains a set of 1000 steady state pairs for each network: (a<sub>i</sub><sup>(j)</sup>, b<sub>i</sub><sup>(j)</sup>), i = 1, ..., 1000, j = N<sub>4</sub>, N<sub>12</sub>, N<sub>13</sub>, N<sub>14</sub>.
- (2) Using parametrizations of the rate constants, a vector of rate constants was obtained for each pair of steady states: k<sub>i</sub><sup>(j)</sup>, i = 1, ..., 1000, j = N<sub>4</sub>, N<sub>12</sub>, N<sub>13</sub>, N<sub>14</sub>. Thus one obtains triplets τ<sub>i</sub><sup>(j)</sup> := (a<sub>i</sub><sup>(j)</sup>, b<sub>i</sub><sup>(j)</sup>, k<sub>i</sub><sup>(j)</sup>), i = 1, ..., 1000, j = N<sub>4</sub>, N<sub>12</sub>, N<sub>13</sub>, N<sub>14</sub> for which networks N<sub>4</sub>, N<sub>12</sub>, N<sub>13</sub> and N<sub>14</sub> show multistationarity (with a<sub>i</sub><sup>(j)</sup>, b<sub>i</sub><sup>(j)</sup>) as steady states and k<sub>i</sub><sup>(j)</sup> as vector of rate constants).
- (3) For every network and every triplet τ<sup>(j)</sup><sub>i</sub>, a numerical continuation of steady states was performed, using A<sub>T</sub> as bifurcation parameter. Bistability was confirmed numerically for each τ<sup>(j)</sup><sub>i</sub>.
- (4) Lower and upper turning point A<sup>(j)</sup><sub>low,i</sub> and A<sup>(j)</sup><sub>up,i</sub> of the numerical continuation were determined (see Fig. 6.5, where turning points are labeled LP, limit point). Using A<sup>(j)</sup><sub>low,i</sub> and A<sup>(j)</sup><sub>up,i</sub>, for each τ<sup>(j)</sup><sub>i</sub>, the relative range of bistability was determined as

$$\Delta_{A_{T,i}}^{(j)} = \frac{|A_{up,i}^{(j)} - A_{low,i}^{(j)}|}{\frac{A_{low,i}^{(j)} + A_{up,i}^{(j)}}{2}}.$$
(6.29)

(5) Finally  $\bar{\Delta}_{A_T}^{(j)}$ , the mean of all  $\Delta_{A_{T,i}}^{(j)}$  was calculated for each network.

The mean  $\bar{\Delta}_{A_T}^{(j)}$  can be interpreted as the average range of total concentration  $A_T$ , where bistability is possible. That is, in the average over all realisations of a particular network, where bistability was confirmed (represented by the triplets  $\tau_i^{(j)}$ ) the relative difference between upper and lower bifurcation



Figure 6.5: Numerical continuation for  $\mathcal{N}_4$  using  $A_T$  (total concentration of A) as bifurcation parameter. Here the value of  $x_1$  (concentration of A) is plotted against  $A_T$ .

point is given by  $\bar{\Delta}_{A_T}^{(j)}$ . The significance of this number is based on the following idea: if a certain network is designed to be a switch, and optimised against noise, its total concentration  $A_T$  should be approximately in the centre of the bistability region, that is close to  $c_b = \frac{A_T^1 + A_T^2}{2}$ . If this is the case, then  $A_T$  can in the average vary by at most  $\pm \frac{\bar{\Delta}_{A_T}^{(j)}}{2}$  without leaving the bistability range. Thus the  $\bar{\Delta}_{A_T}^{(j)}$  seem to be well suited to asses the robustness of the switch-like behaviour against fluctuations of protein concentration. The values of  $\bar{\Delta}_{A_T}^{(j)}$  for  $\mathcal{N}_4$ ,  $\mathcal{N}_{12}$ ,  $\mathcal{N}_{13}$  and  $\mathcal{N}_{14}$  are given in Table 6.5, while Figure 6.6 contains the distribution of the  $\Delta_{A_{T,i}}^{(j)}$  for each network. Apart from network  $\mathcal{N}_{13}$ , all networks have

	$\mathcal{N}_4$	$\mathcal{N}_{12}$	$\mathcal{N}_{13}$	$\mathcal{N}_{14}$
$\bar{\Delta}_{A_T}^{(j)}$	0.2562	0.2708	0.0786	0.2773

Table 6.5: Mean values  $\bar{\Delta}_{A_T}^{(j)}$ 

an average value of  $\approx 25$  % for  $\bar{\Delta}_{A_r}^{(j)}$ . Network  $\mathcal{N}_{13}$  shows a much lower value ( $\approx 8$  %, see Table 6.5). Note that in case of  $\mathcal{N}_{13}$ , a single distributive double-step activation mechanism is responsible for multistationarity. In the other cases either two double-step activation mechanisms (for network  $\mathcal{N}_4$ ) or an explicit autocatalytic activation (for networks  $\mathcal{N}_{12}$  and  $\mathcal{N}_{14}$ ) are responsible for multistationarity. These results thus suggest that, as a mechanism to cause multistationarity, an autocatalytic activation is stronger (i.e. more robust to concentration fluctuations) than a distributive activation, since one of the earlier leads to similar ranges of  $\bar{\Delta}_{A_r}^{(j)}$  as two of the latter.



Figure 6.6: Distribution of  $\Delta_{A_{T,i}}^{(j)}$  for  $\mathcal{N}_{12}$ ,  $\mathcal{N}_{4}$ ,  $\mathcal{N}_{13}$ , and  $\mathcal{N}_{14}$ .

### Chapter 7

## Multistationarity and Beyond

While in the previous chapters biochemical reaction networks have been analysed with respect to the qualitative phenomenon multistationarity, the topic of this chapter is a different qualitative phenomenon: the occurrence of limit cycles. As it turns out, for some biochemical reaction networks it is possible to determine analytically points in state space ('critical states') and in parameter space ('critical parameters'), where the Jacobian has a zero eigenvalue. Thus it is, for example, possible to determine critical points (states and parameters), where necessary conditions for saddle node bifurcations and certain bifurcations of codimension one (e.g. Takens-Bogdanov type bifurcations) are satisfied. The occurrence of a Takens-Bogdanov bifurcation can lead to limit cycles, while the only qualitative behaviour associated with saddle-node bifurcations is multistationarity.

This chapter is organized as follows: in Section 7.1 the Jacobian of  $f(x, k) := Y I_a v(k, x)$  is analyzed with respect to zero eigenvalues and necessary and sufficient conditions for the existence of a saddlenode bifurcation are given. In Section 7.2 the results are applied to network  $\mathcal{N}_4$  for the activation of an MAPK, where the Takens-Bogdanov type bifurcation is used to determine states and parameters, where the ODEs defined by  $\mathcal{N}_4$  exhibit limit cycles.

### 7.1 The Jacobian matrix

Let J(x,k) denote the Jacobian matrix of  $f(x,k) := Y I_a \operatorname{diag}(k) \phi(x)$  as defined in Chapter 2, that is  $J(x,k) := D_x f(x,k)$ . The structure of f(x,k) leads to a special form of J(x,k) [35]:

$$J(x,k) = N \operatorname{diag}(k) \operatorname{diag}(\phi(x)) Y^{(L)^{T}} \operatorname{diag}\left(\frac{1}{x}\right),$$
(7.1)

for x > 0 where  $\frac{1}{x}$  is shorthand for  $\left(\frac{1}{x_1}, \ldots, \frac{1}{x_n}\right)^T$ . In the remainder of this chapter the following assumption regarding J(x, k) is made:

**Assumption 3.** The image of J(x,k) is equal to the image of the matrix N, that is  $\operatorname{im} (J(x,k)) = \operatorname{im} (N)$ .

Note that the set of networks that satisfy Assumption 3 is neither empty nor academic, as it includes, for example,  $N_4$ ,  $N_5$  and  $N_6$ .

### 7.1.1 Zero eigenvalues of the Jacobian

The following notation will be used: as usual, let  $S = [Y I_a]$  and let  $S \in \mathbb{R}^{n \times s}$ ,  $W \in \mathbb{R}^{n \times (n-s)}$  be two matrices whose columns are orthonormal bases for im  $(Y I_a)$  and im  $(Y I_a)^{\perp}$ , respectively, with [61]:

$$\mathcal{S} = [S] = \operatorname{im} (J(x,k)),$$
$$\mathcal{S}^{\perp} := [W] = \operatorname{im} (N)^{\perp} = \ker (N^T)$$

Then x can be written as

$$x = \xi (y, z) = S y + W z$$

with  $y = S^T x \in \mathbb{R}^s$ ,  $z = W^T x \in \mathbb{R}^{n-s}$ . In the new coordinates y and z the system of ODEs  $\dot{x} = f(x,k)$  reads

$$\dot{y} = S^T \dot{x} = S^T f(\xi(y,z),k) =: g(y,z,k),$$
  
 $\dot{z} = W^T \dot{x} = 0$ 
(7.2)

showing the invariance of the subspace im  $(Y I_a)$ . Consider the reduced system

$$\dot{y} = g(y, z, k) = g(y, \nu)$$
(7.3)

with  $\nu = (z, k)$  as parameter vector. The Jacobian  $\mathbf{D}_{u} g$  is then given by

$$\mathbf{D}_{y} g(y, z, k) = S^{T} \mathbf{D}_{x} f\left(\xi\left(y, z\right), k\right) S$$

$$(7.4)$$

As im (N) is invariant under the flow of  $\dot{x} = f(x, k)$ , as the equation  $\dot{z} = 0$  in (7.2) shows,  $\mathbf{D}_x f(x, k)$ always has n - s eigenvalues  $\lambda_0 = 0$ . Values of y, z and k where  $\mathbf{D}_y g(y, z, k)$  has a zero eigenvalue are thus candidates for bifurcation points.

The following lemma, based on the fundamental result im  $(A)^{\perp} = \ker (A^T)$  for any  $(m \times n)$ -matrix A and Fredholm's Alternative [61], will be used to link zero eigenvalues of  $\mathbf{D}_y g(y, z, k)$  to zero eigenvalues of  $J \{\xi (y, z), k\}$ . As for example in [64], an eigenvalue  $\lambda$  is called *defective*, if its *algebraic multiplicity*  $m_{alg}(\lambda)$  is greater than its geometric multiplicity  $m_{geo}(\lambda)$  (i.e. if the multiplicity of  $\lambda$  as a root of the characteristic polynomial is greater than the number of linear independent eigenvectors corresponding to  $\lambda$ ). Recall

**Fact 6.**  $\lambda_0 = 0$  is a defective eigenvalue of a matrix  $A \in \mathbb{R}^{n \times n}$  iff there exists an  $x \neq 0$  with  $x \in \text{im}(A) \cap \ker(A)$ .

**Fact 7.**  $\lambda_0 = 0$  is a defective eigenvalue of a matrix  $A \in \mathbb{R}^{n \times n}$  of order r iff there exists a nontrivial  $x - w.l.o.g. \ 0 \neq x \in \operatorname{im}(A^T) - with$ 

$$A^{j} x \neq 0, \ j = 0, 1, \dots, r \quad and \ A^{r+1} x = 0.$$

Such an x will be called **principal vector of order** r and **principal vector of maximal order** r, if, in addition,  $x \notin im(A)$  holds.

**Lemma 7.** Let  $A \in \mathbb{R}^{n \times n}$  be a matrix of rank s < n. Let S and W be orthonormal bases for im (A) and ker  $(A^T)$ , respectively, and let V be a basis for ker (A). Consider the matrix  $B_1 := S^T A S \in \mathbb{R}^{n \times s}$ .

- 1.  $\lambda_0 = 0$  is an eigenvalue of  $B_1$  iff  $\lambda_0 = 0$  is a defective eigenvalue of A.
- 2.  $\lambda_0 = 0$  is a defective eigenvalue of A iff  $\Gamma = W^T V$  is singular.
- 3. If x is a principal vector of order r for A then any y with Sy = Ax is a principal vector of order r-1 for  $B_1$  note that y is to be taken as  $y = S^T Ax \in \text{im}(B_1^T)$ .
- 4. If y is a principal vector of order r for  $B_1$  then any x with Sy = Ax is a principal vector of order r + 1 for A note that x can be taken to be  $x^{\#} = A^{\#}S y \in im(A^T)$  with the Moore–Penrose inverse  $A^{\#}$  of A.
- 5.  $\lambda_0 = 0$  is an algebraically simple eigenvalue of  $B_1$  if and only if  $\lambda_0 = 0$  is a defective eigenvalue of A with  $m_{alg}(\lambda_0) = m_{geo}(\lambda_0) + 1$ .
- λ<sub>0</sub> = 0 is an algebraically simple eigenvalue of B<sub>1</sub> if and only if B<sub>1</sub> has one-dimensional left and right nullspace [β<sup>T</sup>] and [b], respectively, with β<sup>T</sup> b ≠ 0.

*Proof.* Apply the orthonormal transformation matrix  $\phi = (S, W) \in \mathbb{R}^{n \times n}$  to obtain  $B = \phi^T A \phi$  with

$$B = \begin{bmatrix} S^T A S & S^T A W \\ 0 & 0 \end{bmatrix} = \begin{bmatrix} B_1 & B_2 \\ 0 & 0 \end{bmatrix}.$$
 (7.5)

Note that one has

$$A = \phi B \phi^{T} = S B_{1} S^{T} + S B_{2} W^{T}$$
(7.6)

and thus

$$A^i S = SB_1^i. \tag{7.7}$$

not an eigenvalue of  $B_1$  at all.

- ad(1) If  $\lambda_0$  is a defective eigenvalue of A then there exists by Fact 6 a nontrivial  $v_0 \in im(A) \cap \ker(A)$ so that  $W^T v_0 = 0$  and  $v_0 = A v_1 = S z$  for suitable nonzero  $v_1$  and z. Thus by (7.6)  $0 = A v_0 = S B_1 S^T v_0 + 0 = S B_1 z$  (as  $S^T S = I_s$ ). As S is of full rank,  $B_1 z = 0$  follows. Vice versa, suppose that there exists  $\zeta \in \mathbb{R}^s$  with  $B_1 \zeta = 0$ ,  $\zeta \neq 0$ . Define  $v := S \zeta \in im(A)$ . Then  $W^T v = 0$  and, again by (7.6),  $Av = S B_1 S^T v + 0 = S B_1 \zeta = 0$ . Thus  $v \in \ker(A)$  so that  $\lambda_0$  is a defective eigenvalue of A by Fact 6.
- ad(2) If  $\lambda_0$  is a defective eigenvalue of A then we employ Fact 7. Choose a nontrivial x with  $y := A x \neq 0$ and  $A y = A^2 x = 0$ . Since y belongs to ker (A) it can be written as y = V z with a nonzero z. Since y belongs to im(A) it is orthogonal to  $[W] = ker(A^T)$ . Hence  $\Gamma y = W^T V z = 0$  for a nonzero z. Vice versa, if z is a nullvector of  $\Gamma = W^T V$  then  $V z \in [V] = ker(A), V z \neq 0$ , and V z is orthogonal to  $[W] = im(A)^{\perp}$ . Hence the nonzero V z belongs to ker(A) and im(A).
- ad(3) If x is given as in 3) the element  $\xi := Ax$  can be written as  $\xi = Sy$  with  $S^T \xi = S^T Ax = S^T Sy = y$ . Because of (7.7) we have  $SB_1^i y = A^i Sy = A^{i+1}x$ . Hence y is a principal vector of order r 1 for  $B_1$  (as S has full column rank).
- ad(4) If y is given as in 4) the element  $\eta := S y$  can be written as  $\eta = A x$ . Because of (7.7) we have as above  $SB_1^i y = A^{i+1} x$  so that x (e.g.  $x = x^{\#}$ ) is a principal vector of order r + 1 for A.
- ad(5) Assume  $\lambda_0$  is a multiple eigenvalue of  $B_1$  with two linearly independent nullvectors  $y_1$  and  $y_2$ . Each  $y_j$  generates a corresponding principal vector  $x_j$  of order 1 for A. Since the  $x_j$  are necessarily linearly independent, A has 0 with  $m_{alg}(\lambda_0) > m_{geo}(\lambda_0) + 1$ . Assume now that  $\lambda_0$  is a defective eigenvalue of  $B_1$  of order  $r \ge 1$ . By 4),  $\lambda_0$  is a defective eigenvalue of A of order  $r + 1 \ge 2$ . Thereby we have shown  $\leftarrow$  of 5). For  $\Rightarrow$  we refer to 3) stating that a principal vector of order 2 for A yields a principal vector of order 1 for  $B_1$ . In case  $\lambda_0 = 0$  is a non-defective eigenvalue of A we recall 1) stating that  $\lambda_0$  is
- ad(6) Note that the statement in 6) is a particular case of the one in 2). The equation  $B_1 y = b$  is solvable for y iff  $z^T B_1 = 0$  implies  $z^T b = 0$ . Since the left nullspace of  $B_1$  is spanned by  $\beta^T$ , the equivalence in 6) follows.

The following Corollary uses Lemma 7 to establish a link between right (left) principal vectors of A and right (left) nullvectors of B.

**Corollary 5.** If B has an algebraically simple eigenvalue  $\lambda_0 = 0$  or, equivalently, if A has  $\lambda_0$  as a defective eigenvalue with  $m_{alg}(\lambda_0) = m_{geo}(\lambda_0) + 1$ , one has the following:

A principal right nullvector a of A yields a right nullvector b of B if  $b = S^T A a$  is chosen (with A a = S b).

A principal left nullvector  $\alpha^T$  of A yields a left nullvector  $\beta^T$  of B if  $\beta^T = \alpha^T S$  is chosen (with  $\alpha^T A = \beta^T S^T A$ ).

Vice versa: Nullvectors b and  $\beta^T$  of B give rise to corresponding principal vectors a and  $\alpha^T$  of A of maximal order 1 via Aa = Sb and  $\alpha^T A = \beta^T S^T A$  respectively, that is  $a = A^{\#} Sb \in \text{im}(A^T)$  and  $\alpha = S\beta \in \text{im}(A^T)$ .

Proof. Consult Lemma 7 for right eigenvectors and principal vectors. Turn to left eigenvectors and principal vectors.

 $F^T \beta^T B = 0$  for a nonzero  $\beta^T$  then  $\alpha^T := \beta^T S^T$  (satisfying  $\alpha^T S = \beta^T$ ) is nonzero. The equalities  $\alpha^T = \beta^T S^T$  and  $\alpha^T A = \beta^T S^T A$  imply  $\alpha^T A \neq 0$  as  $S^T A$  has full row-rank (proof by contradiction). Furthermore one deduces  $\alpha^T A^2 = \beta^T S^T A S S^T A = \beta^T B S^T A = 0$ .

Vice versa, given a left principal vector  $\alpha^T \neq 0$  with  $\alpha^T A \neq 0$  and  $\alpha^T A^2 = 0$  we take the nonzero  $\alpha^T S = \beta^T$  (with  $\alpha^T = \beta^T S^T + \sigma^T$  with  $\sigma \in \ker(A^T)$ ) and verify  $\beta^T B = \beta^T S^T A S = \alpha^T A S = 0$  (as  $A^T \alpha \in \ker(A^T) = [S]^{\perp}$ ).

As discussed above, Lemma 7 and Corollary 5 establish a link between spectral properties of the Jacobian  $A = J(\xi(y, z), k)$  and the Jacobian  $B = \mathbf{D}_{y}g(y, z, k)$  of the reduced system (7.3).

### 7.1.2 Saddle-node bifurcations

In this section the focus is on saddle-node bifurcations. The following well-known theorem gives necessary and sufficient conditions for a saddle-node bifurcation of the system  $\dot{y} = g(y, z, k)$  as defined in (7.3). Let  $\mu$  be the component of the parameter vector  $\nu = (z, k)$  that will be used as bifurcation parameter.

**Theorem 5 (see e.g. [51], p. 330 and [7], p. 497).** Suppose  $(y_0, \nu_0)$  is a zero of g and suppose that the  $s \times s$  matrix  $\mathbf{D}_y g(y_0, \nu_0)$  has an algebraically simple eigenvalue  $\lambda_0 = 0$  with right eigenvector b and left eigenvector  $\beta^T$ . Furthermore suppose that the following conditions are satisfied:

$$\beta^T \mathbf{D}_{\mu} g(y_0, \nu_0) \neq 0, \quad \beta^T \left[ \mathbf{D}_y^2 g(y_0, \nu_0)(b, b) \right] \neq 0.$$
 (7.8)

Then there is a smooth curve of zeroes of g passing through  $(y_0, \nu_0)$ . Depending on the signs of the expressions in (7.8), there are no or two zeroes near  $y_0$  for  $\mu \neq \mu_0$  whereas the other components of  $\nu_0$  remain fixed.

In the remainder of this section it is examined, when conditions (7.8) are satisfied for the system (7.3) in terms of the Jacobian  $J(\xi(y,z), k)$ . First note:

(N1) From Lemma 5 follows that  $\mathbf{D}_{y} g(y_{0}, \nu_{0})$  has an algebraically simple eigenvalue  $\lambda_{0} = 0$  if and only if  $J(\xi(y, z), k)$  has  $\lambda_{0}$  as a defective eigenvalue with  $m_{alg}(\lambda_{0}) = m_{geo}(\lambda_{0}) + 1$ .

Now, if  $\mathbf{D}_{y} g(y_0, \nu_0)$  has an algebraically simple eigenvalue  $\lambda_0 = 0$ , observe the following:

(N2) Ad  $\beta^T \mathbf{D}_{\mu} g(y_0, \mu_0) \neq 0$ : if  $\mu$  is any rate constant  $k_i$ , then

$$\mathbf{D}_{\mu} g(y_0, \mu_0) = S^T \mathbf{D}_{k_i} f(x, k) = S^T N \operatorname{diag} (\phi(x_0)) e_i$$

and therefore

$$\beta^T \mathbf{D}_{\mu} g(y_0, \mu_0) = \phi_i(x_0) \beta^T S^T N e_i \neq 0$$

for at least one i (as  $\phi_i(x_0) > 0$  and  $[S] = \operatorname{im}(N)$ ).

(N3) Ad  $\beta^T \left[ \mathbf{D}_y^2 g(y_0, \nu_0)(b, b) \right] \neq 0$ : From Lemma 7 and Corollary 5 follows

$$\beta^{T} \left[ \mathbf{D}_{y}^{2} g(y_{0}, z_{0}, k_{0})(b, b) \right] \neq 0 \Leftrightarrow \alpha^{T} \left[ \mathbf{D}_{x}^{2} f(\xi(y_{0}, z_{0}), k_{0})(A \, a, A \, a) \right] \neq 0$$
(7.9)

with  $A := J(\xi(y, z), k)$ , left and right principal vectors  $\alpha^T$  and a of  $J(\xi(y, z), k)$ . To see this, note that Aa = Sb, thus the evaluation of  $\mathbf{D}_y^2 g a t(b, b)$  and of  $\mathbf{D}_x^2 f a t(Aa, Aa)$  amounts to the same. Since  $\alpha^T S = \beta^T$  (or equivalently  $\alpha^T = \beta^T S^T + [W^T]$ ) and since – by the particular form of  $f - \mathbf{D}_x^2 f(\xi(y_0, z_0), k_0)(Aa, Aa) \subset \operatorname{im}(N)$  the equivalence (7.9) holds true.

As consequence of this discussion the following fact concerning the originally given system (2.4), (2.5) holds:

**Fact 8.** The system (2.4), (2.5) has a saddle-node bifurcation at  $(x_0, k_0)$  (within the plane  $W^T x = W^T x_0 =: c$ ) if the following conditions are satisfied:

(a)  $\lambda_0 = 0$  is a defective eigenvalue of  $J(x_0, k_0)$  with  $m_{alg}(\lambda_0) = m_{geo}(\lambda_0) + 1$ .

(b)  $\alpha^T \left[ \mathbf{D}_x^2 f(x_0, k_0)(Aa, Aa) \right] \neq 0$  is satisfied for left and right principal vectors  $w_0^T$  and  $v_0$  of  $J(x_0, k_0)$ .

Proof. By the present assumptions,  $\mathbf{D}_y g(y_0, z_0, k_0)$  has  $\lambda_0 = 0$  as a simple eigenvalue at  $y_0 = S^T x_0, z_0 = W^T x_0$  and  $k_0$ . By assumption (a) and Note (N2) from above the condition  $\beta^T \mathbf{D}_\mu g(y_0, \mu_0) \neq 0$  is satisfied for at least one choice of  $\mu = k_i$ . Combined with Note (N3) and the assumption (b), all conditions of Theorem 5 are satisfied and the reduced system (7.3) has thus a saddle-node bifurcation at  $y_0 = S^T x_0, z_0 = W^T x_0$  and  $k_0$ . As system (7.3) is on  $\{x : W^T x = W^T x_0 = z_0\}$  equivalent to the original system (2.4), this has a saddle-node bifurcation at  $(x_0, k_0)$  as well.

### 7.2 Application to a double-phosphorylation mechanism

In this section the consequences of Fact 8 for network  $\mathcal{N}_4$  are determined. As in Section 4.2.1, let  $E \in \mathbb{R}^{9\times 6}$  be a set of generators for the pointed polyhedral cone ker $(Y I_a) \cap \mathbb{R}^9_{\geq 0}$ . Then, as in Section 3.1, the following relation must hold for a solution  $(x_0, k_0)$  to  $Y I_a v(k_0, x_0) = 0$  (with  $x_0 > 0, k_0 > 0$ ):

$$\operatorname{diag}(k_0) \phi(x_0) = E \lambda \tag{7.10}$$

for some positive  $\lambda \in \mathbb{R}^6$ . By (7.1) and (7.10) the Jacobian  $J(x_0, k_0)$  evaluated at this zero state is thus  $N \operatorname{diag}(E \lambda) Y^{(L)T} \operatorname{diag}(x_0^{-1})$ . One has to determine nonzero vectors v with  $J(x_0, k_0) v = 0$ . With  $\gamma := \operatorname{diag}(x_0^{-1}) v$  one is thus looking for  $\gamma$ 's with

$$J(\lambda) \gamma = 0, \quad J(\lambda) := N \operatorname{diag}(E \lambda) Y^{(L)^{I}}.$$

Thus diag  $(E \lambda) Y^{(L)T} \gamma \in \ker(N)$  and therefore  $Y^{(L)T} \gamma = \operatorname{diag}((E \lambda)^{-1}) E \alpha$ , for some  $\alpha \in \mathbb{R}^6$ . This leads to the following set of equations (note the similarity to (4.1a)–(4.1l)):

$$\gamma_1 + \gamma_2 = \frac{\alpha_1 + \alpha_5}{\lambda_1 + \lambda_5}, \quad \gamma_3 = \frac{\alpha_1}{\lambda_1}, \quad \gamma_3 = \frac{\alpha_5}{\lambda_5}$$
(7.11a)

$$\gamma_2 + \gamma_4 = \frac{\alpha_2 + \alpha_6}{\lambda_2 + \lambda_6}, \quad \gamma_5 = \frac{\alpha_2}{\lambda_2}, \quad \gamma_5 = \frac{\alpha_6}{\lambda_6}$$
 (7.11b)

$$\gamma_6 + \gamma_7 = \frac{\alpha_3 + \alpha_6}{\lambda_3 + \lambda_6}, \quad \gamma_8 = \frac{\alpha_3}{\lambda_3}, \quad \gamma_8 = \frac{\alpha_6}{\lambda_6} \tag{7.11c}$$

$$\gamma_4 + \gamma_7 = \frac{\alpha_4 + \alpha_5}{\lambda_4 + \lambda_5}, \quad \gamma_9 = \frac{\alpha_4}{\lambda_4}, \quad \gamma_9 = \frac{\alpha_5}{\lambda_5}$$
(7.11d)

The system (7.11a) - (7.11d) is solvable, if

$$\frac{\alpha_1}{\lambda_1} = \frac{\alpha_5}{\lambda_5} = \frac{\alpha_1 + \alpha_5}{\lambda_1 + \lambda_5} = \frac{\alpha_4}{\lambda_4} = \frac{\alpha_4 + \alpha_5}{\lambda_4 + \lambda_5} =: \kappa_1$$

and

$$\frac{\alpha_2}{\lambda_2} = \frac{\alpha_6}{\lambda_6} = \frac{\alpha_2 + \alpha_6}{\lambda_2 + \lambda_6} = \frac{\alpha_3}{\lambda_3} = \frac{\alpha_4 + \alpha_5}{\lambda_4 + \lambda_5} =: \kappa_2.$$

Using  $\kappa_1$  and  $\kappa_2$  and solving for  $\gamma$  yields:

$$\gamma = \gamma_1 (1, -1, 0, 1, 0, 1, -1, 0, 0)^T + \kappa_1 (0, 1, 1, -1, 0, -2, 2, 0, 1)^T + \kappa_2 (0, 0, 0, 1, 1, 2, -1, 1, 0)^T.$$
(7.12)

Note that  $\gamma$  as defined in (7.12) is equivalent to  $\mu$  as in (4.5): both define the same linear subspace  $\mathcal{M}_{\mathcal{N}_4}$ . Denote this by

$$\gamma = M_{\mathcal{N}_4} \begin{bmatrix} \gamma_1 \\ \kappa_1 \\ \kappa_2 \end{bmatrix},$$

with  $M_{\mathcal{N}_4}$  as in (4.6a). Now the right null space of  $J(x_0, k_0)$  can be represented as  $[V(x_0)] = [\operatorname{diag}(x_0) \ M_{\mathcal{N}_4}]$  with  $V(x_0)$  given by

$$V(x_0) = \operatorname{diag}(x_0) M_{\mathcal{N}_4} = \begin{bmatrix} x_{10} & 0 & 0 \\ -x_{20} & 0 & x_{20} \\ 0 & 0 & x_{30} \\ x_{40} & x_{40} & -x_{40} \\ 0 & x_{50} & 0 \\ x_{60} & 2x_{60} & -2x_{60} \\ -x_{70} & -x_{70} & 2x_{70} \\ 0 & x_{80} & 0 \\ 0 & 0 & x_{90} \end{bmatrix} .$$
(7.13)

Since  $J(x_0, k_0) \in \mathbb{R}^{9\times 9}$  has a 3-dimensional nullspace and thus a 6-dimensional range and since, by (7.1), the range of  $J(x_0, k_0)$  belongs to im (N), Assumption 3 is satisfied for  $\mathcal{N}_4$ .

### 7.2.1 Zero eigenvalues

To check whether  $J(x_0, k_0)$  has  $\lambda_0 = 0$  has a defective eigenvalue consider the matrix  $\Gamma$  in Lemma 7, now given by  $\Gamma(x_0) = W^T V(x_0)$ . It can be obtained as

$$\Gamma(x_0) = \begin{bmatrix} -x_{20} & x_{50} & x_{20} + x_{30} \\ -x_{70} & -x_{70} + x_{80} & 2x_{70} + x_{90} \\ x_{10} + x_{40} + x_{60} & x_{40} + x_{50} + 2x_{60} + x_{80} & x_{30} - x_{40} - 2x_{60} + x_{90} \end{bmatrix}.$$
 (7.14)

 $\Gamma(x_0)$  is singular if and only if

$$\Gamma(x_0) \ \pi_0 = W^T V(x_0) \ \pi_0 = 0 \tag{7.15}$$

has a nontrivial solution  $\pi_0 \in \mathbb{R}^3$ . Note that

$$\Gamma(x_0) \ \pi_0 = W^T \ \text{diag}(x_0) \ M_{\mathcal{N}_4} \ \pi_0 = W^T \ \text{diag}(M_{\mathcal{N}_4} \ \pi_0) \ x_0$$

and that [W] is the orthogonal complement of the stoichiometric subspace  $S_{\mathcal{N}_4}$ , that is  $W^T S = 0$ , where  $S \in \mathbb{R}^{9 \times 6}$  is an orthonormal basis for  $S_{\mathcal{N}_4}$ . To obtain  $x_0$ , the following observation is useful: if  $u \in \mathbb{R}^9$  is a solution to  $W^T u = 0$ , then  $\tilde{u} = \text{diag}\left(\frac{1}{M\pi}\right) u$  satisfies  $W^T \text{diag}(M\pi) \tilde{u} = 0$  (for nonzero  $\pi \in \mathbb{R}^3$ ). The following results establish parametrizations of points where  $\Gamma(x_0)$  is singular:

**Corollary 6.** Points  $x_0 \in \mathbb{R}^9$  where  $\Gamma(x_0)$  is singular are given by

$$x_0 = \frac{S\,\rho_0}{M_{\mathcal{N}_4}\,\pi_0},\tag{7.16a}$$

with  $(M_{N_4} \pi_0)_i \neq 0, i = 1, ..., 9$ . Right Eigenvectors  $\gamma_0$  of  $J(x, \lambda)$  with  $W^T \gamma_0 = 0$  are given by

$$\gamma_0 = S \rho_0$$
 (7.16b)

Proof.  $x_0 = \frac{S \rho_0}{M_{\mathcal{N}_4} \pi_0}$  follows from the previous discussion. Recall that by (7.12)  $\gamma_0 = diag(x_0) \ M_{\mathcal{N}_4} \pi_0$ and thus  $\gamma_0 = diag\left(\frac{S \rho_0}{M_{\mathcal{N}_4} \pi_0}\right) \ M_{\mathcal{N}_4} \pi_0 = S \rho_0$ . Note that  $W^T \gamma_0 = 0$  by  $[W] = S_{\mathcal{N}_4}^{\perp}$ .

For biochemical reaction networks only positive  $x_0$  are of interest. To obtain those, the condition

$$W^{T} \operatorname{diag}(M_{N_{4}} \pi_{0}) x_{0} = 0$$

must hold for positive  $x_0$ . The following Corollary can be established:

Corollary 7. The equation

### $W^T \operatorname{diag}\left(M_{\mathcal{N}_4} \pi_0\right) x_0 = 0$

has a positive solution  $x_0 \in \mathbb{R}^9_{>0}$  (for some  $\pi_0 \in \mathbb{R}^3$ ), if and only if an orthant  $\delta$  of  $\mathbb{R}^9$  exists, where

- (i)  $S_{\mathcal{N}_4} \cap \mathbb{R}^9_\delta \neq \emptyset$
- (ii)  $\mathcal{M}_{\mathcal{N}_4} \cap \mathbb{R}^9_{\delta} \neq \emptyset$ .

In this case collect generators of  $S_{\mathcal{N}_4} \cap \mathbb{R}^9_{\delta}$  and  $\mathcal{M}_{\mathcal{N}_4} \cap \mathbb{R}^9_{\delta}$  in  $E^{\delta}_{S_{\mathcal{N}_4}}$  and  $E^{\delta}_{\mathcal{M}_{\mathcal{N}_4}}$ . Then

$$x_0 = \frac{E_{S_{N_4}}^{\delta}\beta}{E_{M_{N_4}}^{\delta}\alpha},$$
(7.17)

for  $\alpha$ ,  $\beta > 0$  of appropriate dimension and  $\left(E_{\mathcal{M}_{\mathcal{N}_4}}^{\delta} \alpha\right)_i \neq 0, i = 1, \ldots, 9.$ 

Note that a different representation of  $x_0$  exists for each orthant, where  $\mathcal{M}_{(\mathcal{N}_4)}^{\delta} \cap \mathbb{R}_{\vartheta}^{q} \neq \emptyset$  and  $\mathcal{S}_{\mathcal{N}_4} \cap \mathbb{R}_{\vartheta}^{q} \neq \emptyset$ . For network  $\mathcal{N}_4$ , 14 such orthants exist (see Section 4.3.1, in particular (4.16) and Table 4.1 for representations of  $\mathcal{M}_{\mathcal{N}_4} \cap \mathbb{R}_{\vartheta_i}^{q}$  and Table 4.2 representations of  $\mathcal{S}_{\mathcal{N}_4} \cap \mathbb{R}_{\vartheta_i}^{q}$ ).

From Table 4.1 and Table 4.2 follows, that, for example for  $\delta_1$ , all solutions  $x_0$  to (7.2.1) can be obtained as:

$$x_{0}^{\delta_{1}} = \left(\frac{\beta_{1} + \beta_{3} + \beta_{4} + \beta_{5}}{\alpha_{1}}, \frac{\beta_{1}}{\alpha_{2}}, \frac{\beta_{2}}{\alpha_{1} + \alpha_{2}}, \frac{\beta_{3}}{\alpha_{2} + \alpha_{3}}, \frac{\beta_{1} + \beta_{2}}{\alpha_{3}}, \frac{\beta_{4}}{\alpha_{3}}, \frac{\beta_{4}}{\alpha_{1} + 2(\alpha_{2} + \alpha_{3})}, \frac{\beta_{5}}{\alpha_{1} + 2\alpha_{2} + \alpha_{3}}, \frac{\beta_{5} + \beta_{6}}{\alpha_{3}}, \frac{\beta_{6}}{\alpha_{1} + \alpha_{2}}\right)^{T}.$$
(7.18)

Finally one obtains the following Corollary:

**Corollary 8.** Given a positive zero  $(x_0, k_0)$  of  $Y I_a v(k_0, x_0) = 0$ , the Jacobian  $\mathbf{D}_{\mu} g(y_0, \nu_0)$  has the eigenvalue  $\lambda_0 = 0$  with  $m_{geo}(\lambda_0) = 1$ , if and only if  $x_0$  can be parametrized as in (7.17) for some orthant  $\delta$ . For a  $x_0$  as in (7.17) to be a component of a zero of  $Y I_a v(k_0, x_0) = 0$ , the vector of rate constants  $k_0$  needs to be of the form (see (3.10a)):

$$k_0 = \operatorname{diag}\left(\phi\left(x_0^{-1}\right)\right) E \lambda. \tag{7.19}$$

**Remark 14.** Equation (7.16a) and equation (7.19) give an explicit representation of the manifold of potential critical points (in the sense of the Jacobian  $\mathbf{D}_{\mu}g(y_0, v_0)$  having a geometrically simple eigenvalue 0). The eigenvalue  $\lambda_0 = 0$  will have algebraic multiplicity 1 too, iff part (6) of Lemma 7 holds true, i.e. if the one-dimensional left nullspace  $[\beta^T]$  is not orthogonal to the right nullspace spanned by  $S^T V \rho$ :

$$\frac{1}{|\beta| |S^T V \rho|} \beta^T S^T V \rho \neq 0.$$
(7.20)

In case the additional hypotheses (7.8) of Theorem 5 are satisfied, saddle-node bifurcations do occur in the reduced system (7.3) and hence in the full system (2.4).

By Fact 8, these bifurcation conditions can be checked with the right eigenvector  $V\rho$  of J and the left principal vector  $\alpha^T$  of J satisfying  $\alpha^T = \beta^T S^T$ .

#### 7.2.2 Saddle-node bifurcations

To illustrate the consequences of Fact 8 and Remark 14, the positive steady states for  $N_4$  obtained in [11] were analyzed. To this end, the parameter vector

$$k = (10.929273, 34.638118, 34.638118, 21.999084, 1.7135134, 1.7135134, 788.27203, 1, 1, 686.21469, 14.271831, 14.271831)^T.$$
(7.21)

and the steady state

$$\begin{split} x^{(1)} &= \left(1.7047107, 0.42617769, 0.11461683, 0.84691022, 2.3169421, 0.73726101, 1.3662e-2, \right.\\ & \left. 3.9701114, 0.27817813 \right)^T \end{split}$$

were used. Numerical continuation of  $x^{(1)}$  using the rate constant  $k_1$  as bifurcation parameter revealed the existence of two positive limit points (c.f. Fig. 7.1):  $\xi_1$  at  $k_1^{(1)} = 10.78047065415938$  and  $\xi_2$  at  $k_1^{(2)} = 16.92069563203855$ :

Evaluated at each of these points, rank( $\Gamma(\xi_i)$ ) = 2 (as expected). Corresponding kernel vectors  $\rho_1$  and  $\rho_2$  are:

$$\rho_1 = \begin{bmatrix} +1.1303636028285451e+00\\ -8.0163523743329912e-02\\ +1.2792783092819293e+00 \end{bmatrix}, \rho_2 = \begin{bmatrix} +3.5343229854352359e-01\\ -5.0805188443278348e-01\\ +8.8274223950851360e-01 \end{bmatrix}$$

The observation rank( $\Gamma(\xi_i)$ ) = 2 implies that  $A_i := J(\xi_i, k^{(i)})$  has  $\lambda_0$  as a defective eigenvalue. (The saddle-node bifurcation occurs at  $(\xi_i, k^{(i)})$ , where  $k^{(i)} = k$  as (7.21), apart from  $k_1^{(i)}$ , which is the value of the bifurcation parameter given above.) Thus  $\mathbf{D}_y g(y, \nu)$  has  $\lambda_0$  as an eigenvalue of geometric multiplicity 1 (Corollary 8). To show that  $m_{alg} = 1$  as well, we use Remark 14 and (7.20): left and right eigenvectors  $\beta_1$  and  $b_1$  of  $S^T A_1 S$  and  $\beta_2$  and  $b_2$  of  $S^T A_2 S$  are

$$\beta_1 = \begin{bmatrix} -4.1964341033838814e-10\\ -1.25040064370758152e-10\\ -3.881092038143e-01\\ -3.881092038340094e-01\\ -3.881092038340094e-01\\ -4.64373323901632e-02\\ -6.64373323901632e-02\\ -6.64373323901632e-02\\ -6.64373323901632e-02\\ -6.64373323901632e-02\\ -8.89750404708077e-01\\ -1.2652490090167902e-01\\ -1.2652490090167902e-01\\ -3.898740640468093577e-01\\ -3.8987406404680932e-01\\ -3.89874064689032e-01\\ -3.89874064689032e-01\\ -3.898740648904689032e-01\\ -3.898740648904689032e-0\\ -3.898740648904689032e-0\\ -3.898740648904689032e-0\\ -3.898740648904689032e-0\\ -3.8987406489042e-0\\ -3.8987406489042e-0\\ -3.898740648904689032e-0\\ -3.898740648904689032e-0\\ -3.898740648904689032e-0\\ -3.898740648904689042e-0\\ -3.89874064890489042e-0\\ -3.89874064890489042e-0\\ -3.89874064890489042e-0\\ -3.89874064890489042e-0\\ -3.89874064890489042e-0\\ -3.89874064890489042e-0\\ -3.89874064890489042e-0\\ -3.89874064890489042e-0\\ -3.89874064890489042e-0\\ -3.89874064890489040489044e-0\\ -3.898740648904489040489044e-0\\ -3.898740648904$$

It is straightforward to verify that  $\beta_1^T b_1 \neq 0$  and  $\beta_2^T b_2 \neq 0$ . Thus  $m_{alg}(\lambda_0) = m_{geo}(\lambda_0) = 1$  for  $\lambda_0$  as an eigenvalue of  $S^T A_i S$  (and thus  $m_{alg}(\lambda_0) = m_{geo}(\lambda_0) + 1$  for  $\lambda_0$  as an eigenvalue of  $A_i$ ). To check hypothesis (b) of Fact 8 note that  $\mathbf{D}_y^2 f(x,k)(v,v)$  evaluated at  $v \in \mathbb{R}^9$  is given by

$$\mathbf{D}_{y}^{2} f(x,k)(v,v) = \left(2 v_{1} v_{2}, 0, 0, 2 v_{2} v_{4}, 0, 0, 2 v_{6} v_{7}, 0, 0, 2 v_{4} v_{7}, 0, 0\right)^{T}.$$

Using this equation together with  $\alpha_i = S \beta_i$  and  $J\left(\xi_i, k^{(i)}\right) a_i = S b_i$  in (7.9), it is straightforward to establish that condition (b) is satisfied for  $(\xi_1, k^{(1)})$  and  $(\xi_1, k^{(2)})$ .

To see the consequences of Corollary 8, choose  $\rho_0 \in \mathbb{R}^6_{>0}$  and  $\pi_0 \in \mathbb{R}^3_{>0}$  and verify numerically that a saddle-node bifurcation occurs at the corresponding pair  $(x^{(a)}, k_0)$  (with  $x^{(a)}$  as in (7.16a) and  $k_0$  as in (7.19)): choose, for example,  $x^{(a)} = (6, 1, 1, 1, 1, 1, 3, 1)^T$  and  $k_0 = (\frac{1}{3}, 1, 1, 2, 1, 1, 2, \frac{1}{3}, \frac{1}{3}, 2, 1, 1)^T$ . Numerical continuation of  $x^{(a)}$  using rate constant  $k_1$  as bifurcation parameter confirms a saddle-node bifurcation at  $x^{(a)}$  (see Fig. 7.2, where  $x_3$  is plotted against  $k_1$ ).



Figure 7.1: Continuation of  $x^{(1)}$  using  $k_1$ . The figure shows  $x_3$ , the third component of the state vector. Along the continuation two positive saddle-node points have been identified ( $\xi_1$  and  $\xi_2$ ).



Figure 7.2: Continuation of  $x^{(a)}$ , using rate constant  $k_1$  as bifurcation parameter. The figure shows  $x_3$ , the third component of the state vector. The starting point of the continuation  $x^{(a)}$  can be identified as saddle-node point.

#### 7.2.3 Bifurcations of higher codimension: Bogdanov-Takens

The ideas discussed in the previous section can be used to determine points where bifurcations of higher codimension can occur. For bifurcations of Bogdanov-Takens type the Jacobian of the reduced system must have a Jordan block of the form  $\begin{bmatrix} 0 & 0 \\ 0 & 0 \end{bmatrix}$ , see e.g. [51, p. 456] or [43, p. 272]).

From Lemma 7 follows that this is the case, iff the Jacobian of the original system has a zero eigenvalue with algebraic multiplicity three and geometric multiplicity one, that is, the following vectors must
exist:

$$J(x, k) \gamma_0 = 0$$
$$J(x, k) \gamma_1 = \gamma_0$$
$$J(x, k) \gamma_2 = \gamma_1$$

These are equivalent to the condition

$$J(x,k) \gamma_0 = 0, W^T \gamma_0 = 0 \text{ and } W^T \gamma_1 = 0.$$
 (7.22)

From Section 7.2.1 it is known that the first condition is equivalent to  $\Gamma(x_0) \pi_0 = 0$  (in this case  $\gamma_0 = \text{diag}(x_0) M_{\mathcal{N}_4} \pi_0$ ). For the remaining conditions recall the special form of the Jacobian

$$J(\lambda, x_0) = J(\lambda) \operatorname{diag}\left(\frac{1}{x_0}\right) \quad \text{with } J(\lambda) = N \operatorname{diag}\left(E\lambda\right) Y^{(L)^T}$$

Using Corollary 6 one obtains  $x_0 = \frac{S \rho_0}{M_{N_4} \pi_0}$  and  $\gamma_0 = S \rho_0$ . Thus a solution  $\gamma_1$  to

$$J(\lambda) \operatorname{diag}\left(\frac{1}{x_0}\right) \gamma_1 = S \rho_0 \tag{7.23}$$

with  $W^T \gamma_1 = 0$  and  $\lambda > 0$  is required. By definition of  $J(\lambda)$ 

$$[S]^{\perp} \subseteq [J(\lambda)]^{\perp},$$

thus (7.23) has a solution. Using  $v := \operatorname{diag}\left(\frac{1}{x_0}\right) \gamma_1$  one obtains the equations

$$J(\lambda) v = S \rho_0 \tag{7.24a}$$

$$W^T \operatorname{diag}\left(x_0\right) v = 0 \tag{7.24b}$$

Using the transformation matrix

one obtains for (7.24a)

$$\begin{bmatrix} I_6 & \tilde{K} \\ 0_{3\times 6} & 0_{3\times 3} \end{bmatrix} v = U S \rho_0$$

where

$$\ker\left(J\left(\lambda\right)\right) = \begin{bmatrix} -\tilde{K} \\ I_3 \end{bmatrix} = M_{\mathcal{N}_4} \pi_1$$

Thus one obtains

$$\gamma_1 = \text{diag}(x_0) [U(\lambda) S \rho_0 + M_{N_4} \pi_1].$$
 (7.25)

Note that  $U(\lambda)$  is invertible for  $\lambda_i > 0$ , as det  $(U(\lambda)) = \frac{1}{\lambda_5(\lambda_1 + \lambda_5)(\lambda_4 + \lambda_5)\lambda_6^2(\lambda_2 + \lambda_6)^2(\lambda_3 + \lambda_6)^2}$ . Thus multiplication with  $U(\lambda)$  is justified. Condition (7.24b) now becomes

$$W^T \operatorname{diag}(x_0) U(\lambda) S \rho_0 + W^T \operatorname{diag}(x_0) M_{\mathcal{N}_4} \pi = 0$$

that is one has to find  $\rho_0 \in I\!\!R^6$  and  $\pi_1 \in I\!\!R^3$  such that

$$Q(x_0, \lambda) \rho_0 + \Gamma(x_0) \pi_1 = 0, \qquad (7.26)$$

where  $\Gamma(x_0)$  as in (7.14) and

$$Q(x_0, \lambda) := W^T \operatorname{diag}(x_0) U(\lambda) S.$$
(7.27)

Recall that  $x_0$  depends on  $\pi_0 \in \mathbb{R}^3$ ,  $\rho_0 \in \mathbb{R}^6$  via (7.16a). Equation (7.26) has the following form

$$\begin{bmatrix} q_1^T \\ 0 \\ q_3^T \end{bmatrix} \rho_0 + \begin{bmatrix} g_1^T \\ g_2^T \\ g_3^T \end{bmatrix} \pi_1 = 0,$$
(7.28)

where  $q_i^T = (q_{i,1}, \ldots, q_{i,6})$  and  $g_i^T = (g_{i_1}, g_{i_2}, g_{i_3})$  are vectors of rational functions of  $\rho_{0_i}$ , i = 1, 2, 3,  $\pi_{0_i}$ ,  $i = 1, \ldots, 6$  and  $\lambda_i$ ,  $i = 1, \ldots, 6$ . Solve in a first step  $g_2^T \pi_1 = 0$  to obtain a solution for  $\pi_1$ :

$$\pi_1 = \begin{bmatrix} 1 & 0 \\ 0 & 1 \\ c_1 & c_2 \end{bmatrix} \begin{pmatrix} \pi_{1_1} \\ \pi_{1_2} \end{pmatrix},$$

with  $c_{1,2}$  rational functions of  $\rho_{0_i}$ , i = 1, 2, 3 and  $\pi_{0_i}$ ,  $i = 1, \ldots, 6$  and  $\pi_{1_{1,2}} \in \mathbb{R}$ . The resulting system is

$$\begin{bmatrix} q_1^{T} \\ 0 \\ q_3^{T} \end{bmatrix} \rho_0 + \begin{bmatrix} g_1^{T} \\ 0 \\ \bar{g}_3^{T} \end{bmatrix} \begin{pmatrix} \pi_{1_1} \\ \pi_{1_2} \end{pmatrix} = 0,$$

$$\begin{bmatrix} \bar{g}_1^{T} \\ 0 \\ \bar{g}_3^{T} \end{bmatrix} := \begin{bmatrix} g_1^{T} \\ g_2^{T} \\ g_3^{T} \end{bmatrix} \begin{bmatrix} 1 & 0 \\ 0 & 1 \\ c_1 & c_2 \end{bmatrix} \begin{pmatrix} \pi_{1_1} \\ \pi_{1_2} \end{pmatrix}.$$

That is

with

$$\begin{bmatrix} \tilde{g}_1^T & q_1^T \\ \tilde{g}_3^T & q_3^T \end{bmatrix} \begin{pmatrix} \pi_{1_1} \\ \pi_{1_2} \\ \rho_0 \end{pmatrix} = 0.$$

Multiplication with an invertible matrix, yields

$$\begin{bmatrix} 1 & g_0 & \tilde{q}_1^T \\ 0 & 0 & \tilde{q}_3^T \end{bmatrix} \begin{pmatrix} \pi_{1_1} \\ \pi_{1_2} \\ \rho_0 \end{pmatrix} = 0.$$

Thus  $\tilde{q}_3^T \rho_0 = 0$  must hold. Note that  $\tilde{q}_3^T \rho_0 = 0$  is a lengthy rational functional of the 15 unknowns  $\rho_{0_i}$ ,  $i = 1, \ldots, 6$ , the  $\pi_{0_i}$ ,  $i = 1, \ldots, 3$  and the  $\lambda_i$ ,  $i = 1, \ldots, 6$  that fills several rows. If one is interested in positive  $x_0$ , the analysis described above is required for each  $\delta$  where  $\mathcal{M}_{\mathcal{N}_4} \cap \mathbb{R}^9_{\delta} \neq \emptyset$  and  $\mathcal{S}_{\mathcal{N}_4} \cap \mathbb{R}^9_{\delta} \neq \emptyset$ : in this case  $x_0^{\delta} = \frac{\mathcal{E}^{\delta}_{\mathcal{N}_4} \rho_0}{\mathcal{E}^{\delta}_{\mathcal{N}_4} \pi_0}$  from (7.17) and  $\gamma_0^{\delta} = \mathcal{E}^{\delta}_{\mathcal{S}} \rho_0$  have to be used. Note that in this case one has to look for **positive**  $\pi_0$  and  $\rho$  and **arbitrary**  $\pi_{1_{1,2}}$ .

To demonstrate the usefulness of this approach, consider  $\pi_{0_i} = 1, i = 1, \ldots, 3, \rho_{0_i} = 1, i = 2, \ldots 6$  and  $\lambda_i = 1, i = 1, \ldots, 5$ . The resulting rational function in  $\rho_{0_1}$  and  $\lambda_6$  is given by:

$$\rho_{0_1} = -\frac{{}^{1908+2937\lambda_6+977\lambda_6^2+40\,\rho_{0_1}^2\,(24+36\,\lambda_6+11\,\lambda_6^2)+8\,\rho_{0_1}\,(417+686\,\lambda_6+231\,\lambda_6^2)}}{{}^{4\lambda_6\,(72+57\,\lambda_6+10\,\rho_{0_1}^2\,(24+3\lambda_6)+8\,\rho_{0_1}\,(23+15\,\lambda_6))}}$$
(7.29a)

Solving for  $\lambda_6$  as a function of  $\rho_{0_1}$  one obtains (as one root):

$$\lambda_{6} = \frac{-^{2937 - 5776} \rho_{0_{1}} - 2176 \rho_{0_{1}}^{2} - 160 \rho_{0_{1}}^{3} - \sqrt{(1169505 + 5047104} \rho_{0_{1}} + 7668736 \rho_{0_{1}}^{2} + 4912832 \rho_{0_{1}}^{3} + 1449216 \rho_{0_{1}}^{4} + 235520 \rho_{0_{1}}^{5} + 25600 \rho_{0_{1}}^{6})}{2 (977 + 2076 \rho_{0_{1}} + 920 \rho_{0_{1}}^{2} + 120 \rho_{0_{1}}^{3})}$$

$$(7.29b)$$

For  $\rho_{0_1} \in (-5, 5)$  equation (7.29b) is plotted in Fig. 7.3. For example, in the interval (-0.721918, -0.634033) one obtains positive values of  $\lambda_6$ . Choosing  $\rho_{0_1} = -0.7$  yields  $\lambda_6 = 0.363018$  and

$$x_0 = \left(2.3, -0.7, \frac{1}{2}, \frac{1}{2}, 0.3, \frac{1}{5}, \frac{1}{4}, 2, \frac{1}{2}\right)^T$$
(7.30a)

and thus

$$k_0 = (-1.24224, 2, 2, -3.89434, 3.33333, 1.21006, 27.2604, 0.5, 0.181509, 16, 2, 2)^T.$$
(7.30b)



Figure 7.3: Plot of (7.29b) for  $\rho_{1_1} \in (-5, 5)$ 

Using this data for a numerical continuation with respect to the conserved moieties, one finds that at  $(x_0, k_0)$  indeed a Takens-Bogdanov bifurcation occurs (that is sufficient conditions for a Takens-Bogdanov bifurcations are satisfied as well; see Figure 7.4, where  $(x_0, k_0)$  are used as initial point of the continuation. The numerical continuation software Matcout finds a Takens-Bogdanov bifurcation after one step.). Note that the bifurcation diagram depicted in Fig. 7.4 is quite involved: in the  $c_1-c_3$  plane



Figure 7.4: Continuation of  $(x_0, k_0)$  as given in (7.30a) and (7.30b) with respect  $c_1$  (total concentration of  $E_1$ ) and  $c_3$  (total concentration of A)

the Takens-Bogdanov bifurcation at (0.1,6.3) is followed by another Takens-Bogdanov bifurcation at (0.098678, 6.2938) and a Generalized-Hopf bifurcation at (0.09455, 6.2859). To gather further numerical evidence for the Takens-Bogdanov bifurcation an initial point in the parameter region where sustained oscillations occur was chosen:

Figure 7.5 shows the numerical integration of the ODES  $x_0$ ,  $k_0$  as given above and  $c_1 = 0.099603388$ . Fig. 7.5 suggest a limit cycle with period T = 0.4. Using this cycle as an initial point for a numerical continuation of limit cycles in Matcont, the bifurcation diagram depicted in Fig. 7.6 was obtained. Here two bifurcations of limit cycles occur: a limit point of cycles (LPC) or fold bifurcation of cycles (see e.g. [43, p. 149]) at  $(c_1, c_3) = (0.099225, 6.3101)$  and at  $(c_1, c_3) = (0.09852, 6.3099)$ . The second fold bifurcation of cycles is immediately followed by a neutral saddle (of limit cycles labeled NS) at  $(c_1, c_3) = (0.098457, 6.3099)$ .

Note that for obtaining Fig. 7.6 the continuation of limit cycles was performed using  $c_1$  and  $c_3$  as bifurcation parameters.

Remark 15. For the numerical analysis used to obtain Fig. 7.5 and Fig. 7.6 each ODE was scaled by

a factor s = 100. This corresponds to a time scale  $t := s \tau$ . Then  $x(t) = x(s \tau)$  and  $x' := \frac{d}{d\tau} x(s \tau) = \frac{d}{d\tau} x(s \tau) \frac{d}{d\tau} x(s \tau) \frac{d}{d\tau} x(s \tau) = s \dot{x}(t) = s \dot{y} I_a v(k, x(t)).$ 

**Remark 16.** Equation (7.29a) has been obtained using generators of  $\mathcal{M}_{\mathcal{N}_4} \cap \mathbb{R}^9_{\delta_1}$  and  $\mathcal{S}_{\mathcal{N}_4} \cap \mathbb{R}^9_{\delta_1}$  for  $\delta_1$  as in (4.16). Obtaining negative values for  $\rho_{0_1}$  is justified in this case, as generators of a cone are a basis for the linear subspace that contains the cone, in this case  $\mathcal{M}_{\mathcal{N}_4}$ . The idea was to try to find solutions to (7.29a) with as many positive components as possible.



Figure 7.5: Numerical integration starting at  $x^1$  using  $k_0$ 



Figure 7.6: Continuation of limit cycles starting with the limit cycle shown in Fig. 7.5

# Chapter 8

# Conclusions

In this thesis multistationarity in (bio)chemical reaction networks with mass action kinetics is analysed. As variables typically describe concentrations of chemical species, only positive steady state solutions are of interest. In general one therefore has to look for positive solutions to a system of polynomial equations with unknown coefficients. In its most general form, these equations are given by (see Chapter 2):

$$Y I_a v(k, x) = 0$$
$$W^T x = c.$$

From these it is possible to derive a system of nonlinear equations with known coefficients:

$$Y^{(L)T} \mu = \ln \frac{E\nu}{E\lambda}.$$

Vectors  $\mu$  and  $\nu > 0, \lambda > 0$ , that are solutions to these equations, define, in general, a parameterization of pairs of steady states a, b with  $Y I_a v(k, a) = Y I_a v(k, b) = 0$  (Lemma 1, Chapter 3). That is, one has established an infinite set of positive solutions to the polynomial equations  $Y I_a v(k, x) = 0$ . But, in general, these solutions do not satisfy the conservation relations  $W^T a = W^T b$ . Once solutions to find a vector  $v \in S$  with sign  $(v) = \text{sign}(\mu)$  (Lemma 3, Theorem 3) in order to find solutions with  $W^T a = W^T b$  (Lemma 3, Theorem 3).

Note that in special cases Theorem 3 contains in fact necessary and sufficient conditions for multistationarity in a given (bio)chemical reaction network. This is the case, if vectors  $\nu$ ,  $\lambda \in \mathbb{R}^n_{>0}$  can be determined independently of vector  $\mu$  (as is, for example the case for the networks discussed in Chapter 4 and 6). In this case one obtains an explicit representation of  $\mu: \mu \in \mathcal{M}$ , a linear subspace of  $\mathbb{R}^n$ . Thus one can test all orthants  $\delta$  of  $\mathbb{R}^n$  for  $S \cap \mathbb{R}^n_{\delta} \neq \emptyset$  and  $\mathcal{M} \cap \mathbb{R}^n_{\delta} \neq \emptyset$ . If a  $\delta$  can be established with  $S \cap \mathbb{R}^n_{\delta} \neq \emptyset$  and  $\mathcal{M} \cap \mathbb{R}^n_{\delta} \neq \emptyset$ , then multistationarity is established. If no  $\delta$  exists, then multistationarity is excluded by Theorem 3. However, if vectors  $\nu > 0$  and  $\lambda > 0$  cannot be determined independently of vector  $\mu$  (as, for example, for the networks discussed in Section 5), then Theorem 3 gives only sufficient conditions for multistationarity: if, for a given  $\mu$ , one can establish a  $v \in S$  with  $\operatorname{sign}(\mu) = \operatorname{sign}(v)$ , then multistationarity is established, if no such  $v \in S$  can be established, there is no guarantee that there does not exist a different  $\mu$  that satisfies this condition.

The usefulness of the methodology developed in Chapter 3 has been demonstrated by applying it to (bio)chemical reaction networks from signal transduction in Chapter 4 and cell cycle control in Chapter 5. Of particular interest in a systems biology oriented context is the potential application for model discrimination (as discussed in Chapter 4.5 and 5.1) and as an explanation for robustness (as discussed in Chapter 6). Note that both is possible, if vectors  $\nu > 0$  and  $\lambda > 0$  can be determined independently of vector  $\mu$ . In this case it is furthermore possible to determine potential bifurcation points analytically (as discussed in Chapter 7). Thus, this thesis closes with one question for further research: which structural properties of a (bio)chemical reaction network allow to determine vectors  $\nu$ ,  $\lambda \in \mathbb{R}^p_{>0}$  can be independently of vector  $\mu$ ?

## Appendix A

# Models for the activation of an MAPK(K)

- Distributive mechanism for phosphorylation and dephos-A.1 phorylation
- A.1.1 Species and complexes of network  $N_4$

Species	$x_i$	Complex	$y_i$
A	$x_1$	$A + E_1$	$y_1$
$E_1$	$x_2$	$A E_1$	$y_2$
$A E_1$	$x_3$	$A_p + E_1$	$y_3$
$A_p$	$x_4$	$A_p E_1$	$y_4$
$A_p E_1$	$x_5$	$A_{pp} + E_1$	$y_5$
$A_{pp}$	$x_6$	$A_{pp} + E_2$	$y_6$
$E_2$	$x_7$	$A_{pp} E_2$	$y_7$
$A_{pp} E_2$	$x_8$	$A_p + E_2$	$y_8$
$A_p E_2$	$x_9$	$A_p E_2$	$y_9$
		$A + E_2$	$y_{10}$

#### A.1.2 Ordinary differential equations

$\dot{x}_1 = -k_1 x_1 x_2 + k_2 x_3 + k_{12} x_9 \tag{A}$	4.1a	a)
---	------	----

$$\dot{x}_2 = -k_1 x_1 x_2 + (k_2 + k_3) x_3 - k_4 x_2 x_4 + (k_5 + k_6) x_5$$
(A.1b)

$$\begin{aligned} \dot{x}_2 &= k_1 x_1 x_2 + (k_2 + k_3) x_3 \\ \dot{x}_3 &= k_1 x_1 x_2 + (-k_2 - k_3) x_3 \\ \dot{x}_4 &= k_3 x_3 - k_4 x_2 x_4 + k_5 x_5 + k_9 x_8 - k_{10} x_4 x_7 + k_{11} x_9 \\ \dot{x}_7 &= k_4 x_9 x_4 + (-k_7 - k_8) x_7 \end{aligned}$$

$$\dot{x}_4 = k_3 x_3 - k_4 x_2 x_4 + k_5 x_5 + k_9 x_8 - k_{10} x_4 x_7 + k_{11} x_9 \tag{A.1d}$$

$$x_5 = k_4 x_2 x_4 + (-k_5 - k_6) x_5 \tag{A.1e}$$

$$\dot{x}_6 = k_6 x_5 - k_7 x_6 x_7 + k_8 x_8 \tag{A.1f}$$

$$\dot{x}_7 = -k_7 x_6 x_7 + (k_8 + k_9) x_8 - k_{10} x_4 x_7 + (k_{11} + k_{12}) x_9 \tag{A.1g}$$

$$\dot{x}_8 = k_7 \, x_6 \, x_7 + (-k_8 - k_9) \, x_8 \tag{A.1h}$$

$$\dot{x}_9 = k_{10} \, x_4 \, x_7 + (-k_{11} - k_{12}) \, x_9 \tag{A.1i}$$

 $x_2 + x_3 + x_5 = c_1$ (A.2a)

(A.1c)

- $x_7 + x_8 + x_9 = c_2$ (A.2b)
- $x_1 + x_3 + x_4 + x_5 + x_6 + x_8 + x_9 = c_3$ (A.2c)

#### A.1.3 Structural data

The ordinary differential equations (A.1a) - (A.1i) can be written as

$$\dot{x} = Y I_a \operatorname{diag}(k) \phi(x) \tag{A.3a}$$

using matrices

and the monomial vector

$$\phi(x) = (x_1 x_2, x_3, x_3, x_2 x_4, x_5, x_5, x_6 x_7, x_8, x_8, x_4 x_7, x_9, x_9)'$$
(A.3d)

A representation  $\phi\left(x\right)=\left(x^{y_{1}^{(L)}},\ldots,x^{y_{r}^{(L)}}\right)'$  can be obtained using

The conservation relations (A.2a) - (A.2c) can be written as

$$W^{T}x = c$$
 (A.4a)

using the matrix

and the vector  $c = (c_1, c_2, c_3)^T$ .

#### A.2 A processive mechanism for phosphorylation and a distributive mechanism for dephosphorylation

A.2.1 Species and complexes of network  $N_5$ 

ſ	Species	$x_i$	Complex	$y_i$
ſ	A	$x_1$	$A + E_1$	$y_1$
	$E_1$	$x_2$	$A E_1$	$y_2$
	$A E_1$	$x_3$	$A_p E_1$	$y_3$
	$A_p E_1$	$x_4$	$A_{pp} + E_1$	$y_4$
	$A_{pp}$	$x_5$	$A_{pp} + E_2$	$y_5$
	$E_2$	$x_6$	$A_{pp} E_2$	$y_6$
	$A_{pp} E_2$	$x_7$	$A_p + E_2$	$y_7$
	$A_p$	$x_8$	$A_p E_2$	$y_8$
l	$A_p E_2$	$x_9$	$A + E_2$	$y_9$

#### A.2.2 Ordinary differential equations

$\dot{x}_1 = -k_1 x_1 x_2 + k_2 x_3 + k_{11} x_9 \tag{A}$	5a	a)	)
---	----	----	---

$$\dot{x}_2 = -k_1 x_1 x_2 + k_2 x_3 + k_5 x_4 \tag{A.5b}$$

$$\dot{x}_3 = k_1 x_1 x_2 + (-k_2 - k_3) x_3 + k_4 x_4 \tag{A.5c}$$

$$\dot{x}_4 = k_3 x_3 + (-k_4 - k_5) x_4$$
 (A.5d)  
 $\dot{x}_5 = k_5 x_5 - k_5 x_5 + k_5 x_5$ 

$$x_5 = k_5 x_4 - k_6 x_5 x_6 + k_7 x_7$$
(A.5e)

$$\begin{aligned} x_6 &= -k_6 \, x_5 \, x_6 + (k_7 + k_8) \, x_7 - k_9 \, x_6 \, x_8 + (k_{10} + k_{11}) \, x_9 \end{aligned} \tag{A.51} \\ \dot{x}_7 &= k_6 \, x_5 \, x_6 + (-k_7 - k_8) \, x_7 \end{aligned}$$

$$\dot{x}_9 = k_9 x_6 x_8 + (-k_{10} - k_{11}) x_9$$
 (A.5i)

 $x_2 + x_3 + x_4 = c_1 \tag{A.6a}$ 

$$x_6 + x_7 + x_9 = c_2 \tag{A.6b}$$

$$x_1 + x_3 + x_4 + x_5 + x_7 + x_8 + x_9 = c_3 \tag{A.6c}$$

#### A.2.3 Structural data

The ordinary differential equations (A.9a) - (A.9h) can be written as

$$\dot{x} = Y I_a \operatorname{diag}(k) \phi(x) \tag{A.7a}$$

using matrices

and the monomial vector

$$\phi(x) = (x_1 x_2, x_3, x_3, x_4, x_4, x_5 x_6, x_7, x_7, x_6 x_8, x_9, x_9)^{\cdot}$$
(A.7d)

A representation  $\phi\left(x\right) = \left(x^{y_{1}^{(L)}}, \dots, x^{y_{r}^{(L)}}\right)'$  can be obtained using

The conservation relations (A.10a) - (A.10c) can be written as

$$W^T x = c \tag{A.8a}$$

using the matrix

and the vector  $c = (c_1, c_2, c_3)^T$ .

#### A.3Processive mechanism for phosphorylation and dephosphorylation

A.3.1 Species and complexes of network  $\mathcal{N}_6$ 

Species	$x_i$	Complex	$y_i$
A	$x_1$	$A + E_1$	$y_1$
$E_1$	$x_2$	$A E_1$	$y_2$
$A E_1$	$x_3$	$A_p E_1$	$y_3$
$A_p E_1$	$x_4$	$A_{pp} + E_1$	$y_4$
$A_{pp}$	$x_5$	$A_{pp} + E_2$	$y_5$
$E_2$	$x_6$	$A_{pp} E_2$	$y_6$
$A_{pp} E_2$	$x_7$	$A_p E_2$	$y_7$
$A_p E_2$	$x_8$	$A + E_2$	$y_8$

#### A.3.2 Ordinary differential equations

$$\dot{x}_1 = -k_1 x_1 x_2 + k_2 x_3 + k_{10} x_8$$
(A.9a)  

$$\dot{x}_2 = -k_1 x_1 x_2 + k_2 x_3 + k_5 x_4$$
(A.9b)

$$\dot{x}_2 = k_1 x_1 x_2 + k_2 x_3 + k_3 x_4$$

$$\dot{x}_3 = k_1 x_1 x_2 - (k_2 + k_3) x_3 + k_4 x_4$$
(A.9c)

$$\dot{x}_4 = k_3 \, x_3 - (k_4 + k_5) \, x_4 \tag{A.9d}$$

$$\dot{x}_5 = k_5 \, x_4 - k_6 \, x_5 \, x_6 + k_7 \, x_7 \tag{A.9e}$$

$$\dot{x}_6 = -k_6 \, x_5 \, x_6 + k_7 \, x_7 + k_{10} \, x_8 \tag{A.9f}$$

$$\dot{x}_7 = k_6 x_5 x_6 - (k_7 + k_8) x_7 + k_9 x_8$$

$$\dot{x}_8 = k_8 x_7 - (k_9 + k_{10}) x_8$$
(A.9g)
(A.9h)

$$x_7 - (k_9 + k_{10}) x_8 \tag{A.9h}$$

$$x_2 + x_3 + x_4 = c_1 \tag{A.10a}$$

$$x_6 + x_7 + x_8 = c_2 \tag{A.10b}$$

$$x_1 + x_3 + x_4 + x_5 + x_7 + x_8 = c_3 \tag{A.10c}$$

#### A.3.3 Structural data

The ordinary differential equations (A.9a) - (A.9h) can be written as

$$\dot{x} = Y I_a \operatorname{diag}(k) \phi(x) \tag{A.11a}$$

using matrices

and the monomial vector

$$\phi(x) = (x_1 x_2, x_3, x_3, x_4, x_4, x_5 x_6, x_7, x_7, x_8, x_8)'$$
(A.11d)

A representation  $\phi(x) = \left(x^{y_1^{(L)}}, \dots, x^{y_r^{(L)}}\right)'$  can be obtained using

The conservation relations (A.10a) - (A.10c) can be written as

$$W^T x = c$$
 (A.12a)

using the matrix

$$W^{T} = \begin{bmatrix} 0 & 1 & 1 & 1 & 0 & 0 & 0 & 0 \\ 0 & 0 & 0 & 0 & 0 & 1 & 1 & 1 \\ 1 & 0 & 1 & 1 & 1 & 0 & 1 & 1 \end{bmatrix}$$
(A.12b)

and the vector  $c = (c_1, c_2, c_3)^T$ .

#### A.4 Open systems

#### A.4.1 Network $N_4 + N_7$

The network is given by

$$\mathbf{A} + \mathbf{E}_{1} \xrightarrow[\mathbf{k}_{3}]{\mathbf{k}_{1}} \mathbf{A} \mathbf{E}_{1} \xrightarrow{\mathbf{k}_{3}} \mathbf{A}_{p} + \mathbf{E}_{1} \xrightarrow[\mathbf{k}_{3}]{\mathbf{k}_{4}} \mathbf{A}_{p} \mathbf{E}_{1} \xrightarrow{\mathbf{k}_{6}} \mathbf{A}_{pp} + \mathbf{E}_{1}$$
$$\mathbf{A}_{pp} + \mathbf{E}_{2} \xrightarrow[\mathbf{k}_{7}]{\mathbf{k}_{5}} \mathbf{A}_{pp} \mathbf{E}_{2} \xrightarrow{\mathbf{k}_{9}} \mathbf{A}_{p} + \mathbf{E}_{2} \xrightarrow{\mathbf{k}_{10}} \mathbf{A}_{p} \mathbf{E}_{2} \xrightarrow{\mathbf{k}_{12}} \mathbf{A} + \mathbf{E}_{2}$$
$$(\mathcal{N}_{15})$$
$$\mathbf{E}_{1} \frac{\mathbf{k}_{13}}{\mathbf{k}_{14}} \mathbf{0}$$

For network  $\mathcal{N}_{15}$  the equation

$$Y^{(L)^T} \mu = \ln \frac{E \nu}{E \lambda}$$

reads

$$\mu_1 + \mu_2 = \ln \frac{\nu_1 + \nu_6}{\lambda_1 + \lambda_6} \tag{A.13}$$

$$\mu_3 = \ln \frac{\nu_1}{\lambda_1} \tag{A.14}$$

$$\mu_3 = \ln \frac{\nu_6}{\lambda_6} \tag{A.15}$$

$$\mu_2 + \mu_4 = \ln \frac{\nu_2 + \nu_7}{\lambda_2 + \lambda_7} \tag{A.16}$$

$$\mu_5 = \ln \frac{\nu_2}{\lambda_2} \tag{A.17}$$

$$\nu_7$$

$$\mu_5 = \ln \frac{1}{\lambda_7} \tag{A.18}$$

$$\mu_6 + \mu_7 = \ln \frac{\nu_3 + \nu_7}{\lambda_3 + \lambda_7} \tag{A.19}$$

$$\mu_8 = \ln \frac{\nu_3}{\lambda_3} \tag{A.20}$$

$$\mu_8 = \ln \frac{\nu_7}{\lambda_7} \tag{A.21}$$

$$\mu_8 = \ln \frac{1}{\lambda_7} \tag{A.21}$$

$$\mu_4 + \mu_7 = \ln \frac{\nu_4 + \nu_6}{1 + \lambda_7} \tag{A.22}$$

$$\mu_9 = \ln \frac{\nu_4}{\lambda_4} \tag{A.23}$$

$$\mu_9 = \ln \frac{\nu_6}{\lambda_6} \tag{A.24}$$

$$0 = \ln \frac{\nu_5}{\lambda_5} \tag{A.25}$$

$$\mu_2 = \ln \frac{\nu_5}{\lambda_5}.\tag{A.26}$$

These equations are solvable, if and only if

$$\ln \frac{\nu_1}{\lambda_1} = \ln \frac{\nu_6}{\lambda_6} \qquad \ln \frac{\nu_2}{\lambda_2} = \ln \frac{\nu_7}{\lambda_7} \qquad \ln \frac{\nu_3}{\lambda_3} = \ln \frac{\nu_7}{\lambda_7} \qquad \ln \frac{\nu_4}{\lambda_4} = \ln \frac{\nu_6}{\lambda_6} \qquad \nu_5 = \lambda_5.$$

That is, if

$$\nu = \left(\lambda_1 \frac{\nu_6}{\lambda_6}, \lambda_2 \frac{\nu_7}{\lambda_7}, \lambda_3 \frac{\nu_7}{\lambda_7}, \lambda_4 \frac{\nu_6}{\lambda_6}, \lambda_5, \nu_6, \nu_7\right)^T,$$

with  $\lambda \in \mathbb{R}^{7}_{>0}$  and  $\nu_{6}, \nu_{7} > 0$  free. Note that it is possible to determine  $\nu$  and  $\lambda \in \mathbb{R}^{7}_{>0}$  independent of  $\mu \in \mathbb{R}^{9}$ . Using  $\nu$  as above, one obtains

$$\ln \frac{\nu_1 + \nu_6}{\lambda_1 + \lambda_6} = \ln \frac{\nu_4 + \nu_4}{\lambda_4 + \lambda_6} = \ln \frac{\nu_6}{\lambda_6}$$

and

$$\ln \frac{\nu_2 + \nu_7}{\lambda_2 + \lambda_7} = \ln \frac{\nu_3 + \nu_7}{\lambda_3 + \lambda_7} = \ln \frac{\nu_7}{\lambda_7}$$

Using  $\kappa_1 := \ln \frac{\nu_6}{\lambda_6}$  and  $\kappa_2 := \ln \frac{\nu_7}{\lambda_7}$  in the right hand side of (A.13) – (A.26) and solving for  $\mu$  yields

$$\mu = \begin{bmatrix} 1 & 0 \\ 0 & 0 \\ 1 & 0 \\ 0 & 1 \\ 0 & 1 \\ -1 & 2 \\ 1 & -1 \\ 0 & 1 \\ 1 & 0 \end{bmatrix} \begin{pmatrix} \kappa_1 \\ \kappa_2 \end{pmatrix}$$

#### A.4.2 Network $N_4 + N_8$

The network is given by

For network  $\mathcal{N}_{16}$  the equation

$$Y^{(L)^T} \mu = \ln \frac{E \nu}{E \lambda}$$

reads

$$\begin{split} \mu_1 + \mu_2 &= \ln \frac{\nu_1 + \nu_6}{\lambda_1 + \lambda_6} \\ \mu_3 &= \ln \frac{\nu_1}{\lambda_1} \\ \mu_3 &= \ln \frac{\nu_6}{\lambda_6} \\ \mu_2 + \mu_4 &= \ln \frac{\nu_2 + \nu_7}{\lambda_2 + \lambda_7} \\ \mu_5 &= \ln \frac{\nu_2}{\lambda_2} \\ \mu_5 &= \ln \frac{\nu_7}{\lambda_7} \\ \mu_6 + \mu_7 &= \ln \frac{\nu_3 + \nu_7}{\lambda_3 + \lambda_7} \\ \mu_8 &= \ln \frac{\nu_3}{\lambda_3} \\ \mu_8 &= \ln \frac{\nu_7}{\lambda_7} \\ \mu_4 + \mu_7 &= \ln \frac{\nu_4 + \nu_6}{\lambda_4 + \lambda_6} \\ \mu_9 &= \ln \frac{\nu_4}{\lambda_6} \\ \mu_9 &= \ln \frac{\nu_6}{\lambda_6} \\ 0 &= \ln \frac{\nu_5}{\lambda_5} \\ \mu_7 &= \ln \frac{\nu_5}{\lambda_5} \\ \end{split}$$

These equations are solvable, if and only if

$$\ln \frac{\nu_1}{\lambda_1} = \ln \frac{\nu_6}{\lambda_6} \qquad \ln \frac{\nu_2}{\lambda_2} = \ln \frac{\nu_7}{\lambda_7} \qquad \ln \frac{\nu_3}{\lambda_3} = \ln \frac{\nu_7}{\lambda_7} \qquad \ln \frac{\nu_4}{\lambda_4} = \ln \frac{\nu_6}{\lambda_6} \qquad \nu_5 = \lambda_5$$

That is, if

$$\nu = \left(\lambda_1 \frac{\nu_6}{\lambda_6}, \lambda_2 \frac{\nu_7}{\lambda_7}, \lambda_3 \frac{\nu_7}{\lambda_7}, \lambda_4 \frac{\nu_6}{\lambda_6}, \lambda_5, \nu_6, \nu_7\right)^T,$$

with  $\lambda \in \mathbb{R}^{7}_{>0}$  and  $\nu_{6}, \nu_{7} > 0$  free. (Note that  $\nu$  as given above is equal to the vector  $\nu$  obtained for network  $\mathcal{N}_{15}$  in Section A.4.1). As network  $\mathcal{N}_{15}$  it is possible to determine  $\nu$  and  $\lambda \in \mathbb{R}^{7}_{>0}$  independent of  $\mu \in \mathbb{R}^{9}$ . Using  $\nu$  as above, one obtains

$$\ln \frac{\nu_1 + \nu_6}{\lambda_1 + \lambda_6} = \ln \frac{\nu_4 + \nu_6}{\lambda_4 + \lambda_6} = \ln \frac{\nu_6}{\lambda_6}$$

and

$$\ln \frac{\nu_2 + \nu_7}{\lambda_2 + \lambda_7} = \ln \frac{\nu_3 + \nu_7}{\lambda_3 + \lambda_7} = \ln \frac{\nu_7}{\lambda_7}.$$

Using  $\kappa_1 := \ln \frac{\nu_6}{\lambda_6}$  and  $\kappa_2 := \ln \frac{\nu_7}{\lambda_7}$  in the right hand side of (A.13) – (A.26) and solving for  $\mu$  yields

$$\mu = \begin{bmatrix} 2 & -1 \\ -1 & 1 \\ 1 & 0 \\ 0 & 1 \\ 0 & 1 \\ 0 & 0 \\ 0 & 1 \\ 1 & 0 \end{bmatrix} \begin{pmatrix} \kappa_1 \\ \kappa_2 \end{pmatrix}$$

#### A.4.3 Network $N_4 + N_9$

The network is given by

$$\begin{split} \mathbf{A} + \mathbf{E}_{1} \underbrace{\overset{\mathbf{k}_{1}}{\overleftarrow{\mathbf{k}_{2}}}}_{\mathbf{k}_{2}} \mathbf{A} \mathbf{E}_{1} \underbrace{\overset{\mathbf{k}_{3}}{\longrightarrow}} \mathbf{A}_{p} + \mathbf{E}_{1} \underbrace{\overset{\mathbf{k}_{4}}{\overleftarrow{\mathbf{k}_{5}}}}_{\mathbf{k}_{5}} \mathbf{A}_{p} \mathbf{E}_{1} \underbrace{\overset{\mathbf{k}_{6}}{\longrightarrow}} \mathbf{A}_{pp} + \mathbf{E}_{1} \\ \mathbf{A}_{pp} + \mathbf{E}_{2} \underbrace{\overset{\mathbf{k}_{7}}{\overleftarrow{\mathbf{k}_{8}}}}_{\mathbf{k}_{8}} \mathbf{A}_{pp} \mathbf{E}_{2} \underbrace{\overset{\mathbf{k}_{9}}{\longrightarrow}} \mathbf{A}_{p} + \mathbf{E}_{2} \underbrace{\overset{\mathbf{k}_{10}}{\overleftarrow{\mathbf{k}_{11}}}}_{\mathbf{k}_{11}} \mathbf{A}_{p} \mathbf{E}_{2} \underbrace{\overset{\mathbf{k}_{12}}{\longrightarrow}} \mathbf{A} + \mathbf{E}_{2} \\ \mathbf{E}_{1} \underbrace{\overset{\mathbf{k}_{13}}{\overleftarrow{\mathbf{k}_{14}}}}_{\mathbf{k}_{14}} \mathbf{0} \underbrace{\overset{\mathbf{k}_{15}}{\overleftarrow{\mathbf{k}_{16}}}}_{\mathbf{k}_{16}} \mathbf{E}_{2} \end{split}$$

For network  $\mathcal{N}_{17}$  the equation

$$Y^{(L)^{T}} \mu = \ln \frac{E \nu}{E \lambda}$$

reads

$$\begin{split} \mu_1 + \mu_2 &= \ln \frac{\nu_1 + \nu_7}{\lambda_1 + \lambda_7} \\ \mu_3 &= \ln \frac{\nu_1}{\lambda_1} \\ \mu_3 &= \ln \frac{\nu_7}{\lambda_7} \\ \mu_2 + \mu_4 &= \ln \frac{\nu_2 + \nu_8}{\lambda_2 + \lambda_8} \\ \mu_5 &= \ln \frac{\nu_8}{\lambda_2} \\ \mu_5 &= \ln \frac{\nu_8}{\lambda_3} \\ \mu_6 + \mu_7 &= \ln \frac{\nu_3 + \nu_8}{\lambda_3 + \lambda_8} \\ \mu_8 &= \ln \frac{\nu_8}{\lambda_3} \\ \mu_8 &= \ln \frac{\nu_8}{\lambda_3} \\ \mu_4 + \mu_7 &= \ln \frac{\nu_4 + \nu_7}{\lambda_4 + \lambda_7} \\ \mu_9 &= \ln \frac{\nu_7}{\lambda_7} \\ \mu_2 &= \ln \frac{\nu_5}{\lambda_5} \\ 0 &= \ln \frac{\nu_6}{\lambda_6} \\ \mu_7 &= \ln \frac{\nu_6}{\lambda_6} \\ \end{split}$$

These equations are solvable, if and only if

$$\ln \frac{\nu_1}{\lambda_1} = \ln \frac{\nu_7}{\lambda_7} \qquad \ln \frac{\nu_2}{\lambda_2} = \ln \frac{\nu_8}{\lambda_8} \qquad \ln \frac{\nu_3}{\lambda_3} = \ln \frac{\nu_8}{\lambda_8} \qquad \ln \frac{\nu_4}{\lambda_4} = \ln \frac{\nu_7}{\lambda_7} \qquad \nu_5 = \lambda_5 \qquad \nu_6 = \lambda_6.$$

Thus one obtains

$$\nu_1 = \lambda_1 \frac{\nu_7}{\lambda_7} \qquad \qquad \nu_2 = \lambda_2 \frac{\nu_8}{\lambda_8}$$
$$\nu_4 = \lambda_4 \frac{\nu_8}{\lambda_8}$$

Note that  $\mu_2 = 0$  and  $\mu_7 = 0$  and thus  $\mu_4 = \ln \frac{\nu_2 + \nu_8}{\lambda_2 + \lambda_8}$  and  $\mu_4 = \ln \frac{\nu_4 + \nu_7}{\lambda_4 + \lambda_7}$  follows. Thus one finally obtains  $\frac{\nu_7}{\lambda_7} = \frac{\nu_8}{\lambda_8}$  and therefore

$$\nu = \left(\lambda_1 \frac{\nu_8}{\lambda_8}, \lambda_2 \frac{\nu_8}{\lambda_8}, \lambda_3 \frac{\nu_8}{\lambda_8}, \lambda_4 \frac{\nu_8}{\lambda_8}, \lambda_5, \lambda_6, \lambda_7 \frac{\nu_8}{\lambda_8}, \nu_8\right)^T,$$

with  $\lambda \in \mathbb{R}^8_{>0}$ ,  $\nu_8 > 0$ , free. As network  $\mathcal{N}_{15}$  it is possible to determine  $\nu$  and  $\lambda \in \mathbb{R}^8_{>0}$  independent of  $\mu \in \mathbb{R}^9$ . Using  $\nu$  as above, one obtains

$$\ln \frac{\nu_1 + \nu_7}{\lambda_1 + \lambda_7} = \ln \frac{\nu_2 + \nu_8}{\lambda_2 + \lambda_8} = \ln \frac{\nu_3 + \nu_8}{\lambda_3 + \lambda_8} = \ln \frac{\nu_4 + \nu_7}{\lambda_4 + \lambda_7} = \ln \frac{\nu_8}{\lambda_8}$$

Using  $\kappa_1 := \ln \frac{\nu_8}{\lambda_8}$  and solving for  $\mu$  yields

$$\mu = \begin{bmatrix} 1, 0, 1, 1, 1, 1, 0, 1, 1 \end{bmatrix}^T \kappa_1$$

#### A.4.4 Network $N_5 + N_7$

The network is given by

$$\mathbf{A} + \mathbf{E}_{1} \underbrace{\stackrel{\mathbf{k}_{1}}{\underset{\mathbf{k}_{2}}{\longrightarrow}} \mathbf{A} \mathbf{E}_{1} \xrightarrow{\underset{\mathbf{k}_{3}}{\overset{\mathbf{k}_{3}}{\longrightarrow}} \mathbf{A}_{p} \mathbf{E}_{1} \xrightarrow{\underset{\mathbf{k}_{5}}{\overset{\mathbf{k}_{5}}{\longrightarrow}} \mathbf{A}_{pp} + \mathbf{E}_{1}}}_{\mathbf{A}_{pp} + \mathbf{E}_{2} \underbrace{\stackrel{\mathbf{k}_{6}}{\underset{\mathbf{k}_{7}}{\longrightarrow}} \mathbf{A}_{p} \mathbf{E}_{2} \xrightarrow{\underset{\mathbf{k}_{8}}{\overset{\mathbf{k}_{8}}{\longrightarrow}} \mathbf{A}_{p} + \mathbf{E}_{2} \underbrace{\stackrel{\mathbf{k}_{9}}{\underset{\mathbf{k}_{10}}{\overleftarrow{\mathbf{k}_{10}}}} \mathbf{A}_{p} \mathbf{E}_{2} \xrightarrow{\underset{\mathbf{k}_{11}}{\overset{\mathbf{k}_{12}}{\longrightarrow}} \mathbf{A} + \mathbf{E}_{2}}_{\mathbf{E}_{1} \underbrace{\stackrel{\mathbf{k}_{12}}{\underset{\mathbf{k}_{13}}{\overset{\mathbf{k}_{12}}{\overleftarrow{\mathbf{k}_{13}}}} \mathbf{0}}$$

For network  $\mathcal{N}_{18}$  the equation

$$Y^{(L)^{T}} \mu = \ln \frac{E \nu}{E \lambda}$$

reads

$$\begin{split} \mu_1 + \mu_2 &= \ln \frac{\nu_1 + \nu_6}{\lambda_1 + \lambda_6} \\ \mu_3 &= \ln \frac{\nu_1}{\lambda_1} \\ \mu_3 &= \ln \frac{\nu_2 + \nu_6}{\lambda_2 + \lambda_6} \\ \mu_4 &= \ln \frac{\nu_2}{\lambda_2} \\ \mu_4 &= \ln \frac{\nu_2}{\lambda_6} \\ \mu_5 + \mu_6 &= \ln \frac{\nu_3 + \nu_6}{\lambda_3 + \lambda_6} \\ \mu_7 &= \ln \frac{\nu_3}{\lambda_6} \\ \mu_7 &= \ln \frac{\nu_6}{\lambda_6} \\ \mu_6 + \mu_8 &= \ln \frac{\nu_4 + \nu_6}{\lambda_4 + \lambda_6} \\ \mu_9 &= \ln \frac{\nu_4}{\lambda_4} \\ \mu_9 &= \ln \frac{\nu_6}{\lambda_5} \\ \mu_2 &= \ln \frac{\nu_5}{\lambda_5} \\ \mu_2 &= \ln \frac{\nu_5}{\lambda_5} \\ \end{split}$$

These equations are solvable, if and only if

$$\ln \frac{\nu_1}{\lambda_1} = \ln \frac{\nu_2 + \nu_6}{\lambda_2 + \lambda_6} \qquad \ln \frac{\nu_2}{\lambda_2} = \ln \frac{\nu_6}{\lambda_6} \qquad \ln \frac{\nu_3}{\lambda_3} = \ln \frac{\nu_6}{\lambda_6} \qquad \ln \frac{\nu_4}{\lambda_4} = \ln \frac{\nu_6}{\lambda_6} \qquad \nu_5 = \lambda_5$$

Thus one obtains

$$\nu_2 = \lambda_2 \frac{\nu_6}{\lambda_6}, \qquad \qquad \nu_3 = \frac{\nu_6}{\lambda_6}, \qquad \qquad \nu_4 = \lambda_4 \frac{\nu_6}{\lambda_6}.$$

Note that  $\frac{\nu_2+\nu_6}{\lambda_2+\lambda_6} = \frac{\nu_6}{\lambda_6}$  and thus  $\nu_1 = \lambda_1 \frac{\nu_6}{\lambda_6}$ , as well. Therefore

$$\nu = \left(\lambda_1 \frac{\nu_6}{\lambda_6}, \lambda_2 \frac{\nu_6}{\lambda_6}, \frac{\nu_6}{\lambda_6}, \lambda_4 \frac{\nu_6}{\lambda_6}, \lambda_5, \nu_6\right)^T$$

with  $\lambda \in \mathbb{R}^6_{>0}$ ,  $\nu_6 > 0$ , free. As for network  $\mathcal{N}_{15}$  it is possible to determine  $\nu$  and  $\lambda \in \mathbb{R}^6_{>0}$  independent of  $\mu \in \mathbb{R}^9$ . Using  $\nu$  as above, one obtains

$$\frac{\nu_1 + \nu_6}{\lambda_1 + \lambda_6} = \frac{\nu_3 + \nu_6}{\lambda_3 + \lambda_6} = \ln \frac{\nu_4 + \nu_6}{\lambda_4 + \lambda_6} = \ln \frac{\nu_6}{\lambda_6}$$

Using  $\kappa_1 := \ln \frac{\nu_6}{\lambda_6}$  and solving for  $\mu$  yields

$$\mu = \begin{bmatrix} 0 & 1 \\ 0 & 0 \\ 0 & 1 \\ 1 & 0 \\ -1 & 1 \\ 0 & 1 \\ 1 & 0 \\ 0 & 1 \end{bmatrix} \begin{pmatrix} \mu_8 \\ \kappa_1 \end{pmatrix}.$$

#### A.4.5 Network $N_5 + N_8$

The network is given by

$$\mathbf{A} + \mathbf{E}_{1} \underbrace{\overset{\mathbf{k}_{1}}{\overleftarrow{\mathbf{k}_{2}}} \mathbf{A} \mathbf{E}_{1} \underbrace{\overset{\mathbf{k}_{3}}{\overleftarrow{\mathbf{k}_{4}}} \mathbf{A}_{p} \mathbf{E}_{1} \underbrace{\overset{\mathbf{k}_{5}}{\longrightarrow} \mathbf{A}_{pp} + \mathbf{E}_{1}}_{\mathbf{A}_{pp} + \mathbf{E}_{2} \underbrace{\overset{\mathbf{k}_{6}}{\overleftarrow{\mathbf{k}_{7}}} \mathbf{A}_{pp} \mathbf{E}_{2} \underbrace{\overset{\mathbf{k}_{8}}{\longrightarrow} \mathbf{A}_{p} + \mathbf{E}_{2} \underbrace{\overset{\mathbf{k}_{9}}{\overleftarrow{\mathbf{k}_{10}}} \mathbf{A}_{p} \mathbf{E}_{2} \underbrace{\overset{\mathbf{k}_{11}}{\longrightarrow} \mathbf{A} + \mathbf{E}_{2}}_{\mathbf{E}_{2} \underbrace{\overset{\mathbf{k}_{12}}{\overleftarrow{\mathbf{k}_{13}}} \mathbf{0}} \mathbf{E}_{2} \underbrace{\overset{\mathbf{k}_{12}}{\overleftarrow{\mathbf{k}_{13}}} \mathbf{0}$$

For network  $\mathcal{N}_{19}$  the equation

$$Y^{(L)^T} \mu = \ln \frac{E \nu}{E \lambda}$$

reads

$$\begin{split} \mu_1 + \mu_2 &= \ln \frac{\nu_1 + \nu_6}{\lambda_1 + \lambda_6} \\ \mu_3 &= \ln \frac{\nu_1}{\lambda_2} \\ \mu_3 &= \ln \frac{\nu_2 + \nu_6}{\lambda_2 + \lambda_6} \\ \mu_4 &= \ln \frac{\nu_2}{\lambda_2} \\ \mu_4 &= \ln \frac{\nu_3}{\lambda_2} \\ \mu_5 + \mu_6 &= \ln \frac{\nu_3 + \nu_6}{\lambda_3 + \lambda_6} \\ \mu_7 &= \ln \frac{\nu_3}{\lambda_6} \\ \mu_7 &= \ln \frac{\nu_4}{\lambda_6} \\ \mu_9 &= \ln \frac{\nu_4 + \nu_6}{\lambda_4 + \lambda_6} \\ \mu_9 &= \ln \frac{\nu_4}{\lambda_6} \\ \mu_9 &= \ln \frac{\nu_5}{\lambda_5} \\ \mu_6 &= \ln \frac{\nu_5}{\lambda_5} \\ \end{split}$$

These equations are solvable, if and only if

$$\ln \frac{\nu_1}{\lambda_1} = \ln \frac{\nu_2 + \nu_6}{\lambda_2 + \lambda_6} \qquad \ln \frac{\nu_2}{\lambda_2} = \ln \frac{\nu_6}{\lambda_6} \qquad \ln \frac{\nu_3}{\lambda_3} = \ln \frac{\nu_6}{\lambda_6} \qquad \ln \frac{\nu_4}{\lambda_4} = \ln \frac{\nu_6}{\lambda_6} \qquad \nu_5 = \lambda_5$$

Note, that these conditions are equivalent to those obtained for A.4.4. Thus one obtains the same solution  $\nu \in \mathbb{R}^{6}_{>0}$ , as in Section A.4.4:

$$\nu = \left(\lambda_1 \frac{\nu_6}{\lambda_6}, \lambda_2 \frac{\nu_6}{\lambda_6}, \frac{\nu_6}{\lambda_6}, \lambda_4 \frac{\nu_6}{\lambda_6}, \lambda_5, \nu_6\right)^T$$

with  $\lambda \in \mathbb{R}^6_{>0}$ ,  $\nu_6 > 0$ , free. As for network  $\mathcal{N}_{15}$  it is possible to determine  $\nu$  and  $\lambda \in \mathbb{R}^6_{>0}$  independent of  $\mu \in \mathbb{R}^9$ . Using  $\nu$  as above, one obtains

$$\frac{\nu_1 + \nu_6}{\lambda_1 + \lambda_6} = \frac{\nu_3 + \nu_6}{\lambda_3 + \lambda_6} = \ln \frac{\nu_4 + \nu_6}{\lambda_4 + \lambda_6} = \ln \frac{\nu_6}{\lambda_6}$$

Using  $\kappa_1 := \ln \frac{\nu_6}{\lambda_6}$  and solving for  $\mu$  yields

$$\mu = \begin{bmatrix} -1 & 1 \\ 1 & 0 \\ 0 & 1 \\ 0 & 1 \\ 0 & 1 \\ 0 & 0 \\ 0 & 1 \\ 0 & 1 \\ 0 & 1 \end{bmatrix} \begin{pmatrix} \mu_2 \\ \kappa_1 \end{pmatrix}.$$

#### A.4.6 Network $N_5 + N_9$

The network is given by

$$\mathbf{A} + \mathbf{E}_{1} \underbrace{\frac{\mathbf{k}_{1}}{\mathbf{k}_{2}}}_{\mathbf{k}_{2}} \mathbf{A} \mathbf{E}_{1} \underbrace{\frac{\mathbf{k}_{3}}{\mathbf{k}_{4}}}_{\mathbf{k}_{4}} \mathbf{A}_{p} \mathbf{E}_{1} \underbrace{\overset{\mathbf{k}_{5}}{\longrightarrow}}_{\mathbf{k}_{pp}} \mathbf{A}_{pp} + \mathbf{E}_{1}$$
$$\mathbf{A}_{pp} + \mathbf{E}_{2} \underbrace{\overset{\mathbf{k}_{6}}{\overleftarrow{\mathbf{k}_{7}}}}_{\mathbf{k}_{7}} \mathbf{A}_{pp} \mathbf{E}_{2} \underbrace{\overset{\mathbf{k}_{8}}{\longrightarrow}}_{\mathbf{k}_{p}} \mathbf{A}_{p} + \mathbf{E}_{2} \underbrace{\overset{\mathbf{k}_{9}}{\overleftarrow{\mathbf{k}_{10}}}}_{\mathbf{k}_{10}} \mathbf{A}_{p} \mathbf{E}_{2} \underbrace{\overset{\mathbf{k}_{11}}{\longrightarrow}}_{\mathbf{k}_{12}} \mathbf{A} + \mathbf{E}_{2}$$
$$(\mathcal{N}_{20})$$
$$\mathbf{E}_{1} \underbrace{\overset{\mathbf{k}_{12}}{\overleftarrow{\mathbf{k}_{13}}}}_{\mathbf{k}_{15}} \mathbf{E}_{2}$$

For network  $\mathcal{N}_{20}$  the equation

$$Y^{(L)T} \mu = \ln \frac{E\nu}{E\lambda}$$

reads

$$\begin{split} \mu_1 + \mu_2 &= \ln \frac{\nu_1 + \nu_7}{\lambda_1 + \lambda_7} \\ \mu_3 &= \ln \frac{\nu_1}{\lambda_1} \\ \mu_3 &= \ln \frac{\nu_2 + \nu_7}{\lambda_2 + \lambda_7} \\ \mu_4 &= \ln \frac{\nu_2}{\lambda_2} \\ \mu_4 &= \ln \frac{\nu_7}{\lambda_7} \\ \mu_5 + \mu_6 &= \ln \frac{\nu_3 + \nu_7}{\lambda_3 + \lambda_7} \\ \mu_7 &= \ln \frac{\nu_3}{\lambda_3} \\ \mu_7 &= \ln \frac{\nu_7}{\lambda_7} \\ \mu_6 + \mu_8 &= \ln \frac{\nu_4 + \nu_7}{\lambda_4 + \lambda_7} \\ \mu_9 &= \ln \frac{\nu_4}{\lambda_4} \\ \mu_9 &= \ln \frac{\nu_5}{\lambda_5} \\ 0 &= \ln \frac{\nu_5}{\lambda_5} \\ 0 &= \ln \frac{\nu_6}{\lambda_6} \\ \mu_6 &= \ln \frac{\nu_6}{\lambda_6} \\ \end{split}$$

These equations are solvable, if and only if

$$\ln\frac{\nu_1}{\lambda_1} = \ln\frac{\nu_2 + \nu_7}{\lambda_2 + \lambda_7} \qquad \ln\frac{\nu_2}{\lambda_2} = \ln\frac{\nu_7}{\lambda_7} \qquad \ln\frac{\nu_3}{\lambda_3} = \ln\frac{\nu_7}{\lambda_7} \qquad \ln\frac{\nu_4}{\lambda_4} = \ln\frac{\nu_7}{\lambda_7} \qquad \nu_5 = \lambda_5 \qquad \nu_6 = \lambda_6.$$

Thus one obtains

$$\nu_2 = \lambda_2 \frac{\nu_7}{\lambda_7}, \qquad \qquad \nu_3 = \frac{\nu_7}{\lambda_7}, \qquad \qquad \nu_4 = \lambda_4 \frac{\nu_7}{\lambda_7}.$$

Note that  $\frac{\nu_2 + \nu_7}{\lambda_2 + \lambda_7} = \frac{\nu_7}{\lambda_7}$  and thus  $\nu_1 = \lambda_1 \frac{\nu_7}{\lambda_7}$ , as well.

$$\nu = \left(\lambda_1 \frac{\nu_7}{\lambda_7}, \lambda_2 \frac{\nu_7}{\lambda_7}, \frac{\nu_7}{\lambda_7}, \lambda_4 \frac{\nu_7}{\lambda_7}, \lambda_5, \lambda_6, \nu_7\right)^T$$

with  $\lambda \in \mathbb{R}^{7}_{>0}$ ,  $\nu_{7} > 0$ , free. As for network  $\mathcal{N}_{15}$  it is possible to determine  $\nu$  and  $\lambda \in \mathbb{R}^{7}_{>0}$  independent of  $\mu \in \mathbb{R}^{9}$ . Using  $\nu$  as above, one obtains

$$\frac{\nu_1 + \nu_7}{\lambda_1 + \lambda_7} = \frac{\nu_3 + \nu_7}{\lambda_3 + \lambda_7} = \ln \frac{\nu_4 + \nu_7}{\lambda_4 + \lambda_7} = \ln \frac{\nu_7}{\lambda_7}$$

Using  $\kappa_1 := \ln \frac{\nu_7}{\lambda_7}$  and solving for  $\mu$  yields

$$\mu = \begin{bmatrix} 1, 0, 1, 1, 1, 0, 1, 1, 1 \end{bmatrix}^T \kappa_1.$$

#### A.4.7 Network $\mathcal{N}_6 + \mathcal{N}_7$

The network is given by

$$\mathbf{A} + \mathbf{E}_{1} \underbrace{\frac{\mathbf{k}_{1}}{\mathbf{k}_{2}}}_{\mathbf{k}_{2}} \mathbf{A} \mathbf{E}_{1} \underbrace{\frac{\mathbf{k}_{3}}{\mathbf{k}_{4}}}_{\mathbf{k}_{9}} \mathbf{A}_{p} \mathbf{E}_{1} \underbrace{\overset{\mathbf{k}_{5}}{\longrightarrow}}_{\mathbf{k}_{9}\mathbf{p}} \mathbf{A}_{p} \mathbf{E}_{1} \underbrace{\overset{\mathbf{k}_{5}}{\longrightarrow}}_{\mathbf{k}_{9}\mathbf{p}} \mathbf{A}_{p} \mathbf{E}_{2} \underbrace{\overset{\mathbf{k}_{1}}{\longrightarrow}}_{\mathbf{k}_{1}\mathbf{p}} \mathbf{A}_{p} \mathbf{E}_{2} \underbrace{\overset{\mathbf{k}_{1}}{\longrightarrow}}_{\mathbf{k}_{1}\mathbf{k}_{1}\mathbf{k}_{1}} \mathbf{A} + \mathbf{E}_{2}$$

$$\mathbf{E}_{1} \underbrace{\overset{\mathbf{k}_{13}}{\mathbf{k}_{14}}}_{\mathbf{k}_{14}\mathbf{k}_{1}\mathbf{k}_{1}} \mathbf{0}$$

For network  $\mathcal{N}_{21}$  the equation

$$Y^{(L)T} \mu = \ln \frac{E\nu}{E\lambda}$$

reads

$$\begin{split} \mu_1 + \mu_2 &= \ln \frac{\nu_1 + \nu_6}{\lambda_1 + \lambda_6} \\ \mu_3 &= \ln \frac{\nu_1}{\lambda_1} \\ \mu_3 &= \ln \frac{\nu_2 + \nu_6}{\lambda_2 + \lambda_6} \\ \mu_4 &= \ln \frac{\nu_2}{\lambda_2} \\ \mu_4 &= \ln \frac{\nu_6}{\lambda_6} \\ \mu_5 + \mu_6 &= \ln \frac{\nu_3 + \nu_6}{\lambda_3 + \lambda_6} \\ \mu_7 &= \ln \frac{\nu_3}{\lambda_3} \\ \mu_7 &= \ln \frac{\nu_4 + \nu_6}{\lambda_4 + \lambda_6} \\ \mu_8 &= \ln \frac{\nu_6}{\lambda_4} \\ \mu_8 &= \ln \frac{\nu_6}{\lambda_5} \\ \mu_2 &= \ln \frac{\nu_5}{\lambda_5} \\ \mu_2 &= \ln \frac{\nu_5}{\lambda_5} \end{split}$$

These equations are solvable, if and only if

$$\ln \frac{\nu_1}{\lambda_1} = \ln \frac{\nu_2 + \nu_6}{\lambda_2 + \lambda_6} \qquad \ln \frac{\nu_2}{\lambda_2} = \ln \frac{\nu_6}{\lambda_6} \qquad \ln \frac{\nu_3}{\lambda_3} = \ln \frac{\nu_4 + \nu_6}{\lambda_4 + \lambda_6} \qquad \ln \frac{\nu_4}{\lambda_4} = \ln \frac{\nu_6}{\lambda_6} \qquad \nu_5 = \lambda_5.$$

Thus one obtains

$$\nu_2 = \lambda_2 \frac{\nu_6}{\lambda_6}, \qquad \qquad \nu_4 = \lambda_4 \frac{\nu_6}{\lambda_6}$$

Note that therefore  $\frac{\nu_2 + \nu_6}{\lambda_2 + \lambda_6} = \frac{\nu_4 + \nu_6}{\lambda_4 + \lambda_6} = \frac{\nu_6}{\lambda_6}$  and thus  $\nu_1 = \lambda_1 \frac{\nu_6}{\lambda_6}$  and  $\nu_3 = \lambda_3 \frac{\nu_6}{\lambda_6}$ . Thus one finally obtains

$$\nu = \left(\lambda_1 \frac{\nu_6}{\lambda_6}, \lambda_2 \frac{\nu_6}{\lambda_6}, \frac{\nu_6}{\lambda_6}, \lambda_4 \frac{\nu_6}{\lambda_6}, \lambda_5, \nu_6\right)^T$$

with  $\lambda \in \mathbb{R}^6_{>0}$ ,  $\nu_6 > 0$ , free. As for network  $\mathcal{N}_{15}$  it is possible to determine  $\nu$  and  $\lambda \in \mathbb{R}^6_{>0}$  independent of  $\mu \in \mathbb{R}^9$ . Using  $\nu$  as above, one obtains

$$\frac{\nu_1 + \nu_6}{\lambda_1 + \lambda_6} = \frac{\nu_3 + \nu_6}{\lambda_3 + \lambda_6} = \frac{\nu_6}{\lambda_6}.$$

Using  $\kappa_1 := \ln \frac{\nu_6}{\lambda_6}$  and solving for  $\mu$  yields

$$\mu = \begin{bmatrix} 0 & 1 \\ 0 & 0 \\ 0 & 1 \\ 0 & 1 \\ -1 & 1 \\ 1 & 0 \\ 0 & 1 \\ 0 & 1 \end{bmatrix} \begin{pmatrix} \mu_6 \\ \kappa_1 \end{pmatrix}$$

#### A.4.8 Network $N_6 + N_8$

The network is given by

For network  $\mathcal{N}_{22}$  the equation

$$Y^{(L)^T} \mu = \ln \frac{E \nu}{E \lambda}$$

reads

$$\mu_1 + \mu_2 = \ln \frac{\nu_1 + \nu_6}{\lambda_1 + \lambda_6}$$
$$\mu_3 = \ln \frac{\nu_1}{\lambda_1}$$
$$\mu_3 = \ln \frac{\nu_2 + \nu_6}{\lambda_2 + \lambda_6}$$
$$\mu_4 = \ln \frac{\nu_2}{\lambda_2}$$
$$\mu_4 = \ln \frac{\nu_6}{\lambda_3 + \lambda_6}$$
$$\mu_5 + \mu_6 = \ln \frac{\nu_3 + \nu_6}{\lambda_3 + \lambda_6}$$
$$\mu_7 = \ln \frac{\nu_3}{\lambda_3}$$
$$\mu_7 = \ln \frac{\nu_4 + \nu_6}{\lambda_4 + \lambda_6}$$
$$\mu_8 = \ln \frac{\nu_6}{\lambda_6}$$
$$0 = \ln \frac{\nu_5}{\lambda_5}$$
$$\mu_6 = \ln \frac{\nu_5}{\lambda_5}$$

These equations are solvable, if and only if

$$\ln \frac{\nu_1}{\lambda_1} = \ln \frac{\nu_2 + \nu_6}{\lambda_2 + \lambda_6} \qquad \ln \frac{\nu_2}{\lambda_2} = \ln \frac{\nu_6}{\lambda_6} \qquad \ln \frac{\nu_3}{\lambda_3} = \ln \frac{\nu_4 + \nu_6}{\lambda_4 + \lambda_6} \qquad \ln \frac{\nu_4}{\lambda_4} = \ln \frac{\nu_6}{\lambda_6} \qquad \nu_5 = \lambda_5.$$

Note, that these conditions are equivalent to those obtained for A.4.7. Thus one obtains the same solution  $\nu \in \mathbb{R}^{6}_{>0}$ , as in Section A.4.7:

$$\nu = \left(\lambda_1 \frac{\nu_6}{\lambda_6}, \lambda_2 \frac{\nu_6}{\lambda_6}, \frac{\nu_6}{\lambda_6}, \lambda_4 \frac{\nu_6}{\lambda_6}, \lambda_5, \nu_6\right)^T$$

with  $\lambda \in \mathbb{R}^6_{>0}$ ,  $\nu_6 > 0$ , free. As for network  $\mathcal{N}_{15}$  it is possible to determine  $\nu$  and  $\lambda \in \mathbb{R}^6_{>0}$  independent of  $\mu \in \mathbb{R}^8$ . Using  $\nu$  as above, one obtains

$$\frac{\nu_1 + \nu_6}{\lambda_1 + \lambda_6} = \frac{\nu_3 + \nu_6}{\lambda_3 + \lambda_6} = \frac{\nu_6}{\lambda_6}$$

Using  $\kappa_1 := \ln \frac{\nu_6}{\lambda_6}$  and solving for  $\mu$  yields

$$\mu = \begin{bmatrix} -1 & 1 \\ 1 & 0 \\ 0 & 1 \\ 0 & 1 \\ 0 & 1 \\ 0 & 0 \\ 0 & 1 \\ 0 & 1 \end{bmatrix} \begin{pmatrix} \mu_2 \\ \kappa_1 \end{pmatrix}.$$

#### A.4.9 Network $N_6 + N_9$

The network is given by

For network  $\mathcal{N}_{23}$  the equation

$$Y^{(L)^{T}} \mu = \ln \frac{E \nu}{E \lambda}$$

reads

$$\begin{array}{l} \mu_{1} + \mu_{2} = \ln \frac{\nu_{1} + \nu_{7}}{\lambda_{1} + \lambda_{7}} \\ \mu_{3} = \ln \frac{\nu_{1}}{\lambda_{1}} \\ \mu_{3} = \ln \frac{\nu_{2} + \nu_{7}}{\lambda_{2} + \lambda_{7}} \\ \mu_{4} = \ln \frac{\nu_{2}}{\lambda_{2}} \\ \mu_{4} = \ln \frac{\nu_{7}}{\lambda_{7}} \\ \mu_{5} + \mu_{6} = \ln \frac{\nu_{3} + \nu_{7}}{\lambda_{3} + \lambda_{7}} \\ \mu_{7} = \ln \frac{\nu_{3}}{\lambda_{3}} \\ \mu_{7} = \ln \frac{\nu_{4}}{\lambda_{3}} \\ \mu_{8} = \ln \frac{\nu_{4}}{\lambda_{4} + \lambda_{7}} \\ \mu_{8} = \ln \frac{\nu_{4}}{\lambda_{7}} \\ \mu_{8} = \ln \frac{\nu_{5}}{\lambda_{5}} \\ 0 = \ln \frac{\nu_{5}}{\lambda_{5}} \\ 0 = \ln \frac{\nu_{5}}{\lambda_{5}} \\ 0 = \ln \frac{\nu_{6}}{\lambda_{6}} \\ \mu_{6} = \ln \frac{\nu_{6}}{\lambda_{6}} \end{array}$$

These equations are solvable, if and only if

$$\ln \frac{\nu_1}{\lambda_1} = \ln \frac{\nu_2 + \nu_7}{\lambda_2 + \lambda_7} \quad \ln \frac{\nu_2}{\lambda_2} = \ln \frac{\nu_7}{\lambda_7} \quad \ln \frac{\nu_3}{\lambda_3} = \ln \frac{\nu_4 + \nu_7}{\lambda_4 + \lambda_7} \quad \ln \frac{\nu_4}{\lambda_4} = \ln \frac{\nu_7}{\lambda_7} \quad \nu_5 = \lambda_5 \quad \nu_6 = \lambda_6.$$

Thus one obtains

$$\nu_2 = \lambda_2 \frac{\nu_7}{\lambda_7}, \qquad \qquad \nu_3 = \lambda_3 \frac{\nu_7}{\lambda_7}, \qquad \qquad \nu_4 = \lambda_4 \frac{\nu_7}{\lambda_7}.$$

and therefore

$$\nu = \left(\lambda_1 \frac{\nu_7}{\lambda_7}, \lambda_2 \frac{\nu_7}{\lambda_7}, \lambda_3 \frac{\nu_7}{\lambda_7}, \lambda_4 \frac{\nu_7}{\lambda_7}, \lambda_5, \lambda_6, \nu_7\right)^T,$$

with  $\lambda \in \mathbb{R}^{7}_{>0}$ ,  $\nu_{7} > 0$ , free. As for network  $\mathcal{N}_{15}$  it is possible to determine  $\nu$  and  $\lambda \in \mathbb{R}^{7}_{>0}$  independent of  $\mu \in \mathbb{R}^{9}$ . Using  $\nu$  as above, one obtains

$$\frac{\nu_1 + \nu_7}{\lambda_1 + \lambda_7} = \frac{\nu_3 + \nu_7}{\lambda_3 + \lambda_7} = \frac{\nu_7}{\lambda_7}.$$

Using  $\kappa_1 := \ln \frac{\nu_7}{\lambda_7}$  and solving for  $\mu$  yields

$$\mu = \begin{bmatrix} 1, 0, 1, 1, 1, 0, 1, 1 \end{bmatrix}^T \kappa_1.$$

## Appendix B

# Models for cell cycle regulation

#### B.1 Binary complex model

Species	$x_i$	Complex	$y_i$
Sic1	$x_1$	0	$y_1$
Sic1P	$x_2$	Sic1	$y_2$
Clb	$x_3$	Sic1P	$y_3$
$Clb \cdot Sic1$	$x_4$	$Sic1 \cdot Clb$	$y_4$
$Clb \cdot Sic1P$	$x_5$	Clb + Sic1	$y_5$
Cdc14	$x_6$	$Clb \cdot Sic1$	$y_6$
$Sic1P \cdot Cdc14$	$x_7$	Clb	$y_7$
$Clb \cdot Sic1P \cdot Cdc14$	$x_8$	Clb + Sic1P	$y_8$
$Sic1 \cdot Clb$	$x_9$	$Clb \cdot Sic1P$	$y_9$
		Sic1P + Cdc14	$y_{10}$
		$Sic1P \cdot Cdc14$	$y_{11}$
		Sic1 + Cdc14	$y_{12}$
		$Clb \cdot Sic1P + Cdc14$	$y_{13}$
		$Clb \cdot Sic1P \cdot Cdc14$	$y_{14}$
		$Clb \cdot Sic1 + Cdc14$	$y_{15}$

#### B.1.1 Species and complexes of network $\mathcal{N}_{10}$

#### B.1.2 Ordinary differential equations

$$\begin{split} \dot{x}_1 &= k_1 - k_2 \, x_1 + k_4 \, x_9 - k_5 \, x_1 \, x_3 - k_6 \, x_1 \, x_3 + k_7 \, x_4 + k_{15} \, x_7 & (B.1a) \\ \dot{x}_2 &= -k_3 \, x_2 + k_9 \, x_9 - k_{10} \, x_2 \, x_3 + k_{11} \, x_5 - k_{13} \, x_2 \, x_6 + k_{14} \, x_7 & (B.1b) \\ \dot{x}_3 &= k_4 \, x_9 - k_5 \, x_1 \, x_3 - k_6 \, x_1 \, x_3 + k_7 \, x_4 + k_8 \, x_4 + k_9 \, x_9 - k_{10} \, x_2 \, x_3 + k_{11} \, x_5 + k_{12} \, x_5 & (B.1c) \\ \dot{x}_4 &= k_6 \, x_1 \, x_3 - k_7 \, x_4 - k_8 \, x_4 + k_{18} \, x_8 & (B.1d) \\ \dot{x}_5 &= k_{10} \, x_2 \, x_3 - k_{11} \, x_5 - k_{12} \, x_5 - k_{16} \, x_5 \, x_6 + k_{17} \, x_8 & (B.1e) \\ \dot{x}_6 &= -k_{13} \, x_2 \, x_6 + k_{14} \, x_7 + k_{15} \, x_7 - k_{16} \, x_5 \, x_6 + k_{17} \, x_8 + k_{18} \, x_8 & (B.1f) \\ \dot{x}_7 &= k_{13} \, x_2 \, x_6 - k_{14} \, x_7 - k_{15} \, x_7 & (B.1g) \\ \dot{x}_8 &= k_{16} \, x_5 \, x_6 - k_{17} \, x_8 - k_{18} \, x_8 & (B.1h) \\ \dot{y}_9 &= -k_4 \, x_9 + k_5 \, x_1 \, x_3 - k_9 \, x_9 & (B.1i) \end{split}$$

$$x_6 + x_7 + x_8 = c_1 \tag{B.2a}$$

$$x_3 + x_4 + x_5 + x_8 + x_9 = c_2 \tag{B.2b}$$

#### B.1.3 Structural data

The ordinary differential equations (B.1a) - (B.1i) can be written as

$$\dot{x} = Y I_a \operatorname{diag}(k) \phi(x)$$

using matrices

			Γ	0	1	0	0	1	0	0	0	0	0	0	1	0	0	0			
				0	0	1	0	0	0	0	1	0	1	0	0	0	0	0			
				0	0	0	0	1	0	1	1	0	0	0	0	0	0	0			
				0	0	0	0	0	1	0	0	0	0	0	0	0	0	1			
		V =	_	õ	Ő	õ	Ő	Ő	0	Ő	Ő	1	Ő	Ő	Ő	1	Ő	0			
		1 -	-	0	0	0	0	0	0	0	0	0	1	0	1	1	0	1	,		
				0	0	0	0	0	0	0	0	0	1	1	1	1	0	0			
				0	0	0	0	0	0	0	0	0	0	1	0	0	0	0			
				0	0	0	0	0	0	0	0	0	0	0	0	0	1	0			
				0	0	0	1	0	0	0	0	0	0	0	0	0	0	0			
			-															-			
		1	1	0		0	0	0	0	2	0	0	0	0	2	0	0	0	0	0	0
	1	-1	0	0		0	0	0		)	0	0	0		)	0	0	0	0	0	0
	ŏ	ő	0	-1		1	0	0	č	j.	-1	0	0	č	Ś	0	0	ő	ő	ő	ő
	0	0	0	1	-	1	-1	1	(	)	0	0	0	(	)	0	0	0	0	0	0
	0	0	0	0		0	1	-1	- 1	L	0	0	0	(	)	0	0	0	0	0	0
T	0	0	0	0		0	0	0	3	L	0	0	0			0	0	0	0	0	0
$I_a =$	0	0	0	0		0	0	0	0	)	1	-1	1	(	)	0	0	0	0	0	0
	0	0	0	0		0	0	0	9	)	0	1	-1	-1		0	0	0	0	0	0
		0	0			0	0	0			0	0	0			- 1	1	1	0	0	8
	l ő	0	0	0		0	0	0	2	Ś	0	0	0		Ś	0	-1	-1	0	0	~ Å
	ŏ	ŏ	ő	ő		ŏ	ő	ő	ò	ś	ő	ő	ő	ò	ś	ő	ŏ	ô	-1	1	ŏ
	0	0	0	0		0	0	0	(	)	0	0	0	(	5	0	0	0	1	-1	-1
	L o	0	0	0		0	0	0	0	)	0	0	0	0	)	0	0	0	0	0	1

and the monomial vector

$$\phi\left(x\right) = \left(1, x_{1}, x_{2}, x_{9}, x_{1} x_{3}, x_{1} x_{3}, x_{4}, x_{4}, x_{9}, x_{2} x_{3}, x_{5}, x_{5}, x_{2} x_{6}, x_{7}, x_{7}, x_{5} x_{6}, x_{8}, x_{8}\right)^{T}$$

A representation  $\phi(x) = \left(x^{y_1^{(L)}}, \dots, x^{y_r^{(L)}}\right)'$  can be obtained using

	0	1	0	0	1	1	0	0	0	0	0	0	0	0	0	0	0	0
	0	0	1	0	0	0	0	0	0	1	0	0	1	0	0	0	0	0
	0	0	0	0	1	1	0	0	0	1	0	0	0	0	0	0	0	0
	0	0	0	0	0	0	1	1	0	0	0	0	0	0	0	0	0	0
$Y^{(L)} =$	0	0	0	0	0	0	0	0	0	0	1	1	0	0	0	1	0	0
	0	0	0	0	0	0	0	0	0	0	0	0	1	0	0	1	0	0
	0	0	0	0	0	0	0	0	0	0	0	0	0	1	1	0	0	0
	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	1	1
	0	0	0	1	0	0	0	0	1	0	0	0	0	0	0	0	0	0
	L																	

The conservation relations (B.2a) and (B.2b) can be written as

$$W^T x = c$$

using the matrix

$$W^T = \left[ \begin{array}{rrrrr} 0 & 0 & 0 & 0 & 0 & 1 & 1 & 1 & 0 \\ 0 & 0 & 1 & 1 & 1 & 0 & 0 & 1 & 1 \end{array} \right]$$

and the vector  $c^T = (c_1, c_2)$ .

#### B.2 Ternary complex model

Species	$x_i$	Complex	$y_i$
Sic1	$x_1$	0	$y_1$
Sic1P	$x_2$	Sic1	$y_2$
Clb	$x_3$	Sic1P	$y_3$
$Clb \cdot Sic1$	$x_4$	Clb + Sic1	$y_4$
$Clb \cdot Sic1P$	$x_5$	$Clb \cdot Sic1$	$y_5$
Cdc14	$x_6$	Clb	$y_6$
$Sic1P \cdot Cdc14$	$x_7$	Clb + Sic1P	$y_7$
$Clb \cdot Sic1P \cdot Cdc14$	$x_8$	$Clb \cdot Sic1P$	$y_8$
$Clb \cdot Sic1 \cdot Clb$	$x_9$	$Clb \cdot Sic1 + Clb$	$y_9$
		$Clb \cdot Sic1 \cdot Clb$	$y_{10}$
		$Clb \cdot Sic1P + Clb$	$y_{11}$
		Sic1P + Cdc14	$y_{12}$
		$Sic1P \cdot Cdc14$	$y_{13}$
		Sic1 + Cdc14	$y_{14}$
		$Clb \cdot Sic1P + Cdc14$	$y_{15}$
		$Clb \cdot Sic1P \cdot Cdc14$	$y_{16}$
		$Clb \cdot Sic1 + Cdc14$	$y_{17}$

B.2.1 Species and complexes of network  $\mathcal{N}_{11}$ 

#### B.2.2 Ordinary differential equations

$$\begin{split} \dot{x}_1 &= k_1 - k_2 \, x_1 - k_4 \, x_1 \, x_3 + k_5 \, x_4 + k_{15} \, x_7 & (B.5a) \\ \dot{x}_2 &= -k_3 \, x_2 - k_7 \, x_2 \, x_3 + k_8 \, x_5 - k_{13} \, x_2 \, x_6 + k_{14} \, x_7 & (B.5b) \\ \dot{x}_3 &= -k_4 \, x_1 \, x_3 + k_5 \, x_4 + k_6 \, x_4 - k_7 \, x_2 \, x_3 + k_8 \, x_5 + k_9 \, x_5 - k_{10} \, x_3 \, x_4 + k_{11} \, x_9 + k_{12} \, x_9 & (B.5c) \\ \dot{x}_4 &= k_4 \, x_1 \, x_3 - k_5 \, x_4 - k_6 \, x_4 - k_{10} \, x_3 \, x_4 + k_{11} \, x_9 + k_{18} \, x_8 & (B.5d) \\ \dot{x}_5 &= k_7 \, x_2 \, x_3 - k_8 \, x_5 - k_9 \, x_5 + k_{12} \, x_9 - k_{16} \, x_5 \, x_6 + k_{17} \, x_8 & (B.5e) \\ \dot{x}_6 &= -k_{13} \, x_2 \, x_6 + k_{14} \, x_7 + k_{15} \, x_7 - k_{16} \, x_5 \, x_6 + k_{17} \, x_8 + k_{18} \, x_8 & (B.5f) \\ \dot{x}_7 &= k_{13} \, x_2 \, x_6 - k_{14} \, x_7 - k_{15} \, x_7 & (B.5g) \\ \dot{x}_8 &= k_{16} \, x_5 \, x_6 - k_{17} \, x_8 - k_{18} \, x_8 & (B.5h) \\ \dot{x}_9 &= k_{10} \, x_3 \, x_4 - k_{11} \, x_9 - k_{12} \, x_9 & (B.5i) \\ \end{split}$$

$$x_6 + x_7 + x_8 = c_1$$
 (B.6a)

$$x_3 + x_4 + x_5 + x_8 + 2x_9 = c_2 \tag{B.6b}$$

#### B.2.3 Structural data

The ordinary differential equations (B.5a) - (B.5i) can be written as

$$\dot{x} = Y I_a \operatorname{diag}(k) \phi(x)$$

using matrices

	0	1	0	1	0	0	0	0	0	0	0	0	0	1	0	0	0	1
	0	0	1	0	0	0	1	0	0	0	0	1	0	0	0	0	0	
	0	0	0	1	0	1	1	0	1	0	1	0	0	0	0	0	0	
	0	0	0	0	1	0	0	0	1	0	0	0	0	0	0	0	1	
Y =	0	0	0	0	0	0	0	1	0	0	1	0	0	0	1	0	0	,
	0	0	0	0	0	0	0	0	0	0	0	1	0	1	1	0	1	
	0	0	0	0	0	0	0	0	0	0	0	0	1	0	0	0	0	
	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	1	0	
	0	0	0	0	0	0	0	0	0	1	0	0	0	0	0	0	0 .	

#### APPENDIX B. MODELS FOR CELL CYCLE REGULATION

and the monomial vector

$$\phi\left(x\right) = \left(1, x_{1}, x_{2}, x_{1} x_{3}, x_{4}, x_{4}, x_{2} x_{3}, x_{5}, x_{5}, x_{3} x_{4}, x_{9}, x_{9}, x_{2} x_{6}, x_{7}, x_{7}, x_{5} x_{6}, x_{8}, x_{8}\right) T$$

A representation  $\phi(x) = \left(x^{y_1^{(L)}}, \dots, x^{y_r^{(L)}}\right)'$  can be obtained using

	0	1	0	1	0	0	0	0	0	0	0	0	0	0	0	0	0	0
	0	0	1	0	0	0	1	0	0	0	0	0	1	0	0	0	0	0
	0	0	0	1	0	0	1	0	0	1	0	0	0	0	0	0	0	0
	0	0	0	0	1	1	0	0	0	1	0	0	0	0	0	0	0	0
$Y^{(L)} =$	0	0	0	0	0	0	0	1	1	0	0	0	0	0	0	1	0	0
	0	0	0	0	0	0	0	0	0	0	0	0	1	0	0	1	0	0
	0	0	0	0	0	0	0	0	0	0	0	0	0	1	1	0	0	0
	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	1	1
	0	0	0	0	0	0	0	0	0	0	1	1	0	0	0	0	0	0

The conservation relations (B.6a) and (B.6b) can be written as

$$W^T x = c$$

using the matrix

and the vector  $c^T = (c_1, c_2)$ . A matrix whose columns are generators of ker $(Y I_a) \cap \mathbb{R}^{18}_{\geq 0}$  is

(B.9)

# Appendix C

# Models of the signal transduction motifs

C.1 Network  $N_{12}$ 

C.1.1	Species	$\operatorname{and}$	compl	exes	of	network	Л	12
-------	---------	----------------------	-------	------	----	---------	---	----

Species	$x_i$	Complex	$y_i$
Α	$x_1$	$A + E_1$	$y_1$
$E_1$	$x_2$	$A E_1$	$y_2$
$A E_1$	$x_3$	$A^{*} + E_{1}$	$y_3$
$A^*$	$x_4$	$A + A^*$	$y_4$
$A A^*$	$x_5$	$A A^*$	$y_5$
$E_2$	$x_6$	$A^* + A^*$	$y_6$
$A^* E_2$	$x_7$	$A^{*} + E_{2}$	$y_7$
		$A^* E_2$	$y_8$
		$A + E_2$	$y_9$

#### C.1.2 Ordinary differential equations

$\dot{x}_1 = -k_1 x_1 x_1$	$x_2 + k_2 x_3 - $	$k_4 x_1 x_4 + k_5 x_5 + k_9 x_7 \tag{(}$	(C.1a	ı)
----------------------------	--------------------	---	-------	----

$$\dot{x}_2 = -k_1 x_1 x_2 + k_2 x_3 + k_3 x_3 \tag{C.1b}$$

$$\dot{x}_3 = k_1 \, x_1 \, x_2 - k_2 \, x_3 - k_3 \, x_3 \tag{C.1c}$$

$$\dot{x}_4 = k_3 x_3 - k_4 x_1 x_4 + k_5 x_5 + 2 k_6 x_5 - k_7 x_4 x_6 + k_8 x_7 \tag{C.1d}$$

$$\dot{x}_5 = k_4 \, x_1 \, x_4 - k_5 \, x_5 - k_6 \, x_5 \tag{C.1e}$$

$$\dot{x}_6 = -k_7 \, x_4 \, x_6 + k_8 \, x_7 + k_9 \, x_7 \tag{C.1f}$$

$$\dot{x}_7 = k_7 x_4 x_6 - k_8 x_7 - k_9 x_7 \tag{C.1g}$$

 $x_2 + x_3 = c_1$  (C.2a)

$$x_6 + x_7 = c_2$$
 (C.2b)

$$x_1 + x_3 + x_4 + 2x_5 + x_7 = c_3 \tag{C.2c}$$

#### C.1.3 Structural data

The ordinary differential equations (C.1a) - (C.1g) can be written as

$$\dot{x} = Y I_a \operatorname{diag}(k) \phi(x)$$

using matrices

and the monomial vector

$$\phi(x) = (x_1 x_2, x_3, x_3, x_1 x_4, x_5, x_5, x_4 x_6, x_7, x_7)^T$$

A representation  $\phi(x) = \left(x^{y_1^{(L)}}, \dots, x^{y_r^{(L)}}\right)^T$  can be obtained using

	1	0	0	1	0	0	0	0	0
	1	0	0	0	0	0	0	0	0
	0	1	1	0	0	0	0	0	0
$Y^{(L)} =$	0	0	0	1	0	0	1	0	0
	0	0	0	0	1	1	0	0	0
	0	0	0	0	0	0	1	0	0
	0	0	0	0	0	0	0	1	1

The conservation relations (C.2a) - (C.2c) can be written as

$$W^T x = c$$

using the matrix

and the vector  $c^{T} = (c_{1}, c_{2}, c_{3}).$ 

#### C.2 Network $\mathcal{N}_{13}$

Species	$x_i$	Complex	$y_i$
A	$x_1$	$A + E_1$	$y_1$
$E_1$	$x_2$	$A E_1$	$y_2$
$A E_1$	$x_3$	$A^{*} + E_{1}$	$y_3$
$A^*$	$x_4$	$A^* E_1$	$y_4$
$A^* E_1$	$x_5$	$A^{**} + E_1$	$y_5$
$A^{**}$	$x_6$	$A^{**} + E_3$	$y_6$
$E_3$	$x_7$	$A^{**}E_3$	$y_7$
$A^{**}E_3$	$x_8$	$A^{*} + E_{3}$	$y_8$
$E_2$	$x_9$	$A^{*} + E_{2}$	$y_9$
$A^* E_2$	$x_{10}$	$A^* E_2$	$y_{10}$
		$A + E_2$	$y_{11}$

C.2.1 Species for and complexes of network  $N_{13}$ 

#### C.2.2 Ordinary differential equations

$$\dot{x}_1 = -k_1 x_1 x_2 + k_2 x_3 + k_{12} x_{10}$$

$$\dot{x}_2 = -k_1 x_1 x_2 + k_2 x_3 + k_3 x_3 - k_4 x_2 x_4 + k_5 x_5 + k_6 x_5$$
(C.5b)

$$\dot{x}_3 = k_1 \, x_1 \, x_2 - k_2 \, x_3 - k_3 \, x_3 \tag{C.5c}$$

$$\dot{x}_4 = k_3 x_3 - k_4 x_2 x_4 + k_5 x_5 + k_9 x_8 - k_{10} x_4 x_9 + k_{11} x_{10}$$
(C.5d)

$$\dot{x}_5 = k_4 \, x_2 \, x_4 - k_5 \, x_5 - k_6 \, x_5 \tag{C.5e}$$

$$\dot{x}_6 = k_6 x_5 - k_7 x_6 x_7 + k_8 x_8 \tag{C.5f}$$

$$\dot{x}_7 = -k_7 x_6 x_7 + k_8 x_8 + k_9 x_8$$
 (C.5g)  
 $\dot{x}_9 = k_7 x_6 x_7 - k_9 x_9 - k_9 x_9$  (C.5b)

$$x_8 = k_7 x_6 x_7 - k_8 x_8 - k_9 x_8 \tag{C.5h}$$

$$\dot{x}_7 = -k_{12} x_7 x_6 x_7 - k_8 x_8 - k_9 x_8 \tag{C.5h}$$

$$x_9 = -\kappa_{10} x_4 x_9 + \kappa_{11} x_{10} + \kappa_{12} x_{10} \tag{(C.51)}$$

$$\dot{x}_{10} = k_{10} \, x_4 \, x_9 - k_{11} \, x_{10} - k_{12} \, x_{10} \tag{C.5j}$$

$$x_2 + x_3 + x_5 = c_1 (C.6a)$$

$$x_7 + x_8 = c_2 (C.0D) (C.0D) (C.6c)$$

$$x_9 + x_{10} - c_3$$
 (C.0c)

$$x_1 + x_3 + x_4 + x_5 + x_6 + x_8 + x_{10} = c_4 \tag{C.6d}$$

#### C.2.3 Structural data

The ordinary differential equations (C.5a) - (C.5j) can be written as

$$\dot{x} = Y I_a \operatorname{diag}(k) \phi(x)$$

using matrices

and the monomial vector

$$\phi\left(x\right) = \left(\,x_{1}\,x_{2},\,x_{3},\,x_{3},\,x_{2}\,x_{4},\,x_{5},\,x_{5},\,x_{6}\,x_{7},\,x_{8},\,x_{8},\,x_{4}\,x_{9},\,x_{10},\,x_{10}\,\right)^{T}$$

A representation  $\phi\left(x\right) = \left(x^{y_{1}^{(L)}}, \dots, x^{y_{r}^{(L)}}\right)^{T}$  can be obtained using

The conservation relations (C.6a) - (C.6d) can be written as

$$W^T x = c$$

using the matrix

	0	1	1	0	1	0	0	0	0	0	1
WT	0	0	0	0	0	0	1	1	0	0	l
vv =	0	0	0	0	0	0	0	0	1	1	
	1	0	1	1	1	1	0	1	0	1	

and the vector  $c^T = (c_1, c_2, c_3, c_4).$ 

#### C.3 Network $N_{14}$

#### C.3.1 Species and complexes of network $N_{14}$

Species	$x_i$	Complex	$y_i$
A	$x_1$	$A + E_1$	$y_1$
$E_1$	$x_2$	$A E_1$	$y_2$
$A E_1$	$x_3$	$A^{*} + E_{1}$	$y_3$
$A^*$	$x_4$	$A^{*} + E_{3}$	$y_4$
$E_3$	$x_5$	$A^* E_3$	$y_5$
$A^* E_3$	$x_6$	$A + E_3$	$y_6$
$A^* A^*$	$x_7$	$2 A^{*}$	$y_7$
$A^{**}$	$x_8$	$A^* A^*$	$y_8$
$A^*A^{**}$	$x_9$	$A^{**} + A^*$	$y_9$
$E_2$	$x_{10}$	$A^* A^{**}$	$y_{10}$
$A^{**}E_2$	$x_{11}$	$2 A^{**}$	$y_{11}$
		$A^{**} + E_2$	$y_{12}$
		$A^{**}E_2$	$y_{13}$
		$A^{*} + E_{2}$	$y_{14}$

#### C.3.2 Ordinary differential equations

$$\begin{split} \dot{x}_1 &= -k_1 \, x_1 \, x_2 + k_2 \, x_3 + k_6 \, x_6 & (C.9a) \\ \dot{x}_2 &= -k_1 \, x_1 \, x_2 + k_2 \, x_3 + k_3 \, x_3 & (C.9b) \\ \dot{x}_3 &= k_1 \, x_1 \, x_2 - k_2 \, x_3 - k_3 \, x_3 & (C.9c) \\ \dot{x}_4 &= k_3 \, x_3 - k_4 \, x_4 \, x_5 + k_5 \, x_6 - 2 \, k_7 \, x_4^2 + 2 \, k_8 \, x_7 + k_9 \, x_7 - k_{10} \, x_4 \, x_8 + k_{11} \, x_9 + k_{15} \, x_{11} & (C.9d) \\ \dot{x}_5 &= -k_4 \, x_4 \, x_5 + k_5 \, x_6 + k_6 \, x_6 & (C.9e) \\ \dot{x}_6 &= k_4 \, x_4 \, x_5 - k_5 \, x_6 - k_6 \, x_6 & (C.9f) \\ \dot{x}_7 &= k_7 \, x_4^2 - k_8 \, x_7 - k_9 \, x_7 & (C.9g) \\ \dot{x}_8 &= k_9 \, x_7 - k_{10} \, x_4 \, x_8 + k_{11} \, x_9 + 2 \, k_{12} \, x_9 - k_{13} \, x_8 \, x_{10} + k_{14} \, x_{11} & (C.9h) \\ \dot{x}_9 &= k_{10} \, x_4 \, x_8 - k_{11} \, x_9 - k_{12} \, x_9 & (C.9i) \\ \dot{x}_{11} &= -k_{13} \, x_8 \, x_{10} - k_{14} \, x_{11} + k_{15} \, x_{11} & (C.9k) \\ \dot{x}_{11} &= k_{13} \, x_8 \, x_{10} - k_{14} \, x_{11} - k_{15} \, x_{11} & (C.9k) \\ \dot{x}_{11} &= k_{13} \, x_8 \, x_{10} - k_{14} \, x_{11} - k_{15} \, x_{11} & (C.9k) \\ \dot{x}_{11} &= k_{13} \, x_8 \, x_{10} - k_{14} \, x_{11} - k_{15} \, x_{11} & (C.9k) \\ \dot{x}_{11} &= k_{13} \, x_8 \, x_{10} - k_{14} \, x_{11} - k_{15} \, x_{11} & (C.9k) \\ \dot{x}_{11} &= k_{13} \, x_8 \, x_{10} - k_{14} \, x_{11} - k_{15} \, x_{11} & (C.9k) \\ \dot{x}_{11} &= k_{13} \, x_8 \, x_{10} - k_{14} \, x_{11} - k_{15} \, x_{11} & (C.9k) \\ \dot{x}_{11} &= k_{13} \, x_8 \, x_{10} - k_{14} \, x_{11} - k_{15} \, x_{11} & (C.9k) \\ \dot{x}_{11} &= k_{13} \, x_8 \, x_{10} - k_{14} \, x_{11} - k_{15} \, x_{11} & (C.9k) \\ \dot{x}_{11} &= k_{13} \, x_8 \, x_{10} - k_{14} \, x_{11} - k_{15} \, x_{11} & (C.9k) \\ \dot{x}_{11} &= k_{13} \, x_8 \, x_{10} - k_{14} \, x_{11} - k_{15} \, x_{11} & (C.9k) \\ \dot{x}_{11} &= k_{13} \, x_8 \, x_{10} - k_{14} \, x_{11} - k_{15} \, x_{11} & (C.9k) \\ \dot{x}_{11} &= k_{13} \, x_8 \, x_{10} - k_{14} \, x_{11} - k_{15} \, x_{11} & (C.9k) \\ \dot{x}_{11} &= k_{13} \, x_8 \, x_{10} - k_{14} \, x_{11} - k_{15} \, x_{11} & (C.9k) \\ \dot{x}_{11} &= k_{13} \, x_8 \, x_{10} - k_{14} \, x_{11} - k_{15} \, x_{11} & (C.9k) \\ \dot{x}_{11} &= k_{13} \, x_8 \, x_{10} - k_{14} \, x_{11} + k_{15} \, x_{11} & (C.9k) \\ \dot{x}$$

$x_2 + x_3 = c_1$	(C.10a)
$x_5 + x_6 = c_2$	(C.10b)
$x_{10} + x_{11} = c_3$	(C.10c)
$x_1 + x_3 + x_4 + x_6 + 2 x_7 + x_8 + 2 x_9 + x_{11} = c_4$	(C.10d)

#### C.3.3 Structural data

The ordinary differential equations (C.9a) - (C.9k) can be written as

$$\dot{x} = Y I_a \operatorname{diag}(k) \phi(x)$$

using matrices

and the monomial vector

$$\phi\left(x\right) = \left(x_{1} x_{2}, x_{3}, x_{3}, x_{4} x_{5}, x_{6}, x_{6}, x_{4}^{2}, x_{7}, x_{7}, x_{4} x_{8}, x_{9}, x_{9}, x_{8} x_{10}, x_{11}, x_{11}\right)^{T}$$

A representation  $\phi(x) = \left(x^{y_1^{(L)}}, \dots, x^{y_r^{(L)}}\right)^T$  can be obtained using

The conservation relations (C.10a) - (C.10d) can be written as

$$W^T x = c$$

using the matrix

and the vector  $c^T = (c_1, c_2, c_3, c_4).$ 

### Appendix D

## An Algorithm to check Lemma 3

#### D.1 Preliminary ideas

The starting point is a software to calculate the extreme rays of a pointed polyhedral cone ker  $(A) \cap \mathbb{R}^n_{>0}$ , where  $A \in \mathbb{R}^{m \times n}$  is an arbitrary  $m \times n$  matrix. It is possible to use this software in order to calculate extreme rays of a different cone ker  $(A) \cap \mathbb{R}^n_{\delta_i}$ . To see this, recall the definition of an extreme ray v of ker  $(A) \cap \mathbb{R}^n_{\delta_i}$  (see Chapter 3):

$$v \in \mathbb{R}^{n}_{\delta_{i}}$$
  
 $A v = 0,$ 

Given u, v with A u = 0 and A v = 0. Then

$$supp(u) \subseteq supp(v) \Rightarrow u = 0 \text{ or } v = \alpha u, \alpha \in \mathbb{R}_{>0}.$$

Further note that any  $v \in \mathbb{R}^n_{\delta_i}$  can be expressed as

$$v = \text{diag}(\delta_i) \ w, \ w \in \mathbb{R}^n_{>0}, \text{ with } w_i = |v_i|, \ i = 1, \dots, n.$$
 (D.1)

Using eq. (D.1) in Av = 0 yields

A diag  $(\delta_i) w = 0.$ 

Then extreme rays of a cone ker  $(A) \cap \mathbb{R}^n_{\delta_i}$  can be defined as

$$v \in \mathbb{R}^{n}_{\delta_{i}}$$
 (D.2a)

given by  $v = diag(\delta_i) w$  with

$$w \in \mathbb{R}^{n}_{\geq 0}$$
 (D.2b)

$$\tilde{A} w = 0$$
, (D.2c)

where  $\widetilde{A} := A \operatorname{diag}(\delta_i)$ . Given  $w^{(1)}, w^{(2)}$  with  $A w^{(1)} = 0$  and  $A w^{(2)} = 0$ . Then

$$\operatorname{supp}\left(w^{(1)}\right) \subseteq \operatorname{supp}\left(w^{(2)}\right) \Rightarrow w^{(1)} = 0 \text{ or } w^{(2)} = \alpha \, u, \, \alpha \in \mathbb{R}_{>0}. \tag{D.2d}$$

If  $\widetilde{E}_1, \ldots, \widetilde{E}_p$  are generators of ker  $(\widetilde{A}) \cap \mathbb{R}_{\geq 0}^n$ , then  $E_i = \operatorname{diag}(\delta_i) \widetilde{E}_i, i = 1, \ldots, p$  are generators of ker  $(A) \cap \mathbb{R}_{\delta_i}^n$ . To see this, consider  $A E_i = A \operatorname{diag}(\delta_i) \widetilde{E}_i = \widetilde{A} \widetilde{E}_i = 0, i = 1, \ldots, p$ . Using this definition it is thus possible to calculate generators of ker  $(A) \cap \mathbb{R}_{\delta_i}^n$  using software designed to calculate generators of ker  $(A) \cap \mathbb{R}_{\geq 0}^n$ .

#### **D.2** Deciding ker $(A) \cap \mathbb{R}^n_{\delta_i} \neq \emptyset$

To decide ker  $(A) \cap \mathbb{R}^n_{\delta_i} \neq \emptyset$  the following algorithm is used: Generate all  $\delta_i$ , i.e. all combinations of n elements from  $\{-1, 0, 1\}$ . This is done using the m-file combin.m, obtainable at http://www.mathworks.com/matlabcentral For all  $\delta_i$ 

- (i)  $\widetilde{A} := \operatorname{diag}(\delta_i) A$
- (ii) calculate generators  $\widetilde{E}_i$ ,  $i = 1, \ldots, p$  of ker  $(A) \cap \mathbb{R}^n_{\delta}$ . If

$$\nexists j \in \{1, \ldots, n\}$$
 with  $j \notin \text{supp}(\widetilde{E}_i), i = 1, \ldots, p$ 

then ker  $(A) \cap \mathbb{R}^n_{\delta_i} \neq \emptyset$ . Thus keep  $\delta_i$ .

The condition in (ii) requires that no entry  $\tilde{E}_{i_j}$  exists that is zero in every generator. Those correspond to generators for a different orthant, namely the one where  $\delta_{i_j} = 0$ .

The above mentioned algorithm is by no means efficient. It could easily be made more efficient by

- Considering that *M* and *S* are subspaces, that is, that μ ∈ *M* ⇒ −μ ∈ *M* and v ∈ *S* ⇒ −v ∈ *S*. Thus, if *M* ∩ ℝ<sup>n</sup><sub>δ</sub> ≠ ∅ and *S* ∩ ℝ<sup>n</sup><sub>δ</sub> ≠ ∅, then *M* ∩ ℝ<sup>n</sup><sub>−δ</sub> ≠ ∅ and *S* ∩ ℝ<sup>n</sup><sub>−δ</sub> ≠ ∅ as well.
- 2. Using the implicit information given in step (ii) that even though ker  $(A) \cap \mathbb{R}^n_{\delta} = \emptyset$ , there exists an orthant  $\overline{\delta}$  with ker  $(A) \cap \mathbb{R}^n_{\overline{\delta}} \neq \emptyset$ . For  $\overline{\delta}$  holds the following

$$\bar{\delta}_j = \begin{cases} \delta_j, \text{ if } \exists i \in \{1, \dots, p\} \text{ with } E_{i_j} \neq 0\\ 0, \text{ else.} \end{cases}$$

The algorithm is available in the matlab file sign\_comp\_subs.m displayed below:

```
function E=sign_comp_subs( W )
% E=sign_comp_subs( W ) returns a matrix E whose columns correspond
% to those orthants of R^n that are sign compatible to the linear
% subspace that is defined by the matrix W: W is the orthogonal
% complement of the subspace under consideration. Each column
% vector e entirely consits of entries e_i=1, e_i=0 or e_i=-1, indicating
% x_i>0, x_i=0 and x_i<0, respectively. Thus each e defines an
% orthant.
n=length(W)
M=combn([-1 0 1], n);
[Mnrows, Mncols] = size(M);
or_count = 1;
for i=1:Mnrows
 disp('Processing orthant:'); M(i,:)
  [elm,num_modes,revs,mode_rates,fix_rates]= ...
      elmodes_calc((diag(M(i,:))*W)',[],[], ones(1,length(W)), ...
                   [],1e-10,'useMex','1');
 if length(fix_rates)==0
    E(:,or_count) = M(i,:)';
   or_count = or_count+1;
 end:
```

end;

The command [elm,...] = elmodes\_calc(...) invokes a routine to calculated the generators of the cone  $diag(M(i,:)) * W \cap \mathbb{R}^n_{>0}$ . It is contained in the software package *CellNetAnalyzer* [32].
## D.3 Determining generators

The following steps were executed to derive the generators for all orthants where  $\mathcal{M} \cap \mathbb{R}^n_{\delta} \neq \emptyset$  and  $\mathcal{S} \cap \mathbb{R}^n_{\delta} \neq \emptyset$ .

- Step 1: Reformulate  $\mathcal{M}$  and  $\mathcal{S}$ : Let  $W = [w_1, \ldots, w_s]$  be a matrix whose column vectors are a basis for  $\mathcal{S}^{\perp}$  and  $B = [b_1, \ldots, b_p]$  be a matrix whose column vectors are a basis for  $\mathcal{M}^{\perp}$ . Then all  $v \in \mathcal{S}$  can be represented as  $\{v \in \mathbb{R}^n \mid W^T v = 0\}$  and all  $\mu \in \mathcal{M}$  as  $\{\mu \in \mathbb{R}^n \mid B' \mu = 0\}$ , i.e.  $v \in \mathcal{S}$  and  $\mu \in \mathcal{M}$  are in a form suitable for the algorithm described above (using either W or B as matrix A).
- Step 1: Determine all  $\delta_i$  that are sign compatible to  $\mathcal{S}$ :  $\Delta^{\mathcal{S}}$ .
- Step 1: Determine all  $\delta_i$  that are sign compatible to  $\mathcal{M}$ :  $\Delta^{\mathcal{M}}$ .
- Step 1: Determine  $\Delta = \Delta^{\mathcal{S}} \cap \Delta^{\mathcal{M}}$ .
- <u>Step 1:</u> For each  $\delta_i \in \Delta$  calculate the generators of  $\mathcal{M} \cap \mathbb{R}^n_{\delta_i}$  and  $\mathcal{S} \cap \mathbb{R}^n_{\delta_i}$  (see the matlab file all\_cones.m).

The following matlab commands were used to perform these steps:

orthi=sign\_comp\_subs(W);

```
orthi_mu=sign_comp_subs(mu);
```

```
k=1;for i=1:length(orthi) for j=1:length(orthi_mu) if
(orthi(:,i)==orthi_mu(:,j)) Erg(:,k)=orthi(:,i);k=k+1;end; end;end;
```

MuCones=all\_cones(Erg,mu)

SCones=all\_cones(Erg,W)

The matlab file all\_cones.m:

```
function ConeSet=all_cones(orth_mat,w)
```

```
[Orows,Ocols]=size(orth_mat);
```

```
ConeSet={};
ConeCounter=1;
for k=1:Ocols
SomeCone = calculate_cone(orth_mat(:,k),w);
ConeSet(ConeCounter)={SomeCone};
ConeCounter = ConeCounter + 1;
end;
```

And the matlab file calculate\_cone.m that calculates generators for a particular cone:

```
function C = calculate_cone( orthant, w)
```

```
[C, num_modes, revs, mode_rates, fix_rates] = elmodes_calc( ...
    (diag(orthant)*w)', [], [], ones(1,length(w')), [], 1e-10, ...
    'useMex', '1');
if (num_modes > 0)
    C=diag(orthant)*C';
else
    C=[];
end;
```

## Bibliography

- [1] Matcont a Matlab continuation package. http://www.matcont.ugent.be/.
- [2] Andres Alonso, Joanna Sasin, Nunzio Bottini, Ilan Friedberg, Iddo Friedberg, Andrei Osterman, Adam Godzik, Tony Hunter, Jack Dixon, and Tomas Mustelin. Protein tyrosine phosphatases in the human genome. *Cell*, 117(6):699–711, Jun 2004.
- [3] T. Ando and R. Brualdi. Sign-central matrices. Linear Algebra and its Applications, 209:283–295, 1994.
- [4] Upinder S. Bhalla and Ravi Iyengar. Emergent properties of networks of biological signaling pathways. Science, 283(5400):381–387, January 1999.
- [5] Upinder S. Bhalla, Prahlad T. Ram, and Ravi Iyengar. MAP kinase phosphatase as a locus of flexibility in a Mitogen-Activated Protein Kinase signaling network. *Science*, 297(5583):1018–1023, August 2002.
- [6] R. Brualdi and B. Shader. Matrices of Sign-solvable Linear Systems. Cambridge University Press, 1995.
- [7] Carmen Chicone. Ordinary Differential Equations with Applications. Springer, 1. edition, 1999.
- [8] Bruce L. Clarke. Stoichiometric network analysis. Cell Biophysics, 12:237–253, 1988.
- [9] Carsten Conradi, Dietrich Flockerzi, Jörg Raisch, and Jörg Stelling. Subnetwork analysis reveals dynamic features of complex (bio)chemical networks. *submitted*, 2007.
- [10] Carsten Conradi, Julio Saez-Rodriguez, Ernst-Dieter Gilles, and Jörg Raisch. Using Chemical Reaction Network Theory to discard a kinetic mechanism hypothesis. In Proceedings of the 2005 Foundations of Systems Biology in Engineerin Conference (FOSBE'05), 2005.
- [11] Carsten Conradi, Julio Saez-Rodriguez, Ernst-Dieter Gilles, and Jörg Raisch. Using Chemical Reaction Network Theory to discard a kinetic mechanism hypothesis. *IEE Proc. Systems Biology*, 152(4):243–248, December 2005.
- [12] Carsten Conradi, Julio Saez-Rodriguez, Ernst-Dieter Gilles, and Jörg Raisch. Chemical Reaction Network Theory ... a tool for systems biology. Proceedings of the 5th MATHMOD, 2006.
- [13] David Cox, John Little, and Donald O'Shea. Ideals, Varieties, and Algorithms: An Introduction to Computational Algebraic Geometry and Communitative Algebra. Undergraduate Texts in Mathematics. Springer-Verlag New York, 2 edition, 1996.
- [14] David Cox, John Little, and Donald O'Shea. Using Algebraic Geometry, volume 185 of Graduate Texts in Mathematics. Springer-Verlag New York, 1998.
- [15] G. Craciun and M. Feinberg. Multiple equilibria in complex chemical reaction networks: extensions to entrapped species models. *IEE Proceedings - Systems Biology*, 153(4):179–186, July 2006.
- [16] Gheorghe Craciun and Martin Feinberg. Multiple equilibria in complex chemical reaction networks: I. the injectivity property. SIAM Journal on Applied Mathematics, 2004.
- [17] Gheorghe Craciun and Martin Feinberg. Multiple equilibria in complex chemical reaction networks: II. the species-reaction graph. SIAM Journal on Applied Mathematics, 66(4):1321–1338, 2006.

- [18] Gheorghe Craciun, Yangzhong Tang, and Martin Feinberg. Understanding bistability in complex enzyme-driven reaction networks. PNAS, 103(23):8697–8702, June 2006.
- [19] John C. Doyle, David L. Alderson, Lun Li, Steven Low, Matthew Roughan, Stanislav Shalunov, Reiko Tanaka, and Walter Willinger. The "robust yet fragile" nature of the Internet. PNAS, 102(41):14497–14502, October 2005.
- [20] H. El-Samad, H. Kurata, J.C. Doyle, C.A. Gross, and M. Khammash. Surviving heat shock: Control strategies for robustness and performance. PNAS, 102(8):2736–2741, February 2005.
- [21] Phillip Ellison and Martin Feinberg. How catalytic mechanisms reveal themselves in multiple steady-state data: I. Basic principles. *Journal of Molecular Catalysis A: Chemical*, 154(1–2):155– 167, March 2000.
- [22] Phillipp Ellison, Martin Feinberg, Ming-Huei Yueb, and Howard Saltsburg. How catalytic mechanisms reveal themselves in multiple steady-state data: II. An ethylene hydrogenation example. *Journal of Molecular Catalysis A: Chemical*, 154(1–2):169–184, March 2000.
- [23] Phillipp Raymond Ellison. The Advanced Deficiency Algorithm and its Applications to Mechanism Discrimination. PhD thesis, The University of Rochester, 1998.
- [24] Martin Feinberg. Chemical reaction network structure and the stability of complex isothermal reactors – I. The Deficiency Zero and Deficiency One theorems. *Chemical Engineering Science*, 42(10):2229–2268, 1987.
- [25] Martin Feinberg. Chemical reaction network structure and the stability of complex isothermal reactors – II. Multiple steady states for networks of Deficiency One. *Chemical Engineering Science*, 43(1):1–25, 1988.
- [26] Martin Feinberg. Necessary and sufficient conditions for detailed balancing in mass action systems of arbitrary complexity. *Chemical Engineering Science*, 44(9):1819–1827, 1989.
- [27] Martin Feinberg. The existence and uniqueness of steady states for a class of chemical reaction networks. Archive for Rational Mechanics and Analysis, 132(4):311–370, 1995.
- [28] Martin Feinberg. Multiple steady states for chemical reaction networks of Deficiency One. Archive for Rational Mechanics and Analysis, 132(4):371–406, 1995.
- [29] Martin Feinberg and Phillip Raymond Ellison. The Chemical Reaction Network Toolbox. http://www.chbmeng.ohio-state.edu/ feinberg/crnt.
- [30] Martin Feinberg and Friedrich J. M. Horn. Dynamics of open chemical systems and the algebraic structure of the underlying reaction network. *Chemical Engineering Science*, 29(3):775–787, March 1974.
- [31] James E. Ferrell. Self-perpetuating states in signal transduction: positive feedback, double-negative feedback and bistability. *Current Opinion in Cell Biology*, 14(2):140–148, April 2002.
- [32] Julien Gagneur and Steffen Klamt. Computation of elementary modes: a unifying framework and the new binary approach. BMC Bioinformatics, 5, November 2004.
- [33] Karin Gatermann and Birkett Huber. A family of sparse polynomial systems arising in chemical reaction systems. Journal of Symbolic Computation, 33(3):275–305, March 2002.
- [34] Karin Gatermann and Pablo A. Parrilo. Symmetry groups, semidefinite programs, and sums of squares. Journal of pure and applied Algebra, 192(1-3):95–128, 2004.
- [35] Karin Gatermann and Matthias Wolfrum. Bernstein's second theorem and viro's method for sparse polynomial systems in chemistry. Advances in Applied Mathematics, 34(2):252–294, February 2005.
- [36] Chris Godsil and Gordon Royle. Algebraic Graph Theory. Springer Verlag, 2001.
- [37] Chi-Ying F. Huang and James E. Ferrell Jr. Ultrasensitivity in the Mitogen-Activated Protein Kinase Cascade. PNAS, 93(19):10078–10083, September 1996.

- [38] Boris N. Kholodenko. Negative feedback and ultrasensitivity can bring about oscillations in the mitogen-activated protein kinase cascades. *European Journal of Biochemistry*, 267(6):1583–1588, March 2000.
- [39] Boris N. Kholodenko. Cell-signalling dynamics in time and space. Nature Reviews Molecular Cell Biology, 7:165–176, March 2006.
- [40] S. Kim and B. Shader. Sign-solvable cone-systems. Linear & Multilinear Algebra, 50(1):23–32, 2002.
- [41] S. Kim, B. Shader, and S. Hwang. On matrices with signed null-spaces. SIAM Journal on Matrix Analysis and Applications, 24(2):570–580, 2002.
- [42] Hiroaki Kitano. Biological robustness. Nature Reviews Genetics, 5(11):826-837, 2004.
- [43] Y. A. Kuznetsov. Elements of Applied Bifurcation Theory. Springer-Verlag, 1995.
- [44] G. Lee and B. Shader. Sign-consistency and solvability of constrained linear systems. *Electronic Journal of Linear Algebra*, 4:1–18, August 1998.
- [45] Nick I. Markevich, Jan B. Hoek, and Boris N. Kholodenko. Signaling switches and bistability arising from multisite phosphorylation in protein kinase cascades. *The Journal of Cell Biology*, 164(3):353–359, 2004.
- [46] M. D. Mendenhall and A. E. Hodge. Regulation of Cdc28 cyclin-dependent protein kinase activity during the cell cycle of the yeast Saccharomyces cerevisiae. *Microbiol Mol Biol Rev*, 62(4):1191– 1243, Dec 1998.
- [47] Mineo Morohashi, Amanda E Winn, Mark T Borisuk, Hamid Bolouri, John Doyle, and Hiroaki Kitano. Robustness as a measure of plausibility in models of biochemical networks. J Theor Biol, 216(1):19–30, May 2002.
- [48] Mineo Morohashi, Amanda E. Winn, Mark T. Borisuk, Hamid Bolouri, John Doyle, and Hiroaki Kitano. Robustness as a measure of plausibility in models of biochemical networks. *Journal of Theoretical Biology*, 216(1):19–30, May 2002.
- [49] T. Mustelin, S. Rahmouni, N. Bottini, and A. Alonso. Role of protein tyrosine phosphatases in t cell activation. *Immunol Rev*, 191:139–47, Feb 2003.
- [50] Fernando Ortega, Jos L. Garcs, Francesc Mas, Boris N. Kholodenko, and Marta Cascante. Bistability from double phosphorylation in signal transduction. *FEBS Journal*, 273(17):3915–3926, September 2006.
- [51] Lawrence Perko. Differential Equations and Dynamical Systems. Springer, 2. edition, 1996.
- [52] Ralph Tyrell Rockafellar. Convex Analysis. Princton University Press, 1970.
- [53] Julio Saez-Rodriguez, Andrea Hammerle-Fickinger, Onkar Dalal, Steffen Klamt, Ernst Dieter Gilles, and Carsten Conradi. On the multistability of signal transduction motifs. *submitted*, 2007.
- [54] Herbert M. Sauro and Boris N. Kholodenko. Quantitative analysis of signaling networks. Progress in Biophysics and Molecular Biology, 86(1):5–43, September 2004.
- [55] Paul M. Schlosser and Martin Feinberg. A theory of multiple steady states in isothermal homogeneous cfstrs with many reactions. *Chemical Engineering Science*, 49(11):1749–1767, June 1994.
- [56] S. Schuster, C. Hilgetag, J. H. Woods, and D. A. Fell. Reaction routes in biochemical reaction systems: Algebraic properties, validated calculation procedure and example from nucleotide metabolism. *Journal of Mathematical Biology*, 45(2):153–181, August 2002.
- [57] R. Seger and E.G. Krebs. The mapk signaling cascade. The FASEB Journal, 9:726–735, 1995.

- [58] E. D. Sontag. Structure and stability of certain chemical networks and applications to the kinetic proofreading model of T-cell receptor signal transduction. *IEEE Transactions on Automatic Control*, 46(7):1028–1047, 2001.
- [59] Jörg Stelling, Ernst Dieter Gilles, and III Francis J. Doyle. Robustness properties of circadian clock architectures. PNAS, 101(36):13210–13215, September 2004.
- [60] Jörg Stelling, Uwe Sauer, Zoltan Szallasi, III Francis J.Doyle, and John Doyle. Robustness of cellular functions. *Cell*, 118(6):675–685, September 2004.
- [61] Gilbert Strang. Linear Algebra and its Applications. Academic Press, 1976.
- [62] Bernd Sturmfels. Solving Systems of Polynomial Equations. American Mathematical Society, 2002.
- [63] John J Tyson, Katherine C Chen, and Bela Novak. Sniffers, buzzers, toggles and blinkers: dynamics of regulatory and signaling pathways in the cell. Curr Opin Cell Biol, 15(2):221–231, Apr 2003.
- [64] J.H. Wilkinson. The Algebraic Eigenvalue Problem. Oxford University Press, 1988.
- [65] Carsten Conradi, Dietrich Flockerzi, and Jörg Raisch. Multistationarity in the activation of an MAPK: parametrizing the relevant region in parameter space. *Mathematical Biosciences*, 211(1):105–131, January 2008.